UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-K

ANNUAL REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the fiscal year ended August 31, 2019

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Transition Period From ______ to _____

Commission file number 001-36759

WALGREENS BOOTS ALLIANCE, INC.

(Exact name of registrant as specified in its charter)

(Endet name of registrant as specifica in its charter)					
Delaware	47-1758322				
(State of incorporation)	(I.R.S. Employer Identification No.)				
108 Wilmot Road, Deerfield, Illinois	60015				
(Address of principal executive offices)	(Zip Code)				
Registrant's telephone number, inc	cluding area code: <u>(847) 315-2500</u>				
Securities registered pursuant to Section 12(b) of the Act:					

Securities registered pursuant to Section 12(b) of the Act:

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common Stock, \$0.01 par value	WBA	The Nasdaq Stock Market LLC
2.875% Walgreens Boots Alliance, Inc. notes due 2020	WBA20	The Nasdaq Stock Market LLC
3.600% Walgreens Boots Alliance, Inc. notes due 2025	WBA25	The Nasdaq Stock Market LLC
2.125% Walgreens Boots Alliance, Inc. notes due 2026	WBA26	The Nasdaq Stock Market LLC
Securities registered pursuant to Section 12(g) of the Act: <u>None</u>		

Indicate by check mark if the registrant is a well-known seasoned issuer, as defined in Rule 405 of the Securities Act. Yes \square No \square Indicate by check mark if the registrant is not required to file reports pursuant to Section 13 or Section 15(d) of the Act. Yes \square No \square

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15(d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports) and (2) has been subject to such filing requirements for the past 90 days. Yes \Box No \Box

Indicate by check mark whether the registrant has submitted electronically every Interactive Data File required to be submitted pursuant to Rule 405 of Regulation S-T (§232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit such files). Yes \square No \square

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, a smaller reporting company or an emerging growth company. See the definitions of "large accelerated filer," "accelerated filer," "smaller reporting company," and "emerging growth company" in Rule 12b-2 of the Exchange Act. (Check one):

Large accelerated filer	Accelerated filer \Box
Non-accelerated filer \Box	Smaller reporting company \Box
	Emerging growth company \Box

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). Yes 🗆 No 🗵

As of February 28, 2019, the aggregate market value of Walgreens Boots Alliance, Inc. common stock held by non-affiliates (based upon the closing transaction price on such date) was approximately \$54.5 billion.

As of September 30, 2019, there were 892,541,146 shares of Walgreens Boots Alliance, Inc. common stock outstanding.

DOCUMENTS INCORPORATED BY REFERENCE

Portions of the definitive proxy statement for our Annual Meeting of Stockholders planned to be held on January 30, 2020 are incorporated by reference into Part III of this Form 10-K as indicated herein.

Walgreens Boots Alliance, Inc. Annual Report on Form 10-K Table of Contents

Part I

		<u>Page</u>
Item 1.	Business	1
Item 1A.	Risk factors	8
Item 1B.	Unresolved staff comments	28
Item 2.	Properties	28
Item 3.	Legal proceedings	29
Item 4.	Mine safety disclosures	29
	Executive officers of the registrant	29

Part II

Item 5.	Market for registrant's common equity, related stockholder matters and issuer purchases of equity securities	32
Item 6.	Selected financial data	33
Item 7.	Management's discussion and analysis of financial condition and results of operations	35
Item 7A.	Quantitative and qualitative disclosure about market risk	56
Item 8.	Financial statements and supplementary data	58
Item 9.	Changes in and disagreements with accountants on accounting and financial disclosure	107
Item 9A.	Controls and procedures	107
Item 9B.	Other information	107

Part III

Item 10.	Directors, executive officers and corporate governance	108
Item 11.	Executive compensation	108
Item 12.	Security ownership of certain beneficial owners and management and related stockholder matters	108
Item 13.	Certain relationships and related transactions and director independence	108
Item 14.	Principal accounting fees and services	108

Part IV

Item 15.	Exhibits and financial statement schedules	109
Item 16.	Form 10-K summary	117
	Signatures	118

On December 31, 2014, Walgreens Boots Alliance, Inc. became the successor of Walgreen Co. ("Walgreens") pursuant to a merger to effect a reorganization of Walgreens into a holding company structure (the "Reorganization"), with Walgreens Boots Alliance, Inc. becoming the parent holding company.

References in this Annual Report on Form 10-K (this "Form 10-K") to the "Company," "we," "us" or "our" refer to Walgreens Boots Alliance, Inc. and its subsidiaries from and after the effective time of the Reorganization on December 31, 2014 and, prior to that time, to the predecessor registrant Walgreens and its subsidiaries, and in each case do not include unconsolidated partially-owned entities, except as otherwise indicated or the context otherwise requires. Our fiscal year ends on August 31, and references herein to "fiscal 2019" refer to our fiscal year ended August 31, 2019.

This Form 10-K includes forward-looking statements made pursuant to the safe harbor provisions of the Private Securities Litigation Reform Act of 1995. See cautionary note regarding forward-looking statements in management's discussion and analysis of financial condition and results of operations in part II, item 7 below.

All trademarks, trade names and service marks used herein are the property of their respective owners.

PART I

Item 1. Business

Overview

Walgreens Boots Alliance, Inc., a Delaware corporation ("Walgreens Boots Alliance"), is a global leader in retail and wholesale pharmacy, touching millions of lives every day through dispensing and distributing medicines, its convenient retail locations, digital platforms and health and beauty products with sales of \$136.9 billion in the fiscal year ended August 31, 2019. We have more than 100 years of trusted health care heritage and innovation in community pharmacy and pharmaceutical wholesaling. Our purpose is to help people across the world lead healthier and happier lives.

Walgreens Boots Alliance is the largest retail pharmacy, health and daily living destination across the United States and Europe. Walgreens Boots Alliance and the companies in which it has equity method investments together have a presence in more than 25¹ countries and employ more than 440,000¹ people. The Company is a global leader in retail and wholesale pharmacy and, together with the companies in which it has equity method investments, has over 18,750¹ stores in 11¹ countries as well as one of the largest global pharmaceutical wholesale and distribution networks, with over 400¹ distribution centers delivering to more than 240,000² pharmacies, doctors, health centers and hospitals each year in more than 20¹ countries. In addition, Walgreens Boots Alliance is one of the world's largest purchasers of prescription drugs and many other health and wellbeing products. The Company's size, scale and expertise will help us to expand the supply of, and address the rising cost of, prescription drugs in the United States and worldwide.

Our portfolio of retail and business brands includes Walgreens, Duane Reade, Boots and Alliance Healthcare, as well as increasingly global health and beauty product brands, such as No7, Soap & Glory, Liz Earle, Botanics, Sleek MakeUP and YourGoodSkin. Our global brands portfolio is enhanced by our in-house product research and development capabilities. We seek to further drive innovative ways to address global health and wellness challenges. Our strategic partnerships with some of the world's leading companies enable us to extend our healthcare solutions and convenience offering to the communities we serve, which we refer to as our "strategic partner strategy." We believe we are well positioned to expand customer offerings in existing markets and become a health and wellbeing partner of choice in emerging markets.

Walgreens Boots Alliance is proud to be a force for good through its contributions to healthy communities, a healthy planet, an inclusive workplace and a sustainable marketplace. The Company is focused on creating a neighborhood health destination and a more modern pharmacy.

Walgreens Boots Alliance was incorporated in Delaware in 2014 and, as described below, is the successor of Walgreen Co., an Illinois corporation, which was formed in 1909 as a successor to a business founded in 1901. Our principal executive offices are located at 108 Wilmot Road, Deerfield, Illinois 60015. Our common stock trades on the Nasdaq Stock Market under the symbol "WBA".

- ¹ As of August 31, 2019, using publicly available information for AmerisourceBergen.
- ² Over 12 month period ending August 31, 2019, using publicly available information for AmerisourceBergen.

- 1 -

Recent Transactions

In fiscal 2019, as part of our strategic partnership strategy, the Company has announced several strategic partnerships and alliances with other leaders in the technology, health and retail sectors. These strategic partnerships, initiatives and alliances are designed to complement and enhance our customer offerings.

On September 19, 2017, the Company announced that it had secured regulatory clearance for an amended and restated asset purchase agreement to purchase 1,932 stores, three distribution centers and related inventory from Rite Aid Corporation ("Rite Aid") for \$4.375 billion in cash and other consideration. Consideration included cash of \$4,157 million and the fair value of the option granted to Rite Aid to become a member of the Company's group purchasing organization, Walgreens Boots Alliance Development GmbH. The purchases of these stores occurred in waves during fiscal 2018 and have been accounted for as business combinations. The transition of the first distribution center and related inventory occurred in September 2018 and the transition of the remaining two distribution centers and related inventory remains subject to closing conditions set forth in the amended and restated asset purchase agreement. Previously, on June 28, 2017, Walgreens Boots Alliance and Rite Aid had terminated an amended agreement and plan of merger pursuant to which the Company had agreed to acquire Rite Aid. Pursuant to such termination, the Company paid Rite Aid a termination fee of \$325 million. The Company also reimbursed \$25 million of transaction costs of Fred's, Inc. in connection with the termination of an asset purchase agreement among the Company, Rite Aid and Fred's, Inc. that was subject to the completion of Rite Aid by the Company. The option granted to Rite Aid to become a member of the Company's group purchasing organization was terminated during fiscal 2019, resulting in recognition of a gain in other income (expense).

On March 31, 2017, Walgreens Boots Alliance and pharmacy benefit manager Prime Therapeutics LLC closed a transaction to form a combined central specialty pharmacy and mail services company, AllianceRx Walgreens Prime, as part of a strategic alliance. AllianceRx Walgreens Prime is consolidated by Walgreens Boots Alliance and reported within the Retail Pharmacy USA division in its financial statements. See note 2, acquisitions, to the Consolidated Financial Statements for further information.

In addition, the Company has completed additional acquisitions, divestitures and strategic initiatives in recent years designed to grow its businesses and enhance its competitive position. Please refer to management's discussion and analysis of financial condition and results of operations in part II, item 7, below and note 2, acquisitions, note 3, exit and disposal activities, and note 5, equity method investments, to the Consolidated Financial Statements included in part II, item 8, below for additional information.

Relationship with AmerisourceBergen

In March 2013, Walgreens, Alliance Boots GmbH ("Alliance Boots") and AmerisourceBergen Corporation ("AmerisourceBergen") announced various agreements and arrangements, including a ten-year pharmaceutical distribution agreement between Walgreens and AmerisourceBergen pursuant to which the Company sources branded and generic pharmaceutical products from AmerisourceBergen in the United States and an agreement which provides AmerisourceBergen the ability to access generic pharmaceutical products through the Company's global sourcing enterprise, Walgreens Boots Alliance Development GmbH. In May 2016, certain of these agreements were extended for three years and are now expected to expire in 2026.

In addition, in March 2013, Walgreens, Alliance Boots and AmerisourceBergen entered into agreements and arrangements pursuant to which the Company has the right, but not the obligation, to purchase a minority equity position in AmerisourceBergen over time through open market purchases and pursuant to warrants to acquire AmerisourceBergen common stock and gain associated representation on AmerisourceBergen's Board of Directors in certain circumstances. Please refer to the Company's Form 8-K filed on March 20, 2013 for more detailed information regarding these agreements and arrangements.

On March 18, 2016, the Company exercised warrants to purchase 22,696,912 shares of AmerisourceBergen common stock at an exercise price of \$51.50 per share for an aggregate exercise price payment of \$1.17 billion. On August 25, 2016, the Company exercised additional warrants to purchase 22,696,912 shares of AmerisourceBergen common stock at an exercise price of \$52.50 per share for an aggregate exercise price payment of \$1.19 billion.

As of August 31, 2019, the Company owned 56,854,867 AmerisourceBergen common shares representing approximately 27% of the outstanding AmerisourceBergen common stock and has designated one member of AmerisourceBergen's board of directors. As of August 31, 2019, the Company can acquire up to an additional 8,398,752 AmerisourceBergen shares in the open market and thereafter designate another member of AmerisourceBergen's board of directors, subject in each case to applicable legal and contractual requirements. The amount of permitted open market purchases is subject to increase or decrease in certain circumstances.

- 2 -

The Company accounts for its investment in AmerisourceBergen using the equity method of accounting, subject to a two-month reporting lag, with the net earnings attributable to the investment classified within the operating income of the Company's Pharmaceutical Wholesale segment.

For more information, see management's discussion and analysis in part II, item 7 and note 5, equity method investments and note 18, related parties to the Consolidated Financial Statements included in part II, item 8, below for additional information.

Industry overview

The retail pharmacy and pharmaceutical wholesale industries across the globe are highly competitive and dynamic and have experienced consolidation and an evolving competitive landscape in recent years. Prescription drugs play a significant role in healthcare and constitute a first line of treatment for many medical conditions. The Company believes the long-term outlook for prescription drug utilization is strong due, in part, to aging populations, increases in life expectancy, increases in the availability of generic drugs, the continued development of innovative drugs that improve quality of life and control healthcare costs and increases in the number of persons with insurance coverage for prescription drugs, including, in the United States, "baby boomers" increasingly becoming eligible for the federally funded Medicare Part D prescription program. Pharmaceutical wholesalers act as a vital link between drug manufacturers and pharmacies and healthcare providers in supplying pharmaceuticals to patients.

The retail pharmacy industry across the globe relies significantly on private and governmental third-party payers. Many private organizations throughout the healthcare industry, including pharmacy benefit management ("PBM") companies and health insurance companies, have consolidated in recent years to create larger healthcare enterprises with greater bargaining power. Third-party payers, including the Medicare Part D plans and the state-sponsored Medicaid and related managed care Medicaid agencies in the United States, can change eligibility requirements or reduce certain reimbursement rates. In addition, in many European countries, the government provides or subsidizes healthcare to consumers and regulates pharmaceutical prices, patient eligibility and reimbursement levels to control costs for the government-sponsored healthcare system. Changes in law or regulation also can impact reimbursement rates and terms. For example, the Patient Protection and Affordable Care Act (the "ACA") was enacted to help control federal healthcare spending, including for prescription drugs, in the United States. These changes generally are expected to reduce Medicaid reimbursements in the United States. State Medicaid programs are also expected to continue to seek reductions in reimbursements. When third-party payers or governmental authorities take actions that restrict eligibility or reduce prices or reimbursement rates, sales and margins in the retail pharmacy industry could be reduced, which would adversely affect industry profitability. In some cases, these possible adverse effects may be partially or entirely offset by controlling inventory costs and other expenses, dispensing more higher margin generics, finding new revenue streams through pharmacy services or other offerings and/or dispensing a greater volume of prescriptions.

These industry dynamics and challenges are continuous and have intensified in recent years. Since the completion of the strategic combination with Alliance Boots in December 2014, the Company has had a continuous focus on operational efficiencies and cost reduction.

Generic prescription drugs have continued to help lower overall costs for customers and third-party payers. The Company expects the utilization of generic pharmaceuticals to continue to increase. In general, in the United States, generic versions of drugs generate lower sales dollars per prescription, but higher gross profit dollars, as compared with patent-protected brand name drugs. The impact on retail pharmacy gross profit dollars can be significant in the first several months after a generic version of a drug is first allowed to compete with the branded version, which is generally referred to as a "generic conversion". In any given year, the number of major brand name drugs that undergo a conversion from branded to generic status can vary and the timing of generic conversions can be difficult to predict, which can have a significant impact on retail pharmacy sales and gross profit dollars. In general, in the United States, specialty prescription business is also growing and generates higher sales dollars per prescription, but lower gross margin, as compared to generic prescription drugs.

The Company expects that market demand, government regulation, third-party reimbursement policies, government contracting requirements and other pressures will continue to cause the industries in which the Company competes to evolve. Pharmacists are on the frontlines of the healthcare delivery system, and the Company believes rising healthcare costs and the limited supply of primary care physicians present opportunities for pharmacists and retail pharmacies to play an even greater role in driving positive outcomes for patients and payers through expanded service offerings such as immunizations and other preventive care, healthcare clinics, pharmacist-led medication therapy management and chronic condition management.

Segments

Our operations are organized into three divisions, which are also our reportable segments:

Retail Pharmacy USA;



- Retail Pharmacy International; and
- Pharmaceutical Wholesale.

For fiscal 2019, our segment sales were: Retail Pharmacy USA, \$104.5 billion; Retail Pharmacy International, \$11.5 billion; and Pharmaceutical Wholesale, \$23.1 billion. Additional information relating to our segments is included in management's discussion and analysis of financial condition and results of operations in part II, item 7 below and in note 16, segment reporting and note 17, sales, to our Consolidated Financial Statements included in part II, item 8 below, which information is incorporated herein by reference.

Retail Pharmacy USA

The Retail Pharmacy USA division (excluding equity method investments) has pharmacy-led health and beauty retail offerings in 50 states, the District of Columbia, Puerto Rico and the U.S. Virgin Islands, each focused on helping people feel happy and healthy. The Company operated 9,277 retail stores in the division as of August 31, 2019. The principal retail pharmacy brands in the division are Walgreens and Duane Reade. The Company is a market leader in the United States and, as of August 31, 2019, approximately 78% of the population of the United States lived within five miles of a Walgreens, Duane Reade or acquired Rite Aid retail pharmacy.

The Company is focused on creating a neighborhood health destination and a more modern pharmacy aligned to a wider range of healthcare services. The Company's services help improve health outcomes for patients and manage costs for payers, including employers, managed care organizations, health systems, PBM companies and the public sector. The Company utilizes its retail network as a channel to provide health and wellness services to its customers and patients, as illustrated by the Company's ability to play a significant role in providing flu vaccines and other immunizations. Additionally, through our strategic partnership strategy, our key collaborations aim to develop new health care delivery models and to improve access to advanced healthcare technologies and solutions. We have taken further steps to develop our neighborhood health destinations, working with our strategic partners.

The Company also provides specialty pharmacy and mail services and offers in-store clinics and other healthcare services throughout the United States, most of which are operated by our healthcare strategic partners. The Company has more than 88,000 healthcare service providers, including pharmacists, pharmacy technicians, nurse practitioners and other health related professionals.

The division provides customers with convenient, omni-channel access to consumer goods and services, including own branded general merchandise, such as Well at Walgreens, NICE!, No7, and Soap & Glory, as well as pharmacy and health and wellness services in communities across the United States. Integrated with the Company's e-commerce platform, the Walgreens mobile application allows customers to refill prescriptions through scan technology, receive notifications when a refill is due and engage with retail functionality, such as photo and shopping features. The division is also implementing new approaches to promotions, product selection and other areas to deliver greater value to its customers in its stores, including through an enhanced beauty offering.

The components of the division's sales are Pharmacy (the sale of prescription drugs and provision of pharmacy-related services) and Retail (the sale of healthcare and retail products including non-prescription drugs, beauty, toiletries and general merchandise). The division's sales are subject to the influence of seasonality, particularly the winter holiday and cough, cold and flu seasons. This seasonality also can affect the division's proportion of sales between Retail and Pharmacy during certain periods. The components of the division's fiscal year sales were as follows:

	Fiscal 2019	Fiscal 2018	Fiscal 2017
Pharmacy	74%	72%	69%
Retail	26%	28%	31%
Total	100%	100%	100%

The Company filled 843.7 million prescriptions (including immunizations) in the division in fiscal 2019. Adjusted to 30-day equivalents, prescriptions filled were 1.2 billion in fiscal 2019. Sales where reimbursement is received from managed care organizations, governmental agencies, PBM companies and private insurance were approximately 97% of the division's fiscal 2019 Pharmacy sales.

The Company fills prescriptions for many state Medicaid public assistance programs. Sales from all such Medicaid plans were approximately 4% of the division's fiscal 2019 sales. Sales from Medicare Part D plans were approximately 20% of the division's fiscal 2019 sales.

The Company's U.S. loyalty program, Balance® Rewards, is designed to reward its most valuable customers and encourage shopping in stores and online. Balance® Rewards members receive special pricing on select products and earn everyday rewards points for purchasing most merchandise that can be instantly redeemed in store or through walgreens.com. As of August 31, 2019, the number of active Balance® Rewards members totaled approximately 90 million. For this purpose, an active member is defined here as someone who has used their card in the last six months.

AmerisourceBergen supplies and distributes a significant amount of generic and branded pharmaceutical products to the division's pharmacies. The Company purchases its non-pharmaceutical merchandise from numerous manufacturers and wholesalers.

The division's sales, gross profit margin and gross profit are impacted by, among other things, both the percentage of prescriptions filled that are generic and the rate at which new generic drugs are introduced to the market. Because any number of factors outside of the Company's control can affect timing for a generic conversion, the Company faces substantial uncertainty in predicting when such conversions will occur and what effect they will have on particular future periods.

The current environment of the Company's pharmacy business also includes ongoing reimbursement pressure and a shift in pharmacy mix towards 90-day at retail (one prescription that is the equivalent of three 30-day prescriptions) and Medicare Part D prescriptions. Further consolidation among generic manufacturers coupled with changes in the number of major brand name drugs anticipated to undergo a conversion from branded to generic status may also result in gross margin pressures within the industry.

The Company continuously faces reimbursement pressure from PBM companies, health maintenance organizations, managed care organizations and other commercial third-party payers; agreements with these payers are regularly subject to expiration, termination or renegotiation. In addition, plan changes with rate adjustments often occur in January and the Company's reimbursement arrangements may provide for rate adjustments at prescribed intervals during their term. The Company experienced lower reimbursement rates in fiscal 2019 as compared to the same period in the prior year. The Company expects these pressures to continue.

The Company has also worked to develop and expand its relationships with commercial third-party payers to enable new and/or improved market access via participation in pharmacy provider networks they offer. The prescription volume impact of new agreements and relationships typically is incremental over time.

The Company's 90-day at retail prescription drug offering is typically at a lower margin than comparable 30-day prescriptions, but provides the Company with the opportunity to increase business with patients with chronic prescription needs while offering increased convenience, helping facilitate improved prescription adherence and resulting in a lower cost to fill the 90-day prescription. Similarly, the growing specialty prescription business, which generates higher sales dollars per prescription, may result in gross margin pressures within the industry, as compared to generic prescription drugs.

Retail Pharmacy International

The Retail Pharmacy International division (excluding equity method investments) has pharmacy-led health and beauty retail businesses in eight countries, each focused on helping people look and feel their best. The Company operated 4,605 retail stores in the division as of August 31, 2019 (see properties in part I, item 2 below for information regarding geographic coverage) and has grown its omni-channel platform, including its online presence, in recent years. The Company's principal retail pharmacy brands are Boots in the United Kingdom, Thailand, Norway, the Republic of Ireland and The Netherlands, Benavides in Mexico and Ahumada in Chile. In Europe, the Company is a market leader and its retail stores are conveniently located and its pharmacists are well placed to provide a significant role in the provision of healthcare services, working closely with other primary healthcare providers in the communities the Company serves.

The Boots omni-channel offering is differentiated from that of competitors due to the product brands the Company owns, such as No7, Boots Pharmaceuticals, Liz Earle, Soap & Glory, Sleek MakeUp, Botanics and 'only at Boots' exclusive products, together with its long established reputation for trust and customer care. The Company's brands portfolio is enhanced by its in-house product research and development capabilities. The Company has recently introduced new beauty brands and beauty halls in key locations. Certain of the product brands of the Company are also sold by third-party retailers.

The Company's retail store networks are typically complemented by online platforms. In the United Kingdom, through the boots.com website and integrated mobile application, the 'order and collect' service allows customers to order from a range of over 33,000 products by 8:00 p.m. and collect from noon the following day from approximately 97% of the United Kingdom's retail stores as of August 31, 2019.

- 5 -

The Boots Advantage Card loyalty program, where customers earn points on purchases for redemption at a later date, continues to be a key element of the Boots offering. As of August 31, 2019, the number of active Boots Advantage Card members totaled approximately 15 million. For this purpose, an active member is defined as someone who has used their card in the last six months.

In addition, Boots in the United Kingdom is one of the leaders in the optical market with 606 practices, of which 165 operated on a franchise basis as of August 31, 2019. Approximately 30% of these optical practices are located in Boots stores with the balance being standalone optical practices.

The components of the division's sales are Pharmacy (typically the sale of prescription drugs and provision of pharmacy-related services, subject to variation in particular jurisdictions depending upon regulatory and other factors) and Retail (primarily the sale of health and beauty products including beauty, toiletries and lifestyle merchandising, non-prescription drugs and, in the United Kingdom, the provision of optical services).

The division's sales are subject to the influence of seasonality, with the second fiscal quarter typically the strongest as a result of the winter holiday period. This seasonality affects the division's proportion of sales between Retail and Pharmacy during certain periods. The components of the division's fiscal year sales were as follows:

	Fiscal 2019	Fiscal 2018	Fiscal 2017
Pharmacy	36%	35%	35%
Retail	64%	65%	65%
Total	100%	100%	100%

The division's Retail sales, gross profit margin and gross profit dollars are impacted by, among other things, the highly competitive nature of the health and beauty category, specifically the Company and its competitors' pricing actions, promotional offers and events and the customer's desire for value and convenience.

The division's Pharmacy sales, gross margin and gross profit dollars are impacted by governmental agencies and other third-party payers seeking to minimize increases in the costs of healthcare, including pharmaceutical drug reimbursement rates. In the United Kingdom, which is the division's largest market for Pharmacy sales, the amount of government funding available for pharmacy services is typically reviewed and agreed with the pharmacy industry on an annual basis.

In addition, performance as measured in U.S. dollars is impacted by the exchange rates used to translate these amounts into U.S. dollars, the exchange rate of British pound sterling being the most significant.

The division's performance and relevant exchange rates are also impacted by the current geopolitical environment, including the political uncertainty in the United Kingdom surrounding "Brexit." For more information relating to this topic, see Risk Factors - Our substantial international business operations subject us to a number of operating, economic, political, regulatory and other international business risks.

Pharmaceutical Wholesale

The Pharmaceutical Wholesale division consists of the Alliance Healthcare pharmaceutical wholesaling and distribution businesses and an equity method investment in AmerisourceBergen. The Pharmaceutical Wholesale division (excluding equity method investments), which mainly operates under the Alliance Healthcare brand, supplies medicines, other healthcare products and related services to more than 115,000 pharmacies, doctors, health centers and hospitals each year from 300 distribution centers in 11 countries, primarily in Europe, as of August 31, 2019.

The distribution of prescription medicines to pharmacists comprises the vast majority of the division's sales. The wholesale businesses seek to provide high core service levels to pharmacists in terms of frequency of delivery, product availability, delivery accuracy, timeliness and reliability at competitive prices. The Company also offers customers innovative added-value services to help pharmacists develop their own businesses. This includes membership of Alphega Pharmacy, the Company's pan-European network for independent pharmacies, which, as of August 31, 2019, had over 6,800 members.

In addition to the wholesale of medicines and other healthcare products, the division's businesses provide services to pharmaceutical manufacturers which are increasingly seeking to gain greater control over their product distribution, while at the same time outsourcing non-core activities. These services include pre-wholesale and contract logistics (mainly under the

- 6 -

Alloga brand), direct deliveries to pharmacies and innovative and specialized healthcare services, covering clinical homecare, medicine support, dispensing services, medicine preparation and clinical trial support (mainly under the Alcura brand).

Combined with local engagement, scale is important in pharmaceutical wholesaling. Walgreens Boots Alliance is one of the largest pharmaceutical wholesalers and distributors in Europe, and it ranks as one of the top three in market share in many of the individual countries in which it operates.

The division's sales, gross profit margin and gross profit dollars are impacted by, among other things, government actions, which typically seek to reduce the growth in prescription drug consumption, reduce reimbursement rates and increase generic drug utilization. A greater proportion of generic drugs, whether as a result of government actions, generic conversions or other factors, typically has an adverse effect on the Company's revenues. However, in the wholesale division, the Company typically earns equal or better gross margins on generic drugs than on branded drugs, although there are exceptions.

Changes in manufacturers' product distribution business models can also impact the division's sales and gross margin. For example, when pharmaceutical drug manufacturers introduce fee-for-service contracts, the Company's sales are reduced even if it is successful in winning these contracts, as the Company only recognizes sales for the amount of the fees charged. Other manufacturer services, including pre-wholesale and contract logistics operations, are typically on a fee-for-service basis.

In addition, performance as measured in U.S. dollars is impacted by the exchange rates used to translate these amounts into U.S. dollars, the exchange rate of British pound sterling and the Euro being the most significant. The division's sales are subject to less seasonality than the Company's other divisions.

The division's performance and relevant exchange rates are also impacted by the current geopolitical environment, including the political uncertainty in the United Kingdom surrounding "Brexit." For more information relating to this topic, see Risk Factors - Our substantial international business operations subject us to a number of operating, economic, political, regulatory and other international business risks.

Intellectual property and licenses

The Company markets products and services under various trademarks, trade dress and trade names and relies on a combination of patent, copyright, trademark, service mark and trade secret laws, as well as contractual restrictions to establish and protect its proprietary rights. The Company owns numerous domain names, holds numerous patents, has registered numerous trademarks and has filed applications for the registration of a number of other trademarks and service marks in various jurisdictions. The Company holds assorted business licenses (such as pharmacy, occupational, liquor and cigarette) having various lives within multiple legal jurisdictions, which are necessary for the normal operation of the business.

Seasonal variations in business

The Company's business is affected by a number of factors including, among others, its sales performance during holiday periods (including particularly the winter holiday season) and during the cough, cold and flu season (the timing and severity of which is difficult to predict), significant weather conditions, the timing of its own or competitor discount programs and pricing actions and the timing of changes in levels of reimbursement from governmental agencies and other third-party payers. See the summary of quarterly results (unaudited) in note 19, supplementary financial information, to the Consolidated Financial Statements included in part II, item 8 below.

Sources and availability of raw materials

Inventories are purchased from numerous domestic and foreign suppliers. The Company does not believe that the loss of any one supplier or group of suppliers under common control would have a material adverse effect on its business or that of any of its divisions.

Working capital practices

Effective inventory management is important to the Company's operations. The Company uses various inventory management techniques, including demand forecasting and planning and various forms of replenishment management. Its working capital needs typically are greater in the months leading up to the winter holiday season. The Company generally finances its inventory and expansion needs with internally-generated funds and short-term debt. For additional information, see the liquidity and capital resources section in management's discussion and analysis of financial condition and results of operations in part II, item 7, below.

Customers

The Company sells to numerous retail and wholesale customers. No single customer accounted for more than 10% of the Company's consolidated sales for any of the periods presented. In fiscal 2019, substantially all of our retail pharmacy sales



were to customers covered by third-party payers (e.g., pharmacy benefit managers, insurance companies and governmental agencies) that agree to pay for all or a portion of a customer's eligible prescription purchases. Two third-party payers, in the Retail Pharmacy USA division, in the aggregate accounted for approximately 22% of the Company's consolidated sales in fiscal 2019.

See note 16, segment reporting, to the Consolidated Financial Statements.

Regulation

In the countries in which the Company does business, the Company is subject to national, state and local laws, regulations and administrative practices concerning retail and wholesale pharmacy operations, including regulations relating to the Company's participation in Medicare, Medicaid and other publicly financed health benefit plans; regulations prohibiting kickbacks, beneficiary inducement and the submission of false claims; the Health Insurance Portability and Accountability Act ("HIPAA"); the ACA; licensure and registration requirements concerning the operation of pharmacies and the practice of pharmacy; and regulations of the U.S. Food and Drug Administration, the U.S. Federal Trade Commission, the U.S. Drug Enforcement Administration and the U.S. Consumer Product Safety Commission, as well as regulations promulgated by comparable foreign, state and local governmental authorities concerning the operation of the Company's businesses. The Company is also subject to laws and regulations relating to licensing, tax, foreign trade, intellectual property, privacy and data protection, currency, political and other business restrictions.

The Company is also governed by national, state and local laws of general applicability in the countries in which it does business, including laws regulating matters of working conditions, health and safety and equal employment opportunity. In connection with the operation of its businesses, the Company is subject to laws and regulations relating to the protection of the environment and health and safety matters, including those governing exposure to, and the management and disposal of, hazardous substances. Environmental protection requirements did not have a material effect on the results of operations or capital expenditures of the Company in fiscal 2019.

Competitive conditions

The industries in which the Company operates are highly competitive. As a leader in the retail pharmacy industry and as a retailer of general merchandise, the Company competes with various local, regional, national and global retailers, including chain and independent pharmacies, mail order prescription providers, grocery stores, convenience stores, mass merchants, online and omni-channel pharmacies and retailers, warehouse clubs, dollar stores and other discount merchandisers. The Company's pharmaceutical wholesale businesses compete with other pharmaceutical wholesalers as well as alternative supply sources such as importers and manufacturers who supply directly to pharmacies. The Company competes primarily on the basis of service, convenience, variety and price. Its geographic dispersion helps mitigate the impact of temporary, localized economic and competitive conditions in individual markets. See "properties" in part I, item 2, below for further information regarding the Company's geographic dispersion.

Employees

As of August 31, 2019, the Company employed approximately 342,000 persons, approximately 110,000 of whom were part-time employees working less than 30 hours per week. The foregoing does not include employees of equity method investments.

Available information

The Company files with the Securities and Exchange Commission (the "SEC") its Annual Report on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and any amendments to those reports, as well as proxy statements and registration statements. In addition, the SEC maintains a website at http://www.sec.gov that contains reports, proxy and information statements and other information regarding issuers, including the Company, that file electronically. The Company makes available free of charge on or through its website at http://investor.walgreensbootsalliance.com its Annual Reports on Form 10-K, Quarterly Reports on Form 10-Q, Current Reports on Form 8-K and amendments to these reports filed or furnished pursuant to Section 13(a) or 15(d) of the Exchange Act of 1934, as amended (the "Exchange Act"), as soon as reasonably practicable after the Company files or furnishes them to the SEC. The contents of the website are not, however, a part of this Form 10-K or the Company's other SEC filings.

Item 1A. Risk factors

In addition to the other information in this report and our other filings with the SEC, you should carefully consider the risks described below, which could materially and adversely affect our business operations, financial condition and results of operations. These risks are not the only risks that we face. Our business operations could also be affected by additional factors that are not presently known to us or that we currently consider to be immaterial.

- 8 -

Reductions in third-party reimbursement levels, from private or governmental agency plans, and potential changes in industry pricing benchmarks for prescription drugs could materially and adversely affect our results of operations.

The substantial majority of the prescriptions we fill are reimbursed by third-party payers, including private and governmental agency payers. The continued efforts of health maintenance organizations, managed care organizations, PBM companies, governmental agencies, and other third-party payers to reduce prescription drug costs and pharmacy reimbursement rates, as well as litigation and other legal proceedings relating to how drugs are priced, may adversely impact our results of operations. In the United States, plan changes with rate adjustments often occur in January and our reimbursement arrangements may provide for rate adjustments at prescribed intervals during their term. In addition, in an environment where some PBM clients utilize narrow or restricted pharmacy provider networks, some of these entities may offer pricing terms that we may not be willing to accept or otherwise restrict our participation in their networks of pharmacy providers.

Changes in political, economic and regulatory influences also may significantly affect healthcare financing and prescription drug reimbursement practices. In the United States, for example, there have been multiple attempts through executive action, legislative action and legal challenges to modify or repeal the ACA, along with additional initiatives focused on increasing transparency and reducing costs in the U.S. healthcare system. We cannot predict whether current or future efforts to modify or repeal these laws will be successful, nor can we predict the impact that such a modification or repeal and any subsequent legislation would have on our business and reimbursement levels. There have also been a number of other proposals and enactments by the federal government and various states to reduce Medicare Part D and Medicaid reimbursement levels in response to budget deficits, and we expect additional proposals in the future. In the event that a third-party payer's budgetary or financial condition deteriorates, they may not be able to pay timely, or may delay payment of, amounts owed to us. There can be no assurance that recent or future changes in prescription drug reimbursement provides or subsidizes healthcare to consumers and regulates pharmaceutical prices, patient eligibility and reimbursement levels to control costs for the government-sponsored healthcare system. Efforts to control healthcare costs, including prescription drug costs, are continuous and reductions in third-party reimbursement levels could materially and adversely affect our results of operations.

In addition, many payers in the United States are increasingly considering new metrics as the basis for reimbursement rates. It is possible that the pharmaceutical industry or regulators may evaluate and/or develop an alternative pricing reference to replace average wholesale price, which is the pricing reference used for many of our contracts. In addition, many state Medicaid fee-for-service programs have established pharmacy network payments on the basis of actual acquisition cost, which could have an impact on reimbursement practices in other commercial and governmental arrangements. Future changes to the pricing benchmarks used to establish pharmaceutical pricing, including changes in the basis for calculating reimbursement by third-party payers, could adversely affect us.

A shift in pharmacy mix toward lower margin plans, products, and programs could adversely affect our results of operations.

Our Retail Pharmacy USA division seeks to grow prescription volume while operating in a marketplace with continuous reimbursement pressure. A shift in the mix of pharmacy prescription volume towards programs offering lower reimbursement rates could adversely affect our results of operations. For example, our Retail Pharmacy USA division has experienced a shift in pharmacy mix towards 90-day at retail in recent years, and specialty pharmacy represents a significant and growing proportion of prescription drug spending in the United States and a correspondingly larger proportion of our revenues. Our 90-day at retail offering for patients with chronic prescription needs typically is at a lower margin than comparable 30-day prescriptions, and specialty pharmacy sales are generally also lower margin. Our Retail Pharmacy USA division also has experienced a shift in pharmacy mix towards Medicare Part D prescriptions in recent years, and that trend may continue. Preferred Medicare Part D networks have increased in number in recent years; however, we do not participate in all such networks. We have accepted lower reimbursement rates in order to secure preferred relationships with Medicare Part D plans serving senior patients with significant pharmacy needs. We also have worked to develop and expand our relationships with commercial third-party payers to enable new and/or improved market access via participation in the pharmacy provider networks they offer. If we are not able to generate additional prescription volume and other business from patients participating in these programs that is sufficient to offset the impact of lower reimbursement, or if the degree or terms of our participation in such preferred networks declines from current levels in future years, our results of operations could be materially and adversely affected.

We derive a significant portion of our sales in the United States from prescription drug sales reimbursed by a limited number of pharmacy benefit management companies.

- 9 -

We derive a significant portion of our sales in the United States from prescription drug sales reimbursed through prescription drug plans administered by a limited number of PBM companies. PBM companies typically administer multiple prescription drug plans that expire at various times and provide for varying reimbursement rates, and often limit coverage to specific drug products on an approved list, known as a formulary, which might not include all of the approved drugs for a particular indication. Contracts with certain PBM companies that we have historically done substantial business with are due for renewal in fiscal 2020. There can be no assurance that we will continue to participate in any particular PBM company's pharmacy provider network in any particular future time period or on terms reasonably acceptable to us. If our participation in the pharmacy provider network for a prescription drug plan administered by one or more of the large PBM companies is restricted or terminated, we expect that our sales would be adversely affected, at least in the short-term. If we are unable to replace any such lost sales, either through an increase in other sales or through a resumption of participation in those plans, our operating results could be materially and adversely affected. If we exit a pharmacy provider network and later resume participation, there can be no assurance that we will achieve any particular level of business on any particular pace, or that all clients of the PBM company will choose to include us again in the pharmacy network for their plans, initially or at all. In addition, in such circumstances we may incur increased marketing and other costs in connection with initiatives to regain former patients and attract new patients covered by such plans.

We could be adversely affected by a decrease in the introduction of new brand name and generic prescription drugs as well as increases in the cost to procure prescription drugs.

The profitability of our pharmacy businesses depends upon the utilization of prescription drugs. Utilization trends are affected by, among other factors, the introduction of new and successful prescription drugs as well as lower-priced generic alternatives to existing brand name drugs. Inflation in the price of drugs also can adversely affect utilization, particularly given the increased prevalence of high-deductible health insurance plans and related plan design changes. New brand name drugs can result in increased drug utilization and associated sales, while the introduction of lower priced generic alternatives typically results in relatively lower sales, but relatively higher gross profit margins. Accordingly, a decrease in the number or magnitude of significant new brand name drugs or generics successfully introduced, delays in their introduction, or a decrease in the utilization of previously introduced prescription drugs, could materially and adversely affect our results of operations.

In addition, if we experience an increase in the amounts we pay to procure pharmaceutical drugs, including generic drugs, it could have a material adverse effect on our results of operations. Our gross profit margins would be adversely affected to the extent we are not able to offset such cost increases. Any failure to fully offset any such increased prices and costs or to modify our activities to mitigate the impact could have a material adverse effect on our results of operations. Additionally, any future changes in drug prices could be significantly different than our expectations.

Consolidation and strategic alliances in the healthcare industry could adversely affect our business operations, competitive positioning, financial condition and results of operations.

Many organizations in the healthcare industry, including PBM companies, have consolidated in recent years to create larger healthcare enterprises with greater bargaining power, which has resulted in greater pricing pressures. If this consolidation trend continues, it could give the resulting enterprises even greater bargaining power, which may lead to further pressure on the prices for our products and services. If these pressures result in reductions in our prices, our businesses would become less profitable unless we are able to achieve corresponding reductions in costs or develop profitable new revenue streams.

New and proposed acquisitions, partnerships and strategic alliances in the healthcare industry also can alter market dynamics and impact our businesses and competitive positioning. For example, in November 2018, CVS Health Corporation, an integrated pharmacy health care company that operates one of the largest retail drugstore chains and PBM companies in the United States, acquired Aetna, Inc., one of the largest diversified health care benefits companies. Changes in the participants in global sourcing enterprises relating to drug procurement, whether as a result of mergers, acquisitions or other transactions, can also have a similar effect on market dynamics and our business. In addition, further consolidation among generic drug manufacturers could lead to generic drug inflation in the future. We expect that market demand, government regulation, third-party reimbursement policies, government contracting requirements, and other pressures will continue to cause the healthcare industry to evolve, potentially resulting in further business consolidations and alliances and increased vertical integration among the industry participants we engage with, and which could, if we are not able to successfully anticipate and respond to evolving industry conditions in a timely and effective manner, materially and adversely impact our business operations, financial condition and results of operations.

Our growth strategy is partially dependent upon our ability to identify and successfully complete acquisitions, joint ventures and other strategic partnerships and alliances.

- 10 -

A significant element of our growth strategy is to identify, pursue and successfully complete and integrate acquisitions, joint ventures and other strategic partnerships and alliances that either expand or complement our existing operations. We have grown significantly through acquisitions in recent years and expect to continue to acquire, partner with or invest in businesses that build on or are deemed complementary to our existing businesses or further our strategic objectives. Due in part to consolidation in the industries in which we compete, there is significant competition for attractive targets and opportunities when available. There can be no assurance that attractive acquisition or other strategic relationship opportunities will be available, that we will be successful in identifying, negotiating and consummating favorable transaction opportunities, or that any such transactions we complete will be successful and justify our investment of financial and other resources.

Acquisitions and other strategic transactions involve numerous risks, including difficulties in successfully integrating the operations and personnel, navigating the necessary regulatory approval requirements, distraction of management from overseeing, and disruption of, our existing operations, difficulties in entering markets or lines of business in which we have no or limited direct prior experience, the possible loss of key employees and customers, and difficulties in achieving the synergies we anticipated. Any failure to select suitable opportunities at fair prices, conduct appropriate due diligence and successfully integrate the acquired company, including particularly when acquired businesses operate in new geographic markets or areas of business, could materially and adversely impact our financial condition and results of operations. These transactions may also cause us to significantly increase our interest expense, leverage and debt service requirements if we incur additional debt to pay for an acquisition or investment, issue common stock that would dilute our current stockholders' percentage ownership, or incur asset write-offs and restructuring costs and other related expenses that could have a material adverse impact on our operating results. Acquisitions, joint ventures and strategic investments also involve numerous other risks, including potential exposure to assumed litigation and unknown environmental and other liabilities, as well as undetected internal control, regulatory or other issues, or additional costs not anticipated at the time the transaction was completed. No assurance can be given that our acquisitions, joint ventures and diffect our business operations, financial condition or results of operations. If we are unable to successfully identify, complete and integrate acquisitions, joint ventures and strategic investments in a timely and effective manner, our business operations and growth strategies could be negatively affected.

Our strategic relationships include outsourcing and similar relationships. We outsource certain business and administrative functions and rely on third parties to perform certain services on our behalf. We rely on these third parties to meet our quality and performance requirements and to timely perform as expected. We periodically negotiate provisions and renewals of these relationships, and there can be no assurance that such terms will remain acceptable to us or such third parties. If our continuing relationship with certain third-party providers is interrupted, or if such third-party providers experience disruptions or do not perform as anticipated, or we experience problems with any transition, we may experience operational difficulties, reputational harm, and increased costs that could materially and adversely affect our business operations and results of operations.

We are exposed to risks related to litigation and other legal proceedings.

We operate in a highly regulated and litigious environment. We are involved in legal proceedings, including litigation, arbitration and other claims, and investigations, inspections, audits, claims, inquiries and similar actions by pharmacy, healthcare, tax and other governmental authorities, including those contained in note 10, commitments and contingencies, to the Consolidated Financial Statements included in part II, item 8 of this Form 10-K. For example, in January 2019, Walgreen Co., on behalf of itself, its subsidiaries and certain identified affiliates, resolved matters regarding certain dispensing practices by entering into, among other things, a Corporate Integrity Agreement with the Office of Inspector General of the United States Department of Health and Human Services. The Corporate Integrity Agreement has a five-year term and provides that Walgreen Co. shall, among other things, continue the compliance program it created to address compliance with federal health care program requirements, provide annual certifications of compliance and provide training and education for certain covered employees. Failure to meet the Corporate Integrity Agreement obligations could have material adverse consequences for us, including reputational harm and monetary penalties for each instance of non-compliance. In addition, in the event of a breach or deliberate violation of the Corporate Integrity Agreement, we could be excluded from participation in federal healthcare programs, or subjected to other significant penalties, which could seriously harm our results of operations, liquidity and financial results.

Legal proceedings, in general, and securities, derivative action and class action and multi-district litigation, in particular, can be expensive and disruptive. Some of these suits may purport or may be determined to be class actions and/or involve parties seeking large and/or indeterminate amounts, including punitive or exemplary damages, and may remain unresolved for several years. For example, we are a defendant in numerous litigation proceedings relating to opioids. From time to time, the Company is also involved in legal proceedings as a plaintiff involving antitrust, tax, contract, intellectual property and other matters. See note 10, commitments and contingencies, for more information.

- 11 -

The Company's financial results may also be adversely affected by the litigation and other legal proceedings of companies in which it has an equity method investment. Any unfavorable outcome or settlement related to these proceedings could have a material adverse effect on the Company's financial results.

Like other companies in the retail pharmacy and pharmaceutical wholesale industries, the Company is subject to extensive regulation by national, state and local government agencies in the United States and other countries in which it operates. There continues to be a heightened level of review and/or audit by regulatory authorities of, and increased litigation regarding, the Company's and the rest of the health care and related industry's business, compliance and reporting practices. As a result, the Company regularly is the subject of government actions of the types described above. In addition, under the qui tam or "whistleblower" provisions of the federal and various state false claims acts, persons may bring lawsuits alleging that a violation of the federal anti-kickback statute or similar laws has resulted in the submission of "false" claims to federal and/or state healthcare programs, including Medicare and Medicaid. After a private party has filed a qui tam action, the government must investigate the private party's claim and determine whether to intervene in and take control over the litigation. These actions may remain under seal while the government makes this determination.

We cannot predict with certainty the outcomes of these legal proceedings and other contingencies, and the costs incurred in litigation can be substantial, regardless of the outcome. Substantial unanticipated verdicts, fines and rulings do sometimes occur. As a result, we could from time to time incur judgments, enter into settlements or revise our expectations regarding the outcome of certain matters, and such developments could harm our reputation and have a material adverse effect on our results of operations in the period in which the amounts are accrued and/or our cash flows in the period in which the amounts are paid. In addition, as a result of governmental investigations or proceedings, the Company may be subject to damages, civil or criminal fines or penalties, or other sanctions, including the possible suspension or loss of licensure and/or suspension or exclusion from participation in government programs. The outcome of some of these legal proceedings and other contingencies could require us to take, or refrain from taking, actions which could negatively affect our operations. Additionally, defending against these lawsuits and proceedings may involve significant expense and diversion of management's attention and resources.

Our business results depend on our ability to successfully manage ongoing organizational change and business transformation and achieve cost savings and operating efficiency initiatives.

Our Board of Directors approved the plans to implement the Transformational Cost Management Program and the Store Optimization Program each as described in management's discussion and analysis of financial condition and results of operations in part II, item 7 below as part of an initiative to reduce costs and increase operating efficiencies. There can be no assurance that we will realize, in full or in part, the anticipated benefits of these programs. Our financial goals assume a level of productivity improvement, including those reflected in the Transformational Cost Management Program, the Store Optimization Program and other business optimization initiatives. If we are unable to deliver these expected productivity improvements, while continuing to invest in business growth, or if the volume and nature of change overwhelms available resources, our business operations and financial results could be materially and adversely impacted. Our ability to successfully manage and execute these initiatives and realize expected savings and benefits in the amounts and at the times anticipated is important to our business success. Any failure to do so, which could result from our inability to successfully execute organizational change and business transformation plans, changes in global or regional economic conditions, competition, changes in the industries in which we compete, unanticipated costs or charges, loss of key personnel and other factors described herein, could have a material adverse effect on our businesses, financial condition and results of operations.

The industries in which we operate are highly competitive and constantly evolving. New entrants to the market, existing competitor actions or other changes in market dynamics could adversely impact us.

The level of competition in the retail pharmacy and pharmaceutical wholesale industries is high. Changes in market dynamics or actions of competitors or manufacturers, including industry consolidation and the emergence of new competitors and strategic alliances, could materially and adversely impact us. Disruptive innovation, or the perception of potentially disruptive innovation, by existing or new competitors could alter the competitive landscape in the future and require us to accurately identify and assess such changes and if required make timely and effective changes to our strategies and business model to compete effectively. For example, in September 2018, online retailer Amazon.com, Inc. acquired PillPack, an online pharmacy with licenses throughout the United States. Some industry analysts have speculated that the PillPack acquisition could provide a platform for Amazon to significantly expand into the market for prescription drugs. Our retail pharmacy businesses face intense competition from local, regional, national and global companies, including other drugstore and pharmacy chains, independent drugstores and pharmacies and retailers, warehouse clubs, dollar stores and other discount merchandisers, some of which are aggressively expanding in markets we serve. Businesses in our Pharmaceutical Wholesale division face intense competition from direct competitors, including national and regional cooperative wholesalers, and

- 12 -

alternative supply sources such as importers and manufacturers who supply directly to pharmacies. Competition may also come from other sources in the future. As competition increases in the markets in which we operate, a significant increase in general pricing pressures could occur, including pressure on the convenience premium of our retail pricing model, which could require us to reevaluate our pricing structures to remain competitive. For example, if we are not able to anticipate and successfully respond to changes in market conditions in our Pharmaceutical Wholesale division, including changes driven by competitors, suppliers or manufacturers and increased competition from national and regional cooperative wholesalers, it could result in a loss of customers or renewal of contracts or arrangements on less favorable terms.

We also could be adversely affected if we fail to identify or effectively respond to changes in market dynamics. As technology, consumer behavior, omnichannel and differential retail models, and market conditions continue to evolve in the United States, it is important that we maintain the relevance of our brand and product and service offerings to customers and patients. In recent years, we announced that we are testing a number of concepts and initiatives designed to position our stores in the United States as convenient neighborhood health destinations for healthcare and retail products and services. The concepts being tested include new approaches to pricing and promotions, product selection, in-store and digital experience, and strategic partnerships that bring new products and services to our customers. We plan to use these pilots to listen to customers, learn and adjust based on feedback, with decisions on the nature and extent of further roll-outs made over time as we gain experience. If our customers are not receptive to these changes, if we are unable to expand successful programs in a timely manner, or we otherwise do not effectively respond to changes in market dynamics, our businesses and financial performance could be materially and adversely affected.

Additionally, specialty pharmacy represents a significant and growing proportion of prescription drug spending in the United States, a significant portion of which is dispensed outside of traditional retail pharmacies. Because our specialty pharmacy business focuses on complex and high-cost medications, many of which are made available by manufacturers to a limited number of pharmacies (so-called limited distribution drugs), that serve a relatively limited universe of patients, the future growth of this business depends to a significant extent upon expanding our ability to access key drugs and successfully penetrate key treatment categories. Accordingly, it is important that we and our affiliates compete effectively in this evolving market, or our business operations, financial condition and results of operations could be materially and adversely affected. To better serve this evolving market, in March 2017, we and Prime Therapeutics LLC, a PBM, closed a transaction to form a combined central specialty pharmacy and mail services company, AllianceRx Walgreens Prime, using an innovative model that seeks to align pharmacy, PBM and health plans to coordinate patient care, improve health outcomes and deliver cost of care opportunities. If this joint venture is not able to compete effectively in this evolving market and successfully adapt to changing market conditions, our business operations, financial condition and results of operations could be materially and adversely affected.

Our substantial international business operations subject us to a number of operating, economic, political, regulatory and other international business risks.

Our substantial international business operations are important to our growth and prospects, including particularly those of our Retail Pharmacy International and Pharmaceutical Wholesale divisions, and are subject to a number of risks, including:

- compliance with a wide variety of foreign laws and regulations, including retail and wholesale pharmacy, licensing, tax, foreign trade, intellectual
 property, privacy and data protection, immigration, currency, political and other business restrictions and requirements and local laws and
 regulations, whose interpretation and enforcement vary significantly among jurisdictions and can change significantly over time;
- additional U.S. and other regulation of non-domestic operations, including regulation under the Foreign Corrupt Practices Act, the UK Bribery Act and other anti-corruption laws;
- potential difficulties in managing foreign operations, mitigating credit risks in foreign markets, enforcing agreements and collecting receivables through foreign legal systems;
- price controls imposed by foreign countries, tariffs, duties or other restrictions on foreign currencies or trade sanctions and other trade barriers imposed by foreign countries that restrict or prohibit business transactions in certain markets, or fluctuations in currency rates;
- impact of recessions and economic slowdowns in economies outside the United States, including foreign currency devaluation, higher interest rates, inflation, and increased government regulation or ownership of traditional private businesses;

- 13 -

- the instability of foreign economies, governments and currencies and unexpected regulatory, economic or political changes in foreign markets; and
- developing and emerging markets may be especially vulnerable to periods of instability and unexpected changes, and consumers in those markets may have relatively limited resources to spend on products and services.

These factors can also adversely affect our payers, vendors and customers in international markets, which in turn can negatively impact our businesses. We cannot assure you that one or more of these factors will not have a material adverse effect on our business operations, results of operation and financial condition.

Many of these factors are subject to change based on changes in political, economic and regulatory influences. For example:

- Our Retail Pharmacy International and Pharmaceutical Wholesale divisions have substantial operations in the United Kingdom and other member countries of the European Union. In June 2016, voters in the United Kingdom approved an advisory referred to as "Brexit," has exacerbated and may further exacerbate many of the risks and uncertainties described above. An original exit date was set for March 29, 2019, but following the UK parliament's rejection of a negotiated outcome, the leaders of the member countries of the European Union have agreed to multiple extensions of the deadline for Brexit. Given the complexity and uncertainty surrounding the potential withdrawal and negotiations, including with respect to terms of trade and customs, there can be no assurance regarding the terms, timing or consummation of any such arrangements, and whether there will be an orderly withdrawal from the European Union or a so-called "hard Brexit" with no continuing UK participation in the EU's Single Market or the EU Customs Union. The proposed withdrawal could, among other potential outcomes, adversely affect the tax, tax treaty, currency, operational, legal and regulatory regimes to which our businesses in the region are subject. The withdrawal could also, among other potential outcomes, disrupt the free movement of goods, services and people between the United Kingdom and the European Union and other parties. Further, uncertainty around and developments regarding the economy of the United Kingdom and the economies of other countries in which we operate and cause significant volatility in currency exchange rates. In the event of a "hard Brexit," the related risks and uncertainties could be further exacerbated; and
- Many of the products we sell are manufactured in whole or in part outside of the United States. In some cases, these products are imported by others and sold to us. In the United States, the Presidential Administration has discussed, and in some cases implemented, changes with respect to certain tax and trade policies, tariffs and other government regulations affecting trade between the United States and other countries. For example, concerns regarding trade relations between the United States and China have escalated during fiscal 2019 and discussions have continued through the date of this report, as both countries imposed significant tariffs on the importation of certain product categories. As a significant portion of our retail products are sourced from China, the imposition on the United States of these and any additional new tariffs on certain goods imported from China have adversely affected costs and could impact the profitability of retail product sales in our Retail Pharmacy USA division. While it is not possible to predict whether or when any additional changes will occur or what form they may take, significant changes in tax or trade policies, tariffs or trade relations between the United States and other countries could result in significant increases in our costs, restrict our access to certain suppliers and adversely impact economic activity. In addition, other countries may change their business and trade policies in anticipation of or in response to increased import tariffs and other changes in United States' trade policy and regulations.

There can be no assurance that any or all of these developments will not have a material adverse effect on our business operations, results of operations and financial condition.

We are exposed to risks associated with foreign currency exchange rate fluctuations.

We operate in 25 countries across the globe, and our significant operations outside of the United States expose us to currency exchange rate fluctuations and related risks, including transaction currency exposures relating to the import and export of goods in currencies other than businesses' functional currencies as well as currency translation exposures relating to profits and net assets denominated in currencies other than the U.S. dollar. We present our financial statements in U.S. dollars and have a significant proportion of net assets and income in non-U.S. dollar currencies, primarily the British pound sterling and the Euro, as well as a range of other foreign currencies. Our results of operations and capital ratios can therefore be sensitive to movements in foreign exchange rates. Due to the constantly changing currency exposures to which we are subject and the

- 14 -

volatility of currency exchange rates, we cannot predict the effect of exchange rate fluctuations upon our future results of operations. In addition, fluctuations in currencies relative to the U.S. dollar may make it more difficult to perform period-to-period comparisons of our reported results of operations. A depreciation of non-U.S. dollar currencies relative to the U.S. dollar could have a significant adverse impact on our results of operations.

We may from time to time, in some instances, enter into foreign currency contracts or other derivative instruments intended to hedge a portion of our foreign currency fluctuation risks, which subjects us to additional risks, such as the risk that counterparties may fail to honor their obligations to us, that could materially and adversely affect us. Additionally, we may (and currently do) use foreign currency debt to hedge some of our foreign currency fluctuation risks. The periodic use of such hedging activities may not offset any or more than a portion of the adverse financial effects of unfavorable movements in foreign exchange rates over the limited time the hedges are in place. We cannot assure you that fluctuations in foreign currency exchange rates, including particularly the strengthening of the U.S. dollar against major currencies or the currencies of large developing countries, will not materially affect our consolidated financial results.

We may experience a significant disruption in our information technology and computer systems.

We rely extensively on our computer systems to manage our ordering, pricing, point-of-sale, pharmacy fulfillment, inventory replenishment, customer loyalty programs, finance and other processes. Our systems are subject to damage or interruption from power outages, facility damage, computer and telecommunications failures, computer viruses, security breaches including credit card or personally identifiable information breaches, vandalism, natural disasters, catastrophic events, human error and potential cyber threats, including malicious codes, worms, phishing attacks, denial of service attacks, ransomware and other sophisticated cyber attacks, and our disaster recovery planning cannot account for all eventualities. If any of our systems are damaged, fail to function properly or otherwise become unavailable, we may incur substantial costs to repair or replace them, and may experience loss or corruption of critical data and interruptions or disruptions and delays in our ability to perform critical functions, which could materially and adversely affect our businesses and results of operations.

In addition, we are currently making, and expect to continue to make, substantial investments in our information technology systems and infrastructure, some of which are significant. Upgrades involve replacing existing systems with successor systems, outsourcing critical information technology to third parties, making changes to existing systems, including the migration of applications to the cloud, or cost-effectively acquiring new systems with new functionality. Implementing new systems carries significant potential risks, including failure to operate as designed, potential loss or corruption of data or information, changes in security processes, cost overruns, implementation delays, disruption of operations, and the potential inability to meet business and reporting requirements. We rely on strategic partners and other service providers to help us with certain significant information technology projects and services. Information technology projects or services frequently are long-term in nature and may take longer to complete and cost more than we expect and may not deliver the benefits we project once they are complete. There can be no assurance that we and our strategic partners and service providers will be able to deliver the required functionality of our information technology functions and there can be no assurance that we and our strategic partners and service providers will be able to systems and outsourcing critical information technology functions and there can be no assurance that we will not experience significant issues with replacing these systems prior to implementation, that our technology functions and there can be no assurance that we will not experience significant issues with our eexisting systems prior to implementation, that our technology initiatives will be successfully deployed as planned or that they will be timely implemented without significant disruption to our operations. We also could be adversely affected by any significant disruption in the systems of third parties we interac

If we or the businesses we interact with do not maintain the privacy and security of sensitive customer and business information, it could damage our reputation and we could suffer a loss of revenue, incur substantial additional costs and become subject to litigation and regulatory scrutiny.

The protection of customer, employee, and company data is critical to our businesses. Cybersecurity and other information technology security risks, such as a significant breach of customer, employee, or company data, could create significant workflow disruption, attract a substantial amount of media attention, damage our customer relationships, reputation and brand, and result in lost sales, fines or lawsuits. Throughout our operations, we receive, retain and transmit certain personal information that our customers and others provide to purchase products or services, fill prescriptions, enroll in promotional programs, participate in our customer loyalty programs, register on our websites, or otherwise communicate and interact with us. In addition, aspects of our operations depend upon the secure transmission of confidential information over public networks. Like other global companies, we and businesses we interact with have experienced threats to data and systems, including by perpetrators of random or targeted malicious cyber-attacks, computer viruses, worms, bot attacks or other

- 15 -

destructive or disruptive software and attempts to misappropriate customer information, including credit card information, and cause system failures and disruptions. Any compromise of our data security systems or of those of businesses with whom we interact, which results in confidential information being accessed, obtained, damaged or used by unauthorized or improper persons, could harm our reputation and expose us to regulatory actions, customer attrition, remediation expenses, and claims from customers, financial institutions, payment card associations and other persons, any of which could materially and adversely affect our business operations, financial condition and results of operations. Because the techniques used to obtain unauthorized access, disable or degrade service, or sabotage systems change frequently and may not immediately produce signs of intrusion, we may be unable to anticipate these techniques or to implement adequate preventative measures. In addition, a security breach could require that we expend substantial additional resources related to the security of information systems and disrupt our businesses.

We depend on and interact with the information technology networks and systems of third-parties for many aspects of our business operations, including payers, strategic partners and cloud service providers. These third parties may have access to information we maintain about our company, operations, customers, employees and vendors, or operating systems that are critical to or can significantly impact our business operations. Like us, these third-parties are subject to risks imposed by data breaches and cyber-attacks and other events or actions that could damage, disrupt or close down their networks or systems. Any expansion of information technology outsourcing, including through arrangements with our strategic partners, may increase vulnerabilities and weaknesses relating to cybersecurity and data management. Security processes, protocols and standards that we have implemented and contractual provisions requiring security measures that we may have sought to impose on such third-parties may not be sufficient or effective at preventing such events, which could result in unauthorized access to, or disruptions or denials of access to, or misuse of, information or systems that are important to our business, including proprietary information, sensitive or confidential data, and other information about our operations, customers, employees and suppliers, including personal information.

The regulatory environment surrounding data security and privacy is increasingly demanding, with the frequent imposition of new and changing requirements across businesses and geographic areas. We are required to comply with increasingly complex and changing data security and privacy regulations in the United States and in other jurisdictions in which we operate that regulate the collection, use and transfer of personal data, including the transfer of personal data between or among countries. In the United States, for example, HIPAA imposes extensive privacy and security requirements governing the transmission, use and disclosure of health information by covered entities in the health care industry, including health care providers such as pharmacies. In addition, the California Consumer Privacy Act, which goes into effect on January 1, 2020, imposes stringent requirements on the use and treatment of "personal information" of California residents, which term is broadly defined to include, among other things, information that identifies, relates to, describes, is capable of being associated with, or could reasonably be linked to a consumer or household. Other U.S. states have enacted, or are proposing similar laws related to the protection of personal data. In addition, the U.S. federal government is considering federal privacy legislation. Outside the United States, many of our business units operate in countries with stringent data protection regulations, and these laws continue to change. For example, the European Union's General Data Protection Regulation, which became effective in May 2018, greatly increased the jurisdictional reach of European Union data protection laws and added a broad array of requirements for handling personal data, including the public disclosure of significant data breaches, and provides for greater penalties for noncompliance. Other countries have enacted or are considering enacting data localization laws that require certain data to stay within their borders. Complying with changing regulatory requirements requires us to incur substantial costs and may require changes to our business practices in certain jurisdictions, any of which could materially and adversely affect our business operations and operating results. We may also face audits or investigations by one or more domestic or foreign government agencies relating to our compliance with these regulations. Compliance with changes in privacy and information security laws and standards may result in significant expense due to increased investment in technology and the development of new operational processes. If we or those with whom we share information fail to comply with these laws and regulations or experience a data security breach, our reputation could be damaged and we could be subject to additional litigation and regulatory risks, particularly to the extent the breach relates to sensitive data. Our security measures may be undermined due to the actions of outside parties, employee error, malfeasance, or otherwise, and, as a result, an unauthorized party may obtain access to our data systems and misappropriate business and personal information. Any such breach or unauthorized access could result in significant legal and financial exposure, damage to our reputation and credibility, and potentially have a material adverse effect on our business operations, financial condition and results of operations.

A significant change in, or noncompliance with, governmental regulations and other legal requirements could have a material adverse effect on our reputation and profitability.

We operate in complex, highly regulated environments around the world and could be materially and adversely affected by changes to applicable legal requirements including the related interpretations and enforcement practices, new legal requirements and/or any failure to comply with applicable regulations. Businesses in our Pharmaceutical Wholesale division

- 16 -

are subject to a range of regulations relating to such things as product margins, product traceability and the conditions under which products must be stored. Our retail pharmacy and health and wellness services businesses are subject to numerous country, state and local regulations including licensing, billing practices, utilization and other requirements for pharmacies and reimbursement arrangements. The regulations to which we are subject include, but are not limited to: country and state registration and regulation of pharmacies and drug discount card programs; dispensing and sale of controlled substances and products containing pseudoephedrine; applicable governmental payer regulations including Medicare and Medicaid; data privacy and security laws and regulations including HIPAA; the ACA or any successor thereto; laws and regulations relating to the protection of the environment and health and safety matters, each of which continues to evolve, including those governing exposure to, and the management and disposal of, hazardous substances; regulations regarding food and drug safety including those of the U.S. Food and Drug Administration ("FDA") and Drug Enforcement Administration ("DEA"), trade regulations including those of the U.S. Federal Trade Commission, and consumer protection and safety regulations including those of the Consumer Product Safety Commission, as well as state regulatory authorities, governing the availability, sale, advertisement and promotion of products we sell as well as our lovalty and drug discount card programs; anti-kickback laws; false claims laws; laws against the corporate practice of medicine; and foreign, national and state laws governing health care fraud and abuse and the practice of the profession of pharmacy. For example, in the United States, the DEA, FDA and various other regulatory authorities regulate the distribution and dispensing of pharmaceuticals and controlled substances. We are required to hold valid DEA and state-level licenses, meet various security and operating standards and comply with the federal and various state controlled substance acts and related regulations governing the sale, dispensing, disposal, holding and distribution of controlled substances. The DEA, FDA and state regulatory authorities have broad enforcement powers, including the ability to seize or recall products and impose significant criminal, civil and administrative sanctions for violations of these laws and regulations. We are also governed by foreign, national and state laws of general applicability, including laws regulating matters of working conditions, health and safety and equal employment opportunity and other labor and employment matters as well as employee benefit, competition and antitrust matters. In addition, we could have significant exposure if we are found to have infringed another party's intellectual property rights.

Changes in laws, regulations and policies and the related interpretations and enforcement practices may alter the landscape in which we do business and may significantly affect our cost of doing business. The impact of new laws, regulations and policies and the related interpretations and enforcement practices generally cannot be predicted, and changes in applicable laws, regulations and policies and the related interpretations and enforcement practices may require extensive system and operational changes, be difficult to implement, increase our operating costs and require significant capital expenditures. Untimely compliance or noncompliance with applicable laws and regulations could result in the imposition of civil and criminal penalties that could adversely affect the continued operation of our businesses, including: suspension of payments from government programs; loss of required government certifications; loss of authorizations to participate in or exclusion from government programs, including the Medicare and Medicaid programs in the United States and the National Health Service in the United Kingdom; loss of licenses; and significant fines or monetary penalties. Any failure to comply with applicable regulatory requirements in the United States or in any of the countries in which we operate could result in significant legal and financial exposure, damage to our reputation and brand, and have a material adverse effect on our business operations, financial condition and results of operations.

We could be adversely affected by violations of anti-bribery, anti-corruption and/or international trade laws.

We are subject to laws concerning our business operations and marketing activities in foreign countries where we conduct business. For example, we are subject to the U.S. Foreign Corrupt Practices Act (the "FCPA"), U.S. export control, anti-money laundering and economic and trade sanction laws, and similar anti-corruption and international trade laws in certain foreign countries, such as the UK Bribery Act, any violation of which could create substantial liability for us and also harm our reputation. The FCPA generally prohibits U.S. companies and their officers, directors, employees, and intermediaries from making improper payments to foreign officials for the purpose of obtaining or retaining business abroad or otherwise obtaining favorable treatment. The FCPA also requires that U.S. public companies maintain books and records that fairly and accurately reflect transactions and maintain an adequate system of internal accounting controls. If we are found to have violated the FCPA, U.S. export control, economic and trade sanctions, or anti-money laundering laws or any other anti-bribery, anti-corruption or international trade laws or to have otherwise conducted business activities inappropriately or unethically, including through acts of bribery, corruption, conflicts of interest, fraud, manipulation of profits, and suspension or debarment of our ability to contract with governmental agencies or receive export licenses. In addition, new initiatives may be proposed from time to time that impact the trading conditions in certain countries or regions, and may include retaliatory duties or trade sanctions. From time to time, we may face audits or investigations by one or more domestic or foreign governmental agencies relating to our international business activities, compliance with which could be costly and time-consuming, and could divert our management and key personnel from our

- 17 -

business operations. An adverse outcome under any such investigation or audit could damage our reputation and subject us to fines or other penalties, which could materially and adversely affect our business operations, financial condition, and results of operations.

Disruption in our global supply chain could negatively impact our businesses.

The products we sell are sourced from a wide variety of domestic and international vendors, and any future disruption in our supply chain or inability to find qualified vendors and access products that meet requisite quality and safety standards in a timely and efficient manner could adversely impact our businesses. The loss or disruption of such supply arrangements for any reason, including for issues such as labor disputes, loss or impairment of key manufacturing sites, inability to procure sufficient raw materials, quality control issues, ethical sourcing issues, a supplier's financial distress, natural disasters, civil unrest or acts of war or terrorism, trade sanctions or other external factors over which we have no control, could interrupt product supply and, if not effectively managed and remedied, have a material adverse impact on our business operations, financial condition and results of operations.

We use a single wholesaler of branded and generic pharmaceutical drugs as our primary source of such products for our Retail Pharmacy USA division.

In March 2013, Walgreens, Alliance Boots and AmerisourceBergen announced various agreements and arrangements, including a ten-year pharmaceutical distribution agreement between Walgreens and AmerisourceBergen pursuant to which we source branded and generic pharmaceutical products from AmerisourceBergen in the United States and an agreement which provides AmerisourceBergen the ability to access generic pharmaceutical products through our global sourcing enterprise. In May 2016, certain of these agreements were extended for three years to now expire in 2026. In addition, in March 2013, Walgreens, Alliance Boots and AmerisourceBergen entered into agreements and arrangements pursuant to which we have the right, but not the obligation, to purchase a minority equity position in AmerisourceBergen and gain associated representation on AmerisourceBergen's board of directors in certain circumstances. As of the date of this report, AmerisourceBergen distributes for our Retail Pharmacy USA division substantially all branded and generic pharmaceutical products. Consequently, our business in the United States may be adversely affected by any operational, financial or regulatory difficulties that AmerisourceBergen experiences. For example, if AmerisourceBergen's operations are seriously disrupted for any reason, whether due to a natural disaster, labor disruption, regulatory action, computer or operational systems or otherwise, it could adversely affect our business in the United States and our results of operations.

Our distribution agreement with AmerisourceBergen is subject to early termination in certain circumstances and, upon the expiration or termination of the agreement, there can be no assurance that we or AmerisourceBergen will be willing to renew the agreement or enter into a new agreement, on terms favorable to us or at all. If such expiration or termination occurred, we believe that alternative sources of supply for most generic and brand-name pharmaceuticals are readily available and that we could obtain and qualify alternative sources, which may include self-distribution in some cases, for substantially all of the prescription drugs we sell on an acceptable basis, such that the impact of any such expiration or termination would be temporary. However, there can be no assurance we would be able to engage alternative supply sources or implement self-distribution processes on a timely basis or on terms favorable to us, or effectively manage these transitions, any of which could adversely affect our business operations, financial condition and results of operations.

The anticipated strategic and financial benefits of our relationship with AmerisourceBergen may not be realized.

As of August 31, 2019, we beneficially owned approximately 27% of the outstanding AmerisourceBergen common stock and had designated one member of AmerisourceBergen's board of directors. In addition, we have the right, but not the obligation, under the transactions contemplated by the Framework Agreement dated as of March 18, 2013 by and among the Company, Alliance Boots and AmerisourceBergen (the "Framework Agreement") to acquire up to an additional 8,398,752 AmerisourceBergen shares in the open market and thereafter designate another member of AmerisourceBergen's board of directors, subject in each case to applicable legal and contractual requirements. There can be no assurance that we will complete any specific level of such potential equity investments in AmerisourceBergen, or that our existing investment, or any future investment if completed, will ultimately be profitable.

The Company accounts for its investment in AmerisourceBergen using the equity method of accounting, subject to a two-month reporting lag, with the net earnings attributable to the investment classified within the operating income of the Company's Pharmaceutical Wholesale segment. The financial performance of AmerisourceBergen, including any charges which may arise relating to its ongoing opioid litigation, will impact the Company's results of operations. Additionally, a substantial and sustained decline in the price of AmerisourceBergen's common stock could trigger an impairment evaluation of

- 18 -

our investment. These considerations may materially and adversely affect the Company's financial condition and results of operations.

Further, our ability to transact in AmerisourceBergen securities is subject to certain restrictions set forth in our agreements with AmerisourceBergen and arising under applicable laws and regulations, which in some circumstances could adversely our ability to transact in AmerisourceBergen securities in amounts and at the times desired. We could also encounter unforeseen costs, circumstances or issues existing or arising with respect to the transactions and collaboration resulting from these agreements. Many of these potential circumstances are outside of our control and any of them could result in increased costs, decreased revenue, decreased synergies and the diversion of management time and attention. If we are unable to achieve our objectives within the anticipated time frame, or at all, the expected benefits may not be realized fully or at all, or may take longer to realize than expected, which could have a material adverse impact on our business operations, financial condition and results of operations.

We may not realize the anticipated benefits of the acquisition of assets from Rite Aid, which could adversely impact our results of operations.

We entered into an agreement to acquire certain Rite Aid stores and distribution centers with the expectation that the transaction will result in various benefits, including, among other things, cost savings and operating efficiencies. The achievement of the anticipated benefits of the transaction is subject to a number of uncertainties, including completion of the pending acquisition of distribution centers and related inventory expected to be completed during fiscal 2020, whether the acquired assets can be integrated into our business in an efficient and effective manner, the possibility of faulty assumptions underlying expectations regarding potential synergies and the integration process, unforeseen expenses or delays, and competitive factors in the marketplace. We can provide no assurance that the anticipated benefits of the transaction, including cost savings and synergies, will be fully realized in the time frame anticipated or at all; the costs or difficulties related to the integration of the acquired assets into our business and operations will not be greater than expected; unanticipated costs, charges and expenses will not result from the transaction; litigation relating to the transaction will not be filed; and the transaction will not cause disruption to the parties' business and operations and relationships with employees and suppliers, payers, customers and other third parties. If one or more of these risks are realized, it could have a material adverse impact on our operating results.

From time to time, we make investments in companies over which we do not have sole control. Some of these companies may operate in sectors that differ from our current operations and have different risks.

From time to time, we make debt or equity investments in companies that we may not control or over which we may not have sole control. For example, while we have a significant equity investment in AmerisourceBergen and have a designee serving on the board of directors of AmerisourceBergen as of the date of this report, we do not have the ability to control day-to-day operations of that company. Some of the businesses in which we have made noncontrolling investments operate in markets or industries that are different from our primary lines of business and/or operate in different geographic markets than we do. For example we have an equity investment in Sinopharm Holding GuoDa Drugstores Co., Ltd., a leading retail pharmacy chain in China. Investments in these businesses, among other risks, subject us to the operating and financial risks of the businesses we invest in and to the risk that we do not have sole control over the operations of these businesses. We rely on the internal controls and financial reporting controls of these entities and their failure to maintain effectiveness or comply with applicable standards may materially and adversely affect us. Investments in entities over which we do not have sole control, including joint ventures and strategic partnerships and alliances, present additional risks such as having differing objectives from our partners or the entities in which we are invested, becoming involved in disputes, or competing with those persons.

Changes in economic conditions could adversely affect consumer buying practices.

Our performance has been, and may continue to be, adversely impacted by changes in global, national, regional or local economic conditions and consumer confidence. These conditions can also adversely affect our key vendors and customers. External factors that affect consumer confidence and over which we exercise no influence include unemployment rates, inflation, levels of personal disposable income, levels of taxes and interest and global, national, regional or local economic conditions, as well as acts of war or terrorism. Changes in economic conditions and consumer confidence could adversely affect consumer preferences, purchasing power and spending patterns, which could lead to a decrease in overall consumer spending as well as in prescription drug and health services utilization and which could be exacerbated by the increasing prevalence of high-deductible health insurance plans and related plan design changes. Our business operations have been adversely affected by challenging market conditions in our Retail Pharmacy International division, particularly our Boots business in the United Kingdom, on account of the economic environment, including the uncertainty associated with Brexit.

- 19 -

In addition, reduced or flat consumer spending may drive us and our competitors to offer additional products at promotional prices. All of these factors could materially and adversely impact our business operations, financial condition and results of operations.

Economic conditions in Europe and certain emerging market countries, together with austerity measures being taken by certain governments, could adversely affect us.

We have significant assets and operations within Europe and certain emerging market countries in our Retail Pharmacy International and Pharmaceutical Wholesale divisions. An economic slowdown within any such markets could adversely affect our businesses in affected regions by reducing the prices our customers may be able or willing to pay for our products and services or by reducing the demand for our products and services, either of which could result in a material adverse impact on our results of operations. In recent years, in response to the economic environment, a number of governments, including the government in the United Kingdom, have announced or implemented austerity measures to reduce healthcare spending for the government-sponsored healthcare systems and constrain overall government expenditures. These measures, which include efforts aimed at reforming healthcare coverage and reducing healthcare costs, continue to exert pressure on the pricing of and reimbursement timelines for pharmaceutical drugs. Any new austerity measures may be similar to or vary in scope and nature from existing austerity measures and could have a material adverse effect on our international business operations and results of operations.

If we do not successfully develop and maintain a relevant omni-channel experience for our customers, our businesses and results of operations could be adversely impacted.

The portion of total consumer expenditures with retailers occurring online and through mobile applications has continued to increase and the pace of this increase could accelerate in the future. Our business has evolved from an in-store experience to interaction with customers across numerous channels, including in-store, online, mobile and social media, among others. Omni-channel and differentiated retail models are rapidly evolving and we must keep pace with changing customer expectations and new developments by our competitors. Our customers are increasingly using computers, tablets, mobile phones, and other devices to comparison shop, determine product availability and complete purchases, as well as to provide immediate public reactions regarding various facets of our operations. We must compete by offering a consistent and convenient shopping experience for our customers regardless of the ultimate sales channel and by investing in, providing and maintaining digital tools for our customers that have the right features and are reliable and easy to use. If we are unable to make, improve, or develop relevant customer-facing technology in a timely manner that keeps pace with technological developments and dynamic customer expectations, our ability to compete and our results of operations could be materially and adversely affected. In addition, if our online activities or our other customer-facing technology systems do not function as designed, we may experience a loss of customer confidence, data security breaches, lost sales, or be exposed to fraudulent purchases, any of which could materially and adversely affect our business operations, reputation and results of operations.

If the merchandise and services that we offer fail to meet customer needs, our sales may be adversely affected.

We could be adversely affected by changes in consumer spending levels and shopping habits and preferences, including attitudes towards our retail and product brands. The success of our retail pharmacy businesses depends on our ability to offer a superior shopping experience, engaging customer service and a quality assortment of available merchandise that differentiates us from other retailers, including enhanced health and beauty product offerings. We must identify, obtain supplies of, and offer to our customers attractive, innovative and high-quality merchandise on a continuous basis. Our products and services must satisfy the needs and desires of our customers, whose preferences may change in the future. For example, our proof of concept initiatives that seek to position our stores in the United States as convenient neighborhood health destinations for healthcare and retail products and services reflect the perceived desires and needs of our target market. However, it is difficult to predict consistently and successfully the products and services our customers will demand. If we misjudge either the demand for products and services we sell or our customers' purchasing habits and tastes, we may be faced with excess inventories of some products and missed opportunities for products and services we chose not to offer. In addition, our sales may decline or we may be required to sell the merchandise we have obtained at lower prices. Failure to timely identify or effectively respond to changing consumer tastes, preferences and spending patterns and evolving demographic mixes in the markets we serve could negatively affect our relationship with our customers and the demand for our products and services, which could materially and adversely impact our results of operations.

Our private brand offerings expose us to various additional risks.

In addition to brand name products, we offer our customers private brand products that in many instances, are not available from other retailers. We seek to continue to grow our private brand offerings as part of our growth strategy, including through



the offering of No7, the introduction of Complete Home and other brands owned or licensed on an exclusive basis, as well as through selective acquisitions. Maintaining consistent product quality, competitive pricing, and availability of our private brand offerings for our customers, as well as the timely development and introduction of new products, is important in differentiating us from other retailers and developing and maintaining customer loyalty. The expansion of our private brand offerings also subjects us to additional risks, such as potential product liability risks and mandatory or voluntary product recalls; our ability to successfully protect our proprietary rights and successfully navigate and avoid claims related to the proprietary rights of third parties; our ability to successfully administer and comply with applicable contractual obligations and regulatory requirements; and other risks generally encountered by entities that source, sell and market exclusive branded offerings for retail. An increase in sales of our private brands may also adversely affect sales of our vendors' products, which, in turn, could adversely affect our relationship with certain of our vendors. Any failure to adequately address some or all of these risks could have a material adverse effect on our reputation, business operations, results of operations and financial condition.

We face significant competition in attracting and retaining talented employees. Further, managing succession for, and retention of, key executives is critical to our success, and failure to do so could have an adverse impact on our future performance.

Our ability to attract, engage and retain qualified and experienced employees, including in executive and other key strategic positions, is essential to meet current and future goals and objectives and there is no guarantee we will be able to attract and retain such employees by offering market competitive compensation, opportunities, empowerment and resources. There can be no assurance that competition among potential employers will not result in increased salaries or other benefits. An inability to retain existing employees or attract additional employees, or an unexpected loss of leadership or loss of key personnel, could have a material adverse effect on our business and results of operations.

In addition, failure to adequately plan for succession of senior management and other key management roles or the failure of key employees to successfully transition into new roles could have a material adverse effect on our business and results of operations. While we have succession plans in place and employment arrangements with certain key executives, these do not guarantee the services of these executives will continue to be available to us.

We are subject to payment-related and other financial services risks that could increase our operating costs, expose us to fraud or theft, subject us to potential liability and potentially disrupt our business operations.

We accept payments using a variety of methods, including cash, checks, credit and debit cards, gift cards and mobile payment technologies such as Apple Pay[™], and we may offer new payment options over time. Acceptance of these payment options subjects us to rules, regulations, contractual obligations and compliance requirements, including payment network rules and operating guidelines, data security standards and certification requirements, and rules governing electronic funds transfers. These requirements and related interpretations may change over time, which could make compliance more difficult or costly. For certain payment methods, including credit and debit cards, we pay interchange and other fees, which could increase over time and raise our operating costs. We rely on third parties to provide payment processing services, including the processing of credit cards, debit cards, and other forms of electronic payment. If these companies become unable to provide these services to us, or if their systems are compromised, it could disrupt our business. The payment methods that we offer also subject us to potential fraud and theft by persons who seek to obtain unauthorized access to or exploit any weaknesses that may exist in the payment systems. If we fail to comply with applicable rules or requirements, or if data is compromised due to a breach or misuse of data relating to our payment systems, we may be liable for costs incurred by payment card issuing banks and other third parties or subject to fines and higher transaction fees, or our ability to accept or facilitate certain types of payments could be impaired. In addition, our reputation could suffer and our customers could lose confidence in certain payment types, which could result in higher costs and/or reduced sales and materially and adversely affect our results of operations.

Additionally, we offer money (wire) transfer services and sell prepaid debit, credit and gift cards at certain business units. These products and services require us to comply with global anti-money laundering laws and regulations. Failure to comply with these laws and regulations could result in fines, sanctions, penalties and damage to our reputation.

Changes in healthcare regulatory environments may adversely affect our businesses.

Political, economic and regulatory influences are subjecting the healthcare industry to significant changes that could adversely affect our results of operations. In recent years, the healthcare industry has undergone significant changes in an effort to reduce costs and government spending. These changes include an increased reliance on managed care; cuts in certain Medicare and Medicaid funding in the United States and the funding of governmental payers in foreign jurisdictions; consolidation of competitors, suppliers and other market participants; and the development of large, sophisticated purchasing groups. We expect the healthcare industry to continue to change significantly in the future. Some of these potential changes, such as a reduction in



governmental funding for certain healthcare services or adverse changes in legislation or regulations governing prescription drug pricing, healthcare services or mandated benefits, may cause customers to reduce the amount of our products and services they purchase or the price they are willing to pay for our products and services. We expect continued governmental and private payer pressure to reduce pharmaceutical pricing. Changes in pharmaceutical manufacturers' pricing or distribution policies could also significantly reduce our profitability.

In the United States, electoral results and changes in political leadership have generated uncertainty with respect to, and could result in, significant changes in legislation, regulation and government policy that could significantly impact our businesses and the health care and retail industries. There have been multiple attempts to repeal, modify or otherwise invalidate all, or certain provisions of, the ACA, which was enacted in 2010 to provide health insurance coverage to millions of previously uninsured Americans through a combination of insurance market reforms, an expansion of Medicaid, subsidies and health insurance mandates. We cannot predict whether current or future efforts to modify these laws and/or adopt new healthcare legislation will be successful, nor can we predict the impact that such a development would have on our business and operating results. Future legislation or rulemaking or other regulatory actions or developments under the ACA or otherwise could adversely impact the number of Americans with health insurance and, consequently, prescription drug coverage, increase regulation of pharmacy services, result in changes to pharmacy reimbursement rates, and otherwise change the way we do business. We cannot predict the timing or impact of any future legislative, rulemaking or other regulatory actions, but any such actions could have a material adverse impact on our results of operations.

We could be adversely affected by product liability, product recall, personal injury or other health and safety issues.

We could be adversely impacted by the supply of defective or expired products, including the infiltration of counterfeit products into the supply chain, errors in re-labeling of products, product tampering, product recall and contamination or product mishandling issues. Through our pharmacies and specialist packaging sites, including through services provided by third-party health care providers, we are also exposed to risks relating to the services we offer. Errors in the dispensing and packaging of pharmaceuticals, including related counseling, and in the provision of other healthcare services could lead to serious injury or death. Product liability or personal injury claims may be asserted against us with respect to any of the products or pharmaceuticals we sell or services we provide. For example, from time to time, the FDA issues statements alerting patients that products in our supply chain may contain impurities or harmful substances, and claims relating to the sale or distribution of such products may be asserted against us or arise from these statements. Our healthcare clinics also increase our exposure to professional liability claims related to medical care. Should a product or other liability issue arise, the coverage limits under our insurance programs and the indemnification amounts available to us may not be adequate to protect us against claims and judgments. We also may not be able to maintain this insurance on acceptable terms in the future. We could suffer significant reputational damage and financial liability if we, or any affiliated entities or third-party healthcare providers that we do business with, experience any of the foregoing health and safety issues or incidents, which could have a material adverse effect on our business operations, financial condition and results of operations.

We have significant outstanding debt; our debt and associated payment obligations could significantly increase in the future if we incur additional debt and do not retire existing debt.

We have outstanding debt and other financial obligations and significant unused borrowing capacity. As of August 31, 2019, we had approximately \$17 billion of outstanding indebtedness, including short-term debt. Our debt level and related debt service obligations could have negative consequences, including:

- requiring us to dedicate significant cash flow from operations to the payment of principal, interest and other amounts payable on our debt, which would reduce the funds we have available for other purposes, such as working capital, capital expenditures, acquisitions, share repurchases and dividends;
- making it more difficult or expensive for us to obtain any necessary future financing for working capital, capital expenditures, debt service requirements, debt refinancing, acquisitions or other purposes;
- reducing our flexibility in planning for or reacting to changes in our industry and market conditions and making us more vulnerable in the event of a
 downturn in our business operations; and
- exposing us to interest rate risk given that a portion of our debt obligations is at variable interest rates.

We may incur or assume significantly more debt in the future, including in connection with acquisitions, strategic investments or joint ventures. If we add new debt and do not retire existing debt, the risks described above could increase.



We also could be adversely impacted by any failure to renew or replace, on terms acceptable to us or at all, existing funding arrangements when they expire, and any failure to satisfy applicable covenants.

Additionally, a portion of our indebtedness bears interest at fluctuating interest rates, some of which is tied to the London interbank offered rate for deposits of U.S. dollars ("LIBOR"). LIBOR tends to fluctuate based on general interest rates, rates set by the Federal Reserve and other central banks, the supply of and demand for credit in the London interbank market and general economic conditions. In July 2017, the United Kingdom Financial Conduct Authority, which regulates LIBOR, announced that it intends to stop compelling banks to submit rates for the calculation of LIBOR after 2021. The expected phase out of LIBOR could cause market volatility or disruption and may adversely affect our access to the capital markets and cost of funding.

Our long-term debt obligations include covenants that may adversely affect our ability, and the ability of certain of our subsidiaries, to incur certain secured indebtedness or engage in certain types of transactions. In addition, our existing credit agreements require us to maintain as of the last day of each fiscal quarter a ratio of consolidated debt to total capitalization not to exceed a certain level. Our ability to comply with these restrictions and covenants may be affected by events beyond our control. If we breach any of these restrictions or covenants and do not obtain a waiver from the lenders, then, subject to applicable cure periods, our outstanding indebtedness could be declared immediately due and payable. This could have a material adverse effect on our business operations and financial condition.

We could be adversely affected by downgrades to our credit ratings or disruptions in our ability to access well-functioning capital markets.

Historically, we have relied on the public debt capital markets to fund portions of our capital investments and access to the commercial paper market and bank credit facilities as part of our working capital management strategy. Our continued access to these markets, and the terms of such access, depend on multiple factors including the condition of debt capital markets, our operating performance, and our credit ratings. The major credit rating agencies have assigned us and our corporate debt investment grade credit ratings. These ratings are based on a number of factors, which include their assessment of our financial strength and financial policies. We benefit from investment grade ratings as they serve to lower our borrowing costs and facilitate our access to a variety of lenders and other creditors, including landlords for our leased stores, on terms that we consider advantageous to our businesses. However, there can be no assurance that any particular rating assigned to us will remain in effect for any given period of time or that a rating will not be changed or withdrawn by a rating agency, if in that rating agency's judgment, future circumstances relating to the basis of the rating so warrant. For more information on our ratings, see "Management's Discussion and Analysis of Financial Condition and Results of Operations - Liquidity and Capital Resources - Credit Ratings." Incurrence of additional debt by us and our operating performance could adversely affect our credit ratings could adversely affect our cost of funds, liquidity, competitive position and access to capital markets or any downgrade of our credit ratings could adversely affect our cost of funds, liquidity, competitive position and access to capital markets and increase the cost of and counterparty risks associated with existing facilities, which could materially and adversely affect our business operations, financial condition, and results of operations.

We may be unable to keep existing store locations or open new locations in desirable places on favorable terms, which could materially and adversely affect our results of operations.

We compete with other retailers and businesses for suitable locations for our stores. Local land use and zoning regulations, environmental regulations and other regulatory requirements may impact our ability to find suitable locations and influence the cost of constructing, renovating and operating our stores. In addition, real estate, zoning, construction and other delays may adversely affect store openings and renovations and increase our costs. Further, changing local demographics at existing store locations may adversely affect revenue and profitability levels at those stores. The termination or expiration of leases at existing store locations may adversely affect us if the renewal terms of those leases are unacceptable to us and we are forced to close or relocate stores. For example, in fiscal 2019, we announced the closure of approximately 200 locations in the United States and approximately 200 stores in the United Kingdom, respectively, under our Transformational Cost Management Program. If we determine to close or relocate a store subject to a lease, we may remain obligated under the applicable lease for the balance of the lease term. If we are unable to maintain our existing store locations or open new locations in desirable places and on favorable terms, our results of operations could be materially and adversely affected.

As a holding company, Walgreens Boots Alliance is dependent on funding from its operating subsidiaries to pay dividends and other obligations.

- 23 -

Walgreens Boots Alliance is a holding company with no business operations of its own. Its assets primarily consist of direct and indirect ownership interests in, and its business is conducted through, subsidiaries which are separate legal entities. As a result, it is dependent on funding from its subsidiaries, including Walgreens and Alliance Boots, to meet its obligations. Additionally, Walgreens Boots Alliance's subsidiaries may be restricted in their ability to pay cash dividends or to make other distributions to Walgreens Boots Alliance, which may limit the payment of cash dividends or other distributions to the holders of Walgreens Boots Alliance common stock. Credit facilities and other debt obligations of Walgreens Boots Alliance, as well as statutory provisions, may further limit the ability of Walgreens Boots Alliance and its subsidiaries to pay dividends. Payments to Walgreens Boots Alliance by its subsidiaries are also contingent upon its subsidiaries' earnings and business considerations. Future Walgreens Boots Alliance dividends will be determined based on earnings, capital requirements, financial condition and other factors considered relevant by its Board of Directors.

Our quarterly results may fluctuate significantly.

Our operating results have historically varied on a quarterly basis, including increased variability in the second quarter of fiscal 2019, and may continue to fluctuate significantly in the future. Factors that may affect our quarterly operating results, some of which are beyond the control of management, include, but are not limited to the timing of the introduction of new generic and brand name prescription drugs; inflation, including with respect to generic drug procurement costs; seasonality, including the timing and severity of the cough, cold and flu season; changes or rates of change in payer reimbursement rates and terms; fluctuations in inventory, energy, transportation, labor, healthcare and other costs; significant acquisitions, dispositions, joint ventures and other strategic initiatives; asset impairment charges, including the performance of and impairment charges related to our equity method investments; the relative magnitude of our LIFO provision in any particular quarter; foreign currency fluctuations; market conditions, including any impact relating to Brexit; prolonged severe weather in key markets; and many of the other risk factors discussed herein. Accordingly, we believe that quarter-to-quarter comparisons of our operating results are not necessarily meaningful and investors should not place undue reliance on the results of any particular quarter as an indication of our future performance.

Our businesses are seasonal in nature, and adverse events during the holiday and cough, cold and flu seasons could adversely impact our operating results.

Our businesses are seasonal in nature, with the second fiscal quarter (December, January and February) typically generating a higher proportion of retail sales and earnings than other fiscal quarters. We purchase significant amounts of seasonal inventory in anticipation of the holiday season. Adverse events, such as deteriorating economic conditions, higher unemployment, higher gas prices, public transportation disruptions, or unanticipated adverse weather, could result in lower-than-planned sales during key selling seasons. For example, frequent or unusually heavy snowfall, ice storms, rainstorms, windstorms or other extreme weather conditions over a prolonged period could make it difficult for our customers to travel to our stores and increase our snow removal and other costs. This could lead to lower sales or to unanticipated markdowns, negatively impacting our financial condition and results of operations. In addition, both prescription and non-prescription drug sales are affected by the timing and severity of the cough, cold and flu season, which can vary considerably from year to year.

We could be adversely impacted by changes in accounting standards and subjective assumptions, estimates and judgments by management related to complex accounting matters.

Generally Accepted Accounting Principles ("GAAP") and related accounting pronouncements, implementation guidelines and interpretations with regard to a wide range of matters that are relevant to our businesses, including, but not limited to, revenue recognition, asset impairment, impairment of goodwill and other intangible assets, inventories, equity method investments, vendor rebates and other vendor consideration, lease obligations, self-insurance liabilities, pension and postretirement benefits, tax matters, unclaimed property laws and litigation and other contingent liabilities are highly complex and involve many subjective assumptions, estimates and judgments. Changes in these rules or their interpretation or changes in underlying assumptions, estimates or judgments could significantly change our reported or expected financial performance or financial condition. For example, changes in accounting standards and the application of existing accounting standards particularly related to the measurement of fair value as compared to carrying value for the Company's reporting units, including goodwill, intangible assets and investments in equity interests, may have an adverse effect on the Company's financial condition and results of operations. Factors that could lead to impairment of goodwill and intangible assets include significant adverse changes in the business climate and declines in the financial condition of a reporting unit. Factors that could lead to impairment of investments in equity interests of the companies in which we invested include a prolonged period of decline in their operating performance or adverse changes in the economic, regulatory and legal environments of the countries in which they operate.

- 24 -

New accounting guidance also may require changes to our processes, accounting systems and internal controls that could increase our operating costs and/or significantly change our financial statements. For example, in February 2016, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2016-02, Leases (Topic 842), which supersedes Topic 840, Leases. This ASU, which became effective for us beginning on September 1, 2019 (fiscal 2020), seeks to increase the transparency and comparability of organizations by recognizing operating lease assets and operating lease liabilities on the balance sheet and disclosing key information about leasing arrangements. The adoption approach for these accounting standards affect the comparability of our consolidated financial statements. Implementing new accounting guidance may require us to make significant changes to and investments in our accounting systems and processes, which could result in significant adverse changes to our financial statements. See, "new accounting pronouncements," within note 1, summary of major accounting policies, to the Consolidated Financial Statements.

We have a substantial amount of goodwill and other intangible assets which could, in the future, become impaired and result in material non-cash charges to our results of operations.

As of August 31, 2019, we had \$27.4 billion of goodwill and other intangible assets on our Consolidated Balance Sheets. We evaluate this goodwill and other indefinite-lived intangible assets for impairment annually during the fourth quarter, or more frequently if an event occurs or circumstances change that could more likely than not reduce the fair value of a reporting unit or indefinite-lived intangible asset below its carrying value. As part of this impairment analysis, we determine fair value for each reporting unit using both the income and market approaches. We determine fair value of indefinite-lived intangible assets using the relief from royalty method and excess earnings method of the income approach. Definite-lived intangible assets are evaluated for impairment if an event occurs or circumstances change that indicate the carrying amount may not be recoverable. Estimated fair values could change if, for example, there are changes in the business climate, changes in the competitive environment, adverse legal or regulatory actions or developments, changes in capital structure, cost of debt and equity, capital expenditure levels, operating cash flows, or market capitalization. For example, during the three months ended August 31, 2019, we completed quantitative impairment analysis for goodwill and indefinite-lived intangible assets as of a June 1, 2019 valuation date. Based on this analysis, the fair value of the Boots reporting unit, within the Retail Pharmacy International segment, is in excess of its carrying value by approximately 9%. As at August 31, 2019, the carrying value of the goodwill of the Boots reporting unit is \$2.6 billion. The fair values of certain indefinite-lived intangibles within the Boots reporting unit exceeded their carrying amounts ranging from approximately 3% to approximately 29%, except for the pharmacy licenses. As at August 31, 2019, the carrying value of these indefinite-lived intangibles within Boots reporting unit is \$6.9 billion. As a result of an annual evaluation, during the three months ended August 31, 2019, the Company recorded an impairment of \$73 million on its pharmacy licenses in the Boots reporting unit. Because of the significance of our goodwill and intangible assets, any future impairment of these assets could result in material non-cash charges to our results of operations, which could have a material adverse effect on our results of operations.

We could be subject to adverse changes in tax laws, regulations and interpretations or challenges to our tax positions.

We are a large corporation with operations in the United States and numerous other jurisdictions around the world. As such, we are subject to tax laws and regulations of the U.S. federal, state and local governments as well as various foreign jurisdictions. We compute our income tax provision based on enacted tax rates in the jurisdictions in which we operate. As the tax rates vary among jurisdictions, a change in earnings attributable to the various jurisdictions in which we operate could result in an unfavorable change in our overall tax provision.

From time to time, changes in tax laws or regulations may be proposed or enacted that could adversely affect our overall tax liability. For example, the U.S. tax legislation enacted on December 22, 2017 represented a significant overhaul of the U.S. federal tax code that impacted our U.S. federal income tax position. There can be no assurance that changes in tax laws or regulations, both within the United States and the other jurisdictions in which we operate, will not materially and adversely affect our effective tax rate, tax payments, financial condition and results of operations. Similarly, changes in tax laws and regulations that impact our customers and counterparties or the economy generally may also impact our financial condition and results of operations.

Tax laws and regulations are complex and subject to varying interpretations, and we are subject to regular review and audit by both domestic and foreign tax authorities. Any adverse outcome of such a review or audit could have a negative impact on our effective tax rate, tax payments, financial condition and results of operations. In addition, the determination of our income tax provision and other tax liabilities requires significant judgment, and there are many transactions and calculations where the ultimate tax determination is uncertain. The ultimate tax determination may differ from the amounts recorded in our financial statements and may materially affect our results of operations in the period or periods for which such determination is made. Any significant failure to comply with applicable tax laws and regulations in all relevant jurisdictions could give rise to substantial penalties and liabilities. Any changes in enacted tax laws (such as the recent U.S. tax legislation), rules or regulatory

- 25 -

or judicial interpretations; or any change in the pronouncements relating to accounting for income taxes could materially and adversely impact our effective tax rate, tax payments, financial condition and results of operations.

Our insurance strategies may expose us to unexpected costs.

We use a combination of insurance and self-insurance to provide for potential liability for workers' compensation; automobile and general liability; property, director and officers' liability; and employee healthcare benefits. Provisions for losses related to self-insured risks generally are based upon actuarially determined estimates. Any actuarial projection of losses is subject to a high degree of variability. Substantial, unanticipated losses or liabilities, including those due to natural disasters or otherwise, as well as changes in legal claims, trends and interpretations, variability in inflation rates, changes in the nature and method of claims settlement, benefit level changes due to changes in applicable laws, insolvency of insurance carriers, and changes in discount rates could all materially and adversely affect our financial condition and results of operations.

We could be adversely impacted by changes in assumptions used in calculating pension assets and liabilities.

We operate certain defined benefit pension plans in the United Kingdom, which were closed to new entrants in 2010, as well as smaller plans in other jurisdictions. The valuation of the pension plans' assets and liabilities depends in part on assumptions, which are primarily based on the financial markets as well as longevity and employee retention rates. This valuation is particularly sensitive to material changes in the value of equity, bond and other investments held by the pension plans, changes in the corporate bond yields which are used in the measurement of the liabilities, changes in market expectations for long-term price inflation, and new evidence on projected longevity rates. Funding requirements and the impact on the statement of earnings relating to these pension plans are also influenced by these factors. Adverse changes in the assumptions used to calculate the value of pension assets and liabilities, including lower than expected pension fund investment returns and/or increased life expectancy of plan participants, or regulatory change could require us to increase the funding of its defined benefit pension plans or incur higher expenses, which would adversely impact our results of operations and financial position.

Certain stockholders may have significant voting influence over matters requiring stockholder approval.

As of September 30, 2019, affiliates of Stefano Pessina, our Executive Vice Chairman and Chief Executive Officer (the "SP Investors"), had sole or shared voting power, directly or indirectly, over an aggregate of approximately 16.2% of our outstanding common stock. The SP Investors have agreed to, for so long as they have the right to designate a nominee for election to the Board, to vote all of their shares of common stock in accordance with the Board's recommendation on matters submitted to a vote of the Company's stockholders (including with respect to the election of directors). The SP Investors' significant interest in our common stock potentially could determine the outcome of matters submitted to a vote by our stockholders. The influence of the SP Investors could result in the Company taking actions that other stockholders do not support or failing to take actions that other stockholders support. As a result, the market price of our common stock could be adversely affected.

Conflicts of interest, or the appearance of conflicts of interest, may arise because certain of our directors and officers are also owners or directors of companies we may have dealings with.

Conflicts of interest, or the appearance of conflicts of interest, could arise between our interests and the interests of the other entities and business activities in which our directors or officers are involved. For example, potential conflicts of interest could arise if a dispute were to arise between the Company and other parties to the Company Shareholders Agreement (as defined herein), including the SP Investors. Mr. Pessina, our Executive Vice Chairman and Chief Executive Officer, indirectly controls Alliance Santé Participations S.A. ("ASP"), a privately-held company which is a party to the Company Shareholders Agreement, and he and his partner Ornella Barra, our Co-Chief Operating Officer, serve as directors of ASP. There are other arrangements between affiliates of Mr. Pessina and the Company, with required disclosures included in the Company's annual proxy statement, including with respect to Alliance Healthcare Italia SpA, which is an entity indirectly owned and controlled by Mr. Pessina and in which the Company has an indirect 9% interest, which operates Boots branded stores in Italy. Conflicts of interest, or the appearance of conflicts of interest, or similar issues could arise in connection with these or other transactions in the future. While our contractual arrangements place restrictions on the parties' conduct in certain situations, and related party transactions are subject to independent review and approval in accordance with our related party transaction approval procedures and applicable law, the potential for a conflict of interest exists and such persons may have conflicts of interest, or the appearance of conflicts of interest, with respect to matters involving or affecting both companies.

Our certificate of incorporation and bylaws, Delaware law and/or our agreements with certain stockholders may impede the ability of our stockholders to make changes to our Board or impede a takeover.



Certain provisions of our certificate of incorporation and bylaws, as well as provisions of the Delaware General Corporation Law (the "DGCL"), could make it difficult for stockholders to change the composition of the Board or discourage, delay, or prevent a merger, consolidation, or acquisitions that stockholders may otherwise consider favorable. These provisions include the authorization of the issuance of "blank check" preferred stock that could be issued by the Board, limitations on the ability of stockholders to call special meetings, and advance notice requirements for nomination for election to the Board or for proposing matters that can be acted upon by stockholders at stockholder meetings. We are also subject to the provisions of Section 203 of the DGCL, which prohibits us, except under specified circumstances, from engaging in any mergers, significant sales of stock or assets, or business combinations with any stockholder or group of stockholders who own 15% or more of our common stock.

Under the Company Shareholders Agreement, the SP Investors are entitled to designate one nominee to the Board (currently Stefano Pessina) for so long as the SP Investors continue to meet certain beneficial ownership thresholds and subject to certain other conditions. Pursuant to the Company Shareholders Agreement, the SP Investors have agreed that, for so long as they have the right to designate a nominee to the Board, they will vote all of their shares of common stock in accordance with the Board's recommendation on matters submitted to a vote of our stockholders (including with respect to the election of directors).

While these provisions do not make us immune from takeovers or changes in the composition of the Board, and are intended to protect our stockholders from, among other things, coercive or otherwise unfair tactics, these provisions could have the effect of making it difficult for stockholders to change the composition of the Board or discouraging, delaying, or preventing a merger, consolidation, or acquisitions that stockholders may otherwise consider favorable.

We cannot guarantee that our stock repurchase program will be fully implemented or that it will enhance long-term stockholder value.

In June 2018, our Board of Directors approved a new stock repurchase program authorizing the repurchase of up to \$10 billion of our common stock. The repurchase program does not have an expiration date and we are not obligated to repurchase a specified number or dollar value of shares, on any particular timetable or at all. There can be no assurance that we will repurchase stock at favorable prices. The repurchase program may be suspended or terminated at any time and, even if fully implemented, may not enhance long-term stockholder value.

The market price of our common stock may be volatile.

The market price of shares of our common stock may be volatile. Broad general economic, political, market and industry factors may adversely affect the market price of the shares, regardless of our actual operating performance. In addition to the other risk factors identified in this Item 1A, factors that could cause fluctuations in the price of the shares include:

- actual or anticipated variations in quarterly operating results and the results of competitors;
- changes in financial estimates by us or by any securities analysts that might cover us;
- conditions or trends in the industry, including regulatory changes or changes in the securities marketplace;
- announcements by us or our competitors of significant acquisitions, strategic partnerships or divestitures;
- · announcements of investigations or regulatory scrutiny of our operations or lawsuits filed against us;
- additions or departures of key personnel;
- issuances or sales of our common stock (or the exercise of related registration rights), including sales of shares by our directors and officers or key
 investors, including the SP Investors and certain other former Alliance Boots stockholders, subject to restrictions in the case of shares held by
 persons deemed to be our affiliates and to certain obligations pursuant to a shareholders agreement (as amended, the "Company Shareholders
 Agreement") with certain of the SP Investors; and
- various other market factors or perceived market factors, including rumors or speculation, whether or not correct, involving or affecting us or our industries, vendors, customers, strategic partners or competitors.

There are a number of additional business risks that could materially and adversely affect our businesses and financial results.



Many other factors could materially and adversely affect our businesses and financial results, including:

- our ability to establish effective advertising, marketing and promotional programs;
- inflation, new or increased taxes, adverse fluctuations in foreign currency exchange rates, changes in market conditions or otherwise;
- natural disasters, civil unrest, severe weather conditions, terrorist activities, global political and economic developments, war, health epidemics or pandemics or the prospect of these events;
- liabilities or expense relating to the protection of the environment, related health and safety matters, environmental remediation or compliance with environmental laws and regulations, including those governing exposure to, and the management and disposal of, hazardous substances;
- the long-term effects of climate change on general economic conditions and the pharmacy industry in particular, along with changes in the supply, demand or available sources of energy and the regulatory and other costs associated with energy production and delivery;
- adverse publicity and potential losses, liabilities and reputational harm stemming from any public incident (whether occurring online, in social media, in our stores or other company facilities, or elsewhere) involving our company, our personnel or our brands, including any such public incident involving our customers, products, services, stores or other property, or those of any of our vendors or other parties with which we do business; and
- negative publicity, even if unwarranted, related to safety or quality, human and workplace rights, or other issues damaging our brand image and corporate reputation, or that of any of our vendors or strategic allies.

Item 1B. Unresolved staff comments

There are no unresolved written comments that were received from the SEC Staff 180 days or more before the end of the fiscal year relating to the Company's periodic or current reports under the Exchange Act.

Item 2. Properties

The following information regarding the Company's properties is provided as of August 31, 2019 and does not include properties of unconsolidated, partiallyowned entities.

The Retail Pharmacy USA division operated 9,277 retail stores and six specialty pharmacies. The Retail Pharmacy International division operated 4,605 retail stores. In addition, the Retail Pharmacy International division also owned or leased 385 standalone Boots Opticians locations. The Company's domestic and international retail locations, which included Boots Opticians and specialty pharmacy locations, covered approximately 149 million square feet. The Company owned approximately 12% and 4% of these Retail Pharmacy USA division and Retail Pharmacy International division locations, respectively. The remaining locations were leased or licensed. For more information on leases, see note 4, leases, to the Consolidated Financial Statements in part II, item 8 of this Form 10-K.

- 28 -

Table of Contents

The following is a breakdown of the Company's retail stores:

	Retail stores
Retail Pharmacy USA:	
United States	9,168
Puerto Rico	108
U.S. Virgin Islands	1
	9,277
Retail Pharmacy International:	
United Kingdom	2,465
Mexico	1,174
Chile	341
Thailand	293
Norway	159
The Republic of Ireland	87
The Netherlands	59
Lithuania	27
	4,605
Walgreens Boots Alliance total	13,882

The Company operated 21 retail distribution centers with a total of approximately 14 million square feet of space, of which 13 locations were owned. Geographically, 16 of these retail distribution centers were located in the United States and five were located outside of the United States. In addition, the Company used public warehouses and third-party distributors to handle certain retail distribution needs. The Company's Retail Pharmacy USA division also operated two prescription mail service facilities which occupied approximately 263 thousand square feet. One of these prescription mail service facilities was leased.

The Company operated 300 pharmaceutical distribution centers located outside of the United States, of which 116 were owned. These pharmaceutical distribution centers occupied approximately 14 million square feet and were operated by the Pharmaceutical Wholesale division, which supplied third-party customers as well as the Retail Pharmacy International division in certain countries.

The Company operated 22 principal office facilities, which occupied approximately 2.5 million square feet. Ten of these principal office facilities were owned, and two of which were located in the United States.

Item 3. Legal proceedings

The information in response to this item is included in note 10, commitments and contingencies, to the Consolidated Financial Statements included in part II, item 8, of this Form 10-K.

Item 103 of SEC Regulation S-K requires disclosure of environmental legal proceedings with, or any such legal proceedings known to be contemplated by, a governmental authority if management reasonably believes that the proceedings involve potential monetary sanctions of \$100,000 or more. The Company is under investigation by certain counties within the State of California for alleged noncompliance with state hazardous waste regulations. Although we cannot predict the outcome of this matter, we do not expect the outcome to have a material adverse effect on our prospects, financial condition, results of operations or cash flows.

Item 4. Mine safety disclosures

Not applicable.

Executive officers of the registrant

The following table sets forth, for each person currently serving as an executive officer of the Company, the name, age (as of October 28, 2019) and office(s) held by such person:

- 29 -

Table of Contents

Name	Age	Office(s) held
James A. Skinner	74	Executive Chairman of the Board
Stefano Pessina	78	Executive Vice Chairman and Chief Executive Officer
Ornella Barra	65	Co-Chief Operating Officer
Alexander W. Gourlay	59	Co-Chief Operating Officer
James Kehoe	56	Executive Vice President and Global Chief Financial Officer
Marco Pagni	57	Executive Vice President, Global Chief Administrative Officer and General Counsel
Kathleen Wilson-Thompson	62	Executive Vice President and Global Chief Human Resources Officer
Heather Dixon	47	Senior Vice President, Global Controller and Chief Accounting Officer

Set forth below is information regarding the principal occupations and employment and business experience over the past five years for each executive officer. Executive officers are elected by, and serve at the discretion of, the Board of Directors. Unless otherwise stated, employment is by Walgreens Boots Alliance.

Mr. Skinner has served as Executive Chairman since January 2015, having served as non-executive Chairman of the Board from July 2012 to January 2015. Mr. Skinner previously served McDonald's Corporation as Vice Chairman from January 2003 to June 2012, as Chief Executive Officer from November 2004 to June 2012 and as a director from 2004 to June 2012. Since 2005, Mr. Skinner has served as a director of Illinois Tool Works Inc. Mr. Skinner served as a director of HP Inc. (f/k/a Hewlett-Packard Company) from July 2013 to November 2015.

Mr. Pessina has served as Chief Executive Officer since July 2015 and as Executive Vice Chairman since January 2015. He served as Acting Chief Executive Officer from January 2015 to July 2015. Previously, he served as Executive Chairman of Alliance Boots from July 2007 to December 2014. Prior to that, Mr. Pessina served as Executive Deputy Chairman of Alliance Boots. Prior to the merger of Alliance UniChem and Boots Group, Mr. Pessina was Executive Deputy Chairman of Alliance UniChem having been its Chief Executive for three years through December 2004. Mr. Pessina was appointed to the Alliance UniChem Board in 1997 when UniChem merged with Alliance Santé, the Franco-Italian pharmaceutical wholesale group which he established in Italy in 1977. Mr. Pessina also serves on the Board of Directors of a number of private companies, and, from 2000 to 2017, served on the Board of Directors of Galenica AG, a publicly-traded Swiss healthcare group.

Ms. Barra has served as Co-Chief Operating Officer since June 2016. She served as Executive Vice President, President and Chief Executive of Global Wholesale and International Retail from December 2014 to June 2016. Previously, she served as the Chief Executive, Wholesale and Brands of Alliance Boots from September 2013 to December 2014 and Chief Executive of the Pharmaceutical Wholesale Division of Alliance Boots from January 2009 to September 2013, and before that, Wholesale & Commercial Affairs Director of Alliance Boots. Since January 2015, Ms. Barra has served as a director of AmerisourceBergen and from April 2013 to April 2019, served as a director of Assicurazioni Generali, the parent company of Generali Group, a global insurance group. Ms. Barra also serves as a director of a number of private companies, and, until February 2015, served as a director of Alliance Boots.

Mr. Gourlay has served as Co-Chief Operating Officer since June 2016. He served as Executive Vice President, President of Walgreens from December 2014 to June 2016. Previously, he served as Executive Vice President, President of Customer Experience and Daily Living of Walgreens from October 2013 to December 2014 and President Elect of Walgreens from September 2014 to December 2014. He served as Chief Executive of the Health & Beauty Division, Alliance Boots, from January 2009 to September 2013, and previously was Managing Director of Boots UK and a member of the Alliance Boots operating committee. He served as a director of Alliance Boots from January 2009 to September 2013.

Mr. Kehoe has served as Executive Vice President and Global Chief Financial Officer since June 2018. Previously, he served Takeda Pharmaceutical Company Limited as Chief Financial Officer and Corporate Officer from June 2016 to March 2018 and as a board director June 2017 to May 2018. He previously served as Executive Vice President and Chief Financial Officer of Kraft Foods Group, Inc. from February 2015 to July 2015. Previously, he worked for Gildan Activewear Inc., a supplier of branded family apparel in Canada, where he served as Executive Vice President and Chief Financial and Administrative Officer earlier in 2015. Prior to that, he was Senior Vice President, Operating Excellence at Mondelēz International, Inc. from November 2013 until December 2014. Mr. Kehoe joined Kraft in 1988 and held a variety of senior-level positions, including serving as Senior Vice President, Corporate Finance from October 2012 to October 2013, and Senior Vice President, Finance of Kraft Foods North America from November 2010 until September 2012.

- 30 -

Mr. Pagni has served as Executive Vice President, Global Chief Administrative Officer and General Counsel since February 2016. He served as Executive Vice President, Global Chief Legal and Administrative Officer from February 2015 to February 2016. Previously, he served as Executive Director and Group Legal Counsel and Chief Administrative Officer of Alliance Boots from 2007 to 2014 and General Counsel and Company Secretary for Alliance Boots from 2006 to 2007, having joined Alliance UniChem, a predecessor company, in the same position in 2003. Prior to this, Mr. Pagni served at McDonald's Corporation for 10 years in a number of senior management positions across the world, including in the U.S. and UK, such as Vice President of International Development, and Vice President, General Counsel, International. Mr. Pagni served as a director of Alliance Boots until February 2015.

Ms. Wilson-Thompson has served as Executive Vice President and Global Chief Human Resources Officer since December 2014. Previously, she served as Senior Vice President and Chief Human Resources Officer of Walgreens from January 2010 to December 2014. Prior to that, she served in a variety of legal and operational positions at Kellogg Company, most recently as Senior Vice President, Global Human Resources from July 2005 to December 2009. She has served as a director of Ashland Global Holdings Inc., a global specialty chemicals company, since 2017, and Tesla, Inc., an automotive and energy company, since 2018 and served as a director of Vulcan Materials Company, a producer of construction aggregates, from 2009 to 2018.

Ms. Dixon has served as Senior Vice President, Global Controller and Chief Accounting Officer since March 2019. Previously, she served as Vice President, Controller and Chief Accounting Officer of Aetna Inc., a health care benefits company from November 2017 to March 2019. She joined Aetna as Vice President, Assistant Controller in August 2016. Prior to joining Aetna, Ms. Dixon served as Vice President, Assistant Controller of PepsiCo, Inc., a global food and beverage company, a position she held from August 2015 to March 2016. Previously, she was with American Express Company, a multinational financial services company, where she held a number of positions of increasing responsibility from 2005 to 2015. Ms. Dixon started her financial career in public accounting, where she held various audit and transaction advisory roles with PricewaterhouseCoopers LLP from 1995 to 2005.

Mr. Pessina and Ms. Barra are partners and share a private residence. There are no other family relationships among any of our directors or executive officers.

PART II

Item 5. Market for registrant's common equity, related stockholder matters and issuer purchases of equity securities

Walgreens Boots Alliance's common stock is listed on the Nasdaq Stock Market under the symbol WBA. As of August 31, 2019, there were approximately 52,800 holders of record of Walgreens Boots Alliance common stock.

The Company has paid cash dividends every quarter since 1933. Future dividends will be determined based on earnings, capital requirements, financial condition and other factors considered relevant by the Walgreens Boots Alliance Board of Directors.

The following table provides information about purchases made by the Company during the quarter ended August 31, 2019 of equity securities that are registered by the Company pursuant to Section 12 of the Exchange Act. Subject to applicable law, share purchases may be made from time to time in open market transactions, privately negotiated transactions including accelerated share repurchase agreements, or pursuant to instruments and plans complying with Rule 10b5-1, among other types of transactions and arrangements.

	Issuer purchases of equity securities					
Period	Total number of shares purchased		erage price paid per share	Total number of shares purchased as part of publicly announced repurchase programs ¹		oximate dollar value of shares hay yet be purchased under the plans or programs ¹
6/1/19 - 6/30/19	2,384,600	\$	51.94	2,384,600	\$	3,753,404,394
7/1/19 - 7/31/19	2,348,488		55.02	2,348,488		3,624,165,659
8/1/19 - 8/31/19	3,498,400		51.97	2,598,400		3,489,712,386
	8,231,488	\$	52.83	7,331,488	\$	3,489,712,386

¹ In June 2018, Walgreens Boots Alliance authorized a stock repurchase program, which authorized the repurchase of up to \$10.0 billion of Walgreens Boots Alliance common stock. This program has no specified expiration date.

- 32 -

Item 6. Selected financial data

Five-Year Summary of Selected Consolidated Financial Data

Walgreens Boots Alliance, Inc. and Subsidiaries (in millions, except per share amounts)

<u>Fiscal year</u>	2019	2018	2017	2016	2015 ⁶
Sales	\$ 136,866	\$ 131,537	\$ 118,214	\$ 117,351	\$ 103,444
Cost of sales	106,790	100,745	89,052	87,477	76,691
Gross profit	30,076	30,792	29,162	29,874	26,753
Selling, general and administrative expenses ¹	25,242	24,694	23,813	23,841	22,349
Equity earnings in AmerisourceBergen ²	164	191	135	37	—
Equity earnings in Alliance Boots ³	—	—	—	_	315
Operating income	4,998	6,289	5,484	6,070	4,719
Gain on previously held equity interest ⁴	_	_	—	_	563
Other income (expense) ^{1, 5}	233	302	62	(330)	634
Earnings before interest and income tax provision	5,231	6,591	5,546	5,740	5,916
Interest expense, net	704	616	693	596	605
Earnings before income tax provision	4,527	5,975	4,853	5,144	5,311
Income tax provision	588	998	760	997	1,056
Post tax earnings from other equity method investments	23	54	8	44	24
Net earnings	3,962	5,031	4,101	4,191	4,279
Net earnings attributable to noncontrolling interests	(20)	7	23	18	59
Net earnings attributable to Walgreens Boots Alliance, Inc.	\$ 3,982	\$ 5,024	\$ 4,078	\$ 4,173	\$ 4,220
Per Common Share					
Net earnings					
Basic	\$ 4.32	\$ 5.07	\$ 3.80	\$ 3.85	\$ 4.05
Diluted	4.31	5.05	3.78	3.82	4.00
Dividends declared	1.778	1.640	1.525	1.455	1.373
Balance Sheet					
Total assets	\$ 67,598	\$ 68,124	\$ 66,009	\$ 72,688	\$ 68,782
Long-term debt	11,098	12,431	12,684	18,705	13,315
Total Walgreens Boots Alliance, Inc. shareholders' equity	23,512	26,007	27,466	29,880	30,861
Noncontrolling interests	641	682	808	401	439
Total equity	\$ 24,152	\$ 26,689	\$ 28,274	\$ 30,281	\$ 31,300

¹ The Company adopted ASU 2017-07, Compensation - Retirement Benefits (Topic 715) as of September 1, 2018 (fiscal 2019) on a retrospective basis. The impact on our previously reported net periodic costs as a result of the retrospective adoption of this standard results in a reclassification from selling, general and administrative expenses to other income (expense) of \$125 million, \$73 million, \$(69) million and \$(51) million for the fiscal years ended August 31, 2018, 2017, 2016 and 2015, respectively.

² Effective March 18, 2016, the Company began accounting for its investment in AmerisourceBergen using the equity method of accounting, subject to a two-month reporting lag. Due to the March 18, 2016 effective date and the two-month reporting lag, the Company's results for the 12 month period ended August 31, 2016 include approximately three and a half months of equity method income relating to its investment in AmerisourceBergen. Similarly, results for the 12 month period ended August 31, 2017 include approximately ten and a half months of equity income reflecting the Company's increased ownership following the exercise on August 25, 2016 of the second tranche of warrants.

³ On August 2, 2012, the Company completed the acquisition of 45% of the issued and outstanding share capital of Alliance Boots in exchange for cash and Company shares. The Company accounted for this investment under the equity method until it completed the acquisition of the remaining 55% of Alliance Boots on December 31, 2014. As a result, fiscal 2015 includes the results of Alliance Boots for eight months (January through August 2015) on a fully consolidated basis and four months (September through December 2014) as equity earnings in Alliance Boots reflecting Walgreens' pre-merger 45% interest.

- 33 -

Table of Contents

- ⁴ In fiscal 2015, as a result of acquiring the remaining 55% interest in Alliance Boots, the Company's previously held 45% interest was remeasured to fair value, resulting in a gain of \$563 million.
- ⁵ Fiscal 2019 primarily reflects gain recognition resulting from the termination of the option granted to Rite Aid to become a member of the Company's group purchasing organization. Fiscal 2018 includes the gain on sale of the Company's equity interest in Premise Health Holding Corp., partially offset by the impairment of the Company's equity method investment in Guangzhou Pharmaceuticals Corporation. In fiscal 2016 and 2015, the Company recorded other income (expense) of \$(517) million and \$779 million, respectively, from fair value adjustments of the AmerisourceBergen warrants and the amortization of the deferred credit associated with the initial value of the warrants. Fiscal 2016 also includes income of \$268 million related to the change in accounting method for the Company's investment in AmerisourceBergen. Fiscal 2015 also includes a \$94 million loss on derivative contracts that were not designated as accounting hedges.
- ⁶ To improve comparability, certain classification changes were made to prior period sales, cost of sales and selling, general and administrative expenses. These changes had no impact on operating income. The reclassifications were made in the fourth quarter of fiscal 2016.

Item 7. Management's discussion and analysis of financial condition and results of operations

The following discussion and analysis of the Company's financial condition and results of operations should be read together with the financial statements and the related notes included elsewhere herein and the description of the Company's business and reportable segments in item 1 above. This discussion contains forward-looking statements that involve risks and uncertainties. Our actual results may differ materially from those discussed in forward-looking statements. Factors that might cause a difference include, but are not limited to, those discussed under cautionary note regarding forward-looking statements below and in risk factors in part I, item 1A of this Form 10-K. References herein to the "Company," "we," "us," or "our" refer to Walgreens Boots Alliance, Inc. and its subsidiaries, and in each case do not include unconsolidated partially-owned entities, except as otherwise indicated or the context otherwise requires.

Certain amounts in the Consolidated Financial Statements and associated notes may not add due to rounding. All percentages have been calculated using unrounded amounts for fiscal 2019.

INTRODUCTION AND SEGMENTS

Walgreens Boots Alliance, Inc. ("Walgreens Boots Alliance") and its subsidiaries are a global leader in retail and wholesale pharmacy. Its operations are conducted through three reportable segments:

- Retail Pharmacy USA;
- Retail Pharmacy International; and
- Pharmaceutical Wholesale

See note 16, segment reporting and note 17, sales further information.

FACTORS AFFECTING OUR RESULTS AND COMPARABILITY

The Company has been, and we expect it to continue to be, affected by a number of factors that may cause actual results to differ from our current expectations. These factors include: financial performance of our equity method investees, including AmerisourceBergen; the influence of certain holidays; seasonality; foreign currency rates; changes in vendor, payer and customer relationships and terms and associated reimbursement pressure; strategic transactions and acquisitions, including the acquisition of stores and other assets from Rite Aid; joint ventures and other strategic collaborations; changes in laws, including the U.S. tax law changes; changes in trade, tariffs, including trade relations between the United States and China, and international relations, including the UK's proposed withdrawal from the European Union and its impact on our operations and prospects and those of our customers and counterparties; the timing and magnitude of cost reduction initiatives, including under our Transformational Cost Management Program (as defined below); fluctuations in variable costs; and general economic conditions in the markets in which the Company operates. These and other factors can affect the Company's operations and net earnings for any period and may cause such results not to be comparable to the same period in previous years. The results presented in this report are not necessarily indicative of future operating results.

TRANSFORMATIONAL COST MANAGEMENT PROGRAM

On December 20, 2018, the Company announced a transformational cost management program that is expected to deliver in excess of \$1.5 billion of annual cost savings by fiscal 2022 (the "Transformational Cost Management Program"). As of the date of this report, the Company now expects to deliver in excess of \$1.8 billion of annual cost savings by fiscal 2022. The Transformational Cost Management Program, which is multi-faceted and includes divisional optimization initiatives, global smart spending, global smart organization and the transformation of the Company's information technology (IT) capabilities, is designed to help the Company achieve increased cost efficiencies. To date, the Company has taken initial actions across all aspects of the Transformational Cost Management Program focus on all reportable segments and the Company's global functions. Divisional optimization within each of the Company's segments includes activities such as optimization of stores which includes plans to close approximately 200 stores in the United Kingdom and approximately 200 locations in the United States.

The Company currently estimates that the Transformational Cost Management Program will result in cumulative pre-tax charges to its GAAP financial results of approximately \$1.9 billion to \$2.4 billion, of which \$1.6 billion to \$2.0 billion are expected to be recorded as exit and disposal activities. The Company estimates that approximately 80% of the cumulative pre-tax charges relating to the Transformational Cost Management Program will result in future cash expenditures, primarily related to lease and other real estate payments, employee severance and technology costs related to the Company's IT transformation.

The Company currently estimates that it will recognize aggregate pre-tax charges to its GAAP financial results related to Transformational Cost Management Program as follows:

- 35 -

Transformational Cost Program Activities

Transformational Cost Program Activities	Range of Charges
Lease obligations and other real estate costs	\$200 to 300 million
Asset impairments ¹	\$400 to 500 million
Employee severance and business transition costs	\$600 to 700 million
Information technology transformation and other exit costs	\$400 to 500 million
Total cumulative pre-tax exit and disposal charges	\$1.6 to 2.0 billion
Other IT transformation costs	\$300 to 400 million
Total estimated pre-tax charges	\$1.9 to 2.4 billion
1	

¹Primarily related to asset write-offs from store closures, information technology and other asset write-offs.

In addition to the impacts discussed above, as a result of the actions related to store closures taken under the Transformational Cost Management Program, the Company expects to record \$508 million of transition adjustments to decrease retained earnings due to the adoption of the new lease accounting standard (Topic 842) that became effective on September 1, 2019. See note 1, summary of major accounting policies, for additional information.

Since the approval of the Transformational Cost Management Program, the Company has recognized aggregate cumulative pre-tax charges to its financial results in accordance with GAAP of \$477 million, of which \$432 million are recorded as exit and disposal activities. See note 3, exit and disposal activities, for additional information.

Costs under the Transformational Cost Management Program, which were primarily recorded in selling, general and administrative expenses and included in the fiscal year ended August 31, 2019 were as follows (in millions):

Twelve Months Ended August 31, 2019	Retai	l Pharmacy USA	Pharmacy rnational	Pl	harmaceutical Wholesale	algreens Boots Alliance, Inc.
Lease obligations and other real estate costs	\$	5	\$ 19	\$	1	\$ 25
Asset impairments ¹		95	67		98	260
Employee severance and business transition costs		42	34		49	125
Information technology transformation and other exit costs		5	10		7	22
Total pre-tax exit and disposal charges	\$	147	\$ 130	\$	154	\$ 432
Other IT transformation costs		42	 3		1	 45
Total pre-tax charges	\$	189	\$ 133	\$	155	\$ 477

¹ Primarily includes write down of leasehold improvements, certain software and inventory.

Transformational Cost Management Program charges are recognized as the costs are incurred over time in accordance with GAAP. The Company treats charges related to the Transformational Cost Management Program as special items impacting comparability of results in its earnings disclosures.

The amounts and timing of all estimates are subject to change until finalized. The actual amounts and timing may vary materially based on various factors. See "cautionary note regarding forward-looking statements" below.

ACQUISITION OF CERTAIN RITE AID ASSETS

On September 19, 2017, the Company announced it had secured regulatory clearance for an amended and restated asset purchase agreement to purchase 1.932 stores, three distribution centers and related inventory from Rite Aid for \$4.375 billion in cash and other consideration. The Company has completed the acquisition of all 1,932 Rite Aid stores and the first distribution center and related inventory, while the transition of the remaining two distribution centers and related inventory remains subject to closing conditions set forth in the amended and restated asset purchase agreement.

The Company expects to complete integration of the acquired stores and related assets by the end of fiscal 2020, at an estimated total cost of approximately \$850 million, which is reported as acquisition-related costs. The Company has recognized cumulative pre-tax charges for the fiscal year 2019 and 2018 of \$295 million and \$221 million, respectively, related to integration of the acquired stores and related assets. In addition, the Company expects to spend approximately \$500 million on store conversions and related activities. The Company expects annual synergies from the transaction of more than \$325 million.

- 36 -

which are expected to be fully realized within four years of the initial closing of this transaction and derived primarily from procurement, cost savings and other operational matters.

The amounts and timing of all estimates are subject to change until finalized. The actual amounts and timing may vary materially based on various factors. See "cautionary note regarding forward-looking statements" below.

STORE OPTIMIZATION PROGRAM

On October 24, 2017, the Company's Board of Directors approved a plan to implement a program (the "Store Optimization Program") to optimize store locations through the planned closure of approximately 600 stores and related assets within the Company's Retail Pharmacy USA segment upon completion of the acquisition of certain stores and related assets from Rite Aid. As of the date of this report, the Company expects to close approximately 750 stores, of which the majority have been closed as part of this program. The actions under the Store Optimization Program commenced in March 2018 and are expected to be complete by the end of fiscal 2020. The Store Optimization Program is expected to result in cost savings of approximately \$350 million per year to be fully delivered by the end of fiscal 2020.

The Company currently estimates that it will recognize cumulative pre-tax charges to its GAAP financial results of approximately \$350 million, including costs associated with lease obligations and other real estate costs and employee severance and other exit costs. The Company expects to incur pre-tax charges of approximately \$160 million for lease obligations and other real estate costs and approximately \$190 million for employee severance and other exit costs. The Company estimates that substantially all of these cumulative pre-tax charges will result in cash expenditures.

Since approval of the Store Optimization Program, the Company has recognized cumulative pre-tax charges to its GAAP financial results of \$296 million, including \$100 million for fiscal 2018 and \$196 million for fiscal 2019, which were primarily recorded within selling, general and administrative expenses. Cumulative pre-tax charges included \$138 million related to lease obligations and other real estate costs and \$158 million in employee severance and other exit costs.

Store Optimization Program charges are recognized as the costs are incurred over time in accordance with GAAP. The Company treats charges related to the Store Optimization Program as special items impacting comparability of results in its earnings disclosures.

The amounts and timing of all estimates are subject to change until finalized. The actual amounts and timing may vary materially based on various factors. See "cautionary note regarding forward-looking statements" below.

U.S. TAX LAW CHANGES

In connection with the U.S. tax law changes enacted in December 2017 and in accordance with SEC Staff Accounting Bulletin 118, the Company completed its analysis of the income tax effects of the U.S. tax law changes during the three months ended February 28, 2019. The incremental net tax benefit recorded upon completion of the analysis of the income tax effects of the U.S. tax law changes was not material to our Consolidated Financial Statements.

While the Company completed its analysis of the income tax effects of the U.S. tax law changes, the final impact of the U.S. tax law changes may differ from this analysis, due to, among other things, technical clarifications from the U.S. Department of the Treasury and the Internal Revenue Service ("IRS"), interpretations of the U.S. tax law changes and actions the Company may take. The Company will continue to evaluate the impact of any future authoritative guidance with respect to the U.S. tax law changes.

During 2019, the U.S. Treasury Department issued regulations to apply retroactively covering certain components of the 2017 U.S. tax law changes. Certain guidance included in these regulations is inconsistent with the Company's interpretation that led to the recognition of \$247 million of tax benefits in prior periods. Despite this new guidance, the Company remains confident in its interpretation of the U.S. tax law changes and intends to defend this position through litigation, if necessary. However, if the Company is ultimately unsuccessful in defending its position, it may be required to reverse all or a portion of the benefits previously recorded.

As the Company repatriates the undistributed earnings of our foreign subsidiaries for use in the United States, the earnings from our foreign subsidiaries will generally not be subject to U.S. federal tax. The Company continuously evaluates the amount of foreign earnings that are not necessary to be permanently reinvested in our foreign subsidiaries.

INVESTMENT IN AMERISOURCEBERGEN

- 37 -

As of August 31, 2019, the Company owned 56,854,867 shares of AmerisourceBergen common stock (representing approximately 27% of its outstanding common stock) and may, subject to certain conditions, acquire up to an additional 8,398,752 AmerisourceBergen shares in the open market.

The Company accounts for its investment in AmerisourceBergen using the equity method of accounting, subject to a two-month reporting lag, with the net earnings attributable to the investment classified within the operating income of the Company's Pharmaceutical Wholesale segment. The financial performance of AmerisourceBergen, including any charges which may arise relating to its ongoing opioid litigation, will impact the Company's results of operations. Additionally, a substantial and sustained decline in the price of AmerisourceBergen's common stock could trigger an impairment evaluation of our investment. These considerations may materially and adversely affect the Company's financial condition and results of operations.

For more information, see "Business - Relationship with AmerisourceBergen" and note 5, equity method investments to the Consolidated Financial Statements.

THE IMPACT OF BREXIT

In June 2016, voters in the United Kingdom approved an advisory referendum to withdraw from the European Union, which proposed exit (and the political, economic and other uncertainties it has raised) is commonly referred to as "Brexit". Since the Brexit vote in June 2016, there has been significant volatility in the global stock markets and currency exchange rates, as well as challenging market conditions in the United Kingdom. In March 2017, the United Kingdom formally started the process to leave the European Union. An original exit date was set for March 29, 2019 but following the UK parliament's rejection of a negotiated outcome, the leaders of the member countries of the European Union have agreed to multiple extensions of the deadline for Brexit and there can be no assurance regarding the terms, timing or consummation of any such arrangements. Although we continue to actively monitor the ongoing potential impacts of Brexit and will seek to minimize its impact on our business, if the UK's membership in the European Union terminates without an agreement, these conditions could continue and there could be increased costs from tariffs on trade between the United Kingdom and European Union and disruptions to the free movement of goods, services and people between the United Kingdom and the European union and other parties. Further, uncertainty around and developments regarding these and related issues has contributed to deteriorating market conditions and could further adversely impact consumer and investor confidence and the economy of the United Kingdom and the economies of other countries in which we operate and cause significant volatility in currency exchange rates. Given the lack of comparable precedent, it is unclear what financial, trade, regulatory and legal implications the withdrawal of the United Kingdom from the European Union will have on our business, particularly European operations; however, Brexit and its related effects could have a material impact on the Company's consolidated financial pos

EXECUTIVE SUMMARY

The following table presents certain key financial statistics for the Company for fiscal 2019, 2018 and 2017:

	(in millions, except per share amounts)			iounts)		
		2019		2018		2017
Sales	\$	136,866	\$	131,537	\$	118,214
Gross profit		30,076		30,792		29,162
Selling, general and administrative expenses ¹		25,242		24,694		23,813
Equity earnings in AmerisourceBergen		164		191		135
Operating income ¹		4,998		6,289		5,484
Adjusted operating income (Non-GAAP measure) ²		6,942		7,679		7,467
Earnings before interest and income tax provision		5,231		6,591		5,546
Net earnings attributable to Walgreens Boots Alliance, Inc.		3,982		5,024		4,078
Adjusted net earnings attributable to Walgreens Boots Alliance, Inc. (Non-GAAP measure) ²		5,529		5,985		5,503
Net earnings per common share – diluted		4.31		5.05		3.78
Adjusted net earnings per common share – diluted (Non-GAAP measure) ²		5.99		6.02		5.10

- 38 -

	Percentage increases (decreases)				
	2019	2018	2017		
Sales	4.1	11.3	0.7		
Gross profit	(2.3)	5.6	(2.4)		
Selling, general and administrative expenses ¹	2.2	3.7	(0.1)		
Operating income ¹	(20.5)	14.7	(9.7)		
Adjusted operating income (Non-GAAP measure) ²	(9.6)	2.8	2.6		
Earnings before interest and income tax provision	(20.6)	18.8	(3.4)		
Net earnings attributable to Walgreens Boots Alliance, Inc.	(20.7)	23.2	(2.3)		
Adjusted net earnings attributable to Walgreens Boots Alliance, Inc. (Non-GAAP measure) ²	(7.6)	8.8	9.9		
Net earnings per common share – diluted	(14.6)	33.6	(1.0)		
Adjusted net earnings per common share – diluted (Non-GAAP measure) ²	(0.5)	18.0	11.1		
		Percent to sales			
	2019	2018	2017		
Gross margin	22.0	23.4	24.7		
Selling, general and administrative expenses ¹	18.4	18.8	20.1		

¹ The Company adopted ASU 2017-07, Compensation - Retirement Benefits (Topic 715) as of September 1, 2018 (fiscal 2019) on a retrospective basis. The impact on our previously reported net periodic costs as a result of the retrospective adoption of this standard results in a reclassification from selling, general and administrative expenses to other income of \$125 million and \$73 million, for the fiscal years ended August 31, 2018 and 2017, respectively. See note 1, summary of major accounting policies, for additional information.

² See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.

WALGREENS BOOTS ALLIANCE RESULTS OF OPERATIONS

The following information summarizes our results of operations for fiscal 2019 compared to fiscal 2018. For discussion related to the results of operations by segment for fiscal 2018 compared to fiscal 2017, refer to Part II, Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations in our fiscal 2018 Form 10-K, which was filed with the United States Securities and Exchange Commission on October 11, 2018.

Fiscal 2019 compared to fiscal 2018

Fiscal 2019 net earnings attributable to Walgreens Boots Alliance decreased 20.7 percent to \$4.0 billion, while diluted net earnings per share decreased 14.6 percent to \$4.31 compared with the prior year. The decrease primarily reflects operating performance, including costs related to the Company's Transformational Cost Management Program, partially offset by certain legal and regulatory accruals in the prior year. Diluted net earnings per share was positively affected by a lower number of shares outstanding compared with the prior year. Net earnings and diluted earnings per share were negatively by 0.8 percentage points and 0.9 percentage points, respectively, as a result of currency translation.

Other income for fiscal 2019 was \$233 million compared to \$302 million for fiscal 2018. Other income for fiscal 2019 primarily reflects gains resulting from the termination of the option granted to Rite Aid to become a member of the Company's group purchasing organization. Results for fiscal 2018 primarily reflect the gain on sale of the Company's equity interest in Premise Health Holding Corp, partially offset by the impairment of the Company's equity method investment in Guangzhou Pharmaceuticals Corporation in the prior year period.

Interest was a net expense of \$704 million and \$616 million in fiscal 2019 and 2018, respectively.

The effective tax rate for fiscal 2019 and 2018 was 13.0% and 16.7%, respectively. The net decrease in the effective tax rate was primarily attributable to the reduced U.S. statutory tax rate as a result of the U.S tax law changes.

Adjusted diluted net earnings per share (Non-GAAP measure) fiscal 2019 compared to fiscal 2018

Adjusted net earnings attributable to Walgreens Boots Alliance in fiscal 2019 decreased 7.6 percent to \$5.5 billion compared with the prior year. Adjusted diluted net earnings per share in fiscal 2019 decreased 0.5 percent to \$5.99 compared with the prior year. Adjusted net earnings and adjusted diluted earnings per share were negatively impacted by 0.9 percentage points and 1.0 percentage points, respectively, as a result of currency translation.

Excluding the impact of currency translation, the decrease in adjusted net earnings for fiscal 2019 primarily reflects a decrease in operating performance, partly offset by a decrease in the adjusted effective tax rate. Adjusted diluted net earnings per share was positively affected by a lower number of shares outstanding compared with the prior year. See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.

RESULTS OF OPERATIONS BY SEGMENT

The following information summarizes our results of operations by segment for fiscal 2019 compared to fiscal 2018. For discussion related to the results of operations by segment for fiscal 2018 compared to fiscal 2017, refer to Part II, Item 7. Management's Discussion and Analysis of Financial Condition and Results of Operations in our fiscal 2018 Form 10-K, which was filed with the United States Securities and Exchange Commission on October 11, 2018.

Retail Pharmacy USA

This division comprises the retail pharmacy business operating in the United States.

	(in mi	(in millions, except location amounts)						
	2019		2018		2017			
Sales	\$ 104,53	2 \$	98,392	\$	87,302			
Gross profit	23,51	1	23,758		22,450			
Selling, general and administrative expenses ¹	19,42	4	18,971		18,356			
Operating income ¹	4,08	8	4,787		4,094			
Adjusted operating income (Non-GAAP measure) ^{1,2}	5,25	5	5,814		5,606			
Number of prescriptions ³	843	7	823.1		764.4			
30 day aquivalent procernitions ^{3,4}	1 150	1	1 004 4		080 7			

30-day equivalent prescriptions ^{3,4}	1,150.1	1,094.4	989.7
Number of locations at period end	9,285	9,569	8,109

	Percent	Percentage increases (decreases)				
	2019	2018	2017			
Sales	6.2	12.7	4.2			
Gross profit	(1.0)	5.8	0.6			
Selling, general and administrative expenses ¹	2.4	3.4	2.5			
Operating income ¹	(14.6)	16.9	(7.3)			
Adjusted operating income (Non-GAAP measure) ^{1,2}	(9.6)	3.7	4.4			
Comparable store sales ⁵	2.0	1.5	2.8			
Pharmacy sales	8.6	17.2	7.3			
Comparable pharmacy sales ⁵	4.0	3.4	4.7			
Retail sales	—	2.4	(2.4)			
Comparable retail sales ⁵	(2.4)	(2.4)	(1.0)			
Comparable number of prescriptions ^{4,5}	(0.1)	0.8	4.0			
Comparable 30-day equivalent prescriptions ^{3,4,5}	3.0	3.5	7.1			
		Percent to sales				
	2019	2018	2017			
Gross margin	22.5	24.1	25.7			

C . 11' 1	and administrative expenses ¹
Selling general	and administrative expenses.
ocinite, general	und dummistidit ve expenses

- 40 -

18.6

19.3

21.0

Table of Contents

- ¹ The Company adopted ASU 2017-07 Topic 715 as of September 1, 2018 (fiscal 2019) on a retrospective basis. The impact for Retail Pharmacy USA on our previously reported net periodic costs as a result of the retrospective adoption of this standard results in a reclassification from selling, general and administrative expenses to other income (expense) of \$109 million and \$101 million, for the fiscal years ended August 31, 2018 and 2017, respectively. See note 1, summary of major accounting policies, for additional information.
- ² See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.
- ³ Includes immunizations.
- ⁴ Includes the adjustment to convert prescriptions greater than 84 days to the equivalent of three 30-day prescriptions. This adjustment reflects that these prescriptions include approximately three times the amount of product days supplied compared to a normal prescription.
- ⁵ Comparable stores are defined as those that have been open for at least twelve consecutive months without closure for seven or more consecutive days and without a major remodel or being subject to a natural disaster in the past twelve months. Relocated stores are not included as comparable stores for the first twelve months after the relocation. Acquired stores are not included as comparable stores for the first twelve months after acquisition or conversion, when applicable, whichever is later. The method of calculating comparable sales varies across the retail industry. As a result, our method of calculating comparable sales may not be the same as other retailers' methods.

Sales fiscal 2019 compared to fiscal 2018

The Retail Pharmacy USA division's sales for fiscal 2019 increased by 6.2% to \$104.5 billion. Sales in comparable stores increased by 2.0% in fiscal 2019. The Company operated 9,285 locations (9,277 retail stores) as of August 31, 2019, compared to 9,569 locations (9,560 retail stores) a year earlier.

Pharmacy sales increased by 8.6% in fiscal 2019 and represented 73.8% of the division's sales. The increase in fiscal 2019 is due to higher brand inflation and growth in central specialty and reflects the impact of the acquired Rite Aid stores. In fiscal 2018, pharmacy sales increased 17.2% and represented 72.2% of the division's sales. Comparable pharmacy sales increased 4.0% in fiscal 2019 compared to an increase of 3.4% in fiscal 2018. The effect of generic drugs, which have a lower retail price, replacing brand name drugs reduced prescription sales by 1.2% in fiscal 2019 compared to a reduction of 1.4% in fiscal 2018. The effect of generics on division sales was a reduction of 0.8% in fiscal 2019 compared to a reduction of 0.9% for fiscal 2018. Third-party sales, where reimbursement is received from managed care organizations, governmental agencies, employers or private insurers, were 97.1% of prescription sales for fiscal 2019 compared to 98.3% for fiscal 2018. The total number of prescriptions (including immunizations) filled in fiscal 2019 was 843.7 million compared to 823.1 million in fiscal 2018. Prescriptions (including immunizations) adjusted to 30-day equivalents were 1,150.1 million in fiscal 2019 compared to 1,094.4 million in fiscal 2018.

Retail sales were flat in fiscal 2019 and were 26.2% of the division's sales. In comparison, fiscal 2018 retail sales increased 2.4% and comprised 27.8% of the division's sales. Comparable retail sales decreased 2.4% in each of fiscal 2019 and fiscal 2018. The decrease in comparable retail sales in fiscal 2019 was primarily due to the continued de-emphasis of tobacco.

Operating income fiscal 2019 compared to fiscal 2018

Retail Pharmacy USA division's operating income for fiscal 2019 decreased 14.6% to \$4.1 billion. The decrease was primarily due to lower gross margin partially offset by a reduction in selling, general and administrative expenses as a percentage of sales.

Gross margin was 22.5% in fiscal 2019 compared to 24.1% in fiscal 2018. Gross margin was negatively impacted in the current fiscal year by pharmacy margins, which were negatively impacted by reimbursement pressure. The decrease in pharmacy margins was partially offset by the favorable impact of procurement efficiencies.

Selling, general and administrative expenses as a percentage of sales were 18.6% in fiscal 2019 compared to 19.3% in fiscal 2018. As a percentage of sales, expenses in the current fiscal year were lower primarily due to a higher mix of specialty sales and lower expenses related to certain legal and regulatory accruals in the prior year.

Adjusted operating income (Non-GAAP measure) fiscal 2019 compared to fiscal 2018

Retail Pharmacy USA division's adjusted operating income for fiscal 2019 decreased 9.6% to \$5.3 billion. The decrease was primarily due to lower gross margins partially offset by a reduction in selling, general, and administrative expenses as a percentage of sales. See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.

- 41 -

Retail Pharmacy International

This division comprises retail pharmacy businesses operating in countries outside of the United States and in currencies other than the U.S. dollar, including the British pound sterling, Euro, Chilean peso and Mexican peso. Therefore, the division's results are impacted by movements in foreign currency exchange rates. See item 7A, quantitative and qualitative disclosure about market risk, foreign currency exchange rate risk, for further information on currency risk.

	(in millions, except location amounts)					
		2019		2018		2017
Sales	\$	11,462	\$	12,281	\$	11,813
Gross profit		4,522		4,958		4,753
Selling, general and administrative expenses ¹		4,084		4,134		3,982
Operating income ¹		438		824		771
Adjusted operating income (Non-GAAP measure) ^{1,2}		747		929		939
Number of locations at period end		4,605		4,767		4,722

	Percent	Percentage increases (decreases)				
	2019	2018	2017			
Sales	(6.7)	4.0	(10.9)			
Gross profit	(8.8)	4.3	(12.5)			
Selling, general and administrative expenses ¹	(1.2)	3.8	(8.4)			
Operating income ¹	(46.8)	6.9	(29.1)			
Adjusted operating income (Non-GAAP measure) ^{1,2}	(19.6)	(1.1)	(22.6)			
Comparable store sales ³	(6.4)	4.7	(10.6)			
Comparable store sales in constant currency ^{3,4}	(1.7)	(1.4)	(0.2)			
Pharmacy sales	(6.4)	4.3	(10.5)			
Comparable pharmacy sales ³	(5.7)	4.7	(10.7)			
Comparable pharmacy sales in constant currency ^{3,4}	(0.9)	(1.2)	(1.0)			
Retail sales	(6.8)	3.8	(11.1)			
Comparable retail sales ³	(6.8)	4.7	(10.6)			
Comparable retail sales in constant currency ^{3,4}	(2.2)	(1.5)	0.2			
		Percent to sales				
	2019	2018	2017			
Gross margin	39.5	40.4	40.2			

Selling, general and administrative expenses¹

¹ The Company adopted ASU 2017-07, Compensation - Retirement Benefits (Topic 715) as of September 1, 2018 (fiscal 2019) on a retrospective basis. The impact for Retail Pharmacy International on our previously reported net periodic costs as a result of the retrospective adoption of this standard results in a reclassification from selling, general and administrative expenses to other income (expense) of \$18 million and \$(30) million, for the fiscal years ended August 31, 2018 and 2017, respectively. See note 1, summary of major accounting policies, for additional information.

35.6

33.7

33.7

² See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.

³ Comparable stores are defined as those that have been open for at least twelve consecutive months without closure for seven or more consecutive days and without a major remodel or being subject to a natural disaster in the past twelve months. Relocated stores are not included as comparable stores for the first twelve months after the relocation. Acquired stores are not included as comparable stores for the first twelve months after the relocation. Acquired stores are not included as comparable stores for the first twelve months after acquisition or conversion, when applicable, whichever is later. The method of calculating comparable sales varies across the retail industry. As a result, our method of calculating comparable sales may not be the same as other retailers' methods.

- 42 -

Table of Contents

⁴ The Company presents certain information related to current period operating results in "constant currency," which is a non-GAAP financial measure. These amounts are calculated by translating current period results at the foreign currency exchange rates used in the comparable period in the prior year. The Company presents such constant currency financial information because it has significant operations outside of the United States reporting in currencies other than the U.S. dollar and this presentation provides a framework to assess how its business performed excluding the impact of foreign currency exchange rate fluctuations. See "--Non-GAAP Measures" below.

Sales fiscal 2019 compared to fiscal 2018

Retail Pharmacy International division's sales for fiscal 2019 decreased 6.7% to \$11.5 billion. Sales in comparable stores decreased 6.4%. The negative impact of currency translation on sales and comparable sales was 4.6 percentage points, and 4.7 percentage points, respectively, and comparable store sales in constant currency decreased 1.7%.

Pharmacy sales decreased 6.4% in fiscal 2019 and represented 35.6% of the division's sales. Comparable pharmacy sales decreased 5.7%. The negative impact of currency translation on pharmacy sales and comparable pharmacy sales was 4.8 percentage points and 4.8 percentage points, respectively. Comparable pharmacy sales in constant currency decreased 0.9 percentage points mainly due to lower National Health Service funding levels and lower volume in Boots UK.

Retail sales decreased 6.8% for fiscal 2019 and represented 64.4% of the division's sales. Comparable retail sales decreased 6.8%. The negative impact of currency translation on each of retail sales and comparable retail sales was 4.5 percentage points and 4.6 percentage points respectively. Comparable retail sales in constant currency decreased 2.2% reflecting lower Boots UK retail sales in a challenging market place.

Operating income fiscal 2019 compared to fiscal 2018

Retail Pharmacy International division's operating income for fiscal 2019 decreased 46.8% to \$438 million. Operating income was negatively impacted by 2.8 percentage points (\$23 million) of currency translation. Excluding the impact of currency translation, the decrease in operating income was primarily due to lower sales and gross margin. The remaining decrease is due to higher selling, general and administrative expenses as a percentage of sales, which includes \$89 million of Transformational Cost Management Program expenses and \$73 million of impairment for indefinite-lived pharmacy licenses.

Gross profit decreased 8.8% in fiscal 2019. Gross profit was negatively impacted by 4.4 percentage points (\$219 million) of currency translation, the remaining decrease was mainly due lower sales and gross margin.

Selling, general and administrative expenses decreased 1.2% from fiscal 2018. Expenses were positively impacted by 4.7 percentage points (\$196 million) as a result of currency translation. As a percentage of sales, selling, general and administrative expenses were 35.6% in fiscal 2019, compared to 33.7% in the prior fiscal year.

Adjusted operating income (Non-GAAP measure) fiscal 2019 compared to fiscal 2018

Retail Pharmacy International division's adjusted operating income for fiscal 2019 decreased 19.6% to \$747 million. Adjusted operating income was negatively impacted by 3.4 percentage points (\$31 million) of currency translation. Excluding the impact of currency translation, the decrease in adjusted operating income was primarily due to lower gross margin and higher selling, general and administrative expenses as a percentage of sales and lower sales. See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.

Pharmaceutical Wholesale

This division includes pharmaceutical wholesale businesses operating in currencies other than the U.S. dollar including the British pound sterling, Euro and Turkish lira. Therefore, the division's results are impacted by movements in foreign currency exchange rates. See item 7A, quantitative and qualitative disclosure about market risk, foreign currency exchange rate risk, for further information on currency risk.

	(in millions)				
	 2019		2018		2017
Sales	\$ 23,053	\$	23,006	\$	21,188
Gross profit	2,041		2,081		1,965
Selling, general and administrative expenses ¹	1,734		1,594		1,481
Equity earnings from AmerisourceBergen	164		191		135
Operating income ¹	471		678		619
Adjusted operating income (Non-GAAP measure) ^{1,2}	939		936		922

- 43 -

	Percent	Percentage increases (decreases)					
	2019	2018	2017				
Sales	0.2	8.6	(6.1)				
Gross profit	(1.9)	5.9	(7.8)				
Selling, general and administrative expenses ¹	8.8	7.6	(6.7)				
Equity earnings from AmerisourceBergen	(14.2)	41.5	264.9				
Operating income ¹	(30.4)	9.5	6.7				
Adjusted operating income (Non-GAAP measure) ^{1,2}	0.4	1.5	30.0				
Comparable sales ³	0.2	8.6	(3.9)				
Comparable sales in constant currency ^{3,4}	8.0	4.2	4.7				
		Percent to sales					
	2019	2018	2017				
Gross margin	8.9	9.0	9.3				
Selling, general and administrative expenses ¹	7.5	6.9	7.0				

¹ The Company adopted ASU 2017-07, Compensation - Retirement Benefits (Topic 715) as of September 1, 2018 (fiscal 2019) on a retrospective basis. The impact for Pharmaceutical Wholesale on our previously reported net periodic costs as a result of the retrospective adoption of this standard results in a reclassification from selling, general and administrative expenses to other income (expense) of \$(2) million and \$2 million, for the fiscal years ended August 31, 2018 and 2017, respectively. See note 1, summary of major accounting policies, for additional information.

- ² See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.
- ³ Comparable Sales are defined as sales excluding acquisitions and dispositions.
- ⁴ The Company presents certain information related to current period operating results in "constant currency," which is a non-GAAP financial measure. These amounts are calculated by translating current period results at the foreign currency exchange rates used in the comparable period in the prior year. The Company presents such constant currency financial information because it has significant operations outside of the United States reporting in currencies other than the U.S. dollar and this presentation provides a framework to assess how its business performed excluding the impact of foreign currency exchange rate fluctuations. See "--Non-GAAP Measures" below.

Sales fiscal 2019 compared to fiscal 2018

Pharmaceutical Wholesale division's sales for the fiscal 2019 increased 0.2% to \$23.1 billion. Comparable sales, which exclude acquisitions and dispositions, increased 0.2%.

Sales and comparable sales were negatively impacted by 7.8 percentage points as a result of currency translation. Comparable sales in constant currency increased 8.0%, reflecting a customer contract change in the UK and growth in emerging markets.

Operating income fiscal 2019 compared to fiscal 2018

Pharmaceutical Wholesale division's operating income for fiscal 2019, which included \$164 million from the Company's share of equity earnings in AmerisourceBergen, decreased 30.4% to \$471 million. Operating income was negatively impacted by 6.1 percentage points (\$41 million) as a result of currency translation. The remaining decrease was due to costs related to the Transformational Cost Management Program in fiscal 2019.

Gross profit decreased 1.9% from the prior fiscal year. Gross profit was negatively impacted by 7.2 percentage points (\$149 million) as a result of currency translation. Excluding the impact of currency translation, the increase in gross profit was primarily due to sales growth, partially offset by lower gross margin.

Selling, general and administrative expenses increased 8.8% from the prior fiscal year. Selling, general and administrative expenses were positively impacted by 6.8 percentage points (\$108 million) as a result of currency translation. Excluding the impact of currency translation, the increase in selling, general and administrative expenses was primarily due to costs related to the Transformational Cost Management Program. As a percentage of sales, selling, general and administrative expenses were 7.5% in fiscal 2019, compared to 6.9% in fiscal 2018.

- 44 -

Adjusted operating income (Non-GAAP measure) fiscal 2019 compared to fiscal 2018

Pharmaceutical Wholesale division's adjusted operating income for fiscal 2019, which included \$397 million from the Company's share of adjusted equity earnings in AmerisourceBergen, increased 0.4% to \$939 million. Adjusted operating income was negatively impacted by 5.5 percentage points (\$51 million) as a result of currency translation.

Excluding the contribution from the Company's share of adjusted equity earnings in AmerisourceBergen and the negative impact of currency translation, adjusted operating income increased 4.3% (\$25 million) over the prior fiscal year, primarily due to higher sales and lower selling, general and administrative expenses as a percentage of sales, partially offset by lower gross margin. See "--Non-GAAP Measures" below for a reconciliation to the most directly comparable financial measure calculated in accordance with GAAP and related disclosures.

NON-GAAP MEASURES

The following information provides reconciliations of the supplemental non-GAAP financial measures, as defined under the rules of the Securities and Exchange Commission, presented herein to the most directly comparable financial measures calculated and presented in accordance with GAAP. The Company has provided the non-GAAP financial measures, which are not calculated or presented in accordance with GAAP, as supplemental information and in addition to the financial measures that are calculated and presented in accordance with GAAP.

These supplemental non-GAAP financial measures are presented because the Company's management has evaluated its financial results both including and excluding the adjusted items or the effects of foreign currency translation, as applicable, and believes that the supplemental non-GAAP financial measures presented provide additional perspective and insights when analyzing the core operating performance of the Company from period to period and trends in its historical operating results. These supplemental non-GAAP financial measures should not be considered superior to, as a substitute for or as an alternative to, and should be considered in conjunction with, the GAAP financial measures presented.

The Company also presents certain information related to current period operating results in "constant currency," which is a non-GAAP financial measure. These amounts are calculated by translating current period results at the foreign currency exchange rates used in the comparable period in the prior year. The Company presents such constant currency financial information because it has significant operations outside of the United States reporting in currencies other than the U.S. dollar and such presentation provides a framework to assess how its business performed excluding the impact of foreign currency exchange rate fluctuations.

	(in millions)													
				Twelve	mo	nths ended August	31, 2019							
	Retail Pharmacy USA			tail Pharmacy International		Pharmaceutical Wholesale	Eliminations		greens Boots liance, Inc.					
Operating income (GAAP) ¹	\$	4,088	\$	438	\$	471	\$ 1	\$	4,998					
Acquisition-related amortization and impairment ²		315		173		78	—		567					
Transformational cost management		189		133		155	—		477					
Acquisition-related costs		300		3		1	—		303					
Adjustments to equity earnings in AmerisourceBergen		_		—		233	—		233					
Store optimization		196		—		—	—		196					
LIFO provision		136		—		—	—		136					
Certain legal and regulatory accruals and settlements ³		31				_	_		31					
Adjusted operating income (Non-GAAP measure) ¹	\$	5,255	\$	747	\$	939	\$ 1	\$	6,942					



						(in millions)			
				Twelve	mont	ths ended August 3	31, 2018		
	Retail I Pharmacy USA			tail Pharmacy nternational	Р	harmaceutical Wholesale	Eliminations		Walgreens Boots Alliance, Inc.
Operating income (GAAP) ¹	\$	4,787	\$	824	\$	678	\$ —		\$ 6,289
Acquisition-related amortization		260		105		83			448
Acquisition-related costs		231		—		—	—		231
Adjustments to equity earnings in AmerisourceBergen				—		175			175
Store optimization		100		—		—	—		100
LIFO provision		84		—		—			84
Certain legal and regulatory accruals and settlements ³		284		—		—			284
Asset recovery		(15)		—		—			(15)
Hurricane-related costs		83		_		—			83
Adjusted operating income (Non-GAAP measure) ¹	\$	5,814	\$	929	\$	936	\$	-	\$ 7,679

						(in millions)								
	Twelve months ended August 31, 2017													
		Retail Retail Pharmacy Pharmaceutical macy USA International Wholesale					Eliminatio	ns	Walgreens as Alliance,					
Operating income (GAAP) ¹	\$	4,094	\$	771	\$	619	\$	_	\$	5,484				
Acquisition-related amortization		152		101		79		—		332				
Acquisition-related costs		474		—		—		—		474				
Adjustments to equity earnings in AmerisourceBergen				—		187		—		187				
LIFO provision		166		—		—		—		166				
Asset recovery		(11)		—		—		—		(11)				
Cost transformation		731		67		37		—		835				
Adjusted operating income (Non-GAAP measure) ¹	\$	5,606	\$	939	\$	922	\$	—	\$	7,467				

¹ The Company adopted new accounting guidance in Accounting Standards Update 2017-07 as of September 1, 2018 (fiscal 2019) on a retrospective basis for the Consolidated Statements of Earnings presentation. This change resulted in reclassification of all the other net cost components (excluding service cost component) of net pension cost and net postretirement benefit cost from selling, general and administrative expenses to other income (expense) with no impact on the Company's net earnings.

² Includes impairment of \$73 million for indefinite-lived pharmacy licenses intangible asset recorded during the three months ended August 31, 2019, in the Boots reporting unit within the Retail Pharmacy International segment.

³ Beginning in the quarter ended August 31, 2018, management reviewed and refined its practice to include all charges related to the matters included in certain legal and regulatory accruals and settlements. This non-GAAP measure is presented on a consistent basis for fiscal year 2019.

- 46 -

			(iı	n millions)		
		2019		2018		2017
Net earnings attributable to Walgreens Boots Alliance, Inc. (GAAP)	\$	3,982	\$	5,024	\$	4,078
Adjustments to operating income:						
Acquisition-related amortization and impairment ¹		567		448		332
Transformational cost management		477		—		—
Acquisition-related costs		303		231		474
Adjustments to equity earnings in AmerisourceBergen		233		175		187
Store optimization		196		100		—
LIFO provision		136		84		166
Certain legal and regulatory accruals and settlements ²		31		284		_
Asset recovery		_		(15)		_
Hurricane-related costs		_		83		(11)
Cost Transformation		—		_		835
Total adjustments to operating income		1,944		1,390		1,983
Adjustments to other income (expense):						
Net investment hedging loss (gain)		18		(21)		48
Gain on sale of equity method investment		—		(322)		—
Impairment of equity method investment		—		178		—
Termination of option granted to Rite Aid		(173)		—		—
Total adjustments to other income (expense)		(155)		(165)		48
Adjustments to interest expense, net:						
Prefunded acquisition financing costs				29		203
Total adjustments to interest expense, net				29		203
Total adjustments to interest expense, net				25		200
Adjustments to income tax provision:						
Equity method non-cash tax		18		25		23
U.S. tax law changes ³		(8)		(125)		_
Tax impact of adjustments ⁴		(291)		(193)		(755)
UK tax rate change ³		_		_		(77)
Total adjustments to income tax provision		(281)		(293)		(809)
Adjustments to post tax equity earnings from other equity method investments:						
Adjustments to equity earnings in other equity method investments ⁵		40		—		_
Total adjustments to post tax equity earnings from other equity method investments		40		—		—
Adjusted net earnings attributable to Walgreens Boots Alliance, Inc. (Non-GAAP	¢.		ф.		φ	
measure)	\$	5,529	\$	5,985	\$	5,503

- 47 -

	2019	2018	2017
Diluted net earnings per common share (GAAP)	\$ 4.31	\$ 5.05	\$ 3.78
Adjustments to operating income	2.10	1.40	1.84
Adjustments to other income (expense)	(0.17)	(0.17)	0.04
Adjustments to interest expense, net	—	0.03	0.19
Adjustments to income tax provision	(0.30)	(0.29)	(0.75)
Adjustments to post tax equity earnings from other equity method investments	0.04	—	
Adjusted diluted net earnings per common share (Non-GAAP measure)	\$ 5.99	\$ 6.02	\$ 5.10
Weighted average common shares outstanding, diluted	923.5	995.0	1,078.5

- ¹ Includes impairment of \$73 million for indefinite-lived pharmacy licenses intangible asset recorded during the three months ended August 31, 2019, in the Boots reporting unit within the Retail Pharmacy International segment.
- ² Beginning in the quarter ended August 31, 2018, management reviewed and refined its practice to include all charges related to the matters included in certain legal and regulatory accruals and settlements. This non-GAAP measure is presented on a consistent basis for fiscal year 2019.
 ³ Discrete tax, only items.
- ³ Discrete tax-only items.
- ⁴ Represents the adjustment to the GAAP basis tax provision commensurate with non-GAAP adjustments.
- ⁵ Beginning in the quarter ended May 31, 2019, management reviewed and refined its practice to reflect the proportionate share of certain equity method investees' non-cash items or unusual or infrequent items consistent with the Company's non-GAAP measures in order to provide investors with a comparable view of performance across periods. These adjustments include acquisition-related amortization and acquisition-related costs and were immaterial for the prior periods presented. Although the Company may have shareholder rights and board representation commensurate with its ownership interests in these equity method investees, adjustments relating to equity method investments are not intended to imply that the Company has direct control over their operations and resulting revenue and expenses. Moreover, these non-GAAP financial measures have limitations in that they do not reflect all revenue and expenses of these equity method investees.

LIQUIDITY AND CAPITAL RESOURCES

Cash and cash equivalents were \$1.0 billion (including \$0.4 billion in non-U.S. jurisdictions) as of August 31, 2019, compared to \$0.8 billion (including \$0.2 billion in non-U.S. jurisdictions) at August 31, 2018. Short-term investment objectives are primarily to minimize risk and maintain liquidity. To attain these objectives, investment limits are placed on the amount, type and issuer of securities. Investments are principally in U.S. Treasury money market funds and AAA-rated money market funds.

The Company's long-term capital policy is to maintain a strong balance sheet and financial flexibility, reinvest in its core strategies, invest in strategic opportunities that reinforce its core strategies and meet return requirements and return surplus cash flow to stockholders in the form of dividends and share repurchases over the long term. In July 2019, the Company's Board of Directors reviewed and refined the Company's dividend policy to set forth the Company's current intention to increase its dividend each year.

Cash provided by operations and the incurrence of debt are the principal sources of funds for expansion, investments, acquisitions, remodeling programs, dividends to stockholders and stock repurchases. Net cash provided by operating activities was \$5.6 billion in fiscal 2019 compared to \$8.3 billion in fiscal 2018 and \$7.3 billion in fiscal 2017. The \$2.7 billion decrease in cash provided by operating activities includes cash outflows relating to the integration of Rite Aid stores, timing of certain legal and regulatory settlements and higher income taxes paid. Changes in income taxes paid are mainly due to the impact of U.S. tax law changes. Changes in accrued expenses and other liabilities, trade accounts payable and trade accounts receivables include cash outflows relating to the integration of Rite Aid stores. Changes in accrued expenses and other liabilities also include the impact of certain legal and regulatory settlements.

Net cash used for investing activities was \$2.3 billion in fiscal 2019 compared to \$5.5 billion in fiscal 2018 and \$0.8 billion in fiscal 2017. Business, investment and asset acquisitions in fiscal 2019 were \$0.7 billion compared to \$4.8 billion in fiscal 2018. Business, investment and asset acquisitions in fiscal 2019 were \$1.7 billion compared to \$1.4 billion in each of fiscal 2018 and fiscal 2017. Capital expenditures by reporting segment were as follows (in millions):

- 48 -

	2019	2018	2017
Retail Pharmacy USA	\$ 1,323	\$ 1,022	\$ 860
Retail Pharmacy International	275	241	384
Pharmaceutical Wholesale	104	104	107
Total	\$ 1,702	\$ 1,367	\$ 1,351

Significant capital expenditures primarily relate to investments in our stores, integration of Rite Aid and information technology projects.

Net cash used for financing activities in fiscal 2019 was \$3.0 billion compared to \$5.3 billion in fiscal 2018 and \$12.9 billion in fiscal 2017. The Company repurchased shares as part of the stock repurchase programs described below and to support the needs of the employee stock plans totaling \$4.2 billion in fiscal 2019 compared to \$5.2 billion in each of fiscal 2018 and fiscal 2017. Proceeds related to employee stock plans were \$174 million in each of fiscal 2019 and fiscal 2017. Cash dividends paid were \$1.6 billion in fiscal 2019 compared to \$1.7 billion in each of fiscal 2018 and fiscal 2017. In fiscal 2019 there were \$12.4 billion in proceeds compared to \$5.9 billion in proceeds in fiscal 2018 primarily from revolving facilities described below and commercial paper debt. There were no proceeds from debt in fiscal 2017. In fiscal 2019 there were \$10.5 billion payments of debt made primarily for revolving facilities and commercial paper debt compared to \$4.9 billion in fiscal 2018 and \$6.2 billion in fiscal 2017.

The Company believes that cash flow from operations, availability under existing credit facilities and arrangements, current cash and investment balances and the ability to obtain other financing, if necessary, will provide adequate cash funds for the foreseeable working capital needs, capital expenditures at existing facilities, pending acquisitions, dividend payments and debt service obligations for at least the next 12 months. The Company's cash requirements are subject to change as business conditions warrant and opportunities arise. The timing and size of any new business ventures or acquisitions that the Company may complete may also impact its cash requirements.

See item 7A, qualitative and quantitative disclosures about market risk, below for a discussion of certain financing and market risks.

Stock repurchase programs

In April 2017, Walgreens Boots Alliance authorized a stock repurchase program (the "April 2017 stock repurchase program"), which authorized the repurchase of up to \$1.0 billion of Walgreens Boots Alliance common stock prior to the program's expiration on December 31, 2017. In May 2017, the Company completed the April 2017 stock repurchase program, purchasing 11.8 million shares. In June 2017, Walgreens Boots Alliance authorized a stock repurchase program, which authorized the repurchase of up to \$5.0 billion of Walgreens Boots Alliance common stock prior to the program's expiration on August 31, 2018, which authorization was increased by an additional \$1.0 billion in October 2017 (as expanded, the "June 2017 stock repurchase program"). In October 2017, the Company completed the June 2017 stock repurchase program, purchasing 77.4 million shares. In June 2018, Walgreens Boots Alliance authorized a stock repurchase program (the "June 2018 stock repurchase program"), which authorized the repurchase program the "June 2018 stock repurchase program"), which authorized the repurchase program may be stock repurchase program?).

The Company purchased 57 million and 72 million shares under stock repurchase programs in fiscal 2019 and 2018 at a cost of \$3.8 billion and \$4.9 billion, respectively. The Company determines the timing and amount of repurchases, including repurchases to offset anticipated dilution from equity incentive plans, based on its assessment of various factors, including prevailing market conditions, alternate uses of capital, liquidity and the economic environment. The Company has repurchased, and may from time to time in the future repurchase, shares on the open market through Rule 10b5-1 plans, which enable the Company to repurchase shares at times when it otherwise might be precluded from doing so under federal securities laws.

Commercial paper

The Company periodically borrows under its commercial paper program and may continue to borrow under it in future periods. The Company had \$2,400 million commercial paper outstanding as of August 31, 2019 and \$430 million as of August 31, 2018. The Company had average daily commercial paper outstanding of \$2.7 billion and \$1.4 billion at a weighted average interest rate of 3.07% and 2.11% for the fiscal years ended August 31, 2019 and 2018 respectively. The Company had no activity under its commercial paper program during fiscal year 2017.

Financing actions



On June 1, 2016, Walgreens Boots Alliance issued in an underwritten public offering \$1.2 billion of 1.750% notes due 2018 (the "2018 notes"), \$1.5 billion of 2.600% notes due 2021 (the "2021 notes"), \$0.8 billion of 3.100% notes due 2023 (the "2023 notes"), \$1.9 billion of 3.450% notes due 2026 (the "2026 notes") and \$0.6 billion of 4.650% notes due 2046 (the "2046 notes"). Because the merger with Rite Aid was not consummated on or prior to June 1, 2017, the 2018 notes, the 2021 notes and the 2023 notes were redeemed on June 5, 2017 under the special mandatory redemption terms of the indenture governing such notes. The 2026 notes and 2046 notes remain outstanding in accordance with their respective terms.

On February 1, 2017, Walgreens Boots Alliance entered into a \$1.0 billion revolving credit facility (as amended, the "February 2017 Revolving Credit Agreement") with the lenders from time to time party thereto and, on August 1, 2017, Walgreens Boots Alliance entered into an amendment agreement thereto. On January 31, 2019, the February 2017 Revolving Credit Agreement matured and the Company paid all amounts due in connection therewith.

On August 24, 2017, Walgreens Boots Alliance entered into a \$1.0 billion revolving credit agreement with the lenders from time to time party thereto (the "August 2017 Revolving Credit Agreement") and a \$1.0 billion term loan credit agreement with Sumitomo Mitsui Banking Corporation (the "2017 Term Loan Credit Agreement"). On November 30, 2018, in connection with the entrance into the November 2018 Credit Agreement (described below), Walgreens Boots Alliance terminated the 2017 Term Loan Credit Agreement in accordance with its terms and as of such date paid all amounts due in connection therewith. On January 31, 2019, the August 2017 Revolving Credit Agreement matured and the Company paid all amounts due in connection therewith.

On August 29, 2018, Walgreens Boots Alliance entered into a revolving credit agreement (the "August 2018 Revolving Credit Agreement") with the lenders and letter of credit issuers from time to time party thereto. The August 2018 Revolving Credit Agreement is an unsecured revolving credit facility with an aggregate commitment in the amount of \$3.5 billion, with a letter of credit subfacility commitment amount of \$500 million. The facility termination date is the earlier of (a) August 29, 2023, subject to extension thereof pursuant to the August 2018 Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the revolving commitments pursuant to the August 2018 Revolving Credit Agreement. Borrowings under the August 2018 Revolving Credit Agreement will bear interest at a fluctuating rate per annum equal to, at Walgreens Boots Alliance's option, the alternate base rate or the Eurocurrency rate, in each case, plus an applicable margin calculated based on Walgreens Boots Alliance's credit ratings. As of August 31, 2019, there were no borrowings under the August 2018 Revolving Credit Agreement.

On November 30, 2018, Walgreens Boots Alliance entered into a credit agreement (as amended the "November 2018 Credit Agreement") with the lenders from time to time party thereto and, on March 25, 2019, the Company entered into an amendment to such credit agreement reflecting certain changes to the borrowing notice provisions thereto. The November 2018 Credit Agreement includes a \$500 million senior unsecured revolving credit facility and a \$500 million senior unsecured term loan facility. The facility termination date is, with respect to the revolving credit facility, the earlier of (a) May 30, 2020 and (b) the date of termination in whole of the aggregate amount of the revolving commitments pursuant to the November 2018 Credit Agreement. Borrowings under the November 2018 Credit Agreement will bear interest at a fluctuating rate per annum equal to, at Walgreens Boots Alliance's option, the alternate base rate or the Eurocurrency rate, in each case, plus an applicable margin calculated based on Walgreens Boots Alliance's credit ratings. As of August 31, 2019, there were \$0.5 billion of borrowings under the November 2018 Credit Agreement.

On December 5, 2018, Walgreens Boots Alliance entered into a \$1.0 billion term loan credit agreement (as amended, the "December 2018 Credit Agreement") with the lenders from time to time party thereto and, on August 9, 2019, the Company entered into an amendment to such credit agreement to permit the Company to borrow, repay and reborrow amounts borrowed thereunder prior to the maturity date. The December 2018 Credit Agreement is a senior unsecured revolving credit facility with a facility termination date of the earlier of (a) January 29, 2021, subject to extension thereof pursuant to the December 2018 Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the December 2018 Credit Agreement. Borrowings under the December 2018 Credit Agreement will bear interest at a fluctuating rate per annum equal to, at Walgreens Boots Alliance's option, the alternate base rate or the Eurocurrency rate, plus an applicable margin of 0.75% in the case of Eurocurrency rate loans. As of August 31, 2019, there were \$0.8 billion of borrowings outstanding under the December 2018 Credit Agreement.

On December 21, 2018, the Company entered into a \$1.0 billion revolving credit agreement (the "December 2018 Revolving Credit Agreement") with the lenders from time to time party thereto. The December 2018 Revolving Credit Agreement is a senior unsecured revolving credit facility with a facility termination date of the earlier of (a) 18 months following January 28, 2019, the date of the effectiveness of the commitments pursuant to the December 2018 Revolving Credit Agreement and (b) the date of termination in whole of the

- 50 -

Table of Contents

aggregate amount of the commitments pursuant to the December 2018 Revolving Credit Agreement. Borrowings under the December 2018 Revolving Credit Agreement will bear interest at a fluctuating rate per annum equal to, at Walgreens Boots Alliance's option, the alternate base rate or the Eurocurrency rate, plus an applicable margin of 0.75% in the case of Eurocurrency rate loans. As of August 31, 2019, there were \$0.3 billion borrowings outstanding under the December 2018 Revolving Credit Agreement.

On January 18, 2019, the Company entered into a \$2.0 billion 364-day revolving credit agreement (the "January 2019 364-Day Revolving Credit Agreement") with the lenders from time to time party thereto. The January 2019 364-Day Revolving Credit Agreement is a senior unsecured 364-day revolving credit facility, with a facility termination date of the earlier of (a) 364 days following January 31, 2019, the date of the effectiveness of the commitments pursuant to the January 364- Day Revolving Credit Agreement, subject to extension thereof pursuant to the January 2019 364-Day Revolving Credit Agreement, subject to extension thereof pursuant to the January 2019 364-Day Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the January 2019 364-Day Revolving Credit Agreement will bear interest at a fluctuating rate per annum equal to, at Walgreens Boots Alliance's option, the alternate base rate or the Eurocurrency rate, in each case, plus an applicable margin calculated based on the Company's credit ratings. As of August 31, 2019, there were no borrowings outstanding under the January 364-Day Revolving Credit Agreement.

On August 30, 2019, the Company entered into three \$500 million revolving credit agreements (together, the "August 2019 Revolving Credit Agreements" and each individually, an "August 2019 Revolving Credit Agreement") with the lenders from time to time party thereto. Each of the August 2019 Revolving Credit Agreements" and each individually, an "August 2019 Revolving credit facilities, with facility termination dates of the earlier of (a) 18 months following August 30, 2019, subject to extension thereof pursuant to the applicable August 2019 Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the applicable August 2019 Revolving Credit Agreement. Borrowings under each of the August 2019 Revolving Credit Agreements will bear interest at a fluctuating rate per annum equal to, at Walgreens Boots Alliance's option, the alternate base rate or the Eurocurrency rate, plus an applicable margin of 0.95% in the case of Eurocurrency rate loans. As of August 31, 2019, there were no borrowings outstanding under the August 2019 Revolving Credit Agreements.

From time to time, the Company may also enter into other credit facilities or financing arrangements.

Debt covenants

Each of the Company's credit facilities described above contain a covenant to maintain, as of the last day of each fiscal quarter, a ratio of consolidated debt to total capitalization not to exceed 0.60:1.00, subject to increase in certain circumstances set forth in the applicable credit agreement. The credit facilities contain various other customary covenants. As of August 31, 2019, the Company was in compliance with all such applicable covenants.

Credit ratings

As of October 25, 2019, the credit ratings of Walgreens Boots Alliance were:

		Commercial	
Rating Agency	Long-Term Debt Rating	Paper Rating	Outlook
Fitch	BBB	F2	Negative
Moody's	Baa2	P-2	Stable
Standard & Poor's	BBB	A-2	Stable

In assessing the Company's credit strength, each rating agency considers various factors including the Company's business model, capital structure, financial policies and financial performance. There can be no assurance that any particular rating will be assigned or maintained. The Company's credit ratings impact its borrowing costs, access to capital markets and operating lease costs. The rating agency ratings are not recommendations to buy, sell or hold the Company's debt securities or commercial paper. Each rating may be subject to revision or withdrawal at any time by the assigning rating agency and should be evaluated independently of any other rating.

AmerisourceBergen relationship

As of August 31, 2019, the Company owned 56,854,867 AmerisourceBergen common shares representing approximately 27% of the outstanding AmerisourceBergen common stock and had designated one member of AmerisourceBergen's board of directors. As of August 31, 2019, the Company can acquire up to an additional 8,398,752 AmerisourceBergen shares in the open market and thereafter designate another member of AmerisourceBergen's board of directors, subject in each case to applicable legal and contractual requirements. The amount of permitted open market purchases is subject to increase or



decrease in certain circumstances. Subject to applicable legal and contractual requirements, share purchases may be made from time to time in open market transactions or pursuant to instruments and plans complying with Rule 10b5-1. See note 5, equity method investments, to the Consolidated Financial Statements included herein for further information.

COMMITMENTS AND CONTINGENCIES

The information set forth in note 10, commitments and contingencies, to the Consolidated Financial Statements included in part II, item 8 of this Form 10-K is incorporated herein by reference.

CRITICAL ACCOUNTING POLICIES

The Consolidated Financial Statements are prepared in accordance with accounting principles generally accepted in the United States of America and include amounts based on management's prudent judgments and estimates. Actual results may differ from these estimates. Management believes that any reasonable deviation from those judgments and estimates would not have a material impact on our consolidated financial position or results of operations. To the extent that the estimates used differ from actual results, however, adjustments to the Consolidated Statements of Earnings and corresponding Consolidated Balance Sheets accounts would be necessary. These adjustments would be made in future periods. Some of the more significant estimates include business combinations, goodwill and indefinite-lived intangible asset impairment, cost of sales and inventory, equity method investments, pension and postretirement benefits and income taxes. The Company uses the following methods to determine its estimates:

Business combinations – The Company accounts for business combinations using the acquisition method of accounting, which requires that once control is obtained, all the assets acquired and liabilities assumed, including amounts attributable to noncontrolling interests, be recorded at their respective fair values at the date of acquisition. The determination of fair values of assets and liabilities acquired requires estimates and the use of valuation techniques when market value is not readily available.

For intangible assets, the Company generally uses the income approach to determine fair value. The income approach requires management to make significant estimates and assumptions. These estimates and assumptions primarily include, but are not limited to: discount rates, terminal growth rates, royalty rates, forecasts of revenue, operating income, depreciation, amortization and capital expenditures. The discount rates applied to the projections reflect the risk factors associated with those projections.

Although the Company believes its estimates of fair value are reasonable, actual financial results could differ from those estimates due to the inherent uncertainty involved in making such estimates. Changes in assumptions concerning future financial results or other underlying assumptions could have a significant impact on the determination of the fair value of the intangible assets acquired.

Judgment is also required in determining the intangible asset's useful life.

Goodwill and indefinite-lived intangible asset impairment – Goodwill and indefinite-lived intangible assets are evaluated for impairment annually during the fourth quarter, or more frequently if an event occurs or circumstances change that could more likely than not reduce the fair value of a reporting unit or intangible asset below its carrying value. As part of the Company's impairment analysis, fair value of a reporting unit is determined using both the income and market approaches. The income approach requires management to estimate a number of factors for each reporting unit, including projected future operating results, economic projections, anticipated future cash flows and discount rates. The market approach estimates fair value using comparable marketplace fair value data from within a comparable industry grouping.

The determination of the fair value of the reporting units requires the Company to make significant estimates and assumptions. These estimates and assumptions primarily include, but are not limited to: the selection of appropriate peer group companies, control premiums appropriate for acquisitions in the industries in which we compete, discount rates, terminal growth rates, forecasts of revenue, operating income, depreciation, amortization and capital expenditures. Although we believe our estimates of fair value are reasonable, actual financial results could differ from those estimates due to the inherent uncertainty involved in making such estimates. Changes in assumptions concerning future financial results or other underlying assumptions could have a significant impact on either the fair value of the reporting units, the amount of any goodwill impairment charge, or both.

The Company also compares the sum of estimated fair values of reporting units to the Company's fair value as implied by the market value of its equity securities. This comparison provides an indication that, in total, assumptions and estimates are reasonable. Future declines in the overall market value of the Company's equity securities may provide an indication that the fair value of one or more reporting units has declined below its carrying value.

- 52 -

The fair values of the Company's reporting units exceeded their carrying amounts ranging from approximately 9% to approximately 269% as of June 1, 2019 valuation date. The fair value of the Boots reporting unit, in the Retail Pharmacy International division, is in excess of its carrying value by approximately 9%. As at August 31, 2019, the carrying value of the goodwill of the Boots reporting unit is \$2.6 billion.

The Company continues to closely monitor industry and market trends and the impact it may have on the Boots reporting unit and indefinite-lived intangibles as the challenging market conditions in the UK continued to impact the operating performance of the Boots reporting unit during the three months ended August 31, 2019.

Indefinite-lived intangible assets are tested by comparing the estimated fair value of the asset to its carrying value. If the carrying value of the asset exceeds its estimated fair value, an impairment loss is recognized and the asset is written down to its estimated fair value. Indefinite-lived intangible assets fair values are estimated using the relief from royalty method and excess earnings method of the income approach. These estimates can be affected by a number of factors including, but not limited to, general economic conditions, availability of market information as well as our profitability. The fair values of indefinite-lived intangibles within the Boots reporting unit exceeded their carrying value amounts ranging from approximately 3% to approximately 29%, except for the pharmacy licenses. As at August 31, 2019, the carrying value of these indefinite-lived intangibles within the Boots reporting unit, which was included in selling, general, and administrative expenses in the Consolidated Statement of Earnings.

See note 6, goodwill and other intangible assets, to the Consolidated Financial Statements for additional information.

Cost of sales and inventory – Cost of sales includes the purchase price of goods and cost of services rendered, store and warehouse inventory loss, inventory obsolescence and supplier rebates. In addition to product costs, cost of sales includes warehousing costs for retail operations, purchasing costs, freight costs, cash discounts and vendor allowances.

Cost of sales for our Retail Pharmacy USA segment is derived based upon point-of-sale scanning information with an estimate for shrinkage and is adjusted based on periodic inventory counts. Inventories are valued at the lower of cost or market determined by the last-in, first-out ("LIFO") method for the Retail Pharmacy USA segment and primarily on a first-in first-out ("FIFO") basis for inventory in the Retail Pharmacy International and Pharmaceutical Wholesale segments.

Equity method investments – The Company uses the equity method of accounting for equity investments in companies if the investment provides the ability to exercise significant influence, but not control, over operating and financial policies of the investee. The Company's proportionate share of the net income or loss of these companies is included in consolidated net earnings. Judgment regarding the level of influence over each equity method investment includes considering key factors such as the Company's ownership interest, legal form of the investee (e.g. limited liability partnership), representation on the board of directors, participation in policy-making decisions and material intra-entity transactions

The Company evaluates equity method investments for impairment whenever events or changes in circumstances indicate that the carrying amount of the investment might not be recoverable. Factors considered by the Company when reviewing an equity method investment for impairment include the length of time (duration) and the extent (severity) to which the fair value of the equity method investment has been less than cost, the investee's financial condition and near-term prospects and the intent and ability to hold the investment for a period of time sufficient to allow for anticipated recovery. An impairment that is other-than-temporary is recognized in the period identified.

Pension and postretirement benefits – The Company has various defined benefit pension plans that cover some of its non-U.S. employees. The Company also has a postretirement healthcare plan that covers qualifying U.S. employees. Eligibility and the level of benefits for these plans vary depending on participants' status, date of hire and or length of service. Pension and postretirement plan expenses and valuations are dependent on assumptions used by third-party actuaries in calculating those amounts. These assumptions include discount rates, healthcare cost trends, long-term return on plan assets, retirement rates, mortality rates and other factors.

In determining long-term rate of return on plan assets assumption, the Company considers both the historical performance of the investment portfolio as well as the long-term market return expectations based on the investment mix of the portfolio. A change in any of these assumptions would have an effect on its pension expense. A 25 basis point increase in the discount rate would result in a decline of \$368.1 million to the Company's pension benefit obligation. A 25 basis point decrease on the expected return on plan assets assumption would increase the Company's pension expense by \$19.1 million.

- 53 -

Table of Contents

The Company funds its pension plans in accordance with applicable regulations. The postretirement healthcare plan is not funded.

Income taxes –The Company is subject to routine income tax audits that occur periodically in the normal course of business. U.S. federal, state, local and foreign tax authorities raise questions regarding the Company's tax filing positions, including the timing and amount of deductions and the allocation of income among various tax jurisdictions. In evaluating the tax benefits associated with the various tax filing positions, the Company records a tax benefit for uncertain tax positions using the highest cumulative tax benefit that is more likely than not to be realized. Adjustments are made to the liability for unrecognized tax benefits in the period in which the Company determines the issue is effectively settled with the tax authorities, the statute of limitations expires for the return containing the tax position or when more information becomes available. The liability for unrecognized tax benefits, including accrued penalties and interest, is primarily included in other non-current liabilities and current income taxes on the Company's Consolidated Balance Sheets and in income tax provision in its Consolidated Statements of Earnings.

In determining its provision for income taxes, the Company uses income, permanent differences between book and tax income and enacted statutory income tax rates. The provision for income taxes rate also reflects its assessment of the ultimate outcome of tax audits in addition to any foreign-based income deemed to be taxable in the United States. Discrete events such as audit settlements or changes in tax laws are recognized in the period in which they occur.

CONTRACTUAL OBLIGATIONS AND COMMITMENTS

The following table lists the Company's contractual obligations and commitments at August 31, 2019 (in millions):

	Payments due by period											
		Total	L	ess than 1 year		1-3 years		3-5 years	Ov	er 5 years		
Operating leases ¹	\$	32,268	\$	3,484	\$	6,447	\$	5,714	\$	16,623		
Purchase obligations:		3,231		2,359		224		100		548		
Open inventory purchase orders		1,708		1,707		1		—		—		
Real estate development		179		179		—		—		—		
Other obligations		1,344		474		223		100		548		
Short-term debt and long-term debt*		16,916		5,752		1,741		1,200		8,223		
Interest payment on short-term debt and long-term debt		4,704		430		761		635		2,878		
Retirement benefit obligations		695		42		81		83		489		
Closed location obligations ¹		1,613		211		359		286		757		
Capital lease obligations ^{*1}		1,093		59		118		115		801		
Finance lease obligations		311		21		41		40		209		
Other liabilities reflected on the balance sheet* ²		1,216		122		529		185		380		
Total	\$	62,047	\$	12,480	\$	10,302	\$	8,358	\$	30,908		

* Recorded on balance sheet.

¹ Amounts do not include certain operating expenses under these leases such as common area maintenance, insurance and real estate taxes, where appropriate. These expenses were \$569 million for the fiscal year ended August 31, 2019.

Includes \$502 million (\$88 million in less than 1 year, \$385 million in 1-3 years and \$29 million in 3-5 years) of unrecognized tax benefits recorded under Accounting Standards Codification Topic 740, Income Taxes. Includes \$352 million (\$67 million in 1-3 years, \$96 million in 3-5 years and \$189 million in over 5 years) of noncurrent income taxes payable as a result of the U.S tax law changes enacted in December 2017.

The information in the foregoing table is presented as of August 31, 2019 and accordingly does not reflect obligations under agreements the Company entered into after that date.

OFF-BALANCE SHEET ARRANGEMENTS

The Company does not have any unconsolidated special purpose entities and, except as described herein, the Company does not have significant exposure to any off-balance sheet arrangements. The term "off-balance sheet arrangement" generally means any transaction, agreement or other contractual arrangement to which an entity unconsolidated with the Company is a party, under which the Company has: (i) any obligation arising under a guarantee contract, derivative instrument or variable interest; or (ii) a retained or contingent interest in assets transferred to such entity or similar arrangement that serves as credit, liquidity or market risk support for such assets.

At August 31, 2019, the Company had \$51 million of guarantees outstanding and no amounts issued under letters of credit.

RECENT ACCOUNTING PRONOUNCEMENTS

See "new accounting pronouncements" within note 1, summary of major accounting policies, to the Consolidated Financial Statements for information regarding recent accounting pronouncements.

CAUTIONARY NOTE REGARDING FORWARD-LOOKING STATEMENTS

This report and other documents that we file or furnish with the SEC contain forward-looking statements that are based on current expectations, estimates, forecasts and projections about our future performance, our business, our beliefs and our management's assumptions. In addition, we, or others on our behalf, may make forward-looking statements in press releases or written statements, on the Company's website or in our communications and discussions with investors and analysts in the normal course of business through meetings, webcasts, phone calls, conference calls and other communications. Some of such forward-looking statements may be based on certain data and forecasts relating to our business and industry that we have obtained from internal surveys, market research, publicly available information and industry publications. Industry publications, surveys and market research generally state that the information they provide has been obtained from sources believed to be reliable, but that the accuracy and completeness of such information is not guaranteed. Statements that are not historical facts are forward-looking statements, including, without limitation, those regarding estimates of and goals for future

- 55 -

financial and operating performance as well as forward-looking statements concerning the expected execution and effect of our business strategies, our costsavings and growth initiatives, pilot programs, strategic partnerships and initiatives, and restructuring activities and the amounts and timing of their expected impact and delivery of estimated cost savings, our amended and restated asset purchase agreement with Rite Aid and the transactions contemplated thereby and their possible timing and effects, our commercial agreement with AmerisourceBergen, the arrangements and transactions contemplated by our framework agreement with AmerisourceBergen and their possible effects, estimates of the impact of developments on our earnings, earnings per share and other financial and operating metrics, cough, cold and flu season, prescription volume, pharmacy sales trends, prescription margins and reimbursement rates, changes in generic prescription drug prices, retail margins, number and location of new store openings, network participation, vendor, payer and customer relationships and terms, possible new contracts or contract extensions, the proposed withdrawal of the United Kingdom from the European Union and its possible effects, competition, economic and business conditions, outcomes of litigation and regulatory matters, the level of capital expenditures, industry trends, demographic trends, growth strategies, financial results, cost reduction initiatives, impairment or other charges, acquisition and joint venture synergies, competitive strengths and changes in legislation or regulations. All statements in the future tense and all statements accompanied by words such as "expect," "likely," "outlook," "forecast," "preliminary," "pilot," "would," "could," "should," "can," "will," "project," "intend," "goal," "guidance," "target," "aim," "continue," "sustain," "synergy," "transform," "accelerate," "model," "long-term," on track," on schedule," "headwind," "tailwind," "believe," "seek," "estimate," "anticipate," "upcomi

These forward-looking statements are not guarantees of future performance and are subject to risks, uncertainties and assumptions, known or unknown, that could cause actual results to vary materially from those indicated or anticipated, including, but not limited to, those relating to the impact of private and public third-party payers' efforts to reduce prescription drug reimbursements, fluctuations in foreign currency exchange rates, the timing and magnitude of the impact of branded to generic drug conversions and changes in generic drug prices, our ability to realize synergies and achieve financial, tax and operating results in the amounts and at the times anticipated, the inherent risks, challenges and uncertainties associated with forecasting financial results of large, complex organizations in rapidly evolving industries, particularly over longer time periods, supply arrangements including our commercial agreement with AmerisourceBergen, the arrangements and transactions contemplated by our framework agreement with AmerisourceBergen and their possible effects, the risks associated with our equity method investment in AmerisourceBergen, circumstances that could give rise to the termination, cross-termination or modification of any of our contractual obligations, the amount of costs, and charges incurred with strategic transactions, whether the costs and charges associated with restructuring initiatives, including the Transformational Cost Management Program and Store Optimization Program, will exceed estimates, our ability to realize expected savings and benefits from cost-savings initiatives, including the Transformational Cost Management Program and Store Optimization Program, restructuring activities and acquisitions and joint ventures in the amounts and at the times anticipated, the timing and amount of any impairment or other charges, the timing and severity of cough, cold and flu season, risks related to pilot programs and new business initiatives and ventures generally, including the risks that anticipated benefits may not be realized, changes in management's plans and assumptions, the risks associated with governance and control matters, the ability to retain key personnel, changes in economic and business conditions generally or in particular markets in which we participate, changes in financial markets, credit ratings and interest rates, the risks relating to the terms, timing and magnitude of any share repurchase activity, the risks associated with international business operations, including the risks associated with the proposed withdrawal of the United Kingdom from the European Union and international trade policies, tariffs, including tariff negotiations between the United States and China, and relations, the risks associated with cybersecurity or privacy breaches related to customer information, changes in vendor, customer and payer relationships and terms, including changes in network participation and reimbursement terms and the associated impacts on volume and operating results, risks related to competition including changes in market dynamics, participants, product and service offerings, retail formats and competitive positioning, risks associated with new business areas and activities, risks associated with acquisitions, divestitures, joint ventures and strategic investments, including those relating to the asset acquisition from Rite Aid, the risks associated with the integration of complex businesses, the impact of regulatory restrictions and outcomes of legal and regulatory matters and risks associated with changes in laws, including those related to the December 2017 U.S. tax law changes, regulations or interpretations thereof. These and other risks, assumptions and uncertainties are described in Item 1A, Risk factors, above and in other documents that we file or furnish with the SEC. Should one or more of these risks or uncertainties materialize, or should underlying assumptions prove incorrect, actual results may vary materially from those indicated or anticipated by such forward-looking statements. Accordingly, you are cautioned not to place undue reliance on these forward-looking statements, which speak only as of the date they are made. Except to the extent required by law, we do not undertake, and expressly disclaim, any duty or obligation to update publicly any forward-looking statement after the date of this report, whether as a result of new information, future events, changes in assumptions or otherwise.

- 56 -

Item 7A. Quantitative and qualitative disclosure about market risk

Interest rate risk

The Company is exposed to interest rate volatility with regard to existing variable-rate debt instruments and future incurrences of fixed or variable-rate debt, which exposure primarily relates to movements in various interest rates, such as U.S treasury rates and commercial paper rates. From time to time, the Company uses interest rate swaps and forward-starting interest rate swaps to hedge its exposure to the impact of interest rate changes on existing debt and future debt issuances respectively, to reduce the volatility of financing costs and, based on current and projected market conditions, achieve a desired proportion of fixed-rate versus floating-rate debt. Generally under these swaps, the Company agrees with a counterparty to exchange the difference between fixed-rate and floating-rate interest amounts based on an agreed upon notional principal amount.

Information regarding the Company's transactions are set forth in note 8, financial instruments, to the Consolidated Financial Statements. These financial instruments are sensitive to changes in interest rates. On August 31, 2019, the Company had no material long-term debt obligations that had floating interest rates. The amounts exclude the impact of any associated derivative contracts.

Foreign currency exchange rate risk

The Company is exposed to fluctuations in foreign currency exchange rates, primarily with respect to the British pound sterling and Euro, and certain other foreign currencies, which may affect its net investment in foreign subsidiaries and may cause fluctuations in cash flows related to foreign denominated transactions. The Company is also exposed to the translation of foreign currency earnings to the U.S. dollar. The Company enters into foreign currency forward contracts to hedge against the effect of exchange rate fluctuations on non-functional currency cash flows. These transactions are almost exclusively less than 12 months in maturity. In addition, the Company enters into foreign currency forward contracts that are not designated in hedging relationships to offset, in part, the impacts of certain intercompany activities (primarily associated with intercompany financing transactions). Under certain market conditions, the Company may seek to protect against possible declines in the reported net investments of our foreign subsidiaries by using foreign currency cross-currency swaps, foreign currency forward-exchange contracts or foreign currency debt.

The Company's foreign currency derivative instruments are sensitive to changes in exchange rates. A hypothetical 1% change in foreign currency exchange rates versus the U.S. dollar would change the fair value of the foreign currency derivatives held as of August 31, 2019, by approximately \$22 million. The foreign currency derivatives are intended to partially hedge anticipated transactions, foreign currency trade payables and receivables and net investments in foreign subsidiaries.

Equity price risk

Changes in AmerisourceBergen common stock price may have a significant impact on the fair value of the equity investment in AmerisourceBergen described in note 5, equity method investments, to the Consolidated Financial Statements. See "--Investment in AmerisourceBergen" above.

Item 8. Financial statements and supplementary data

WALGREENS BOOTS ALLIANCE, INC. AND SUBSIDIARIES CONSOLIDATED BALANCE SHEETS At August 31, 2019 and 2018 (in millions, except shares and per share amounts)

	2019	2018
Assets		
Current assets:		
Cash and cash equivalents	\$ 1,023	\$ 785
Accounts receivable, net	7,226	6,573
Inventories	9,333	9,565
Other current assets	1,118	923
Total current assets	18,700	 17,846
Non-current assets:		
Property, plant and equipment, net	13,478	13,911
Goodwill	16,560	16,914
Intangible assets, net	10,876	11,783
Equity method investments (see note 5)	6,851	6,610
Other non-current assets	1,133	1,060
Total non-current assets	48,899	 50,278
Total assets	\$ 67,598	\$ 68,124
Liabilities and equity		
Current liabilities:		
Short-term debt	\$ 5,738	\$ 1,966
Trade accounts payable (see note 18)	14,341	13,566
Accrued expenses and other liabilities	5,474	5,862
Income taxes	216	273
Total current liabilities	 25,769	 21,667
Non-current liabilities:		
Long-term debt	11,098	12,431
Deferred income taxes	1,785	1,815
Other non-current liabilities	4,795	5,522
Total non-current liabilities	 17,678	 19,768
Commitments and contingencies (see note 10)		
Equity:		
Preferred stock \$.01 par value; authorized 32 million shares, none issued	_	_
Common stock \$.01 par value; authorized 3.2 billion shares; issued 1,172,513,618 at August 31, 2019 and 2018	12	12
Paid-in capital	10,639	10,493
Retained earnings	35,815	33,551
Accumulated other comprehensive loss	(3,897)	(3,002)
Treasury stock, at cost; 277,126,116 shares at August 31, 2019 and 220,380,200 shares at August 31, 2018	(19,057)	(15,047)
Total Walgreens Boots Alliance, Inc. shareholders' equity	23,512	 26,007
Noncontrolling interests	641	682
Total equity	24,152	 26,689
Total liabilities and equity	\$ 67,598	\$ 68,124

The accompanying notes to Consolidated Financial Statements are an integral part of these Statements.

WALGREENS BOOTS ALLIANCE, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EQUITY For the years ended August 31, 2019, 2018 and 2017 (in millions, except shares)

Equity attributable to Walgreens Boots Alliance, Inc.

	Equity attributable to Walgreens Boots Alliance, Inc.															
	Common stock shares	sto	nmon ock ount		Freasury stock amount		Paid-in capital		Employee stock loan receivable	Accumul other comprehe income (nsive		tained mings	ľ	Noncontrolling interests	Total equity
August 31, 2016	1,082,986,591	\$	12	\$	(4,934)	\$	10,111	\$	(1)	\$ (2,992) \$	5	27,684	\$	401	\$ 30,281
Net earnings	—		—		—		—		—		—		4,078		23	4,101
Other comprehensive income (loss), net of tax	—		—				_		—		(59)		—		(36)	(95)
Dividends declared and distributions	—		—				_		—		—		(1,625)		(98)	(1,723)
Treasury stock purchases	(64,589,677)		—		(5,220)		_		—		_		_		_	(5,220)
Employee stock purchase and option plans	5,452,156				183		34		1		—		_		_	218
Stock-based compensation	_				_		91		_		_		_		—	91
Noncontrolling interests acquired and arising on business combinations	_		_		_		103		_		_		_		518	621
August 31, 2017	1,023,849,070	\$	12	\$	(9,971)	\$	10,339	\$	—	\$ (3,051) \$	5	30,137	\$	808	\$ 28,274
Net earnings	_		_		_		_		_		_		5,024		7	5,031
Other comprehensive income (loss), net of tax	_				_		_		_		49		—		1	50
Dividends declared and distributions	_		—		_		_		_		_		(1,610)		(138)	(1,748)
Treasury stock purchases	(76,069,557)				(5,228)		_		_		—		—		_	(5,228)
Employee stock purchase and option plans	4,353,905		—		152		22		_		_		—		_	174
Stock-based compensation	_				_		130		_		_		—		_	130
Noncontrolling interests contribution and other	_		_		_		2		_		_		_		4	6
August 31, 2018	952,133,418	\$	12	\$	(15,047)	\$	10,493	\$	—	\$ (3,002) \$	5	33,551	\$	682	\$ 26,689
Net earnings	_		_		—		_		_		_		3,982		(20)	3,962
Other comprehensive income (loss), net of tax	_				_		_		_		(896)		_		(13)	(909)
Dividends declared and distributions	_				_		_		_		—		(1,629)		(3)	(1,632)
Treasury stock purchases	(61,723,456)		_		(4,160)		_		_		_		_		_	(4,160)
Employee stock purchase and option plans	4,977,540				150		24		_		—		_		_	174
Stock-based compensation	_				_		119		_		_		—		_	119
Adoption of new accounting standards	_		_		—				_				(88)		_	(88)
Noncontrolling interests contribution and other	_		_		_		3		_		_		(1)		(5)	(3)
August 31, 2019	895,387,502	\$	12	\$	(19,057)	\$	10,639	\$	_	\$ (3,897) \$	5	35,815	\$	641	\$ 24,152

The accompanying notes to Consolidated Financial Statements are an integral part of these Statements.

- 59 -

WALGREENS BOOTS ALLIANCE, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF EARNINGS For the years ended August 31, 2019, 2018 and 2017 (in millions, except per share amounts)

	2019	2018	2017
Sales	\$ 136,866	\$ 131,537	\$ 118,214
Cost of sales	106,790	100,745	89,052
Gross profit	30,076	30,792	 29,162
Selling, general and administrative expenses	25,242	24,694	23,813
Equity earnings in AmerisourceBergen	164	191	135
Operating income	4,998	6,289	5,484
Other income	233	302	62
Earnings before interest and income tax provision	 5,231	6,591	5,546
Interest expense, net	704	616	693
Earnings before income tax provision	4,527	5,975	4,853
Income tax provision	588	998	760
Post tax earnings from other equity method investments	23	54	8
Net earnings	 3,962	 5,031	 4,101
Net (loss) earnings attributable to noncontrolling interests	(20)	7	23
Net earnings attributable to Walgreens Boots Alliance, Inc.	\$ 3,982	\$ 5,024	\$ 4,078
Net earnings per common share:			
Basic	\$ 4.32	\$ 5.07	\$ 3.80
Diluted	\$ 4.31	\$ 5.05	\$ 3.78
Weighted average common shares outstanding:			
Basic	921.5	991.0	1,073.5
Diluted	923.5	995.0	1,078.5

The accompanying notes to Consolidated Financial Statements are an integral part of these Statements.

- 60 -

WALGREENS BOOTS ALLIANCE, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME For the years ended August 31, 2019, 2018 and 2017 (in millions)

	2019	2018		2017
Comprehensive income:	 	 		
Net earnings	\$ 3,962	\$ 5,031	\$	4,101
Other comprehensive income (loss), net of tax:				
Pension/postretirement obligations	(149)	240		73
Unrealized gain on cash flow hedges	60	3		4
Unrecognized (loss) on available-for-sale investments				(2)
Share of other comprehensive income (loss) of equity method investments	(1)	5		(1)
Currency translation adjustments	(820)	(198)		(169)
Total other comprehensive income (loss)	(909)	 50		(95)
Total comprehensive income	3,053	5,081		4,006
Comprehensive income (loss) attributable to noncontrolling interests	(33)	8		(13)
Comprehensive income attributable to Walgreens Boots Alliance, Inc.	\$ 3,086	\$ 5,073	\$	4,019

The accompanying notes to Consolidated Financial Statements are an integral part of these Statements.

- 61 -

WALGREENS BOOTS ALLIANCE, INC. AND SUBSIDIARIES CONSOLIDATED STATEMENTS OF CASH FLOWS For the years ended August 31, 2019, 2018 and 2017 (in millions)

	2019	2018	2017
Cash flows from operating activities:	¢	¢ = 004	.
Net earnings	\$ 3,962	\$ 5,031	\$ 4,101
Adjustments to reconcile net earnings to net cash provided by operating activities:	0.000	1 550	
Depreciation and amortization	2,038	1,770	1,654
Gain on previously held equity interest		(337)	
Deferred income taxes	100	(322)	(434)
Stock compensation expense	119	130	91
Equity earnings from equity method investments	(187)		(143)
Other	302	296	364
Changes in operating assets and liabilities:			
Accounts receivable, net	(789)	(391)	(153)
Inventories	141	331	98
Other current assets	(112)	(22)	
Trade accounts payable	954	1,352	1,684
Accrued expenses and other liabilities	(374)	287	(128)
Income taxes	(406)	694	44
Other non-current assets and liabilities	(154)	(311)	77
Net cash provided by operating activities	5,594	8,263	7,255
Cash flows from investing activities:			
Additions to property, plant and equipment	(1,702)	(1,367)	(1,351)
Proceeds from sale leaseback transactions	3	_	444
Proceeds from sale of other assets	117	655	59
Business, investment and asset acquisitions, net of cash acquired	(741)	(4,793)	(88)
Other	16	4	93
Net cash used for investing activities	(2,307)	(5,501)	(843)
Cash flows from financing activities:			
Net change in short-term debt with maturities of 3 months or less	536	586	33
Proceeds from debt	12,433	5,900	_
Payments of debt	(10,461)	(4,890)	(6,196)
Stock purchases	(4,160)		(5,220)
Proceeds related to employee stock plans	174	174	217
Cash dividends paid	(1,643)	(1,739)	(1,723)
Other	75	(98)	(45)
Net cash used for financing activities	(3,047)		(12,934)
Effect of exchange rate changes on cash, cash equivalents and restricted cash	(9)	11	26
Changes in cash, cash equivalents and restricted cash			
Net increase (decrease) in cash, cash equivalents, and restricted cash	232	(2,522)	(6,496)
Cash, cash equivalents and restricted cash at beginning of period	975	3,496	9,992
Cash, cash equivalents and restricted cash at organisming or period	\$ 1,207	\$ 975	\$ 3,496
1 · · · · · · · · · · · · · · · · · · ·			

The accompanying notes to Consolidated Financial Statements are an integral part of these Statements.

WALGREENS BOOTS ALLIANCE, INC. AND SUBSIDIARIES NOTES TO CONSOLIDATED FINANCIAL STATEMENTS

Note 1. Summary of major accounting policies

Organization

Walgreens Boots Alliance and its subsidiaries are a global leader in retail and wholesale pharmacy. Its operations are conducted through three reportable segments: Retail Pharmacy USA, Retail Pharmacy International and Pharmaceutical Wholesale. See note 16, segment reporting and note 17, sales, for further information.

Basis of presentation

The Consolidated Financial Statements include all subsidiaries in which the Company holds a controlling interest. The Company uses the equity-method of accounting for equity investments in less than majority-owned companies if the investment provides the ability to exercise significant influence. All intercompany transactions have been eliminated.

The preparation of financial statements in accordance with accounting principles generally accepted in the United States of America ("GAAP") requires management to use judgment in the application of accounting policies, including making estimates and assumptions. The Company bases its estimates on the information available at the time, its experience and various other assumptions believed to be reasonable under the circumstances. Adjustments may be made in subsequent periods to reflect more current estimates and assumptions about matters that are inherently uncertain. Actual results may differ.

The influence of certain holidays, seasonality, foreign currency rates, changes in vendor, payer and customer relationships and terms, strategic transactions including acquisitions, changes in laws and general economic conditions in the markets in which the Company operates and other factors on the Company's operations and net earnings for any period may not be comparable to the same period in previous years.

Certain amounts in the Consolidated Financial Statements and associated notes may not add due to rounding. All percentages have been calculated using unrounded amounts for fiscal 2019.

Cash and cash equivalents

Cash and cash equivalents include cash on hand and highly liquid investments with an original maturity of three months or less. Credit and debit card receivables, which generally settle within two to seven business days, of \$90 million and \$127 million were included in cash and cash equivalents at August 31, 2019 and 2018, respectively.

Restricted cash

The Company is required to maintain cash deposits with certain banks which consist of deposits restricted under contractual agency agreements and cash restricted by law and other obligations. At August 31, 2019 and 2018, the amount of such restricted cash was \$184 million and \$190 million, respectively, and is reported in other current assets on the Consolidated Balance Sheets.

Accounts receivable

Accounts receivable are stated net of allowances for doubtful accounts. Accounts receivable balances primarily consist of trade receivables due from customers, including amounts due from third-party payers (e.g., pharmacy benefit managers, insurance companies and governmental agencies), clients and members. Trade receivables were \$6.0 billion and \$5.4 billion at August 31, 2019 and August 31, 2018, respectively. Other accounts receivable balances, which consist primarily of receivables from vendors and manufacturers, including receivables from AmerisourceBergen (see note 18, related parties), were \$1.2 billion at August 31, 2019 and 2018.

Charges to allowance for doubtful accounts are based on estimates of recoverability using both historical write-offs and specifically identified receivables. The allowance for doubtful accounts for trade receivables at August 31, 2019 and August 31, 2018 was \$95 million and \$75 million, respectively.

Inventories

The Company values inventories on a lower of cost and net realizable value or market basis. Inventories include product costs, inbound freight, direct labor, warehousing costs for retail pharmacy operations, overhead costs relating to the manufacture and distribution of products and vendor allowances not classified as a reduction of advertising expense.

The Company's Retail Pharmacy USA segment inventory is accounted for using the last-in-first-out ("LIFO") method. The total carrying value of the segment inventory accounted for under the LIFO method was \$6.6 billion and \$6.7 billion at



August 31, 2019 and 2018, respectively. At August 31, 2019 and 2018, Retail Pharmacy USA segment inventory would have been greater by \$3.2 billion and \$3.0 billion, respectively, if they had been valued on a lower of first-in-first-out ("FIFO") cost and net realizable value.

The Company's Retail Pharmacy International and Pharmaceutical Wholesale segments' inventory is primarily accounted for using the FIFO method. The total carrying value of the inventory for Retail Pharmacy International and Pharmaceutical Wholesale segments was \$2.7 billion and \$2.8 billion at August 31, 2019 and 2018, respectively.

Property, plant and equipment

Property, plant and equipment are stated at cost less accumulated depreciation. Major repairs, which extend the useful life of an asset, are capitalized; routine maintenance and repairs are charged against earnings. Depreciation is provided on a straight-line basis over the estimated useful lives of owned assets. Leasehold improvements, equipment under capital lease and capital lease properties are amortized over their respective estimate of useful life or over the term of the lease, whichever is shorter. The majority of the Company's fixtures and equipment uses the composite method of depreciation. Therefore, gains and losses on retirement or other disposition of such assets are included in earnings only when an operating location is closed, substantially remodeled or impaired. The following table summarizes the Company's property, plant and equipment (in millions) and estimated useful lives (in years):

	Estimated useful life	2019	2018
Land and land improvements	20	\$ 3,507	\$ 3,593
Buildings and building improvements	3 to 50	8,023	7,874
Fixtures and equipment	3 to 20	9,786	9,750
Capitalized system development costs and software	3 to 8	2,770	2,464
Capital lease properties		703	743
		24,789	24,424
Less: accumulated depreciation and amortization		11,310	10,513
Balance at end of year		\$ 13,478	\$ 13,911

The Company capitalizes application development stage costs for internally developed software. These costs are amortized over a three to eight year period. Amortization expense for capitalized system development costs and software was \$273 million in fiscal 2019, \$254 million in fiscal 2018 and \$245 million in fiscal 2017. Unamortized costs were \$1.5 billion at August 31, 2019 and 2018.

Depreciation and amortization expense for property, plant and equipment including capitalized system development costs and software was \$1.5 billion in fiscal 2019, \$1.4 billion in fiscal 2018 and \$1.3 billion in fiscal 2017.

Leases

Initial terms for leased premises in the U.S. are typically 15 to 25 years, followed by additional terms containing renewal options at five-year intervals, and may include rent escalation clauses. Non-U.S. leases are typically for shorter terms and may include cancellation clauses or renewal options. The commencement date of all lease terms is the earlier of the date the Company becomes legally obligated to make rent payments or the date the Company has the right to control the property. In addition to minimum fixed rentals, some leases provide for contingent rentals based upon a portion of sales.

Capital leases are recognized within property, plant and equipment and as a capital lease liability within accrued expenses and other liabilities and other noncurrent liabilities. Operating leases are expensed on a straight line basis over the lease term.

See note 4, leases, for further information.

Business combinations

The Company allocates the fair value of purchase consideration to the tangible and intangible assets purchased and the liabilities assumed on the basis of their fair values at the date of acquisition. The determination of fair values of assets acquired and liabilities assumed requires estimates and the use of valuation techniques when a market value is not readily available. Any excess of purchase price over the fair value of net tangible and intangible assets acquired is allocated to goodwill. If the Company obtains new information about facts and circumstances that existed as of the acquisition date during the measurement period, which may be up to one year from the acquisition date, the Company may record adjustments to the assets acquired and liabilities assumed.

Goodwill and other intangible assets

Goodwill represents the excess of the purchase price over the fair value of assets acquired and liabilities assumed in business combinations. Acquired intangible assets are recorded at fair value.

Goodwill and indefinite-lived intangible assets are evaluated for impairment annually during the fourth quarter, or more frequently if an event occurs or circumstances change that could more likely than not reduce the fair value of a reporting unit or intangible asset below its carrying value. As part of the Company's impairment analysis, fair value of a reporting unit is determined using both the income and market approaches. The income approach requires management to estimate a number of factors for each reporting unit, including projected future operating results, economic projections, anticipated future cash flows and discount rates. The market approach estimates fair value using comparable marketplace fair value data from within a comparable industry grouping.

Finite-lived intangible assets are amortized on a straight-line basis over their estimated useful lives. See note 6, goodwill and other intangible assets, for additional disclosure regarding the Company's intangible assets.

Equity method investments

The Company uses the equity method of accounting for equity investments in companies if the investment provides the ability to exercise significant influence, but not control, over operating and financial policies of the investee. The Company's proportionate share of the net income or loss of these companies is included in consolidated net earnings. Judgment regarding the level of influence over each equity method investment includes considering key factors such as the Company's ownership interest, legal form of the investee (e.g. limited liability partnership), representation on the board of directors, participation in policy-making decisions and material intra-entity transactions.

The Company evaluates equity method investments for impairment whenever events or changes in circumstances indicate that the carrying amount of the investment might not be recoverable. Factors considered by the Company when reviewing an equity method investment for impairment include the length of time (duration) and the extent (severity) to which the fair value of the equity method investment has been less than cost, the investee's financial condition and near-term prospects and the intent and ability to hold the investment for a period of time sufficient to allow for anticipated recovery. An impairment that is other-than-temporary is recognized in the period identified.

See note 5, equity method investments, for further information.

Financial instruments

The Company uses derivative instruments to hedge its exposure to interest rate and currency risks arising from operating and financing activities. In accordance with its risk management policies, the Company does not hold or issue derivative instruments for trading or speculative purposes.

Derivatives are recognized on the Consolidated Balance Sheets at their fair values. When the Company becomes a party to a derivative instrument and intends to apply hedge accounting, it formally documents the hedge relationship and the risk management objective for undertaking the hedge which includes designating the instrument for financial reporting purposes as a fair value hedge, a cash flow hedge, or a net investment hedge. The accounting for changes in fair value of a derivative instrument depends on whether the Company had designated it in a qualifying hedging relationship and on the type of hedging relationship. The Company applies the following accounting policies:

- Changes in the fair value of a derivative designated as a fair value hedge, along with the gain or loss on the hedged asset or liability attributable to the hedged risk, are recorded in the Consolidated Statements of Earnings in the same line item, generally interest expense, net.
- Changes in the fair value of a derivative designated as a cash flow hedge are recorded in accumulated other comprehensive income (loss) in the Consolidated Statements of Comprehensive Income and reclassified into earnings in the period or periods during which the hedged item affects earnings and is presented in the same line item as the earnings effect of the hedged item.
- Changes in the fair value of a derivative designated as a hedge of a net investment in a foreign operation are recorded in cumulative translation adjustments within accumulated other comprehensive income (loss) in the Consolidated Statements of Comprehensive Income. Recognition in earnings of amounts previously recorded in cumulative translation adjustments is limited to circumstances such as complete or substantially complete liquidation of the net investment in the hedged investments in foreign operations.

- 65 -

Changes in the fair value of a derivative not designated in a hedging relationship are recognized in the Consolidated Statements of Earnings.

Cash receipts or payments on a settlement of a derivative contract are reported in the Consolidated Statements of Cash Flows consistent with the nature of the underlying hedged item.

For derivative instruments designated as hedges, the Company assesses, both at the hedge's inception and on an ongoing basis, whether the derivatives that are used in hedging transactions are highly effective in offsetting changes in fair values or cash flows of hedged items. Highly effective means that cumulative changes in the fair value of the derivative are between 80% and 125% of the cumulative changes in the fair value of the hedged item. In addition, when the Company determines that a derivative is not highly effective as a hedge, hedge accounting is discontinued. When it is probable that a hedged forecasted transaction will not occur, the Company discontinues hedge accounting for the affected portion of the forecasted transaction and reclassifies any gains or losses in accumulated other comprehensive income (loss) to earnings in the Consolidated Statement of Earnings. When a derivative in a hedge relationship is terminated or the hedged item is sold, extinguished or terminated, hedge accounting is discontinued prospectively.

Liabilities for facility closings

The Company provides for future costs related to closed locations. The liability is based on the present value of future rent obligations and other related costs (net of estimated sublease rent) to the first lease option date. The liability for facility closings, including locations closed under the Company's restructuring actions, was \$993 million as of August 31, 2019 and \$964 million as of August 31, 2018. See note 4, leases, for further information.

Pension and postretirement benefits

The Company has various defined benefit pension plans that cover some of its non-U.S. employees. The Company also has a postretirement healthcare plan that covers qualifying U.S. employees. Eligibility and the level of benefits for these plans vary depending on participants' status, date of hire and or length of service. Pension and postretirement plan expenses and valuations are dependent on assumptions used by third-party actuaries in calculating those amounts. These assumptions include discount rates, healthcare cost trends, long-term return on plan assets, retirement rates, mortality rates and other factors.

The Company funds its pension plans in accordance with applicable regulations. The Company records the service cost component of net pension cost and net postretirement benefit cost in selling, general and administrative expenses. The Company records all other net cost components of net pension cost and net postretirement benefit cost in other income (expense). The postretirement healthcare plan is not funded.

See note 13, retirement benefits, for further information.

Noncontrolling interests

The Company presents noncontrolling interests as a component of equity on its Consolidated Balance Sheets and reports the portion of its earnings or loss for noncontrolling interest as net earnings attributable to noncontrolling interests in the Consolidated Statements of Earnings.

Currency

Assets and liabilities of non-U.S. dollar functional currency operations are translated into U.S. dollars at end-of-period exchange rates while revenues, expenses and cash flows are translated at average monthly exchange rates over the period. Equity is translated at historical exchange rates and the resulting cumulative translation adjustments are included as a component of accumulated other comprehensive income (loss) in the Consolidated Balance Sheets.

Assets and liabilities not denominated in the functional currency are remeasured into the functional currency at end-of-period exchange rates, except for nonmonetary balance sheet amounts, which are remeasured at historical exchange rates. Revenues and expenses are recorded at average monthly exchange rates over the period, except for those expenses related to nonmonetary balance sheet amounts, which are remeasured at historical exchange rates. Gains or losses from foreign currency remeasurement are generally included in selling, general and administrative expenses within the Consolidated Statements of Earnings. For all periods presented, there were no material operational gains or losses from foreign currency transactions.

Commitments and contingencies

On a quarterly basis, the Company assesses its liabilities and contingencies for outstanding legal proceedings and reserves are established on a case-by-case basis for those legal claims for which management concludes that it is probable that a loss will be incurred and that the amount of such loss can be reasonably estimated. Substantially all of these contingencies are subject to significant uncertainties and, therefore, determining the likelihood of a loss and/or the measurement of any loss can be



complex. With respect to litigation and other legal proceedings where the Company has determined that a loss is reasonably possible, the Company may be unable to estimate the amount or range of reasonably possible loss due to the inherent difficulty of predicting the outcome of and uncertainties regarding such litigation and legal proceedings. The Company's assessments are based on estimates and assumptions that have been deemed reasonable by management, but that may prove to be incomplete or inaccurate, and unanticipated events and circumstances may occur that might cause the Company to change those estimates and assumptions. Therefore, it is possible that an unfavorable resolution of one or more pending litigation or other contingencies could have a material adverse effect on the Company's Consolidated Financial Statements in a future fiscal period. Management's assessment of current litigation and other legal proceedings, including the corresponding accruals, could change because of the discovery of facts with respect to legal actions or other proceedings pending against the Company which are not presently known. Adverse rulings or determinations by judges, juries, governmental authorities or other parties could also result in changes to management's assessment of current liabilities and contingencies. Accordingly, the ultimate costs of resolving these claims may be substantially higher or lower than the amounts reserved. See note 10, commitments and contingencies, for further information.

Revenue recognition

Sales are recognized at an amount that reflects the consideration to which the Company expects to be entitled in exchange for transferring control of goods or services to the customer. Sales are reported on the gross amount billed to a customer less discounts if it has earned revenue as a principal from the sale of goods and services. Sales are reported on the net amount retained (that is, the amount billed to the customer less the amount paid to a vendor) if it has earned a commission or a fee as an agent.

Retail Pharmacy USA and Retail Pharmacy International

The Company recognizes revenue, net of taxes and expected returns, at the time it sells merchandise or dispenses prescription drugs to the customer. The Company estimates revenue based on expected reimbursements from third-party payers (e.g., pharmacy benefit managers, insurance companies and governmental agencies) for dispensing prescription drugs. The estimates are based on all available information including historical experience and are updated to actual reimbursement amounts.

Pharmaceutical Wholesale

Wholesale revenue is recognized, net of taxes and expected returns, upon shipment of goods, which is generally also the day of delivery.

Loyalty programs and gift cards

The Company's loyalty rewards programs represent a separate performance obligation and are accounted for using the deferred revenue approach. When goods are sold, the transaction price is allocated between goods sold and loyalty points awarded based upon the relative standalone selling price. The revenue allocated to the loyalty points is recognized upon redemption. Loyalty programs breakage is recognized as revenue based on the redemption pattern.

Customer purchases of gift cards are not recognized as revenue until the card is redeemed. Gift card breakage (i.e., unused gift card) is recognized as revenue based on the redemption pattern.

Cost of sales

Cost of sales includes the purchase price of goods and cost of services rendered, store and warehouse inventory loss, inventory obsolescence and supplier rebates. In addition to product costs, cost of sales includes warehousing costs for retail operations, purchasing costs, freight costs, cash discounts and vendor allowances.

Vendor allowances and supplier rebates

Vendor allowances are principally received as a result of purchases, sales or promotion of vendors' products. Allowances are generally recorded as a reduction of inventory and are recognized as a reduction of cost of sales when the related merchandise is sold. Allowances received for promoting vendors' products are offset against advertising expense and result in a reduction of selling, general and administrative expenses to the extent of advertising costs incurred, with the excess treated as a reduction of inventory costs.

Rebates or refunds received by the Company from its suppliers, mostly in cash, are considered as an adjustment of the prices of the supplier's products purchased by the Company.

Selling, general and administrative expenses

Selling, general and administrative expenses mainly consist of salaries and employee costs, occupancy costs, depreciation and amortization, credit and debit card fees and expenses directly related to stores. In addition, other costs included are headquarters' expenses, advertising costs (net of vendor advertising allowances), wholesale warehousing costs and insurance.

Advertising costs

Advertising costs, which are reduced by the portion funded by vendors, are expensed as incurred or when services have been received. Net advertising expenses, which are included in selling, general and administrative expenses, were \$585 million in fiscal 2019, \$665 million in fiscal 2018 and \$571 million in fiscal 2017.

Impairment of long-lived assets

The Company tests long-lived assets for impairment whenever events or circumstances indicate that a certain asset or asset group may be impaired. Once identified, the amount of the impairment is computed by comparing the carrying value of the assets to the fair value, which is primarily based on the discounted estimated future cash flows. Impairment charges included in selling, general and administrative expenses were \$260 million in fiscal 2019. Impairment charges recognized in fiscal 2018 and 2017 were \$57 million and \$234 million, respectively.

Stock compensation plans

Stock based compensation is measured at fair value at the grant date. The Company grants stock options, performance shares and restricted units to the Company's non-employee directors, officers and employees. The Company recognizes compensation expense on a straight-line basis over the substantive service period. See note 12, stock compensation plans, for more information on the Company's stock-based compensation plans.

Insurance

The Company obtains insurance coverage for catastrophic exposures as well as those risks required by law to be insured. In general, the Company's U.S. subsidiaries retain a significant portion of losses related to workers' compensation, property, comprehensive general, pharmacist and vehicle liability, while non-U.S. subsidiaries manage their exposures through insurance coverage with third-party carriers. Management regularly reviews the probable outcome of claims and proceedings, the expenses expected to be incurred, the availability and limits of the insurance coverage and the established accruals for liabilities. Liabilities for losses are recorded based upon the Company's estimates for both claims incurred and claims incurred but not reported. The provisions are estimated in part by considering historical claims experience, demographic factors and other actuarial assumptions.

Income taxes

The Company accounts for income taxes according to the asset and liability method. Under this method, deferred tax assets and liabilities are recognized based upon the estimated future tax consequences attributable to differences between the financial statement carrying amounts of existing assets and liabilities are measured pursuant to tax laws using rates expected to apply to taxable income in the years in which those temporary differences are expected to be recovered or settled. The effect on deferred tax assets and liabilities of a change in tax rate is recognized in income in the period that includes the enactment date. Valuation allowances are established when necessary to reduce deferred tax assets to the amounts more likely than not to be realized.

In determining the provision for income taxes, the Company uses income, permanent differences between book and tax income, the relative proportion of foreign and domestic income, enacted statutory income tax rates, projections of income subject to Subpart F rules and unrecognized tax benefits related to current year results. Discrete events such as the assessment of the ultimate outcome of tax audits, audit settlements, recognizing previously unrecognized tax benefits due to lapsing of the applicable statute of limitations, recognizing or de-recognizing benefits of deferred tax assets due to future year financial statement projections and changes in tax laws are recognized in the period in which they occur.

The Company is subject to routine income tax audits that occur periodically in the normal course of business. U.S. federal, state, local and foreign tax authorities raise questions regarding the Company's tax filing positions, including the timing and amount of deductions and the allocation of income among various tax jurisdictions. In evaluating the tax benefits associated with the various tax filing positions, the Company records a tax benefit for uncertain tax positions using the highest cumulative tax benefit that is more likely than not to be realized. Adjustments are made to the liability for unrecognized tax benefits in the period in which the Company determines the issue is effectively settled with the tax authorities, the statute of limitations expires for the return containing the tax position or when more information becomes available.

Earnings per share

The dilutive effect of outstanding stock options on earnings per share is calculated using the treasury stock method. Stock options are anti-dilutive and excluded from the earnings per share calculation if the exercise price exceeds the average market price of the common shares. Outstanding options to purchase common shares that were anti-dilutive and excluded from earnings per share totaled 14.9 million, 10.1 million and 3.9 million in fiscal 2019, 2018 and 2017, respectively.



New accounting pronouncements

Adoption of new accounting pronouncements

Presentation of net periodic pension cost and net periodic postretirement benefit cost

In March 2017, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2017-07, Compensation - Retirement Benefits (Topic 715): Improving the Presentation of Net Periodic Pension Cost and Net Periodic Postretirement Benefit Cost. This ASU requires an employer to report the service cost component of net periodic pension cost and net periodic postretirement benefit cost in the same line item in the statement of earnings as other compensation costs arising from services rendered by the related employees during the period. All other net cost components are required to be presented in the statement of earnings separately from the service cost component and outside a subtotal of income from operations. Additionally, the line item used in the statement of earnings to present the other net cost components must be disclosed in the notes to the financial statements. The Company adopted this new accounting guidance as of September 1, 2018 (fiscal 2019) on a retrospective basis and the adoption did not have a material impact on the Company's results of operations, cash flows or financial position. The impact on our previously reported net periodic costs as a result of the retrospective adoption of this standard results in a reclassification from selling, general and administrative expenses to other income (expense) of \$125 million and \$73 million for the fiscal years ended August 31, 2018 and 2017, respectively.

The following is a reconciliation of the effect of the reclassification of all other net cost components (excluding service cost component) of net pension cost and net postretirement benefit cost from selling, general and administrative expenses to other income (expense) in the Company's Consolidated Statements of Earnings (in millions):

	As reported		Adjustments		As revised	
Twelve months ended August 31, 2018						
Selling, general and administrative expenses	\$ 24,569	\$	125	\$	24,694	
Operating income	6,414		(125)		6,289	
Other income (expense)	177		125		302	
Twelve months ended August 31, 2017						
Selling, general and administrative expenses	\$ 23,740	\$	73	\$	23,813	
Operating income	5,557		(73)		5,484	
Other income (expense)	(11)		73		62	

Restricted cash

In November 2016, the FASB issued ASU 2016-18, Statement of Cash Flows (Topic 230): Restricted Cash. This ASU requires that a statement of cash flows explain the change during the period in the total of cash, cash equivalents, and amounts generally described as restricted cash or restricted cash equivalents. Therefore, amounts generally described as restricted cash and restricted cash equivalents should be included with cash and cash equivalents when reconciling the beginning-of-period and end-of-period total amounts shown on the Statement of Cash Flows. The Company adopted this new accounting guidance as of September 1, 2018 (fiscal 2019) on a full retrospective basis and the adoption did not have a material impact on the Company's Statement of Cash Flows.

The following is a reconciliation of the effect on the relevant line items on the Consolidated Statements of Cash Flows for the twelve months ended August 31, 2018 and 2017 as a result of adopting this new accounting guidance (in millions):

- 69 -

	As reported		Adjustments		As revised
Twelve months ended August 31, 2018					
Trade accounts payable	\$ 1,323	\$	29	\$	1,352
Accrued expenses and other liabilities	281		6		287
Other non-current assets and liabilities	(275)		(36)		(311)
Net cash provided by operating activities	8,265		(2)		8,263
Effect of exchange rate changes on cash, cash equivalents and restricted cash	15		(4)		11
Net increase (decrease) in cash, cash equivalents and restricted cash	(2,516)		(6)		(2,522)
Cash, cash equivalents and restricted cash at beginning of period	3,301		195		3,496
Cash, cash equivalents and restricted cash at end of period	\$ 785	\$	190	\$	975

	As reported Adjustments		As revised		
Twelve months ended August 31, 2017					
Trade accounts payable	\$ 1,690	\$	(6)	\$	1,684
Accrued expenses and other liabilities	(128)		—		(128)
Other non-current assets and liabilities	67		10		77
Net cash provided by operating activities	7,251		4		7,255
Effect of exchange rate changes on cash, cash equivalents and restricted cash	20		6		26
Net increase (decrease) in cash, cash equivalents and restricted cash	(6,506)		10		(6,496)
Cash, cash equivalents and restricted cash at beginning of period	9,807		185		9,992
Cash, cash equivalents and restricted cash at end of period	\$ 3,301	\$	195	\$	3,496

Tax accounting for intra-entity asset transfers

In October 2016, the FASB issued ASU 2016-16, Income Taxes (Topic 740): Intra-Entity Transfers of Assets Other Than Inventory. This ASU prohibits the recognition of current and deferred income taxes for an intra-entity asset transfer until the asset has been sold to an outside party. In addition, interpretations of this guidance have developed in practice for transfers of certain intangible and tangible assets. This prohibition on recognition is an exception to the principle of comprehensive recognition of current and deferred income taxes. To more faithfully represent the economics of intra-entity asset transfers, the amendments in this ASU require that entities recognize the income tax consequences of an intra-entity transfer of an asset other than inventory when the transfer occurs. This ASU does not change the pre-tax effects of an intra-entity asset transfer under Topic 810, Consolidation, or for an intra-entity transfer of inventory. The Company adopted this new accounting guidance as of September 1, 2018 (fiscal 2019) on a modified retrospective basis and the adoption did not have a material impact on the Company's results of operations.

Classification of certain cash receipts and cash payments

In August 2016, the FASB issued ASU 2016-15, Statement of Cash Flows (Topic 230): Classification of Certain Cash Receipts and Cash Payments. This ASU addresses the classification of certain specific cash flow issues including debt prepayments or extinguishment costs, settlement of certain debt instruments, contingent consideration payments made after a business combination, proceeds from the settlement of certain insurance claims, and distributions received from equity method investees. The Company adopted this new accounting guidance as of September 1, 2018 (fiscal 2019) and adoption did not have a material impact on the Company's Statement of Cash Flows.

Revenue recognition on contracts with customers

In May 2014, the FASB issued ASU 2014-09, Revenue from Contracts with Customers (Topic 606). Subsequently, the FASB issued additional ASUs, which further clarify this guidance. This ASU provides a single principles-based revenue recognition model with a five-step analysis of transactions to determine when and how revenue is recognized. The core principle is that a company should recognize revenue to depict the transfer of promised goods or services to customers in an amount that reflects

the consideration to which the entity expects to be entitled in exchange for those goods or services. The Company adopted this new accounting guidance on September 1, 2018 (fiscal 2019) using the modified retrospective transition approach for all contracts and the adoption did not have a material impact on the Company's results of operations. The adoption mainly resulted in changes to recognition of revenues related to loyalty programs and gift cards breakage. Prior to adoption, the Company used the cost approach to account for loyalty programs. Upon adoption, the Company uses the deferred revenue approach. Prior to adoption, gift card breakage was primarily recognized at point of sale. Upon adoption, all gift card breakage is recognized based on the redemption pattern. The changes in accounting for loyalty programs and gift card breakage resulted in a cumulative transition adjustment of \$98 million in retained earnings. See note 17, sales, for additional disclosures. See the updated accounting policy for revenue recognition and loyalty programs in the revenue recognition section above.

Classification and measurement of financial instruments

In January 2016, the FASB issued ASU 2016-01, Financial Instruments - Overall (Subtopic 825-10): Recognition and Measurement of Financial Assets and Financial Liabilities. Subsequently, the FASB issued additional ASUs, which further clarify this guidance. This ASU requires equity investments (except those under the equity method of accounting or those that result in the consolidation of an investee) to be measured at fair value with changes in fair value recognized in net income. However, an entity may choose to measure equity investments that do not have readily determinable fair values at cost less impairment, if any, and changes resulting from observable price changes in orderly transactions for the identical or a similar investment of the same issuer. This simplifies the impairment assessment of equity investments previously held at cost. Separate presentation of financial assets and liabilities by measurement category is required. The Company adopted this new accounting guidance as of September 1, 2018 (fiscal 2019) and adoption did not have a material impact on the Company's results of operations, cash flows or financial position. The new guidance was applied on a modified retrospective basis, with the exception of the amendments related to the measurement alternative for equity investments without readily determinable fair values, which was applied on a prospective basis.

New accounting pronouncements not yet adopted

Investments - equity securities

In April 2019, the FASB issued ASU 2019-04, Codification Improvements to Financial Instruments-Credit Losses (Topic 326), Derivatives and Hedging (Topic 815), and Financial Instruments (Topic 825). This extensive ASU provides clarifications for three topics related to financial instruments accounting, some of which apply to the Company. For example, this ASU clarifies the disclosure requirements that apply to equity securities without a readily determinable fair value for which the measurement alternative is elected. This ASU is effective for fiscal years beginning after December 15, 2019 (fiscal 2021). The adoption of this ASU is not expected to have a significant impact on the Company's results of operations, cash flows or financial position.

Collaborative arrangements

In November 2018, the FASB issued ASU 2018-18, Collaborative Arrangements (Topic 808). This ASU clarifies the interaction between Topic 808, Collaborative Arrangements, and Topic 606, Revenue from Contracts with Customers. This ASU is effective for fiscal years beginning after December 15, 2019 (fiscal 2021). The adoption of this ASU is not expected to have a significant impact on the Company's results of operations, cash flows or financial position.

Financial instruments - hedging and derivatives

In October 2018, the FASB issued ASU 2018-16, Derivatives and Hedging (Topic 815): Inclusion of the Secured Overnight Financing Rate (SOFR) Overnight Index Swap (OIS) Rate as Benchmark Interest Rate for Hedge Accounting Purposes. This ASU permits use of the OIS rate based on the SOFR as a U.S. benchmark interest rate for hedge accounting purposes. This ASU is effective for fiscal years beginning after December 15, 2018 (fiscal 2020), and interim periods within those fiscal years, with early adoption permitted. The new guidance must be applied on a prospective basis. The adoption of this ASU is not expected to have a significant impact on the Company's results of operations, cash flows or financial position.

Intangibles – goodwill and other – internal-use software

In August 2018, the FASB issued ASU 2018-15, Intangibles-Goodwill and Other- Internal-Use Software (Subtopic 350-40). This ASU addresses customer's accounting for implementation costs incurred in a cloud computing arrangement that is a service contract and also adds certain disclosure requirements related to implementation costs incurred for internal-use software and cloud computing arrangements. The amendment aligns the requirements for capitalizing implementation costs incurred in a hosting arrangement that is a service contract with the requirements for capitalizing implementation costs incurred in a hosting arrangement that is a service contract with the requirements for capitalizing implementation costs incurred to develop or obtain internal-use software (and hosting arrangements that include an internal-use software license). This ASU is effective for fiscal years beginning after December 15, 2019 (fiscal 2021), and interim periods within those fiscal years, with early adoption permitted. The amendments in this ASU can be applied either retrospectively or prospectively to all implementation costs incurred after the date of adoption. The Company is evaluating the effect of adopting this new accounting guidance, but does not expect adoption will have a material impact on the Company's financial position or results of operations.

Compensation - retirement benefits - defined benefit plans

In August 2018, the FASB issued ASU 2018-14, Compensation - Retirement benefits (Topic 715-20). This ASU amends ASC 715 to add, remove and clarify disclosure requirements related to defined benefit pension and other postretirement plans. The ASU eliminates the requirement to disclose the amounts in accumulated other comprehensive income expected to be recognized as part of net periodic benefit cost over the next year. The ASU also removes the disclosure requirements for the effects of a one-percentage-point change on the assumed health care costs and the effect of this change in rates on service cost, interest cost and the benefit obligation for postretirement health care benefits. This ASU is effective for fiscal years ending after December 15, 2020 (fiscal 2022) and must be applied on a retrospective basis. The Company is evaluating the effect of adopting this new accounting guidance, but does not expect adoption will have a material impact on the Company's financial position.

Fair value measurement

In August 2018, the FASB issued ASU 2018-13, Fair Value Measurement (Topic 820). The ASU eliminates such disclosures as the amount of and reasons for transfers between Level 1 and Level 2 of the fair value hierarchy. The ASU adds new disclosure requirements for Level 3 measurements. This ASU is effective for fiscal years beginning after December 15, 2019 (fiscal 2021), and interim periods within those fiscal years, with early adoption permitted for any eliminated or modified disclosures. The Company is evaluating the effect of adopting this new accounting guidance, but does not expect adoption will have a material impact on the Company's disclosures.

Contributions made

In June 2018, the FASB issued ASU 2018-08, Clarifying the Scope and the Accounting Guidance for Contributions Received and Contributions Made (Topic 958). This ASU clarifies and improves guidance about whether a transfer of assets (or reduction of liabilities) is a contribution or an exchange transaction, and whether a contribution is conditional. The ASU applies to all entities, including business entities, that receive or make contributions of cash or other assets, including promises to give. This ASU is effective for fiscal years beginning after December 15, 2018 (fiscal 2020) and must be applied on a prospective basis in the period of adoption. The Company is evaluating the effect of adopting this new accounting guidance, but does not expect adoption will have a material impact on the Company's financial position or results of operations.

Compensation - stock compensation

In June 2018, the FASB issued ASU 2018-07, Compensation-Stock Compensation (Topic 718). This ASU eliminated most of the differences between accounting guidance for share-based compensation granted to nonemployees and the guidance for share-based compensation granted to employees. The ASU supersedes the guidance for nonemployees and expands the scope of the guidance for employees to include both. This ASU is effective for annual periods beginning after December 15, 2018 (fiscal 2020), and interim periods within those years. The Company is evaluating the effect of adopting this new accounting guidance, but does not expect adoption will have a material impact on the Company's financial position.

Accounting for reclassification of certain tax effects from accumulated other comprehensive income

In February 2018, the Financial Accounting Standards Board ("FASB") issued Accounting Standards Update ("ASU") 2018-02, Income Statement-Reporting Comprehensive Income (Topic 220): Reclassification of Certain Tax Effects from Accumulated Other Comprehensive Income. This ASU addresses the income tax effects of items in accumulated other comprehensive income ("AOCI") which were originally recognized in other comprehensive income, rather than in income from continuing operations. Specifically, it permits a reclassification from AOCI to retained earnings for the adjustment of deferred taxes due to the reduction of the historical corporate income tax rate to the newly enacted corporate income tax rate resulting from the U.S. tax law changes enacted in December 2017. It also requires certain disclosures about these reclassifications. This ASU is effective for fiscal years beginning after December 15, 2018 (fiscal 2020), and interim periods within those fiscal years, with early adoption permitted. The new guidance must be applied either on a prospective basis in the period of adoption or retrospectively to each period (or periods) in which the effect of the change in the U.S. federal corporate income tax rate in the U.S. tax law changes are recognized. The Company does not expect this adoption will have a material impact on the Company's financial position.

Financial instruments - credit losses

In June 2016, the FASB issued ASU 2016-13: Measurement of Credit Losses on Financial Instruments (Topic 326), which amends the Board's guidance on the impairment of financial instruments. The ASU adds to U.S. GAAP an impairment model that is based on expected losses rather than incurred losses, which is known as the current expected credit loss ("CECL") model. The CECL model applies to most debt instruments (other than those measured at fair value), trade and other receivables, financial guarantee contracts, and loan commitments. This ASU is effective for fiscal years beginning after December 15, 2019 (fiscal 2021), and interim periods within those fiscal years, with early adoption permitted. The Company is evaluating the effect of adopting this new accounting guidance, but does not expect adoption will have a material impact on the Company's financial position or results of operations.

- 72 -

Leases

In February 2016, the FASB issued ASU 2016-02, Leases (Topic 842), which supersedes Topic 840, Leases. Subsequently, the FASB has issued additional ASUs which further clarify this guidance. This ASU increases the transparency and comparability of organizations by requiring the capitalization of substantially all leases on the balance sheet and disclosures of key information about leasing arrangements. Under this new guidance, at the lease commencement date, a lessee recognizes a right- of-use asset and lease liability, which is initially measured at the present value of the future lease payments. For income statement purposes, a dual model was retained for lessees, requiring leases to be classified as either operating or finance leases. Under the operating lease model, lease expense is recognized on a straight-line basis over the lease term. Under the finance lease model, interest on the lease liability is recognized separately from amortization of the right-of-use asset. In addition, a new ASU was issued in July 2018, to provide relief to companies from restating the comparative periods.

The Company adopted this new accounting standard on September 1, 2019 on a modified retrospective basis and applied the new standard to all leases through a cumulative-effect adjustment to beginning retained earnings. As a result, comparative financial information has not been restated and continues to be reported under the accounting standards in effect for those periods. The Company elected a package of practical expedients permitted under the transition guidance, which among other things, allows the carryforward of historical lease classification. The adoption of this new accounting standard is expected to result in recognition of lease liabilities of approximately \$24 billion and recognition of right-of-use assets of approximately \$22 billion net of liabilities for facility closing, deferred rent, favorable lease interest intangible asset, unfavorable lease interest liability, lease incentives and prepaid rent as of August 31, 2019. The adoption is also expected to result in a decrease to retained earnings of approximately \$0.4 billion due to transition date impairment of right-of-use assets related to previously impaired long-lived assets of approximately \$0.8 billion net of tax, partially offset by de-recognition of deferred gains on historical sale and leaseback transactions of approximately \$0.4 billion net of tax.

Note 2. Acquisitions

Acquisition of certain Rite Aid assets

On September 19, 2017, the Company announced that it had secured regulatory clearance for an amended and restated asset purchase agreement to purchase 1,932 stores, three distribution centers and related inventory from Rite Aid for \$4.375 billion in cash and other consideration. The purchases of these stores have been accounted for as business combinations and occurred in waves during fiscal 2018. The Company purchased 1,932 stores for total cash consideration of \$4.2 billion for the fiscal year ended August 31, 2018.

As of May 31, 2019, the Company completed the analysis to assign fair values for assets acquired and liabilities assumed for the acquired stores. During the fiscal year ended August 31, 2019, the Company recorded certain measurement period adjustments based on additional information primarily to other non-current liabilities, intangible assets and deferred income taxes, which did not have a material impact on goodwill. The following table summarizes the consideration paid and the amounts of identified assets acquired and liabilities assumed for purchase of 1,932 stores as of the fiscal year ended August 31, 2019.

Consideration	\$ 4,330
Identifiable assets acquired and liabilities assumed	
Inventories	\$ 1,171
Property, plant and equipment	490
Intangible assets	2,039
Accrued expenses and other liabilities	(55)
Deferred income taxes	291
Other non-current liabilities	(937)
Total identifiable net assets	\$ 2,999
Goodwill	\$ 1,331

- 73 -

The identified definite-lived intangible assets were as follows:

	Weighted-average useful	Weighted-average useful life						
Definite-lived intangible assets	(in years)	Amou	ınt (in millions)					
Customer relationships	12	\$	1,800					
Favorable lease interests	10		219					
Trade names	2		20					
Total		\$	2,039					

Consideration includes cash of \$4,157 million and the fair value of the option granted to Rite Aid to become a member of the Company's group purchasing organization, Walgreens Boots Alliance Development GmbH. The fair value for this option was determined using the income approach methodology. The fair value estimates are based on the market compensation for such services and appropriate discount rate, as relevant, that market participants would consider when estimating fair values. During fiscal 2019, this option was terminated resulting in recognition of a gain in other income (expense).

The goodwill of \$1,331 million arising from the business combinations primarily reflects the expected operational synergies and cost savings generated from the Store Optimization Program as well as the expected growth from new customers. See note 3, exit and disposal activities, for additional information. The goodwill was allocated to the Retail Pharmacy USA segment. Substantially all of the goodwill recognized is expected to be deductible for income tax purposes.

The fair value for customer relationships was determined using the multi-period excess earnings method, a form of the income approach. Real property fair values were determined using primarily the income approach and sales comparison approach. The fair value measurements of the intangible assets are based on significant inputs not observable in the market and thus represent Level 3 measurements. The fair value estimates for the intangible assets are based on projected discounted cash flows, historical and projected financial information and attrition rates, as relevant, that market participants would consider when estimating fair values.

The following table presents supplemental unaudited pro forma consolidated sales for the fiscal year ended 2018 and 2017 as if all 1,932 stores were acquired on September 1, 2016. Pro forma net earnings of the Company, assuming these purchases had occurred at the beginning of each period presented, would not be materially different from the results reported. See note 3, exit and disposal activities, for additional information. The unaudited pro forma information has been prepared for comparative purposes only and is not intended to be indicative of what the Company's results would have been had the purchases occurred at the beginning of the periods presented or results which may occur in the future.

(in millions)	2018 ¹		2017	
Sales	\$	135,503	\$	127,893

¹ Impacted by store closures due to the Store Optimization Program.

Actual sales from acquired Rite Aid stores for the fiscal year ended 2018 included in the Consolidated Statement of Earnings are as follows:

(in millions)	2018
Sales	\$ 5,112

The 1,932 Rite Aid stores acquired did not have a material impact on net earnings of the Company for the fiscal year ended 2018.

The Company acquired the first distribution center and related inventory for cash consideration of \$61 million during the twelve months ended August 31, 2019. The transition of the remaining two distribution centers and related inventory remains subject to closing conditions set forth in the amended and restated asset purchase agreement.

AllianceRx Walgreens Prime

On March 31, 2017, Walgreens Boots Alliance and pharmacy benefit manager Prime Therapeutics LLC ("Prime") closed a transaction to form a combined central specialty pharmacy and mail services company AllianceRx Walgreens Prime, as part of a strategic alliance. AllianceRx Walgreens Prime is consolidated by Walgreens Boots Alliance and reported within the Retail Pharmacy USA division in its financial statements. The Company accounted for this acquisition of Prime's specialty pharmacy

- 74 -

and mail services business as a business combination involving noncash purchase consideration of \$720 million consisting of the issuance of an equity interest in AllianceRx Walgreens Prime.

The Company has completed the purchase accounting for the AllianceRx Walgreens Prime transaction. The following table summarizes the consideration for the acquisition and the amounts of identified assets acquired and liabilities assumed at the date of the transaction (in millions).

Total consideration	\$ 720
Identifiable assets acquired and liabilities assumed	
Accounts receivable	\$ 217
Inventories	149
Property, plant and equipment	11
Intangible assets	331
Trade accounts payable	(90)
Accrued expenses and other liabilities	 (1)
Total identifiable net assets	617
Goodwill	\$ 103

The identified intangible assets primarily include payer contracts. These contracts are estimated to have a weighted average useful life of 15 years. The goodwill of \$103 million arising from the transaction consists of expected purchasing synergies, operating efficiencies by benchmarking performance and applying best practices across the combined company, consolidation of operations, reductions in selling, general and administrative expenses and combining workforces. Substantially all of the goodwill recognized is not expected to be deductible for income tax purposes.

In accordance with ASC Topic 810, Consolidation, the noncontrolling interest was recognized based on its proportionate interest in the identifiable net assets of AllianceRx Walgreens Prime. The difference between the carrying amount of the noncontrolling interest and the fair value recognized as consideration in the business combination is recognized as additional paid in capital.

Pro forma net earnings and sales of the Company, assuming the acquisition had occurred at the beginning of each period presented, would not be materially different from the results reported. The acquisition did not have a material impact on net earnings or sales of the Company for fiscal 2017.

Other acquisitions

The Company acquired certain prescription files and related pharmacy inventory in Retail Pharmacy USA for the aggregate purchase price of \$304 million during the twelve months ended August 31, 2019.

Note 3. Exit and disposal activities

Transformational Cost Management Program

On December 20, 2018, the Company announced a transformational cost management program that is expected to deliver in excess of \$1.5 billion of annual cost savings by fiscal 2022 (the "Transformational Cost Management Program"). As of the date of this report, the Company now expects to deliver in excess of \$1.8 billion of annual cost savings by fiscal 2022. The Transformational Cost Management Program, which is multi-faceted and includes divisional optimization initiatives, global smart spending, global smart organization and the transformation of the Company's information technology (IT) capabilities, is designed to help the Company achieve increased cost efficiencies. To date, the Company has taken initial actions across all aspects of the Transformational Cost Management Program focus on all reportable segments and the Company's global functions. Divisional optimization within each of the Company's segments includes activities such as optimization of stores which includes plans to close approximately 200 stores in the United Kingdom and approximately 200 locations in the United States.

The Company currently estimates that the Transformational Cost Management Program will result in cumulative pre-tax charges to its GAAP financial results of approximately \$1.9 billion to \$2.4 billion, of which \$1.6 billion to \$2.0 billion are expected to be recorded as exit and disposal activities.



Costs related to exit and disposal activities under the Transformational Cost Management Program, which were primarily recorded in selling, general and administrative expenses included in the fiscal year ended August 31, 2019 were as follows (in millions):

Twelve Months Ended August 31, 2019	Retai	il Pharmacy USA		ail Pharmacy iternational	Р	harmaceutical Wholesale	algreens Boots Alliance, Inc.
Lease obligations and other real estate costs	\$	5	\$	19	\$	1	\$ 25
Asset impairment ¹		95		67		98	260
Employee severance and business transition costs		42		34		49	125
Information technology transformation and other exit costs		5		10		7	22
Total pre-tax exit and disposal charges	\$	147	\$	130	\$	154	\$ 432
¹ Primarily includes write down of leasehold improvements cer	tain softw	are and invent	orv				

Primarily includes write down of leasehold improvements, certain software and inventory.

The changes in liabilities and assets related to the exit and disposal activities under Transformational Cost Management Program include the following (in millions):

	severance exit costs	Asset Impairments	Lease obligations and other real estate costs	Information technology transformation and other exit costs	Total
Balance at August 31, 2018	\$ _	\$ —	\$ _	\$ —	\$ —
Costs	125	260	25	22	432
Payments	(69)	—	(8)	(13)	(90)
Other - non cash		(260)	—	(6)	(265)
Currency translation adjustments	1	—	—	—	1
Balance at August 31, 2019	\$ 57	\$ —	\$ 17	\$ 4	\$ 78

Store Optimization Program

On October 24, 2017, the Company's Board of Directors approved a plan to implement a program (the "Store Optimization Program") to optimize store locations through the planned closure of approximately 600 stores and related assets within the Company's Retail Pharmacy USA segment upon completion of the acquisition of certain stores and related assets from Rite Aid. As of the date of this report, the Company expects to close approximately 750 stores, of which the majority have been closed as part of this program. The actions under the Store Optimization Program commenced in March 2018 and are expected to be complete by end of fiscal 2020.

The Company currently estimates that it will recognize cumulative pre-tax charges to its GAAP financial results of approximately \$350 million, including costs associated with lease obligations and other real estate costs, employee severance and other exit costs. The Company expects to incur pre-tax charges of approximately \$160 million for lease obligations and other real estate costs and approximately \$190 million for employee severance and other exit costs. The Company estimates that substantially all of these cumulative pre-tax charges will result in cash expenditures.

Since approval of the Store Optimization Program, the Company has recognized cumulative pre-tax charges to its financial results in accordance with GAAP totaling \$296 million, which were primarily recorded within selling, general and administrative expenses. These charges included \$138 million related to lease obligations and other real estate costs and \$158 million employee severance and other exit costs.

- 76 -

Costs related to the Store Optimization Program, which were primarily recorded in selling, general and administrative expenses for the Company's Retail Pharmacy USA segment included in the fiscal years ended August 31, 2019 and 2018, are as follows (in millions):

	20	019	:	2018
Lease obligations and other real estate costs	\$	119	\$	19
Employee severance and other exit costs		77		81
Total costs	\$	196	\$	100

The changes in liabilities related to the Store Optimization Program for the fiscal years ended August 31, 2019 and 2018 include the following (in millions):

	ease obligations and other real estate costs	sev	Employee rerance and other exit costs	Total
Balance at August 31, 2017	\$ _	\$	_	\$ _
Costs	19		81	100
Payments	(18)		(60)	(78)
Other - non cash ¹	307		—	307
Balance at August 31, 2018	\$ 308	\$	21	\$ 329
Costs	119		77	196
Payments	(171)		(69)	(240)
Other - non cash ¹	152		(7)	144
Balance at August 31, 2019	\$ 407	\$	22	\$ 429

¹ Primarily represents unfavorable lease liabilities from acquired Rite Aid stores.

Cost Transformation Program

On April 8, 2015, the Walgreens Boots Alliance Board of Directors approved a plan to implement a restructuring program (the "Cost Transformation Program") as part of an initiative to reduce costs and increase operating efficiencies. The Cost Transformation Program implemented and built on the cost-reduction initiative previously announced by the Company on August 6, 2014 and included plans to close stores across the United States; reorganize corporate and field operations; drive operating efficiencies; and streamline information technology and other functions. The actions under the Cost Transformation Program focused primarily on the Retail Pharmacy USA segment, but included activities from all segments. The Company completed the Cost Transformation Program in the fourth quarter of fiscal 2017.

The changes in liabilities related to the Cost Transformation Program include the following (in millions):

	Severance and other business transition and exit Real estate costs costs				Total
Balance at August 31, 2017	\$	521	\$ 79	\$	600
Payments		(139)	(68))	(207)
Other - non cash		32	(3))	29
Currency translation adjustments			(1)	(1)
Balance at August 31, 2018	\$	414	\$ 7	\$	421
Payments		(86)	(4)	(90)
Other - non cash		59	—		59
Balance at August 31, 2019	\$	387	\$ 3	\$	390

- 77 -

Total costs by segment, which were primarily recorded in selling, general and administrative expenses included in the fiscal year ended August 31, 2017 are as follows (in millions):

Fiscal year ended 2017	Pharmacy USA	etail Pharmacy International	Р	harmaceutical Wholesale	/algreens Boots Alliance, Inc.
Asset impairments	\$ 272	\$ 21	\$	2	\$ 295
Real estate costs	372	—		—	372
Severance and other business transition and exit costs	87	46		35	168
Total costs	\$ 731	\$ 67	\$	37	\$ 835

Note 4. Leases

The following table is a summary of the future minimum lease payments as of August 31, 2019 (in millions):

	Finan	ce lease				
	obli	gation	Capital lease		Oper	ating lease ¹
2020	\$	21	\$	59	\$	3,484
2021		21		60		3,323
2022		20		58		3,124
2023		20		57		2,943
2024		20		58		2,771
Later		209		801		16,623
Total minimum lease payments	\$	311	\$	1,093	\$	32,268

¹ Includes \$1.6 billion of minimum rental commitments on closed locations

The capital and finance lease amounts include \$683 million of imputed interest. Total minimum lease payments have not been reduced by minimum sublease rentals of \$344 million due in the future under non-cancelable subleases.

The Company continuously evaluates its real estate portfolio in conjunction with its capital needs. Historically, the Company has entered into several saleleaseback transactions. In fiscal 2019, proceeds from sale-leaseback transactions were not material. In fiscal 2018, the Company did not record any proceeds from sale-leaseback transactions. In fiscal 2017, the Company recorded proceeds from sale-leaseback transactions of \$444 million.

In fiscal 2019, 2018 and 2017, the Company recorded charges of \$185 million, \$129 million and \$394 million, respectively, for closed or relocated facilities. These charges are reported in selling, general and administrative expenses in the Consolidated Statements of Earnings.

The changes in liability for facility closings and related lease termination charges include the following (in millions):

	2019	2018
Balance at beginning of period	\$ 964	\$ 718
Provision for present value of non-cancelable lease payments on closed facilities	90	52
Changes in assumptions	56	19
Accretion expense	39	58
Other - non cash ¹	160	338
Cash payments, net of sublease income	(316)	(221)
Balance at end of period	\$ 993	\$ 964

¹ Primarily unfavorable lease liabilities from acquired Rite Aid stores.

Rental expense, which includes common area maintenance, insurance and taxes, where appropriate, was as follows (in millions):

- 78 -

	20	19	2018	2017		
Minimum rentals	\$	3,622	\$ 3,447	\$	3,259	
Contingent rentals		74	68		59	
Less: sublease rental income		(66)	(67)		(55)	
	\$	3,631	\$ 3,448	\$	3,263	

Note 5. Equity method investments

Equity method investments as of August 31, 2019 and 2018 were as follows (in millions, except percentages):

		201	.9		2018			
	Carry	ing value	Ownership percentage	Carr	ying value	Ownership percentage		
AmerisourceBergen	\$	5,211	27%	\$	5,138	26%		
Others		1,640	8% - 50%		1,472	8% - 50%		
Total	\$ 6,851			\$	6,610			

AmerisourceBergen Corporation ("AmerisourceBergen") investment

As of August 31, 2019 and 2018, the Company owned 56,854,867 AmerisourceBergen common shares, representing approximately 27% and 26% of the outstanding AmerisourceBergen common stock, respectively. The Company accounts for its equity investment in AmerisourceBergen using the equity method of accounting, with the net earnings attributable to the Company's investment being classified within the operating income of its Pharmaceutical Wholesale segment. Due to the timing and availability of financial information of AmerisourceBergen, the Company accounts for this equity method investment on a financial reporting lag of two months. Equity earnings from AmerisourceBergen are reported as a separate line in the Consolidated Statements of Earnings. The financial performance of AmerisourceBergen, including any charges which may arise relating to its ongoing opioid litigation, will impact the Company's results of operations.

The Level 1 fair market value of the Company's equity investment in AmerisourceBergen common stock at August 31, 2019 and 2018 were \$4.7 billion and \$5.1 billion, respectively. As of August 31, 2019, the Company's investment in AmerisourceBergen carrying value exceeded its proportionate share of the net assets of AmerisourceBergen by \$4.4 billion. This premium of \$4.4 billion was recognized as part of the carrying value in the Company's equity investment in AmerisourceBergen. The difference was primarily related to goodwill and the fair value of AmerisourceBergen intangible assets.

Other investments

The Company's other equity method investments include its investments in Guangzhou Pharmaceuticals Corporation ("Guangzhou Pharmaceuticals") and Nanjing Pharmaceutical Corporation Limited, the Company's pharmaceutical wholesale investments in China; its investment in Sinopharm Holding Guoda Drugstores Co., Ltd., the Company's retail pharmacy investment in China, the Company's investment in BioScrip (resulting from its merger with Option Care Inc.) and PharMerica Corporation in the United States.

The Company reported \$23 million, \$53 million and \$8 million of post-tax equity earnings from other equity method investments, including equity method investments classified as operating, for the fiscal years ended 2019, 2018 and 2017, respectively.

During the fiscal year ended August 31, 2018, the Company recorded an impairment of \$170 million in its equity interest in Guangzhou Pharmaceuticals, which was included in other income (expense) in the Consolidated Statement of Earnings. The fair value of the Company's equity interest in Guangzhou Pharmaceuticals was determined using the proposed sale price and thus represents Level 3 measurement. During the fiscal year ended August 31, 2018, the Company completed the sale of a 30 percent interest in Guangzhou Pharmaceuticals to its joint venture partner Guangzhou Baiyunshan Pharmaceutical Holdings resulting in a \$172 million reduction in carrying value and a \$8 million cumulative translation adjustment loss. In July 2018, the Company completed the sale of its minority equity interest in Premise Health Holding Corp., resulting in an after-tax gain on disposition of \$245 million and a reduction in carrying value of \$76 million.

- 79 -

Summarized financial information

Summarized financial information for the Company's equity method investments in aggregate is as follows:

Balance sheet (in millions)

		Year endeo	Year ended August 31,				
	2019			2018			
Current assets	\$	36,523	\$	34,493			
Non-current assets		15,710		14,971			
Current liabilities		35,857		34,055			
Non-current liabilities		9,633		8,759			
Shareholders' equity ¹		6,743		6,650			

Statements of earnings (in millions)

		Year ended August 31,							
	2019 2018					2017			
Sales	\$	197,237	\$	179,887	\$	164,844			
Gross profit		7,516		6,875		5,958			
Net earnings		1,037		1,315		1,040			
Share of earnings from equity method investments		187		245		143			

¹ Shareholders' equity at August 31, 2019 and 2018 includes \$411 million and \$445 million, respectively, related to noncontrolling interests.

The summarized financial information for equity method investments has been included on an aggregated basis for all investments as reported at the end of each fiscal year end.

Note 6. Goodwill and other intangible assets

Goodwill and indefinite-lived intangible assets are evaluated for impairment annually or more frequently if an event occurs or circumstances change that could more likely than not reduce the fair value of a reporting unit or intangible asset below its carrying value. Based on this annual evaluation as of June 1, 2019 valuation date, the fair values of the Company's reporting units exceeded their carrying amounts ranging from approximately 9% to approximately 269%. The fair value of the Boots reporting unit, within the Retail Pharmacy International segment, is in excess of its carrying value by approximately 9%. As at August 31, 2019, the carrying value of the goodwill of the Boots reporting unit is \$2.6 billion. The fair values of certain indefinite-lived intangibles within the Boots reporting unit exceeded their carrying amounts ranging from approximately 29%, except for the pharmacy licenses. As at August 31, 2019, the carrying value of these indefinite-lived intangibles within the Boots reporting unit is \$6.9 billion. As a result of an annual evaluation, during the three months ended August 31, 2019, the Company recorded an impairment of \$73 million on its pharmacy licenses in the Boots reporting unit, which was included in selling, general and administrative expenses in the Consolidated Statement of Earnings. The fair value of pharmacy licenses was determined using excess earnings method and thus represents Level 3 measurement.

The Company continues to closely monitor industry and market trends and the impact it may have on the Boots reporting unit and indefinite-lived intangibles as the challenging market conditions in the UK continued to impact the operating performance of the Boots reporting unit during the three months ended August 31, 2019.

The fair value for each reporting unit is estimated using both the income and market approaches. The income approach requires management to estimate a number of factors for each reporting unit, including projected future operating results, economic projections, anticipated future cash flows and discount rates. The market approach estimates fair value using comparable marketplace fair value data from within a comparable industry grouping. The determination of the fair value of the reporting units requires us to make significant estimates and assumptions. These estimates and assumptions primarily include, but are not limited to: the selection of appropriate peer group companies, control premiums appropriate for acquisitions in the industries in which we compete, discount rates, terminal growth rates, forecasts of revenue, operating income, depreciation, amortization and capital expenditures. Indefinite-lived intangible assets fair values are estimated using the relief from royalty method and excess earnings method of the income approach. These estimates can be affected by a number of factors including, but not

- 80 -

limited to, general economic conditions, availability of market information as well as our profitability. Although the Company believes its estimates of fair value are reasonable, actual financial results could differ from those estimates due to the inherent uncertainty involved in making such estimates. Changes in assumptions concerning future financial results or other underlying assumptions could have a significant impact on either the fair value of the reporting units, the amount of the goodwill impairment charge, or both.

Changes in the carrying amount of goodwill by reportable segment consist of the following activity (in millions):

	Retail	etail Pharmacy USA				Pharmaceutical Wholesale	Walgreens Boots Alliance, Inc.
August 31, 2017	\$	9,139	\$	3,392	\$	3,101	\$ 15,632
Acquisitions		1,344		—		4	1,348
Currency translation adjustments		—		(22)		(44)	(66)
August 31, 2018	\$	10,483	\$	3,370	\$	3,061	\$ 16,914
Acquisitions		8				_	 8
Dispositions		_		(9)		—	(9)
Currency translation adjustments		—		(182)		(171)	(353)
August 31, 2019	\$	10,491	\$	3,179	\$	2,890	\$ 16,560

In fiscal 2019, the Company completed the analysis to assign fair values for assets acquired and liabilities assumed for the acquired Rite Aid stores resulting in decreases to goodwill of \$13 million. In fiscal 2018, the Company purchased 1,932 stores from Rite Aid for total consideration of \$4.3 billion, resulting in an increase of \$1,344 million to goodwill and \$2,054 million to intangible assets during the year. See note 2, acquisitions, for additional information.

The carrying amount and accumulated amortization of intangible assets consists of the following (in millions):

	Augu	August 31, 2018		
Gross amortizable intangible assets				
Customer relationships and loyalty card holders ¹	\$	4,290	\$	4,235
Favorable lease interests and non-compete agreements		654		680
Trade names and trademarks		461		489
Purchasing and payer contracts		382		390
Total gross amortizable intangible assets		5,787		5,794
Accumulated amortization				
Customer relationships and loyalty card holders ¹	\$	1,262	\$	997
Favorable lease interests and non-compete agreements		410		359
Trade names and trademarks		250		206
Purchasing and payer contracts		99		78
Total accumulated amortization		2,021		1,640
Total amortizable intangible assets, net	\$	3,766	\$	4,154
Indefinite-lived intangible assets				
Trade names and trademarks	\$	5,232	\$	5,557
Pharmacy licenses		1,878		2,072
Total indefinite-lived intangible assets	\$	7,110	\$	7,629
Total intangible assets, net	\$	10,876	\$	11,783

¹ Includes purchased prescription files.

- 81 -

Amortization expense for intangible assets was \$552 million, \$493 million and \$385 million in fiscal 2019, 2018 and 2017, respectively.

Estimated future annual amortization expense for the next five fiscal years for intangible assets recorded at August 31, 2019 is as follows (in millions):

		2020		2020		2020		2020 2021		2021	2022			2023	2024		
Estimated annual amortization expense	\$	478	\$	433	\$	412	\$	378	\$	357							

Note 7. Debt

Debt consists of the following (all amounts are presented in millions of U.S. dollars and debt issuances are denominated in U.S. dollars, unless otherwise noted)

	I	August 31, 2019	August 31, 2018
Short-term debt ¹			
Commercial paper	\$	2,400	\$ 430
Credit facilities ²		1,624	999
\$8 billion note issuance ^{3,4}			
2.700% unsecured notes due 2019		1,250	_
\$1 billion note issuance ⁵			
5.250% unsecured notes due 2019 ⁶			249
Other ⁷		464	288
Total short-term debt	\$	5,738	\$ 1,966
Long-term debt ¹			
\$6 billion note issuance ^{3,4}			
3.450% unsecured notes due 2026	\$	1,890	\$ 1,888
4.650% unsecured notes due 2046		591	590
\$8 billion note issuance ^{3,4}			
2.700% unsecured notes due 2019			1,248
3.300% unsecured notes due 2021		1,247	1,245
3.800% unsecured notes due 2024		1,992	1,990
4.500% unsecured notes due 2034		495	495
4.800% unsecured notes due 2044		1,492	1,492
£700 million note issuance ^{3,4}			
2.875% unsecured Pound sterling notes due 2020		488	517
3.600% unsecured Pound sterling notes due 2025		365	387
€750 million note issuance ^{3,4}			
2.125% unsecured Euro notes due 2026		824	868
\$4 billion note issuance ^{3,5}			
3.100% unsecured notes due 2022		1,197	1,196
4.400% unsecured notes due 2042		493	492
Other ⁸		25	23
Total long-term debt, less current portion	\$	11,098	\$ 12,431

¹ Carrying values are presented net of unamortized discount and debt issuance costs, where applicable, and foreign currency denominated debt has been translated using the spot rates at August 31, 2019 and 2018, respectively.

² Credit facilities primarily include debt outstanding under the various credit facilities described in more detail below.

³ The \$6 billion, \$8 billion, £0.7 billion, €0.75 billion, and \$4 billion note issuances as of August 31, 2019 had a fair value and carrying value of \$2.6 billion and \$2.5 billion, \$6.8 billion and \$6.5 billion, \$0.9 billion and \$0.9 billion, \$0.9 billion and \$0.8 billion, \$1.7 billion and \$1.7 billion, respectively. The fair values of the notes outstanding are Level 1 fair value



measures and determined based on quoted market price and translated at the August 31, 2019 spot rate, as applicable. The fair values and carrying values of these issuances do not include notes that have been redeemed or repaid as of August 31, 2019.

- ⁴ Notes are unsubordinated debt obligations of Walgreens Boots Alliance and rank equally in right of payment with all other unsecured and unsubordinated indebtedness of Walgreens Boots Alliance from time to time outstanding.
- ⁵ Notes are senior debt obligations of Walgreen Co. and rank equally with all other unsecured and unsubordinated indebtedness of Walgreen Co. On December 31, 2014, Walgreens Boots Alliance fully and unconditionally guaranteed the outstanding notes on an unsecured and unsubordinated basis. The guarantee, for so long as it is in place, is an unsecured, unsubordinated debt obligation of Walgreens Boots Alliance and will rank equally in right of payment with all other unsecured and unsubordinated indebtedness of Walgreens Boots Alliance.
- ⁶ Includes interest rate swap fair market value adjustments. See note 9, fair value measurements, for additional fair value disclosures.
- ⁷ Other short-term debt represents a mix of fixed and variable rate debt with various maturities and working capital facilities denominated in various currencies.
- ⁸ Other long-term debt represents a mix of fixed and variable rate debt in various currencies with various maturities.

At August 31, 2019, the future maturities of short-term and long-term debt, excluding debt discounts and issuance costs and financing and capital lease obligations (see note 4, leases, for the future lease obligation maturities), consisted of the following (in millions):

	Amount
2020	\$ 5,752
2021	491
2022	1,250
2023	1,200
2024	—
Later	8,223
Total estimated future maturities	\$ 16,916

August 2019 Revolving Credit Agreements

On August 30, 2019, the Company entered into three \$500 million revolving credit agreements (together, the "August 2019 Revolving Credit Agreements" and each individually, an "August 2019 Revolving Credit Agreement") with the lenders from time to time party thereto. Each of the August 2019 Revolving Credit Agreement" are senior unsecured revolving credit facilities, with facility termination dates of the earlier of (a) 18 months following August 30, 2019, subject to extension thereof pursuant to the applicable August 2019 Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the applicable August 2019 Revolving Credit Agreement. As of August 31, 2019, there were no borrowings outstanding under the August 2019 Revolving Credit Agreements.

January 2019 364-Day Revolving Credit Agreement

On January 18, 2019, the Company entered into a \$2.0 billion 364-day revolving credit agreement (the "January 2019 364-Day Revolving Credit Agreement") with the lenders from time to time party thereto. The January 2019 364-Day Revolving Credit Agreement is a senior unsecured 364-day revolving credit facility, with a facility termination date of the earlier of (a) 364 days following January 31, 2019, the date of the effectiveness of the commitments pursuant to the January 364-Day Revolving Credit Agreement, subject to extension thereof pursuant to the January 2019 364-Day Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the January 2019 364-Day Revolving Credit Agreement. As of August 31, 2019, there were no borrowings outstanding under the January 364-Day Revolving Credit Agreement.

December 2018 Revolving Credit Agreement

On December 21, 2018, the Company entered into a \$1.0 billion revolving credit agreement (the "December 2018 Revolving Credit Agreement") with the lenders from time to time party thereto. The December 2018 Revolving Credit Agreement is a senior unsecured revolving credit facility with a facility termination date of the earlier of (a) 18 months following January 28, 2019, the date of the effectiveness of the commitments pursuant to the December 2018 Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the December 2018 Revolving Credit Agreement. As of August 31, 2019, there were \$0.3 billion borrowings outstanding under the December 2018 Revolving Credit Agreement.

- 83 -

December 2018 Credit Agreement

On December 5, 2018, Walgreens Boots Alliance entered into a \$1.0 billion term loan credit agreement (as amended, the "December 2018 Credit Agreement") with the lenders from time to time party thereto and, on August 9, 2019, the Company entered into an amendment to such credit agreement to permit the Company to borrow, repay and reborrow amounts borrowed thereunder prior to the maturity date. The December 2018 Credit Agreement is a senior unsecured revolving facility with a facility termination date of the earlier of (a) January 29, 2021, subject to extension thereof pursuant to the December 2018 Credit Agreement and (b) the date of termination in whole of the aggregate amount of the commitments pursuant to the December 2018 Credit Agreement. As of August 31, 2019, there were \$0.8 billion of borrowings outstanding under the December 2018 Credit Agreement.

November 2018 Credit Agreement

On November 30, 2018, the Company entered into a credit agreement (as amended, the "November 2018 Credit Agreement") with the lenders from time to time party thereto and, on March 25, 2019, the Company entered into an amendment to such credit agreement reflecting certain changes to the borrowing notice provisions thereto. The November 2018 Credit Agreement includes a \$500 million senior unsecured revolving credit facility and a \$500 million senior unsecured term loan facility. The facility termination date is, with respect to the revolving credit facility, the earlier of (a) May 30, 2020 and (b) the date of termination in whole of the aggregate amount of the revolving commitments pursuant to the November 2018 Credit Agreement and, with respect to the term loan facility, the earlier of (a) May 30, 2020 and (b) the date of acceleration of all term loans pursuant to the November 2018 Credit Agreement. As of August 31, 2019, there were \$0.5 billion borrowings outstanding under the November 2018 Credit Agreement.

August 2018 Revolving Credit Agreement

On August 29, 2018, the Company entered into a revolving credit agreement (the "August 2018 Revolving Credit Agreement") with the lenders and letter of credit issuers from time to time party thereto. The August 2018 Revolving Credit Agreement is an unsecured revolving credit facility with an aggregate commitment in the amount of \$3.5 billion, with a letter of credit subfacility commitment amount of \$500 million. The facility termination date is the earlier of (a) August 29, 2023, subject to extension thereof pursuant to the August 2018 Revolving Credit Agreement and (b) the date of termination in whole of the aggregate amount of the revolving commitments pursuant to the August 2018 Revolving Credit Agreement. As of August 31, 2019, there were no borrowings outstanding under the August 2018 Revolving Credit Agreement.

August 2017 Credit Agreements

On August 24, 2017, the Company entered into a \$1.0 billion revolving credit agreement with the lenders from time to time party thereto (the "August 2017 Revolving Credit Agreement") and a \$1.0 billion term loan credit agreement with Sumitomo Mitsui Banking Corporation (the "2017 Term Loan Credit Agreement"). On November 30, 2018, in connection with the entrance into the November 2018 Credit Agreement, the Company terminated the 2017 Term Loan Credit Agreement in accordance with its terms and as of such date paid all amounts due in connection therewith. On January 31, 2019, the August 2017 Revolving Credit Agreement matured and the Company paid all amounts due in connection therewith.

February 2017 Revolving Credit Agreement

On February 1, 2017, the Company entered into a \$1.0 billion revolving credit facility (as amended, the "February 2017 Revolving Credit Agreement") with the lenders from time to time party thereto and, on August 1, 2017, the Company entered into an amendment agreement thereto. On January 31, 2019, the February 2017 Revolving Credit Agreement matured and the Company paid all amounts due in connection therewith.

Debt covenants

Each of the Company's credit facilities described above contain a covenant to maintain, as of the last day of each fiscal quarter, a ratio of consolidated debt to total capitalization not to exceed 0.60:1.00, subject to increase in certain circumstances set forth in the applicable credit agreement. The credit facilities contain various other customary covenants.

Commercial paper

The Company periodically borrows under its commercial paper program and may borrow under it in future periods. The Company had average daily commercial paper outstanding of \$2.7 billion at a weighted average interest rate of 3.07% for the fiscal year ended August 31, 2019. The Company had average daily commercial paper outstanding of \$1.4 billion at a weighted average interest rate of 2.11% for the fiscal year ended August 31, 2018.

Interest

Interest paid was \$676 million in fiscal 2019, \$577 million in fiscal 2018 and \$643 million in fiscal 2017.

Note 8. Financial instruments

- 84 -

The Company uses derivative instruments to manage its exposure to interest rate and foreign currency exchange risks.

The Company has non-U.S. dollar denominated net investments and uses foreign currency denominated financial instruments, specifically foreign currency derivatives and foreign currency denominated debt, to hedge its foreign currency risk.

The Company utilizes foreign currency forward contracts and other foreign currency derivatives to hedge significant committed and highly probable future transactions and cash flows denominated in currencies other than the functional currency of the Company or its subsidiaries.

The Company uses interest rate swaps from time to time to manage the interest rate exposure associated with some of its fixed-rate debt and designates them as fair value hedges. From time to time, the Company uses forward starting interest rate swaps to hedge its interest rate exposure of some of its anticipated debt issuances.

The notional amount and fair value of derivative instruments outstanding were as follows (in millions):

August 31, 2019	I	Notional	tional Fair value		Location in Consolidated Balance Sheets
Derivatives designated as hedges:					
Cross currency interest rate swap	\$	800	\$	73	Other non-current assets
Foreign currency forwards		18		1	Other current assets
Derivatives not designated as hedges:					
Foreign currency forwards		3,485		87	Other current assets
Foreign currency forwards		707		6	Other current liabilities
August 31, 2018	I	Notional	F	air value	Location in Consolidated Balance Sheets
August 31, 2018 Derivatives designated as hedges:	11	Notional	F	air value	Location in Consolidated Balance Sheets
5	1 \$	Notional 250	F \$	air value	Location in Consolidated Balance Sheets Other non-current liabilities
Derivatives designated as hedges:				air value	
Derivatives designated as hedges: Interest rate swaps	\$	250		r <mark>air value</mark> 1 —	Other non-current liabilities
Derivatives designated as hedges: Interest rate swaps Foreign currency forwards	\$	250		air value 1 - 52	Other non-current liabilities

Net investment hedges

The Company uses cross currency interest rate swaps as hedges of net investments in subsidiaries with non-U.S. dollar functional currencies. For qualifying net investment hedges, changes in the fair value of the derivatives are recorded in the currency translation adjustment within accumulated other comprehensive income (loss).

Derivatives not designated as hedges

The Company enters into derivative transactions that are not designated as accounting hedges. These derivative instruments are economic hedges of foreign currency risks. The income and (expense) due to changes in fair value of these derivative instruments were recognized in earnings as follows (in millions):

	Location in Consolidated Statements of Earnings	2019		2018	2017		
Foreign currency forwards	Selling, general and administrative expense	\$ 139	\$	17	\$	11	
Foreign currency forwards	Other income (expense)	(18)		22		(48)	

Derivatives credit risk

Counterparties to derivative financial instruments expose the Company to credit-related losses in the event of counterparty nonperformance, and the Company regularly monitors the credit worthiness of each counterparty.

Derivatives offsetting

The Company does not offset the fair value amounts of derivative instruments subject to master netting agreements in the Consolidated Balance Sheets.

Note 9. Fair value measurements

- 85 -

The Company measures certain assets and liabilities in accordance with ASC Topic 820, Fair Value Measurements and Disclosures, which defines fair value as the price that would be received for an asset or paid to transfer a liability in an orderly transaction between market participants on the measurement date. In addition, it establishes a fair value hierarchy that prioritizes observable and unobservable inputs used to measure fair value into three broad Levels:

Level 1 - Quoted prices in active markets that are accessible at the measurement date for identical assets and liabilities. The fair value hierarchy gives the highest priority to Level 1 inputs.

- Level 2 Observable inputs other than quoted prices in active markets.
- Level 3 Unobservable inputs for which there is little or no market data available. The fair value hierarchy gives the lowest priority to Level 3 inputs.

Assets and liabilities measured at fair value on a recurring basis were as follows (in millions):

	Augu	ıst 31, 2019		Level 1	Level 2	Level 3	
Assets:							
Money market funds ¹	\$	217	\$	217	\$ —	\$	—
Investments in equity securities ²		5		5	—		—
Foreign currency forwards ³		88		_	88		—
Cross Currency interest rate swaps ⁴		73			73		_
Liabilities:							
Foreign currency forwards ³		6			6		_
	Augu	ıst 31, 2018	_	Level 1	Level 2	 Level 3	
Assets:	Augu	ıst 31, 2018		Level 1	 Level 2	 Level 3	
Assets: Money market funds ¹	Augu \$		\$	Level 1	\$	\$ Level 3	
			\$		\$	\$ Level 3	
Money market funds ¹			\$		\$	\$ Level 3	
Money market funds ¹ Available-for-sale investments ²		227 1	\$		\$ 	\$ Level 3	
Money market funds ¹ Available-for-sale investments ² Foreign currency forwards ³		227 1	\$		\$ 	\$ Level 3	

¹ Money market funds are valued at the closing price reported by the fund sponsor.

² Fair values of quoted investments are based on current bid prices as of August 31, 2019 and 2018.

³ The fair value of forward currency contracts is estimated by discounting the difference between the contractual forward price and the current available forward price for the residual maturity of the contract using observable market rates.

⁴ The fair value of interest rate swaps and cross currency interest rate swaps is calculated by discounting the estimated future cash flows based on the applicable observable yield curves. See note 8, financial instruments, for additional information.

There were no transfers between Levels in fiscal 2019 or 2018.

The Company reports its debt instruments under the guidance of ASC Topic 825, Financial Instruments, which requires disclosure of the fair value of the Company's debt in the footnotes to the Consolidated Financial Statements. Unless otherwise noted, the fair value for all notes was determined based upon quoted market prices and therefore categorized as Level 1. See note 7, debt, for further information. The carrying values of accounts receivable and trade accounts payable approximated their respective fair values due to their short-term nature.

Note 10. Commitments and contingencies

The Company is involved in legal proceedings, including litigation, arbitration and other claims, and investigations, inspections, subpoenas, audits, claims, inquiries and similar actions by pharmacy, healthcare, tax and other governmental authorities, arising in the normal course of the Company's business, including the matters described below. Legal proceedings, in general, and securities, class action and multi-district litigation, in particular, can be expensive and disruptive. Some of these suits may purport or may be determined to be class actions and/or involve parties seeking large and/or indeterminate amounts, including punitive or exemplary damages, and may remain unresolved for several years. From time to time, the Company is

- 86 -

also involved in legal proceedings as a plaintiff involving antitrust, tax, contract, intellectual property and other matters. Gain contingencies, if any, are recognized when they are realized.

Like other companies in the retail pharmacy and pharmaceutical wholesale industries, the Company is subject to extensive regulation by national, state and local government agencies in the United States and other countries in which it operates. There continues to be a heightened level of review and/or audit by regulatory authorities of, and increased litigation regarding, the Company's and the rest of the health care and related industry's business, compliance and reporting practices. As a result, the Company regularly is the subject of government actions of the types described above. The Company also may be named from time to time in qui tam actions initiated by private third parties. In such actions, the private parties purport to act on behalf of federal or state governments, allege that false claims have been submitted for payment by the government and may receive an award if their claims are successful. After a private party has filed a qui tam action, the government must investigate the private party's claim and determine whether to intervene in and take control over the litigation. These actions may remain under seal while the government makes this determination. If the government declines to intervene, the private party may nonetheless continue to pursue the litigation on his or her own purporting to act on behalf of the government.

The results of legal proceedings, including government investigations, are often uncertain and difficult to predict, and the costs incurred in these matters can be substantial, regardless of the outcome. With respect to litigation and other legal proceedings where the Company has determined that a loss is reasonably possible, the Company is unable to estimate the amount or range of reasonably possible loss due to the inherent difficulty of predicting the outcome of and uncertainties regarding such litigation and legal proceedings. The Company believes that its defenses and assertions in pending legal proceedings have merit and does not believe that any of these pending matters, after consideration of applicable reserves and rights to indemnification, will have a material adverse effect on the Company's consolidated financial position. However, substantial unanticipated verdicts, fines and rulings do sometimes occur. As a result, the Company could from time to time incur judgments, enter into settlements or revise its expectations regarding the outcome of certain matters, and such developments could have a material adverse effect on its results of operations in the period in which the amounts are paid. In addition, as a result of governmental investigations or proceedings, the Company may be subject to damages, civil or criminal fines or penalties, or other sanctions, including the possible suspension or loss of licensure and/or suspension or exclusion from participation in government programs.

On December 29, 2014, a putative shareholder filed a derivative action in federal court in the Northern District of Illinois against certain current and former directors and officers of Walgreen Co., and Walgreen Co. as a nominal defendant, arising out of certain public statements the Company made regarding its former fiscal 2016 goals. The action asserts claims for breach of fiduciary duty, waste and unjust enrichment. On April 10, 2015, the defendants filed a motion to dismiss. On May 18, 2015, the case was stayed in light of a securities class action that was filed on April 10, 2015. After a ruling issued on September 30, 2016 in the securities class action, which is described below, on November 3, 2016, the Court entered a stipulation and order extending the stay until the resolution of the securities class action.

On April 10, 2015, a putative shareholder filed a securities class action in federal court in the Northern District of Illinois against Walgreen Co. and certain former officers of Walgreen Co. The action asserts claims for violation of the federal securities laws arising out of certain public statements the Company made regarding its former fiscal 2016 goals. On June 16, 2015, the Court entered an order appointing a lead plaintiff. Pursuant to the Court's order, lead plaintiff filed a consolidated class action complaint on August 17, 2015, and defendants moved to dismiss the complaint on October 16, 2015. On September 30, 2016, the Court issued an order granting in part and denying in part defendants' motion to dismiss. Defendants filed their answer to the court granted plaintiffs' motion on March 29, 2018 and merits discovery is proceeding. On December 19, 2018, plaintiffs filed a first amended complaint and defendants moved to dismiss the new complaint on February 19, 2019. On September 23, 2019, the Court issued an order granting in part defendants' motion to dismiss.

On December 11, 2017, purported Rite Aid shareholders filed an amended complaint in a putative class action lawsuit in the United States District Court for the Middle District of Pennsylvania (the "M.D. Pa. action") arising out of transactions contemplated by the merger agreement between the Company and Rite Aid. The amended complaint alleged that the Company and certain of its officers made false or misleading statements regarding the transactions. The Court denied the Company's motion to dismiss the amended complaint on April 15, 2019. The Company filed an answer and affirmative defenses, discovery commenced, and plaintiffs have filed a motion for class certification. The Company's response to that motion is due on November 21, 2019.

- 87 -

In June 2019, a Fred's, Inc. shareholder filed a nearly identical lawsuit to the M.D. Pa. action in the United States District Court for the Western District of Tennessee, except naming Fred's, Inc. and one of its former officers along with the Company and certain of its officers. Lead plaintiffs will file an amended complaint on or before November 4, 2019.

As previously disclosed, the Company was also named as a defendant in a putative class action lawsuit similar to the M.D. Pa. action filed in State of Pennsylvania in the Court of Common Pleas of Cumberland County, which was terminated by the court for lack of prosecution in November 2018.

In December 2017, the United States Judicial Panel on Multidistrict Litigation consolidated numerous cases filed against an array of defendants by various plaintiffs such as counties, cities, hospitals, Indian tribes and others, alleging claims generally concerning the impacts of widespread opioid abuse. The consolidated multidistrict litigation ("MDL"), captioned In re National Prescription Opiate Litigation (MDL No. 2804), is pending in the U.S. District Court for the Northern District of Ohio. The Company is named as a defendant in a subset of the cases included in this MDL. The first trial in the MDL that had been scheduled for October 2019 was terminated. The MDL court will address next steps in a scheduling conference that has not yet been set. The Company also has been named as a defendant in numerous lawsuits brought in state courts relating to opioid matters. The relief sought by various plaintiffs is compensatory and punitive damages, as well as injunctive relief. Additionally, the Company has received from the Attorney Generals of several states subpoenas, civil investigative demands, and/or other requests concerning opioid matters.

On September 28, 2018, the Company announced that it had reached an agreement with the SEC to fully resolve an investigation into certain forward-looking financial goals and related disclosures by Walgreens. The disclosures at issue were made prior to the strategic combination with Alliance Boots and the merger pursuant to which Walgreens Boots Alliance became the parent holding company on December 31, 2014. The settlement does not involve any of the Company's current officers or executives, nor does it allege intentional or reckless conduct by the Company. In agreeing to the settlement, the Company neither admitted nor denied the SEC's allegations. Pursuant to the agreement with the SEC, the Company consented to the SEC's issuance of an administrative order, and the Company paid a \$34.5 million penalty, which was fully reserved for in the Company's Consolidated Financial Statements as of August 31, 2018.

On January 22, 2019, the Company announced that it had reached an agreement to resolve a civil investigation involving allegations under the False Claims Act by a United States Attorney's Office, working in conjunction with several states, regarding certain dispensing practices. Pursuant to the agreement, the Company paid \$209.2 million to the United States and the various states involved in the matter, substantially all of which was reserved for in the Company's Consolidated Financial Statements as of November 30, 2018.

Note 11. Income taxes

U.S. tax law changes

In connection with the U.S. tax law changes enacted in December 2017 and in accordance with SEC Staff Accounting Bulletin 118 (SAB 118), the Company completed its analysis of the income tax effects of the U.S. tax law changes during the three months ended February 28, 2019. The incremental net tax benefit recorded upon completion of the analysis of the income tax effects of the U.S. tax law changes was not material to our Consolidated Financial Statements.

In fiscal 2018, as a direct result of the U.S. tax law changes, the Company performed preliminary analysis and recorded a provisional estimated net tax benefit of \$125 million. This provisional net tax benefit arose from a benefit of \$648 million from re-measuring the Company's net U.S. deferred tax liabilities, partially offset by the Company's accrual for the transition tax and other U.S. tax law changes of \$523 million. In fiscal 2019, the Company finalized its provisional estimates and recognized incremental net income tax expense, which was not material to our Consolidated Financial Statements, upon completing the accounting related to the U.S. tax law changes, primarily related to the refinement of the transition tax expense on the one-time mandatory repatriation of previously untaxed earnings of foreign subsidiaries.

The U.S. tax law changes include broad and complex changes. Among other things, the U.S. tax law changes reduced the federal corporate tax rate from 35% to 21% effective January 1, 2018 and required companies to immediately accrue for a one-time transition tax on certain un-repatriated earnings of foreign subsidiaries, which is generally payable over an eight-year period. The U.S. tax law changes modified the taxation of foreign earnings, repealed the deduction for domestic production activities, limited interest deductibility and established a global intangible low tax income (GILTI) regime.

The lower corporate income tax rate of 21% became effective January 1, 2018 and was phased in as a result of our August 31 fiscal year-end. This resulted in a blended U.S. statutory federal tax rate of approximately 26% for fiscal 2018. This statutory

- 88 -

federal tax rate is 21% for fiscal 2019 and subsequent fiscal years, which provided a benefit to the Company's fiscal 2019 tax provision of approximately \$89 million.

During 2019, the U.S. Treasury Department issued regulations to apply retroactively covering certain components of the 2017 U.S. tax law changes. Certain guidance included in these regulations is inconsistent with the Company's interpretation that led to the recognition of \$247 million of tax benefits in prior periods. Despite this new guidance, the Company remains confident in its interpretation of the U.S. tax law changes and intends to defend this position through litigation, if necessary. However, if the Company is ultimately unsuccessful in defending its position, it may be required to reverse all or a portion of the benefits previously recorded.

By establishing the GILTI regime, the Act created a minimum tax on certain foreign sourced earnings. The taxability of the foreign earnings and the applicable tax rates are dependent on future events. The Company's accounting policy for the minimum tax on foreign sourced earnings is to report the tax effects on the basis that the minimum tax will be recognized in tax expense in the year it is incurred as a period expense.

As the Company repatriates the undistributed earnings of its foreign subsidiaries for use in the United States, the earnings from its foreign subsidiaries will generally not be subject to U.S. federal tax. The Company continuously evaluates the amount of foreign earnings that are not necessary to be permanently reinvested in its foreign subsidiaries.

The components of earnings before income tax provision were (in millions):

	2019		2018	2017
U.S.	\$ 1,89	8 5	\$ 3,292	\$ 1,953
Non–U.S.	2,62	29	2,683	2,900
Total	\$ 4,52	27 5	\$ 5,975	\$ 4,853

The provision for income taxes consists of the following (in millions):

	2019	2018	2017
Current provision			
Federal	\$ 201	\$ 866	\$ 759
State	46	103	45
Non–U.S.	241	353	390
	488	 1,322	 1,194
Deferred provision			
Federal – tax law change	_	(648)	_
Federal – excluding tax law change	151	304	(306)
State	4	78	(24)
Non–U.S. – tax law change		_	(80)
Non–U.S. – excluding tax law change	(55)	(58)	(24)
	100	(324)	(434)
Income tax provision	\$ 588	\$ 998	\$ 760

- 89 -

The difference between the statutory federal income tax rate and the effective tax rate is as follows:

	2019	2018	2017
Federal statutory rate	21.0 %	25.7 %	35.0 %
State income taxes, net of federal benefit	0.9	2.3	0.3
Foreign income taxed at non-U.S. rates	(2.1)	(12.2)	(11.8)
Non-taxable income	(3.5)	(5.2)	(5.3)
Non-deductible expenses	0.5	2.1	1.5
Transition tax	—	12.4	_
Tax law changes	(0.4)	(10.9)	(1.6)
Change in valuation allowance ¹	1.9	8.7	0.7
Tax credits	(4.8)	(6.9)	(2.9)
Other	(0.5)	0.7	(0.2)
Effective income tax rate	13.0 %	16.7 %	15.7 %

¹ Net of changes in related tax attributes.

The deferred tax assets and liabilities included in the Consolidated Balance Sheets consist of the following (in millions):

	2019	2018
Deferred tax assets:		
Compensation and benefits	133	152
Insurance	90	74
Accrued rent	219	271
Allowance for doubtful accounts	13	27
Tax attributes	6,687	2,351
Stock compensation	45	44
Deferred income	115	110
Other	78	44
	 7,380	3,073
Less: valuation allowance	6,638	2,226
Total deferred tax assets	 742	847
Deferred tax liabilities:		
Accelerated depreciation	475	603
Inventory	394	301
Intangible assets	1,116	1,234
Equity method investment	481	459
	2,466	2,597
Net deferred tax liabilities	\$ 1,724	\$ 1,750

As of August 31, 2019, the Company has recorded deferred tax assets for tax attributes of \$6.7 billion, primarily reflecting the benefit of \$3.1 billion in U.S. federal, \$159 million in state and \$23.7 billion in non-U.S. ordinary and capital losses. In addition, these deferred tax assets include \$61 million of income tax credits. Of these deferred tax assets, \$6.4 billion will expire at various dates from 2020 through 2036. The residual deferred tax assets of \$248 million have no expiry date.

The Company believes it is more likely than not that the benefit from certain deferred tax assets will not be realized. In recognition of this risk, the Company has recorded a valuation allowance of \$6.6 billion against those deferred tax assets as of August 31, 2019.

Income taxes paid, net of refunds were \$0.9 billion, \$0.6 billion and \$1.1 billion for fiscal years 2019, 2018 and 2017, respectively.

- 90 -

ASC Topic 740, Income Taxes, provides guidance regarding the recognition, measurement, presentation and disclosure in the financial statement of tax positions taken or expected to be taken on a tax return, including the decision whether to file in a particular jurisdiction. As of August 31, 2019, unrecognized tax benefits of \$499 million were reported as long-term liabilities on the Consolidated Balance Sheets while an immaterial amount is reported as current tax liabilities. Both of these amounts include interest and penalties, when applicable.

The following table provides a reconciliation of the total amounts of unrecognized tax benefits (in millions):

	2019	:	2018	2017
Balance at beginning of year	\$ 456	\$	409	\$ 269
Gross increases related to tax positions in a prior period	33		123	151
Gross decreases related to tax positions in a prior period	(53)		(15)	(36)
Gross increases related to tax positions in the current period	26		29	33
Settlements with taxing authorities	(2)		(87)	(2)
Currency	—			(1)
Lapse of statute of limitations	(5)		(3)	(5)
Balance at end of year	\$ 455	\$	456	\$ 409

At August 31, 2019, 2018 and 2017, \$311 million, \$331 million and \$286 million, respectively, of unrecognized tax benefits would favorably impact the effective tax rate if recognized. During the next twelve months, based on current knowledge, it is reasonably possible the amount of unrecognized tax benefits could decrease by up to \$16 million due to anticipated tax audit settlements and the expirations of statutes of limitations associated with tax positions related to multiple tax jurisdictions.

The Company recognizes interest and penalties in the income tax provision in its Consolidated Statements of Earnings. At August 31, 2019 and August 31, 2018, the Company had accrued interest and penalties of \$47 million and \$87 million, respectively. For the year ended August 31, 2019, the amount reported in income tax expense related to interest and penalties was \$40 million income tax benefit.

The Company files a consolidated U.S. federal income tax return as well as income tax returns in various states and multiple foreign jurisdictions. It is generally no longer under audit examinations for U.S. federal income tax purposes for any years prior to fiscal 2014. With few exceptions, it is no longer subject to state and local income tax examinations by tax authorities for years before fiscal 2007. In foreign tax jurisdictions, the Company is generally no longer subject to examination by the tax authorities in the United Kingdom prior to 2015, Luxembourg prior to 2013, in Germany prior to 2014, in France prior to 2008 and in Turkey prior to 2014.

The Company has received tax holidays from Swiss cantonal income taxes relative to certain of its Swiss operations. The income tax holidays are expected to extend through September 2022. The holidays had a beneficial impact of \$127 million and \$127 million (inclusive of capital GILTI tax cost) during fiscal 2019 and 2018, respectively. This benefit is primarily included as part of the foreign income taxed at non-U.S. rates line in the effective tax rate reconciliation table above.

At August 31, 2019, it is not practicable for the Company to determine the amount of the unrecognized deferred tax liability it has with respect to temporary differences related to investments in foreign subsidiaries and foreign corporate joint ventures that are essentially permanent in duration.

Note 12. Stock compensation plans

The Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Omnibus Plan"), which became effective in fiscal 2013, provides for incentive compensation to the Company's non-employee directors, officers and employees and consolidates several previously existing equity compensation plans into a single plan.

The Company grants stock options, performance shares and restricted units under the Omnibus Plan. Performance shares issued under the Omnibus Plan offer performance-based incentive awards and equity-based awards to key employees. The fair value of each performance share granted assumes that performance goals will be achieved at 100 percent. If such goals are not met, no compensation expense is recognized and any recognized compensation expense is reversed. Restricted stock units are also equity-based awards with performance requirements that are granted to key employees. The performance shares and restricted stock unit awards are both subject to restrictions as to continuous employment except in the case of death, normal retirement or total and permanent disability.

- 91 -

Total stock-based compensation expense for fiscal 2019, 2018 and 2017 was \$119 million, \$130 million and \$91 million, respectively. The recognized tax benefit was \$43 million, \$43 million and \$78 million for fiscal 2019, 2018 and 2017, respectively. Unrecognized compensation cost related to non-vested awards at August 31, 2019 was \$172 million, which will be recognized over three years.

Note 13. Retirement benefits

The Company sponsors several retirement plans, including defined benefit plans, defined contribution plans and a postretirement health plan.

Defined benefit pension plans (non-U.S. plans)

The Company has various defined benefit pension plans outside the United States. The principal defined benefit pension plan is the Boots Pension Plan (the "Boots Plan"), which covers certain employees in the United Kingdom. The Boots Plan is a funded final salary defined benefit plan providing pensions and death benefits to members. The Boots Plan was closed to future accrual effective July 1, 2010, with pensions calculated based on salaries up until that date. The Boots Plan is governed by a trustee board, which is independent of the Company. The plan is subject to a full funding actuarial valuation on a triennial basis.

The investment strategy of the principal defined benefit pension plan is to hold approximately 85% of its assets in a diverse portfolio which aims to broadly match the characteristics of the plan's liabilities by investing in bonds, derivatives and other fixed income assets, with the remainder invested in predominantly return-seeking assets. Interest rate and inflation rate swaps are also employed to complement the role of fixed and index-linked bond holdings in liability risk management.

The following tables present classes of defined benefit pension plan assets by fair value hierarchy (in millions):

	August 31, 2	019	Level 1	Level 2	Level 3
Equity securities:					
Equity securities ¹	\$	1,006	\$ —	\$ 1,005	\$ —
Debt securities:					
Fixed interest government bonds ²		489	148	341	_
Index linked government bonds ²		3,861	3,824	37	—
Corporate bonds ³		2,390	1	2,389	_
Real estate:					
Real estate ⁴		471		—	471
Other:					
Other investments ⁵		914	63	516	335
Total	\$	9,131	\$ 4,036	\$ 4,288	\$ 806

- 92 -

	Au	ıgust 31, 2018	Level 1		Level 2		Level 2		Level 2	
Equity securities:										
Equity securities ¹	\$	1,030	\$	—	\$	1,030	\$	—		
Debt securities:										
Fixed interest government bonds ²		901		_		901		_		
Index linked government bonds ²		2,880		—		2,880		—		
Corporate bonds ³		2,542		—		2,536		6		
Real estate:										
Real estate ⁴		501		—		—		501		
Other:										
Other investments ⁵		822		64		531		227		
Total	\$	8,676	\$	64	\$	7,878	\$	734		

¹ Equity securities, which mainly comprise investments in commingled funds, are valued based on quoted prices and are primarily exchange-traded. Securities for which official close or last trade pricing on an active exchange is available are classified as Level 1 investments. If closing prices are not available, or the investments are in a commingled fund,

- securities are valued at the last quoted bid price and typically are categorized as Level 2 investments.
 ² Debt securities: government bonds comprise fixed interest and index linked bonds issued by central governments and are valued based on quotes received from independent pricing services or from dealers who make markets in such securities. Pricing services utilize pricing which considers readily available inputs such as the yield or price of bonds of comparable quality, coupon, maturity and type, as well as dealer-supplied prices. Certain government bonds which were categorized as Level 2 investments in fiscal 2018 were deemed to be classified as Level 1 in fiscal 2019.
- ³ Debt securities: corporate bonds comprise bonds issued by corporations in both segregated and commingled funds and are valued using recently executed transactions, or quoted market prices for similar assets and liabilities in active markets, or for identical assets and liabilities in markets that are not active. If there have been no market transactions in a particular fixed income security, its fair value is calculated by pricing models that benchmark the security against other securities with actual market prices. Certain corporate bonds which were categorized as Level 2 investments in fiscal 2018 were deemed to be classified as Level 1 in fiscal 2019.
- ⁴ Real estate comprises investments in certain property funds which are valued based on the underlying properties. These properties are valued using a number of standard industry techniques such as cost, discounted cash flows, independent appraisals and market based comparable data. Real estate investments are categorized as Level 3 investments. Changes in Level 3 investments during fiscal 2019 were driven by actual return on plan assets still held at August 31, 2019 and purchases during the year.
- ⁵ Other investments mainly comprise cash and cash equivalents, derivatives and direct private placements. Cash is categorized as a Level 1 investment and cash in commingled funds is categorized as Level 2 investments. Cash equivalents are valued using observable yield curves, discounting and interest rates and are categorized as Level 2 investments. Derivatives which are exchange-traded and for which market quotations are readily available are valued at the last reported sale price or official closing price as reported by an independent pricing service on the primary market, or exchange on which they are traded, and are categorized as Level 1 investments. Over-the-counter derivatives typically are valued by independent pricing services and are categorized as Level 3 investments. Direct private placements are typically bonds valued by reference to comparable bonds and are categorized as Level 3 investments. Changes in Level 3 investments during fiscal 2019 were primarily driven by purchases during the year.

Components of net periodic pension costs for the defined benefit pension plans (in millions):

	Boots and other pension plans							
		2019		2018		2017		
Service costs	\$	4	\$	5	\$	5		
Interest costs		196		193		174		
Expected returns on plan assets/other		(246)		(209)		(146)		
Total net periodic pension (income) cost	\$	(46)	\$	(11)	\$	33		

- 93 -

Change in benefit obligations for the defined benefit pension plans (in millions):

	2019	2018
Benefit obligation at beginning of year	\$ 8,293	\$ 8,880
Service costs	4	5
Interest costs	196	193
Amendments/other	22	(4)
Net actuarial loss (gain)	1,212	(466)
Benefits paid	(363)	(398)
Currency translation adjustments	 (530)	 83
Benefit obligation at end of year	\$ 8,834	\$ 8,293

Change in plan assets for the defined benefit pension plans (in millions):

	2019	2018
Plan assets at fair value at beginning of year	\$ 8,676	\$ 8,980
Employer contributions	38	65
Benefits paid	(363)	(398)
Return on assets/other	1,333	(55)
Currency translation adjustments	(552)	84
Plan assets at fair value at end of year	\$ 9,131	\$ 8,676

Amounts recognized in the Consolidated Balance Sheets (in millions):

	2019		20)18
Other non-current assets	\$	486	\$	554
Accrued expenses and other liabilities		(6)		(7)
Other non-current liabilities		(183)		(164)
Net asset (liability) recognized at end of year	\$	297	\$	383

Cumulative pre-tax amounts recognized in accumulated other comprehensive (income) loss (in millions):

	2019	2018
Net actuarial (gain) loss	\$ 92	\$ (27)
Prior service cost	24	
Total	116	(27)
	 -	

The projected benefit obligation, accumulated benefit obligation and fair value of plan assets for all pension plans, including accumulated benefit obligations in excess of plan assets, at August 31 were as follows (in millions):

	2019		2018
Projected benefit obligation	\$	8,834 \$	8,293
Accumulated benefit obligation		8,823	8,285
Fair value of plan assets		9,131	8,676

Pension plans with projected benefit obligation and accumulated benefit obligations in excess of plan assets were not material for periods presented.

Estimated future benefit payments for the next 10 years from defined benefit pension plans to participants are as follows (in millions):

- 94 -

Estimated future benefit

	P	ayments
2020	\$	261
2021		253
2022		265
2023		276
2024		288
2025-2029		1,629

The assumptions used in accounting for the defined benefit pension plans were as follows:

	2019	2018
Weighted-average assumptions used to determine benefit obligations		
Discount rate	1.80%	2.67%
Rate of compensation increase	2.91%	2.68%
Weighted-average assumptions used to determine net periodic benefit cost		
Discount rate	1.58%	2.12%
Expected long-term return on plan assets	3.10%	2.27%
Rate of compensation increase	2.68%	2.64%
Nate of compensation increase	2.0878	2.0470

Based on current actuarial estimates, the Company plans to make contributions of \$27 million to its defined benefit pension plans in fiscal 2020 and expects to make contributions beyond 2020, which will vary based upon many factors, including the performance of the defined benefit pension plan assets.

Defined contribution plans

The principal retirement plan for U.S. employees is the Walgreen Profit-Sharing Retirement Trust, to which both the Company and participating employees contribute. The Company's contribution is in the form of a guaranteed match which is made pursuant to the applicable plan document approved by the Walgreen Co. Board of Directors. Plan activity is reviewed periodically by certain Committees of the Walgreens Boots Alliance Board of Directors. The profit-sharing provision was an expense of \$239 million, \$217 million and \$221 million in fiscal 2019, 2018 and 2017, respectively. The Company's contributions were \$234 million, \$366 million and \$220 million in fiscal 2019, 2018 and 2017, respectively.

The Company also has certain contract based defined contribution arrangements. The principal one is the Alliance Healthcare & Boots Retirement Savings Plan, which is United Kingdom based and to which both the Company and participating employees contribute. The cost recognized in the Consolidated Statement of Earnings was \$124 million in fiscal 2019, \$142 million in fiscal 2018 and \$112 million in fiscal 2017.

Postretirement healthcare plan

The Company provides certain health insurance benefits to retired U.S. employees who meet eligibility requirements, including age, years of service and date of hire. The costs of these benefits are accrued over the service life of the employee. An amendment to this plan during the fourth quarter of fiscal 2018 resulted in a reduction in the benefit plan obligation of \$201 million and the recognition of a curtailment gain of \$112 million. An amendment during the third quarter of fiscal 2017 resulted in the recognition of a curtailment gain of \$109 million. The Company's postretirement health benefit plan obligation was \$161 million and \$146 million in fiscal 2019 and 2018, respectively and is not funded. The expected benefit to be paid net of the estimated federal subsidy during fiscal year 2020 is \$8 million.

Note 14. Capital stock

In connection with the Company's capital policy, the Board of Directors has authorized share repurchase programs. In April 2017, Walgreens Boots Alliance authorized a stock repurchase program (the "April 2017 stock repurchase program"), which authorized the repurchase of up to \$1.0 billion of Walgreens Boots Alliance common stock prior to the program's expiration on December 31, 2017. In May 2017, the Company completed the April 2017 stock repurchase program, purchasing 11.8 million shares. In June 2017, Walgreens Boots Alliance authorized a new stock repurchase program, which authorized the repurchase of up to \$5.0 billion of Walgreens Boots Alliance common stock prior to the program. Stock prior to the program stock prior to the program.



authorization was increased by an additional \$1.0 billion in October 2017 (as expanded, the "June 2017 stock repurchase program"). The June 2017 stock repurchase program was completed in October 2017. In June 2018, Walgreens Boots Alliance authorized a new stock repurchase program, which authorized the repurchase of up to \$10.0 billion of Walgreens Boots Alliance common stock, which program has no specified expiration date. The Company purchased 57 million and 72 million shares under stock repurchase programs in fiscal 2019 and 2018 at a cost of \$3.8 billion and \$4.9 billion, respectively.

The Company determines the timing and amount of repurchases based on its assessment of various factors including prevailing market conditions, alternate uses of capital, liquidity, the economic environment and other factors. The timing and amount of these purchases may change at any time and from time to time. The Company has repurchased, and may from time to time in the future repurchase, shares on the open market through Rule 10b5-1 plans, which enable a company to repurchase shares at times when it otherwise might be precluded from doing so under insider trading laws.

In addition, the Company continued to repurchase shares to support the needs of the employee stock plans. Shares totaling \$339 million were purchased to support the needs of the employee stock plans during fiscal 2019 as compared to \$289 million and \$457 million in fiscal 2018 and fiscal 2017, respectively. At August 31, 2019, 25 million shares of common stock were reserved for future issuances under the Company's various employee benefit plans.

Note 15. Accumulated other comprehensive income (loss)

The following is a summary of net changes in accumulated other comprehensive income by component and net of tax for fiscal 2019, 2018 and 2017 (in millions):

	r	Pension/post- retirement obligations		Unrecognized gain (loss) on available- for-sale investments		Unrealized gain (loss) on cash flow hedges		Share of OCI of equity method investments		Cumulative translation adjustments	Total
Balance at August 31, 2016	\$	(212)	\$	2	\$	(37)	\$	(1)	\$	(2,744)	\$ (2,992)
Other comprehensive income (loss) before reclassification adjustments		(34)		(2)		_		(1)		(133)	(170)
Amounts reclassified from AOCI ¹		109		_		5		_		_	114
Tax benefit (provision)		(2)		—		(1)		—		—	(3)
Net change in other comprehensive income (loss)		73		(2)		4		(1)		(133)	(59)
Balance at August 31, 2017	\$	(139)	\$	_	\$	(33)	\$	(2)	\$	(2,877)	\$ (3,051)
Other comprehensive income (loss) before reclassification adjustments		417		_		_		(4)		(207)	206
Amounts reclassified from AOCI ¹		(120)		—		4		11		8	(97)
Tax benefit (provision)		(57)		—		(1)		(2)		—	(60)
Net change in other comprehensive income (loss)		240		_		3		5		(199)	49
Balance at August 31, 2018	\$	101	\$	—	\$	(30)	\$	3	\$	(3,076)	\$ (3,002)
Other comprehensive income (loss) before reclassification adjustments		(162)		_		74		(1)		(801)	(889)
Amounts reclassified from AOCI		(17)		_		5		_		_	(12)
Tax benefit (provision)		30		—		(19)				(6)	5
Net change in other comprehensive income (loss)		(149)		_		60		(1)		(807)	(896)
Balance at August 31, 2019	\$	(48)	\$	_	\$	31	\$	3	\$	(3,884)	\$ (3,897)

¹ Includes amendment to U.S. postretirement healthcare plan resulting in a curtailment gain. See note 13, retirement benefits.

Note 16. Segment reporting

The Company has aligned its operations into three reportable segments: Retail Pharmacy USA, Retail Pharmacy International and Pharmaceutical Wholesale. The operating segments have been identified based on the financial data utilized by the Company's Chief Executive Officer (the chief operating decision maker) to assess segment performance and allocate resources among the Company's operating segments. The chief operating decision maker uses adjusted operating income to assess segment profitability. The chief operating decision maker does not use total assets by segment to make decisions regarding resources; therefore, the total asset disclosure by segment has not been included.

Retail Pharmacy USA

The Retail Pharmacy USA segment consists of the Walgreens business, which includes the operation of retail drugstores, convenient care clinics and mail and central specialty pharmacy services. Sales for the segment are principally derived from the sale of prescription drugs and a wide assortment of retail products, including health and wellness, beauty and personal care and consumables and general merchandise.

Retail Pharmacy International

The Retail Pharmacy International segment consists of pharmacy-led health and beauty retail businesses and optical practices. These businesses include Boots branded stores in the United Kingdom, Thailand, Norway, the Republic of Ireland and the Netherlands, Benavides in Mexico and Ahumada in Chile. Sales for the segment are principally derived from the sale of prescription drugs and health and wellness, beauty and personal care and other consumer products.

Pharmaceutical Wholesale

The Pharmaceutical Wholesale segment consists of the Alliance Healthcare pharmaceutical wholesaling and distribution businesses and an equity method investment in AmerisourceBergen. Wholesale operations are located in the United Kingdom, Germany, France, Turkey, Spain, the Netherlands, Egypt, Norway, Romania, Czech Republic and Lithuania. Sales for the segment are principally derived from wholesaling and distribution of a comprehensive offering of brand-name pharmaceuticals (including specialty pharmaceutical products) and generic pharmaceuticals, health and beauty products, home healthcare supplies and equipment and related services to pharmacies and other healthcare providers.

The results of operations for each reportable segment includes procurement benefits and an allocation of corporate-related overhead costs. The "Eliminations" column contains items not allocable to the reportable segments, as the information is not utilized by the chief operating decision maker to assess segment performance and allocate resources.

- 97 -

The following table reflects results of operations of the Company's reportable segments (in millions):

Walgreens Boots Alliance, Inc.\$6,94Depreciation and amortization:Retail Pharmacy USA\$1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$2,03		2018	
Retail Pharmacy USA\$104,53Retail Pharmacy International11,46Pharmaceutical Wholesale23,05Eliminations1(2,18)Walgreens Boots Alliance, Inc.\$136,86Adjusted Operating income:2**Retail Pharmacy USA\$5,25Retail Pharmacy International*74Pharmaceutical Wholesale93*Eliminations1*93Eliminations1**Walgreens Boots Alliance, Inc.\$6,94Depreciation and amortization:*42Retail Pharmacy USA\$1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$\$2,03			2017
Retail Pharmacy International11,46Pharmaceutical Wholesale23,05Eliminations1(2,18Walgreens Boots Alliance, Inc.\$ 136,86Adjusted Operating income:2Retail Pharmacy USA\$ 5,25Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations193Walgreens Boots Alliance, Inc.\$ 6,94Depreciation and amortization:\$ 1,45Retail Pharmacy USA\$ 1,45Retail Pharmacy USA\$ 1,45Retail Pharmacy USA\$ 1,45Walgreens Boots Alliance, Inc.\$ 2,03Walgreens Boots Alliance, Inc.\$ 2,03			
Pharmaceutical Wholesale23,05Eliminations1(2,18)Walgreens Boots Alliance, Inc.\$ 136,86Adjusted Operating income:2\$ 136,86Adjusted Operating income:2\$ 5,25Retail Pharmacy USA\$ 5,25Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations193Walgreens Boots Alliance, Inc.\$ 6,94Depreciation and amortization:1Retail Pharmacy USA\$ 1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03		98,392	\$ 87,302
Eliminational AntionalCapitalEliminations1(2,18)Walgreens Boots Alliance, Inc.\$ 136,86Adjusted Operating income:2*Retail Pharmacy USA\$ 5,25Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations1*Walgreens Boots Alliance, Inc.\$ 6,94Depreciation and amortization:*Retail Pharmacy USA\$ 1,45Retail Pharmacy USA\$ 1,45Retail Pharmacy USA\$ 1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03	-	12,281	11,813
Walgreens Boots Alliance, Inc.\$136,86Adjusted Operating income:2*136,86Adjusted Operating income:2**Retail Pharmacy USA\$5,25Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations193Walgreens Boots Alliance, Inc.\$6,94Depreciation and amortization:*42Retail Pharmacy USA\$1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$\$2,03	3	23,006	21,188
Adjusted Operating income:2Retail Pharmacy USA\$ 5,25Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations193Walgreens Boots Alliance, Inc.\$ 6,94Depreciation and amortization:93Retail Pharmacy USA\$ 1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03))	(2,142)	(2,089)
Retail Pharmacy USA\$5,25Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations193Walgreens Boots Alliance, Inc.\$Retail Pharmacy USA\$Retail Pharmacy USA\$Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$\$2,03	5\$	131,537	\$ 118,214
Retail Pharmacy International74Pharmaceutical Wholesale93Eliminations193Walgreens Boots Alliance, Inc.\$ 6,94Depreciation and amortization:*Retail Pharmacy USA\$ 1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03			
Pharmaceutical Wholesale93Eliminations1\$Walgreens Boots Alliance, Inc.\$Depreciation and amortization:*Retail Pharmacy USA\$Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$2,03	5\$	5,814	\$ 5,606
Eliminations1Walgreens Boots Alliance, Inc.\$Depreciation and amortization:Retail Pharmacy USA\$Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$2,03	7	929	939
Walgreens Boots Alliance, Inc.\$6,94Depreciation and amortization:Retail Pharmacy USA\$1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$2,03)	936	922
Depreciation and amortization:Retail Pharmacy USA\$ 1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03	1		
Retail Pharmacy USA\$ 1,45Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03	2 \$	7,679	\$ 7,467
Retail Pharmacy International42Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$ 2,03			
Pharmaceutical Wholesale15Walgreens Boots Alliance, Inc.\$\$2,03	9 \$	1,196	\$ 1,090
Walgreens Boots Alliance, Inc. \$ 2,03)	419	414
)	155	150
	3 \$	1,770	\$ 1,654
Capital expenditures:			
Retail Pharmacy USA\$ 1,32	3 \$	1,022	\$ 860
Retail Pharmacy International 27	5	241	384
Pharmaceutical Wholesale 10	1	104	107
Walgreens Boots Alliance, Inc.\$ 1,70	2 \$	1,367	\$ 1,351

The following table reconciles adjusted operating income to operating income (in millions):

	For the years ending August 31,						
		2019	2018	2017			
Adjusted operating income ²	\$	6,942	\$ 7,679	\$	7,467		
Acquisition-related amortization and impairment ³		(567)	(448)		(332)		
Transformational cost management		(477)	—		—		
Acquisition-related costs		(303)	(231)		(474)		
Adjustments to equity earnings in AmerisourceBergen		(233)	(175)		(187)		
Store optimization		(196)	(100)		_		
LIFO provision		(136)	(84)		(166)		
Certain legal and regulatory accruals and settlements ⁴		(31)	(284)		_		
Asset recovery			15		11		
Hurricane-related costs		—	(83)		_		
Cost Transformation		—	—		(835)		
Operating income ²	\$	4,998	\$ 6,289	\$	5,484		

¹ Eliminations relate to intersegment sales between the Pharmaceutical Wholesale and the Retail Pharmacy International segments.

- 98 -

- ² The Company adopted new accounting guidance in Accounting Standards Update 2017-07 as of September 1, 2018 (fiscal 2019) on a retrospective basis for the Consolidated Statements of Earnings presentation. See note 1, summary of major accounting policies, for further information.
- ³ Includes impairment of \$73 million for indefinite-lived pharmacy licenses intangible asset recorded during the three months ended August 31, 2019, in the Boots reporting unit within the Retail Pharmacy International segment.
- ⁴ Beginning in the quarter ended August 31, 2018, management reviewed and refined its practice to include all charges related to the matters included in certain legal and regulatory accruals and settlements. This non-GAAP measure is presented on a consistent basis for fiscal year 2019.

No single customer accounted for more than 10% of the Company's consolidated sales for any of the periods presented. Substantially all of our retail pharmacy sales are to customers covered by third-party payers (e.g., pharmacy benefit managers, insurance companies and governmental agencies) that agree to pay for all or a portion of a customer's eligible prescription purchases. In the Retail Pharmacy USA segment, two third-party payers, in the aggregate accounted for approximately 22% of the Company's consolidated sales in fiscal 2019 and three third-party payers, in the aggregate accounted for approximately 32% of the Company's consolidated sales in fiscal 2018. No third-party payer accounted for more than 10% of the Company's consolidated sales in fiscal 2017.

Geographic data for sales is as follows (in millions):

	2019			2018	2017	
United States	\$	104,532	\$	98,392	\$	87,302
United Kingdom		12,729		13,297		12,552
Europe (excluding the United Kingdom)		17,009		17,594		16,224
Other		2,597		2,254		2,136
Sales	\$	136,866	\$	131,537	\$	118,214

Geographic data for long-lived assets, defined as property, plant and equipment, is as follows (in millions):

	2019	2018
United States	\$ 10,598	\$ 10,678
United Kingdom	2,162	2,458
Europe (excluding the United Kingdom)	521	576
Other	197	199
Total long-lived assets	\$ 13,478	\$ 13,911

- 99 -

Note 17. Sales

The following table summarizes the Company's sales by segment and by major source (in millions):

	For the years ending August 31,					
	2019		2018			2017
Retail Pharmacy USA						
Pharmacy	\$	77,192	\$	71,055	\$	60,608
Retail		27,340		27,337		26,695
Total		104,532		98,392		87,302
Retail Pharmacy International						
Pharmacy		4,080		4,360		4,180
Retail		7,382		7,921		7,633
Total		11,462		12,281		11,813
Pharmaceutical Wholesale		23,053		23,006		21,188
Eliminations ¹		(2,180)		(2,142)		(2,089)
Walgreens Boots Alliance, Inc.	\$	136,866	\$	131,537	\$	118,214

¹ Eliminations relate to intersegment sales between the Pharmaceutical Wholesale and the Retail Pharmacy International segments.

Contract balances with customers

Contract liabilities primarily represent the Company's obligation to transfer additional goods or services to a customer for which the Company has received consideration, for example the Company's Balance Rewards[®] and Boots Advantage Card loyalty programs. Under such programs, customers earn reward points on purchases for redemption at a later date. See note 1, summary of major accounting policies, for further information on receivables from contracts with customers.

Note 18. Related parties

The Company has a long-term pharmaceutical distribution agreement with AmerisourceBergen pursuant to which the Company sources branded and generic pharmaceutical products from AmerisourceBergen principally for its U.S. operations. Additionally, AmerisourceBergen receives sourcing services for generic pharmaceutical products.

Related party transactions with AmerisourceBergen (in millions):

	2019	2018	2017
Purchases, net	\$ 57,429	\$ 53,161	\$ 43,571
Trade accounts payable, net	\$ 6,484	\$ 6,274	\$ 4,384
Trade accounts payable, net	\$ 6,484	\$ 6,274	\$ 4,384

- 100 -

Note 19. Supplementary financial information

Summary of Quarterly Results (Unaudited) (in millions, except per share amounts)

Quarter ended November February May August **Fiscal year** Fiscal 2019 Sales \$ 33,793 \$ 34,528 \$ 34,591 \$ 33,954 \$ 136,866 Gross profit 7,641 7,754 7,453 7,228 30,076 Net earnings attributable to Walgreens Boots Alliance, Inc. 3,982 1,123 1,156 1,025 677 Net earnings per common share: \$ \$ \$ \$ 4.32 Basic 1.18 1.25 \$ 1.13 0.75 Diluted 1.18 1.24 1.13 0.75 4.31 Cash dividends declared per common share \$ 0.440 \$ 0.440 \$ 0.440 \$ 0.458 \$ 1.778 Fiscal 2018 Sales \$ 30,740 \$ 33,021 \$ 34,334 \$ 33,442 \$ 131,537 Gross profit 7,341 7,780 30,792 8,096 7,575 Net earnings attributable to Walgreens Boots Alliance, Inc. 821 1,349 1,342 1,512 5,024 Net earnings per common share: \$ \$ Basic 0.82 \$ 1.35 \$ \$ 5.07 1.36 1.55 Diluted 0.81 1.36 1.35 1.55 5.05 \$ 0.400 \$ 0.400 \$ 0.400 \$ 0.440 \$ Cash dividends declared per common share 1.640

- 101 -

Management's Report on Internal Control

Our management is responsible for establishing and maintaining adequate internal control over financial reporting, as such term is defined in Exchange Act Rule 13a-15(f). Under the supervision and with the participation of our management, including our principal executive officer and principal financial officer, we conducted an evaluation of the effectiveness of our internal control over financial reporting based on the framework in *Internal Control - Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). As permitted by the SEC, our assessment of internal control over financial reporting of equity method investees. However, our assessment of internal control over financial reporting of equity method investees. However, our assessment that are recorded in the Consolidated Financial Statements, including controls over the selection of accounting methods for our investments, the recognition of equity method earnings and losses and the determination, valuation and recording of our investment account balances.

Based on our evaluation, management concluded that our internal control over financial reporting was effective as of August 31, 2019. Deloitte & Touche LLP, the Company's independent registered public accounting firm, has audited our internal control over financial reporting, as stated in its report which is included herein.

/s/ Stefano Pessina

/s/ James Kehoe

James Kehoe Executive Vice President and Global Chief Financial Officer

Stefano Pessina Executive Vice Chairman and Chief Executive Officer

October 28, 2019

- 102 -

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and the Board of Directors of Walgreens Boots Alliance, Inc.

Opinion on the Financial Statements

We have audited the accompanying consolidated balance sheets of Walgreens Boots Alliance, Inc. and subsidiaries (the "Company") as of August 31, 2019 and 2018, the related consolidated statements of earnings, comprehensive income, equity, and cash flows for each of the three years in the period ended August 31, 2019, and the related notes (collectively referred to as the "financial statements"). In our opinion, the financial statements present fairly, in all material respects, the financial position of the Company as of August 31, 2019 and 2018, and the results of its operations and its cash flows for each of the three years in the period ended August 31, 2019, in conformity with accounting principles generally accepted in the United States of America.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the Company's internal control over financial reporting as of August 31, 2019, based on criteria established in Internal Control - Integrated Framework (2013) issued by the Committee of Sponsoring Organizations of the Treadway Commission and our report dated October 28, 2019, expressed an unqualified opinion on the Company's internal control over financial reporting.

Basis for Opinion

These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on the Company's financial statements based on our audits. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audits in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether the financial statements are free of material misstatement, whether due to error or fraud. Our audits included performing procedures to assess the risks of material misstatement of the financial statements, whether due to error or fraud, and performing procedures that respond to those risks. Such procedures included examining, on a test basis, evidence regarding the amounts and disclosures in the financial statements. Our audits also included evaluating the accounting principles used and significant estimates made by management, as well as evaluating the overall presentation of the financial statements. We believe that our audits provide a reasonable basis for our opinion.

Critical Audit Matters

The critical audit matters communicated below are matters arising from the current-period audit of the financial statements that were communicated or required to be communicated to the audit committee and that (1) relate to accounts or disclosures that are material to the financial statements and (2) involved our especially challenging, subjective, or complex judgments. The communication of critical audit matters does not alter in any way our opinion on the financial statements, taken as a whole, and we are not, by communicating the critical audit matters below, providing separate opinions on the critical audit matters or on the accounts or disclosures to which they relate.

Goodwill and Indefinite-Lived Intangible Assets Impairment Evaluation - Boots Reporting Unit and Boots Indefinite-lived Intangible Assets - Refer to Notes 1 and 6 to the financial statements

Critical Audit Matter Description

The Company's evaluation of goodwill and indefinite-lived intangible assets for impairment involves the comparison of the fair value of each reporting unit or asset to its carrying value. The Company uses the income and the market approaches to estimate the fair value of its reporting units in its goodwill impairment analysis. The income approach requires management to make a number of assumptions for each reporting unit including annual assumptions on future revenue growth, earnings before interest, taxes, depreciation and amortization (EBITDA) margins and discount rates. The market approach requires management to estimate fair value using comparable marketplace fair value data from within a comparable industry group. The Company primarily uses the multi-period excess earnings model and the relief from royalty model to estimate the fair value of the indefinite-lived intangible assets. Changes in assumptions or the selection of companies in the comparable industry group could have a significant impact on the valuation of the reporting units and the amount of a goodwill or indefinite-lived intangible asset impairment charge, if any.

- 103 -

The Company's goodwill balance was \$16.6 billion as of August 31, 2019, of which \$2.6 billion was allocated to the Boots Reporting Unit. Management's estimate of the fair value of the Boots Reporting Unit is in excess of its carrying value by approximately 9% and, therefore, no impairment was recognized for the year ended August 31, 2019.

The Company's indefinite-lived intangible assets balance was \$7.1 billion as of August 31, 2019, of which \$6.9 billion represented Boots indefinite-lived intangible assets. The fair value of the Boots indefinite-lived intangible assets is in excess of their carrying value by approximately 3% to approximately 29%, except for the Pharmacy License intangible, for which the Company recorded a \$73 million impairment for the year ended August 31, 2019.

Management has made significant judgments to estimate the fair value of the Boots Reporting Unit and the Boots indefinite-lived intangible assets. Given the small difference between their fair values and carrying values, performing audit procedures to evaluate the reasonableness of management's estimates and assumptions of future revenue growth, EBITDA margins, the selection of the discount rate, the selection of the royalty rates for the Boots trade name indefinite-lived intangible assets, and the market multiple selected for the Boots Reporting Unit, specifically due to the sensitivity of the Boots Reporting Unit and Boots indefinite-lived intangible assets to changes in the British economy, required a high degree of auditor judgment and an increased extent of effort, including the need to involve our fair value specialists.

How the Critical Audit Matter Was Addressed in the Audit

Our audit procedures related to the fair value of goodwill for the Boots Reporting Unit and the Boots indefinite-lived intangible assets included the following, among others:

- We tested the effectiveness of controls over the goodwill and intangible asset impairment analyses, including those over the development of forecasts of
 future revenues, EBITDA margins, and the selection of royalty rates, market multiples, and discount rates.
- We evaluated management's ability to accurately forecast future revenues and EBITDA margins by comparing actual results to management's historical forecasts.
- We evaluated the reasonableness of management's forecasts of future revenues and EBITDA margins by performing certain procedures, including:
- Comparing the forecasts to internal communications to management and the Board of Directors.
- Comparing the forecasts to third-party economic research, including discussions with our economic and industry specialists.
- We performed sensitivity analyses to evaluate the risk of impairment if key assumptions are changed.
- We evaluated, with the assistance of our fair value specialists, the (1) valuation methodology used for the Boots Reporting Unit goodwill and the Boots indefinite-lived intangible assets and (2) the reasonableness of the related discount rates, by performing certain procedures, including:
 - Comparing the valuation methodologies used to generally accepted valuation practices for each asset type.
 - Evaluating the appropriateness of the Company's selection of companies in its industry comparable group for comparability to the Boots Reporting Unit.
 - Testing the source information underlying the determination of the discount rate and the mathematical accuracy of the calculation.
 - Developing an independent discount rate and comparing it to the discount rate selected by management.

Income Taxes - Uncertain Tax Positions - Refer to Notes 1 and 11 to the financial statements

Critical Audit Matter Description

The Company has a complex legal structure involving numerous domestic and foreign locations with constantly changing tax laws and regulations. The Company's management is required to interpret and apply these tax laws and regulations in determining the amount of its income tax liability and provision. When an uncertain tax position is identified by management, the Company must evaluate whether it is more likely to be sustained than not on the basis of its technical merits. The Company recognizes a benefit for tax positions using the highest cumulative tax benefit that is more likely than not to be realized. The Company establishes a liability for unrecognized tax benefits that do not meet this threshold. The evaluation of each uncertain tax position requires management to apply specialized skill, knowledge, and significant judgment related to the identified position. The Company's liability for unrecognized tax benefits as of August 31, 2019 was \$455 million.

Because of the numerous taxing jurisdictions in which the Company files its tax returns and the complexity of tax laws and regulations, auditing uncertain tax positions and the determination of whether the more likely than not threshold was met requires a high degree of auditor judgment and increased extent of effort, including the involvement of our income tax specialists.

How the Critical Audit Matter Was Addressed in the Audit

Our audit procedures related to unrecognized tax benefits included the following, among others:

- We tested the effectiveness of controls over income taxes, including those over identifying uncertain tax positions and measuring liabilities.
- We evaluated, with the assistance of our tax specialists, a selection of underlying tax positions to evaluate the more likely than not principle as it applied to the specific underlying tax position.
- We evaluated, with the assistance of our tax specialists, the Company's unrecognized tax positions by performing the following:
- Obtaining management and third-party opinions or memoranda regarding the analysis of uncertain tax positions and identifying the key judgments and evaluating whether the analysis was consistent with our interpretation of the relevant laws and regulations.
- Evaluating the basis for certain intercompany transactions, such as transfer pricing, by comparison to economic studies performed by management and third-party data.
- Evaluating the matters raised by tax authorities in former and ongoing tax audits and considering the implications of these matters on open tax years.
- Assessing changes and interpretation of applicable tax law.

/s/ DELOITTE & TOUCHE LLP

Chicago, Illinois October 28, 2019 We have served as the Company's auditor since 2002.

- 105 -

REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Shareholders and the Board of Directors of Walgreens Boots Alliance, Inc.

Opinion on Internal Control over Financial Reporting

We have audited the internal control over financial reporting of Walgreens Boots Alliance, Inc. and subsidiaries (the "Company") as of August 31, 2019, based on criteria established in *Internal Control - Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). In our opinion, the Company maintained, in all material respects, effective internal control over financial reporting as of August 31, 2019, based on criteria established in *Internal Control - Integrated Framework (2013)* issued by the Committee of Sponsoring Organizations of the Treadway Commission (COSO). In our opinion, the Company maintained, in all material respects, effective internal control over financial reporting as of August 31, 2019, based on criteria established in *Internal Control - Integrated Framework (2013)* issued by COSO.

We have also audited, in accordance with the standards of the Public Company Accounting Oversight Board (United States) (PCAOB), the consolidated financial statements as of and for the year ended August 31, 2019, of the Company and our report dated October 28, 2019, expressed an unqualified opinion on those financial statements.

Basis for Opinion

The Company's management is responsible for maintaining effective internal control over financial reporting and for its assessment of the effectiveness of internal control over financial reporting, included in the accompanying Management's Report on Internal Control. Our responsibility is to express an opinion on the Company's internal control over financial reporting based on our audit. We are a public accounting firm registered with the PCAOB and are required to be independent with respect to the Company in accordance with the U.S. federal securities laws and the applicable rules and regulations of the Securities and Exchange Commission and the PCAOB.

We conducted our audit in accordance with the standards of the PCAOB. Those standards require that we plan and perform the audit to obtain reasonable assurance about whether effective internal control over financial reporting was maintained in all material respects. Our audit included obtaining an understanding of internal control over financial reporting, assessing the risk that a material weakness exists, testing and evaluating the design and operating effectiveness of internal control based on the assessed risk, and performing such other procedures as we considered necessary in the circumstances. We believe that our audit provides a reasonable basis for our opinion.

Definition and Limitations of Internal Control over Financial Reporting

A company's internal control over financial reporting is a process designed to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles. A company's internal control over financial reporting includes those policies and procedures that (1) pertain to the maintenance of records that, in reasonable detail, accurately and fairly reflect the transactions and dispositions of the assets of the company; (2) provide reasonable assurance that transactions are recorded as necessary to permit preparation of financial statements in accordance with generally accepted accounting principles, and that receipts and expenditures of the company are being made only in accordance with authorizations of management and directors of the company; and (3) provide reasonable assurance regarding prevention or timely detection of unauthorized acquisition, use, or disposition of the company's assets that could have a material effect on the financial statements.

Because of its inherent limitations, internal control over financial reporting may not prevent or detect misstatements. Also, projections of any evaluation of effectiveness to future periods are subject to the risk that controls may become inadequate because of changes in conditions, or that the degree of compliance with the policies or procedures may deteriorate.

/s/ DELOITTE & TOUCHE LLP

Chicago, Illinois October 28, 2019

- 106 -

Item 9. Changes in and disagreements with accountants on accounting and financial disclosure

None.

Item 9A. Controls and procedures

Evaluation of disclosure controls and procedures

Management conducted an evaluation of the effectiveness of the design and operation of our disclosure controls and procedures as of the end of the period covered by this Form 10-K. The controls evaluation was conducted under the supervision and with the participation of the Company's management, including its Chief Executive Officer ("CEO") and Chief Financial Officer ("CFO"). Based upon the controls evaluation, our CEO and CFO have concluded that, as of the end of the period covered by this report, our disclosure controls and procedures were effective to provide reasonable assurance that information required to be disclosed in our Exchange Act reports is recorded, processed, summarized and reported within the time periods specified by the SEC, and that such information is accumulated and communicated to management, including the CEO and CFO, as appropriate to allow timely decisions regarding required disclosure.

Report on internal control over financial reporting

Management's report on internal control over financial reporting and the report of Deloitte & Touche LLP, the Company's independent registered public accounting firm, related to their assessment of the effectiveness of internal control over financial reporting are included in Part II, Item 8 of this Form 10-K and are incorporated in this Item 9A by reference.

Changes in internal control over financial reporting

In connection with the evaluation pursuant to Exchange Act Rule 13a-15(d) of the Company's internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f)) by the Company's management, including its CEO and CFO, no changes during the quarter ended August 31, 2019 were identified that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting. We are implementing a new enterprise resource planning (ERP) system which affects many of our financial processes. This project is expected to improve the efficiency and effectiveness of certain financial and business transaction processes, as well as the underlying systems environment. The new ERP system will be a significant component of our internal control over financial reporting.

Inherent limitations on effectiveness of controls

Our management, including the CEO and CFO, do not expect that our disclosure controls and procedures or our internal control over financial reporting will prevent or detect all errors and all fraud. A control system, no matter how well designed and operated, can provide only reasonable, not absolute, assurance that the control system's objectives will be met. The design of a control system must reflect the fact that there are resource constraints, and the benefits of controls must be considered relative to their costs. Further, because of the inherent limitations in all control systems, no evaluation of controls can provide absolute assurance that misstatements due to error or fraud will not occur or that all control issues and instances of fraud, if any, within the Company have been detected. These inherent limitations include the realities that judgments in decision-making can be faulty and that breakdowns can occur because of simple error or mistake. Controls can also be circumvented by the individual acts of some persons, by collusion of two or more people, or by management override of the controls. The design of any system of controls is based in part on certain assumptions about the likelihood of future events, and there can be no assurance that any design will succeed in achieving its stated goals under all potential future conditions. Projections of any evaluation of controls effectiveness to future periods are subject to risks. Over time, controls may become inadequate because of changes in conditions or deterioration in the degree of compliance with policies or procedures.

Item 9B. Other information

On October 22, 2019, the Compensation and Leadership Performance Committee of the Board of Directors of the Company approved forms of grant agreement for future use under the amended and restated Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan. These forms are attached to this Annual Report on Form 10-K as: Exhibit 10.3 (form of Performance Share Award agreement); Exhibit 10.6 (form of Stock Option Award agreement); Exhibit 10.20 (form of Restricted Stock Unit Award agreement); Exhibit 10.29 (form of Stock Option Award agreement under UK Sub-plan); Exhibit 10.10 (form of Performance Share Award agreement for CEO); Exhibit 10.10 (form of Stock Option Award agreement for CEO); Exhibit 10.19 (form of Restricted Stock Unit Award agreement for CEO); and Exhibit 10.19 (form of Restricted Stock Unit Award agreement for Executive Chairman) hereto, and each of which are incorporated herein by reference.

On October 22, 2019, the Compensation and Leadership Performance Committee also approved an Amended and Restated Executive Deferred Profit-Sharing Plan that, among other things, allows for participant investment elections that are separate from the election opportunities available under the Walgreen Profit-Sharing Retirement Plan. The Committee also approved an extension of the Assignment Agreement of Alex Gourlay. The Amended and Restated Executive Deferred Profit-Sharing Plan and the extension of the Assignment are attached as Exhibit 10.43 and Exhibit 10.62, respectively, to this Annual Report on Form 10-K and are incorporated herein by reference.

- 107 -

PART III

Item 10. Directors, executive officers and corporate governance

The information required by Item 10, with the exception of the information relating to the executive officers of the Company, which is presented in part I above under the heading "Executive Officers of the Registrant," is incorporated herein by reference to the following sections of the Company's Proxy Statement relating to its next Annual Meeting of Stockholders (the "Proxy Statement"): Proposal–1 Election of Directors; Governance; and Delinquent Section 16(a) Reports.

The Company has adopted a Code of Conduct and Business Ethics applicable to all employees, officers and directors that incorporates policies and guidelines designed to deter wrongdoing and to promote honest and ethical conduct and compliance with applicable laws and regulations. The Company has also adopted a Code of Ethics for CEO and Financial Executives. This Code applies to and has been signed by the Chief Executive Officer, the Chief Financial Officer and the Controller. The Company intends to promptly disclose on its website in accordance with applicable rules required disclosure of changes to or waivers, if any, of the Code of Ethics for CEO and Financial Executives or the Code of Conduct and Business Ethics for directors and executive officers.

Charters of all committees of the Company's Board of Directors, as well as the Company's Corporate Governance Guidelines and Code of Ethics for CEO and Financial Executives and Code of Conduct and Business Ethics, are available on the Company's website at investor.walgreensbootsalliance.com or, upon written request and free of charge, in printed hardcopy form. Written requests should be sent to Walgreens Boots Alliance, Inc., Attention: Investor Relations, Mail Stop #1833, 108 Wilmot Road, Deerfield, Illinois 60015.

Item 11. Executive compensation

The information required by Item 11 is incorporated herein by reference to the following sections of the Company's Proxy Statement: Director Compensation; Executive Compensation; and Governance.

The material incorporated herein by reference to the material under the caption "Compensation Committee Report" in the Proxy Statement shall be deemed furnished, and not filed, in this Form 10-K and shall not be deemed incorporated by reference into any filing under the Securities Act of 1933, as amended, or the Exchange Act as a result of this furnishing, except to the extent that the Company specifically incorporates it by reference.

Item 12. Security ownership of certain beneficial owners and management and related stockholder matters

The information required by Item 12 is incorporated herein by reference to the following sections of the Company's Proxy Statement: Security Ownership of Certain Beneficial Owners and Management; and Equity Compensation Plan Information.

Item 13. Certain relationships and related transactions and director independence

The information required by Item 13 is incorporated herein by reference to the following sections of the Company's Proxy Statement: Related Party Transactions; Director Independence; and Governance.

Item 14. Principal accounting fees and services

The information required by Item 14 is incorporated herein by reference to the following section of the Company's Proxy Statement: Independent Registered Public Accounting Firm Fees and Services.

- 108 -

PART IV

Item 15. Exhibits and financial statement schedules

(a) Documents filed as part of this report:

(1) *Financial statements.* The following financial statements, supplementary data and reports of independent public accountants appear in part II, item 8 of this Form 10-K and are incorporated herein by reference.

Consolidated Balance Sheets at August 31, 2019 and 2018 Consolidated Statements of Equity, Earnings, Comprehensive Income and Cash Flows for the years ended August 31, 2019, 2018 and 2017 Notes to Consolidated Financial Statements Management's Report on Internal Control Report of Independent Registered Public Accounting Firm

(2) Financial statement schedules and supplementary information

Schedules I, II, III, IV and V are not submitted because they are not applicable or not required or because the required information is included in the Financial Statements referenced in (1) above or the notes thereto.

(3) *Exhibits.* Exhibits 10.1 through 10.62 constitute management contracts or compensatory plans or arrangements required to be filed as exhibits pursuant to Item 15(b) of this Form 10-K.

The agreements included as exhibits to this report are included to provide information regarding their terms and not intended to provide any other factual or disclosure information about the Company or the other parties to the agreements. The agreements may contain representations and warranties by each of the parties to the applicable agreement that were made solely for the benefit of the other parties to the applicable agreement, and:

- should not in all instances be treated as categorical statements of fact, but rather as a way of allocating the risk to one of the parties if those statements prove to be inaccurate;
- may have been qualified by disclosures that were made to the other party in connection with the negotiation of the applicable agreement, which disclosures are not necessarily reflected in the agreement;
- may apply standards of materiality in a way that is different from what may be viewed as material to you or other investors; and
- were made only as of the date of the applicable agreement or such other date or dates as may be specified in the agreement and are subject to more recent developments.

Accordingly, these representations and warranties may not describe the actual state of affairs as of the date they were made or at any other time.

(b) Exhibits

.

<u>Exhibit</u> <u>No.</u>	Description	SEC Document Reference
<u>2.1</u> *	Purchase and Option Agreement by and among Walgreen Co., Alliance Boots GmbH and AB Acquisitions Holdings Limited dated June 18, 2012 and related annexes.	Incorporated by reference to Annex B-1 to the proxy statement/prospectus forming a part of the Registration Statement on Form S-4 (File No. 333-198768) filed with the SEC pursuant to Rule 424(b)(3) on November 24, 2014.
<u>2.2</u> *	Amendment No. 1 to Purchase and Option Agreement and Walgreen Co. Shareholders Agreement, dated August 5, 2014, by and among Walgreen Co., Alliance Boots GmbH, AB Acquisitions Holdings Limited, Walgreen Scotland Investments LP, KKR Sprint (European II) Limited, KKR Sprint (2006) Limited and KKR Sprint (KPE) Limited, Alliance Santé Participations S.A., Stefano Pessina and Kohlberg Kravis Roberts & Co. L.P.	Incorporated by reference to Annex B-2 to the proxy statement/prospectus forming a part of the Registration Statement on Form S-4 (File No. 333-198768) filed with the SEC pursuant to Rule 424(b)(3) on November 24, 2014.

- 109 -

```
Table of Contents
```

<u>2.3</u>	Agreement and Plan of Merger, dated October 17, 2014, by and among Walgreen Co., Walgreens Boots Alliance, Inc. and Ontario Merger Sub, Inc.	Incorporated by reference to Annex A to the proxy statement/prospectus forming a part of the Registration Statement on Form S-4 (File No. 333-198768) filed with the SEC pursuant to Rule 424(b)(3) on November 24, 2014.
<u>2.4</u>	Amendment No. 1 to Agreement and Plan of Merger, dated December 23, 2014, by and among Walgreen Co., Walgreens Boots Alliance, Inc. and Ontario Merger Sub, Inc.	Incorporated by reference to Exhibit 2.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on December 24, 2014.
<u>2.5</u>	Amendment No. 2 to Agreement and Plan of Merger, dated December 29, 2014, by and among Walgreen Co., Walgreens Boots Alliance, Inc. and Ontario Merger Sub, Inc.	Incorporated by reference to Exhibit 2.3 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2014 (File No. 1-36759) filed with the SEC on December 30, 2014.
<u>2.6</u> *	Amended and Restated Asset Purchase Agreement, dated as of September 18, 2017, by and among Walgreens Boots Alliance, Inc., Walgreen Co. and Rite Aid Corporation	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on September 19, 2017.
<u>3.1</u>	Amended and Restated Certificate of Incorporation of Walgreens Boots Alliance, Inc.	Incorporated by reference to Exhibit 3.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K12B (File No. 1-36759) filed with the SEC on December 31, 2014.
<u>3.2</u>	Amended and Restated By-laws of Walgreens Boots Alliance, Inc.	Incorporated by reference to Exhibit 3.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on June 10, 2016.
<u>4.1</u> **	Indenture, dated as of July 17, 2008, between Walgreen Co. and Wells Fargo Bank, National Association, as trustee.	Incorporated by reference to Exhibit 4.3 to Walgreen Co.'s registration statement on Form S-3ASR (File No. 333-152315) filed with the SEC on July 14, 2008.
<u>4.2</u>	Form of Walgreen Co. 3.100% Note due 2022.	Incorporated by reference to Exhibit 4.4 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on September 13, 2012.
<u>4.3</u>	Form of Walgreen Co. 4.400% Note due 2042.	Incorporated by reference to Exhibit 4.5 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on September 13, 2012.
<u>4.4</u>	Form of Guarantee of Walgreens Boots Alliance, Inc.	Incorporated by reference to Exhibit 4.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K12B (File No. 1-36759) filed with the SEC on December 31, 2014.
<u>4.5</u>	Indenture dated November 18, 2014 among Walgreens Boots Alliance, Inc. and Wells Fargo Bank, National Association, as trustee.	Incorporated by reference to Exhibit 4.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 18, 2014.
<u>4.6</u>	Form of 2.700% Notes due 2019.	Incorporated by reference to Exhibit 4.4 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 18, 2014.
<u>4.7</u>	Form of 3.300% Notes due 2021.	Incorporated by reference to Exhibit 4.5 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 18, 2014.
<u>4.8</u>	Form of 3.800% Notes due 2024.	Incorporated by reference to Exhibit 4.6 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 18, 2014.
<u>4.9</u>	Form of 4.500% Notes due 2034.	Incorporated by reference to Exhibit 4.7 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 18, 2014.
<u>4.10</u>	Form of 4.800% Notes due 2044.	Incorporated by reference to Exhibit 4.8 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 18, 2014.

- 110 -

Table of Contents

<u>4.11</u>	Form of 2.875% Notes due 2020 (£).	Incorporated by reference to Exhibit 4.2 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 20, 2014.
<u>4.12</u>	Form of 3.600% Notes due 2025 (£).	Incorporated by reference to Exhibit 4.3 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 20, 2014.
<u>4.13</u>	Form of 2.125% Notes due 2026 (€).	Incorporated by reference to Exhibit 4.4 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 20, 2014.
<u>4.14</u>	Indenture, dated as of December 17, 2015, between Walgreens Boots Alliance, Inc. and Wells Fargo Bank, National Association, as trustee	Incorporated by reference to Exhibit 4.1 to the Registration Statement on Form S-3 (File No. 333-208587) filed with the SEC on December 17, 2015.
<u>4.15</u>	Form of 3.450% Notes due 2026	Incorporated by reference to Exhibit 4.5 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on June 1, 2016.
<u>4.16</u>	Form of 4.650% Notes due 2046	Incorporated by reference to Exhibit 4.6 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on June 1, 2016.
<u>4.17</u>	Walgreen Co. Shareholders Agreement, dated as of August 2, 2012, among Walgreen Co., Stefano Pessina, KKR Sprint (European II) Limited, KKR Sprint (2006) Limited and KKR Sprint (KPE) Limited, Alliance Santé Participations S.A., Kohlberg Kravis Roberts & Co. L.P. and certain other investors party thereto.	Incorporated by reference to Exhibit 4.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on August 6, 2012.
<u>4.18</u>	Amendment No. 1 to Purchase and Option Agreement and Walgreen Co. Shareholders Agreement, dated August 5, 2014, by and among Walgreen Co., Alliance Boots GmbH, AB Acquisitions Holdings Limited, Walgreen Scotland Investments LP, KKR Sprint (European II) Limited, KKR Sprint (2006) Limited and KKR Sprint (KPE) Limited, Alliance Santé Participations S.A., Stefano Pessina and Kohlberg Kravis Roberts & Co. L.P.	Incorporated by reference to Annex B-2 to the proxy statement/prospectus forming a part of the Registration Statement on Form S-4 (File No. 333-198768) filed with the SEC pursuant to Rule 424(b)(3) on November 24, 2014.
<u>4.19</u>	Amendment No. 2 to Purchase and Option Agreement and Walgreen Co. Shareholders Agreement, dated December 31, 2014, as Amended by Amendment No. 1, dated as of August 5, 2014, by and among Walgreen Co., Alliance Boots GmbH, AB Acquisitions Holdings Limited, Ontario Holdings WBS Limited, KKR Sprint (European II) Limited, KKR Sprint (2006) Limited and KKR Sprint (KPE) Limited, Alliance Santé Participations S.A., Stefano Pessina and Kohlberg Kravis Roberts & Co. L.P.	Incorporated by reference to Exhibit E to the Schedule 13D filed by Alliance Santé Participations S.A. (File No. 005-88481) filed with the SEC on December 31, 2014).
<u>4.20</u>	Description of Registered Securities.	Filed herewith.
<u>10.1</u>	Walgreens Boots Alliance, Inc. Management Incentive Plan (as amended and restated effective July 1, 2016).	Incorporated by reference to Exhibit 10.2 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2016 (File No. 1-36759) filed with the SEC on October 20, 2016.
<u>10.2</u>	Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (as amended and restated).	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on January 19, 2018.
<u>10.3</u>	Form of Performance Share Award agreement (effective October 2019).	Filed herewith
<u>10.4</u>	Form of Performance Share Award agreement (effective October 2018).	Incorporated by reference to Exhibit 10.2 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2018 (File No. 1-36759) filed with the SEC on December 20, 2018.

<u>10.5</u>	Form of Performance Share Award agreement (effective October 2017).	Incorporated by reference to Exhibit 10.5 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2017 (File No. 1-36759) filed with the SEC on October 25, 2017.
<u>10.6</u>	Form of Stock Option Award agreement (effective October 2019).	Filed herewith.
<u>10.7</u>	Form of Stock Option Award agreement (effective October 2018).	Incorporated by reference to Exhibit 10.3 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2018 (File No. 1-36759) filed with the SEC on December 20, 2018.
<u>10.8</u>	Form of Stock Option Award agreement (effective October 2017).	Incorporated by reference to Exhibit 10.7 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2017 (File No. 1-36759) filed with the SEC on October 25, 2017.
<u>10.9</u>	Form of Stock Option Award agreement (effective July 2016).	Incorporated by reference to Exhibit 10.8 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2016 (File No. 1-36759) filed with the SEC on October 20, 2016.
<u>10.10</u>	Form of Performance Share Award agreement for CEO (November 2019).	Filed herewith.
<u>10.11</u>	Form of Performance Share Award agreement for CEO (November 2018).	Incorporated by reference to Exhibit 10.5 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2018 (File No. 1-36759) filed with the SEC on December 20, 2018.
<u>10.12</u>	Form of Performance Share Award agreement for CEO (November 2017).	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2017 (File No. 1-36759) filed with the SEC on January 4, 2018.
<u>10.13</u>	Form of Performance Share Award agreement for CEO (November 2016).	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2016 (File No. 1-36759) filed with the SEC on January 5, 2017.
<u>10.14</u>	Form of Stock Option Award agreement for CEO (November 2019).	Filed herewith.
<u>10.15</u>	Form of Stock Option Award agreement for CEO (November 2018).	Incorporated by reference to Exhibit 10.6 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2018 (File No. 1-36759) filed with the SEC on December 20, 2018.
<u>10.16</u>	Form of Stock Option Award agreement for CEO (November 2017).	Incorporated by reference to Exhibit 10.2 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2017 (File No. 1-36759) filed with the SEC on January 4, 2018.
<u>10.17</u>	Form of Stock Option Award agreement for CEO (November 2016).	Incorporated by reference to Exhibit 10.2 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2016 (File No. 1-36759) filed with the SEC on January 5, 2017.
<u>10.18</u>	Form of Restricted Stock Unit Award agreement for CEO (November 2019).	Filed herewith.
<u>10.19</u>	Form of Restricted Stock Unit Award agreement for Executive Chairman (November 2019).	Filed herewith.
<u>10.20</u>	Form of Restricted Stock Unit agreement (effective October 2019)	Filed herewith.
<u>10.21</u>	Form of Restricted Stock Unit Award agreement for Executive Chairman (November 2018).	Incorporated by reference to Exhibit 10.7 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2018 (File No. 1-36759) filed with the SEC on December 20, 2018.
<u>10.22</u>	Form of Restricted Stock Unit Award agreement for Executive Chairman (November 2017).	Incorporated by reference to Exhibit 10.3 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2017 (File No. 1-36759) filed with the SEC on January 4, 2018.

<u>10.23</u>	Form of Restricted Stock Unit Award agreement for Executive Chairman (November 2016).	Incorporated by reference to Exhibit 10.3 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2016 (File No. 1-36759) filed with the SEC on January 5, 2017.	
<u>10.24</u>	Form of Restricted Stock Unit Award agreement (September 2019).	Filed herewith.	
<u>10.25</u>	Form of Restricted Stock Unit Award agreement for James Kehoe (June 2018).	Incorporated by reference to Exhibit 10.18 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K (File No. 1-36759) filed with the SEC on October 11, 2018.	
<u>10.26</u>	Form of Amendment to Stock Option Award agreements.	Incorporated by reference to Exhibit 10.11 to Walgreen Co.'s Annual Report on Form 10-K for the fiscal year ended August 31, 2014 (File No. 1-00604) filed with the SEC on October 20, 2014.	
<u>10.27</u>	Amendments to certain Omnibus Plan Award agreements (October 2018).	Incorporated by reference to Exhibit 10.7 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on October 26, 2018.	
<u>10.28</u>	UK Sub-Plan under the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan.	Incorporated by reference to Exhibit 10.16 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K (File No. 1-36759) filed with the SEC on October 28, 2015.	
<u>10.29</u>	Form of Stock Option Award agreement under UK Sub-plan (effective October 2019).	Filed herewith.	
<u>10.30</u>	Form of Stock Option Award agreement under UK Sub-plan (effective October 2018).	Incorporated by reference to Exhibit 10.4 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2018 (File No. 1-36759) filed with the SEC on December 20, 2018.	
<u>10.31</u>	Form of Stock Option Award agreement under UK Sub-plan (effective October 2017).	Incorporated by reference to Exhibit 10.24 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2017 (File No. 1-36759) filed with the SEC on October 25, 2017.	
<u>10.32</u>	Form of Stock Option Award agreement under UK Sub-plan (effective July 2016).	Incorporated by reference to Exhibit 10.23 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2016 (File No. 1-36759) filed with the SEC on October 20, 2016.	
<u>10.33</u>	Walgreen Co. Long-Term Performance Incentive Plan (amendment and restatement of the Walgreen Co. Restricted Performance Share Plan).	Incorporated by reference to Exhibit 10.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on January 11, 2007.	
<u>10.34</u>	Walgreen Co. Long-Term Performance Incentive Plan Amendment No. 1 (effective January 10, 2007).	Incorporated by reference to Exhibit 10.2 to Walgreen Co.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2007 (File No. 1-00604).	
<u>10.35</u>	Walgreen Co. Long-Term Performance Incentive Plan Amendment No. 2.	Incorporated by reference to Exhibit 10.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on April 14, 2011.	
<u>10.36</u>	Form of Restricted Stock Unit Award agreement (August 15, 2011 grants).	Incorporated by reference to Exhibit 10.5 to Walgreen Co.'s Annual Report on Form 10-K for the fiscal year ended August 31, 2011 (File No. 1-00604).	
<u>10.37</u>	Walgreen Co. Executive Stock Option Plan (as amended and restated effective January 13, 2010).	Incorporated by reference to Exhibit 99.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on January 20, 2010.	
<u>10.38</u>	Walgreen Co. 2002 Executive Deferred Compensation/Capital Accumulation Plan.	Incorporated by reference to Exhibit 10(g) to Walgreen Co.'s Annual Report on Form 10-K for the fiscal year ended August 31, 2002 (File No. 1-00604).	
<u>10.39</u>	Amendment to the Walgreen Co. 2002 et. al. Executive Deferred Compensation/Capital Accumulation Plans.	Incorporated by reference to Exhibit 10.3 to Walgreen Co.'s Quarterly Report on Form 10-Q for the fiscal quarter ended February 28, 2009 (File No. 1-00604).	

- 113 -

<u>10.40</u>	Walgreen Co. 2006 Executive Deferred Compensation/Capital Accumulation Plan (effective January 1, 2006).	Incorporated by reference to Exhibit 10(b) to Walgreen Co.'s Quarterly Report on Form 10-Q for the fiscal quarter ended November 30, 2005 (File No. 1-00604).	
<u>10.41</u>	Walgreen Co. 2011 Executive Deferred Compensation Plan.	Incorporated by reference to Exhibit 10.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on November 12, 2010.	
<u>10.42</u>	Amendment No. 1 to the Walgreen Co. 2011 Executive Deferred Compensation Plan.	Incorporated by reference to Exhibit 10.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on January 19, 2011.	
<u>10.43</u>	Walgreens Boots Alliance, Inc. Executive Retirement Savings Plan (as amended and restated effective January 1, 2020).	Filed herewith.	
<u>10.44</u>	Share Walgreens Stock Purchase/Option Plan (effective October 1, 1992), as amended.	Incorporated by reference to Exhibit 10(d) to Walgreen Co.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2003 (File No. 1-00604).	
<u>10.45</u>	Share Walgreens Stock Purchase/Option Plan Amendment No. 4 (effective July 15, 2005), as amended.	Incorporated by reference to Exhibit 10(h)(ii) to Walgreen Co.'s Annual Report on Form 10-K for the fiscal year ended August 31, 2005 (File No. 1-00604).	
<u>10.46</u>	Share Walgreens Stock Purchase/Option Plan Amendment No. 5 (effective October 11, 2006).	Incorporated by reference to Exhibit 10(b) to Walgreen Co.'s Quarterly Report on Form 10-Q for the quarter ended November 30, 2006 (File No. 1-00604).	
<u>10.47</u>	Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan (as amended and restated effective August 6, 2019).	Filed herewith.	
<u>10.48</u>	Offer Letter agreement between Stefano Pessina and Walgreens Boots Alliance, Inc.	Incorporated by reference to Exhibit 10.29 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2015 (File No. 1-36759) filed with the SEC on April 9, 2015.	
<u>10.49</u>	Offer letter agreement dated as of March 6, 2018 between James Kehoe and Walgreens Boots Alliance, Inc.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on March 8, 2018.	
<u>10.50</u>	Service Agreement between Boots UK Limited and Alex Gourlay, dated January 29, 2009.	Incorporated by reference to Exhibit 10.18 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2015 (File No. 1-36759) filed with the SEC on April 9, 2015.	
<u>10.51</u>	Letter Agreement between Alliance Boots Management Services Limited and Alex Gourlay, dated June 28, 2010.	Incorporated by reference to Exhibit 10.19 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2015 (File No. 1-36759) filed with the SEC on April 9, 2015.	
<u>10.52</u>	Employment Agreement between Alliance UniChem Plc and Ornella Barra dated December 10, 2002.	Incorporated by reference to Exhibit 10.20 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2015 (File No. 1-36759) filed with the SEC on April 9, 2015.	
<u>10.53</u>	Agreement among Alliance Boots plc, Alliance UniChem Plc and Ornella Barra, dated July 31, 2006.	Incorporated by reference to Exhibit 10.21 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2015 (File No. 1-36759) filed with the SEC on April 9, 2015.	
<u>10.54</u>	Novation of Service Agreement among Alliance Boots Holdings Limited, Alliance Boots Management Services MC S.A.M and Ornella Barra, dated June 1, 2013.	Incorporated by reference to Exhibit 10.22 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2015 (File No. 1-36759) filed with the SEC on April 9, 2015.	
<u>10.55</u>	Walgreens Boots Alliance, Inc. Long-Term Global Assignment Relocation Policy	Incorporated by reference to Exhibit 10.68 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K (File No. 1-36759) filed with the SEC on October 28, 2015.	

- 114 -

Table of Contents

<u>10.56</u>	Secondment Agreement dated September 27, 2013 between Alliance Boots Management Services Limited and Walgreen Co.	Incorporated by reference to Exhibit 10.52 to Walgreen Co.'s Annual Report on Form 10-K for the fiscal year ended August 31, 2013 (File No. 1-00604).
<u>10.57</u>	Assignment Letter dated September 27, 2013 between Alexander Gourlay and Alliance Boots Management Services Ltd.	Incorporated by reference to Exhibit 10.53 to Walgreen Co.'s Annual Report on Form 10-K for the fiscal year ended August 31, 2013 (File No. 1-00604).
<u>10.58</u>	Extension, dated January 27, 2016, to Assignment Letter between Alexander Gourlay and Walgreens Boots Alliance Services Limited (formerly Alliance Boots Management Services Ltd.).	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on February 1, 2016.
<u>10.59</u>	Extension, dated as of March 27, 2017, to Assignment Letter between Alexander Gourlay and Walgreens Boots Alliance Services Limited (formerly Alliance Boots Management Services Ltd.).	Incorporated by reference to Exhibit 10.6 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2017 (File No. 1-36759) filed with the SEC on April 5, 2017.
<u>10.60</u>	Extension, dated as of July 13, 2017, to Assignment Letter between Alexander Gourlay and Walgreens Boots Alliance Services Limited (formerly Alliance Boots Management Services Ltd.).	Incorporated by reference to Exhibit 10.62 to Walgreens Boots Alliance, Inc.'s Annual Report on Form 10-K for the year ended August 31, 2017 (File No. 1-36759) filed with the SEC on October 25, 2017.
<u>10.61</u>	Extension, dated as of August 1, 2018, to Assignment Letter between Alexander Gourlay and Walgreens Boots Alliance Services Limited.	Incorporated by reference to Exhibit 10.8 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 1-36759) filed with the SEC on October 26, 2018.
<u>10.62</u>	Extension, effective as of October 22, 2019, to Assignment Letter between Alexander Gourlay and Walgreens Boots Alliance Services Limited.	Filed herewith.
<u>10.63</u>	Shareholders' Agreement, dated as of August 2, 2012, by and among Alliance Boots GmbH, AB Acquisition Holdings Limited and Walgreen Co.	Incorporated by reference to Exhibit 10.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on August 6, 2012.
<u>10.64</u>	Framework Agreement, dated as of March 18, 2013, by and among Walgreen Co., Alliance Boots GmbH and AmerisourceBergen Corporation.	Incorporated by reference to Exhibit 10.1 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on March 20, 2013.
<u>10.65</u>	Shareholders Agreement, dated as of March 18, 2013, by and among Walgreen Co., Alliance Boots GmbH and AmerisourceBergen Corporation.	Incorporated by reference to Exhibit 10.2 to Walgreen Co.'s Current Report on Form 8-K (File No. 1-00604) filed with the SEC on March 20, 2013.
<u>10.66</u>	Revolving Credit Agreement, dated as of August 29, 2018, by and among Walgreens Boots Alliance, Inc., the lenders and letter of credit issuers from time to time party thereto, Wells Fargo Bank, National Association, as administrative agent, and the joint lead arrangers, joint bookrunners and co-syndication.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on August 30, 2018.
<u>10.67</u>	Credit Agreement, dated as of November 30, 2018, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Sumitomo Mitsui Banking Corporation, as sole lead arranger and administrative agent.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on December 6, 2018.
<u>10.68</u>	Term Loan Credit Agreement, dated as of December 5, 2018, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Wells Fargo Bank, National Association, as administrative agent.	Incorporated by reference to Exhibit 10.2 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on December 6, 2018.
<u>10.69</u>	Revolving Credit Agreement, dated as of December 21, 2018, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Bank of America, N.A., as administrative agent.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on December 26, 2018.
<u>10.70</u>	364-Day Revolving Credit Agreement, dated as of January 18, 2019, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Mizuho Bank, Ltd., as administrative agent.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on January 22, 2019.

- 115 -

```
Table of Contents
```

	Amendment No. 1 to Credit Agreement, dated as of March 25, 2019, by and between Walgreens Boots Alliance, Inc. and Sumitomo Mitsui Banking Corporation, as sole lead arranger and administrative agent, amending that certain Credit Agreement, dated as of November 30, 2018, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto, and Sumitomo Mitsui Banking Corporation, as sole lead arranger and administrative agent.	Incorporated by reference to Exhibit 10.5 to Walgreens Boots Alliance, Inc.'s Quarterly Report on Form 10-Q for the quarter ended February 28, 2019 (File No. 1-36759) filed with the SEC on April 2, 2019.
	Amendment No. 1 to Term Loan Credit Agreement, dated as of August 9, 2019, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Wells Fargo Bank, National Association, as administrative agent, amending that certain Term Loan Credit Agreement, dated as of December 5, 2018, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Wells Fargo Bank, National Association, as administrative agent.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on August 12, 2019
	Revolving Credit Agreement, dated as of August 30, 2019, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and HSBC Bank USA, N.A., as administrative agent, and HSBC Securities (USA), Inc., as sole lead arranger.	Incorporated by reference to Exhibit 10.1 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on September 4, 2019.
	Revolving Credit Agreement, dated as of August 30, 2019, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and Citibank, N.A., as administrative agent and sole lead arranger.	Incorporated by reference to Exhibit 10.2 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on September 4, 2019.
	Revolving Credit Agreement, dated as of August 30, 2019, by and among Walgreens Boots Alliance, Inc., the lenders from time to time party thereto and UniCredit Bank AG, New York Branch, as administrative agent.	Incorporated by reference to Exhibit 10.3 to Walgreens Boots Alliance, Inc.'s Current Report on Form 8-K (File No. 001-36759) filed with the SEC on September 4, 2019.
<u>21</u>	Subsidiaries of the Registrant.	Filed herewith.
	Consent of Deloitte & Touche LLP.	Filed herewith.
	Certification of the Chief Executive Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	Filed herewith.
	Certification of the Chief Financial Officer pursuant to Section 302 of the Sarbanes-Oxley Act of 2002.	Filed herewith.
	Certification of the Chief Executive Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350.	Furnished herewith.
	Certification of the Chief Financial Officer pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, 18 U.S.C. Section 1350.	Furnished herewith.
101.INS	XBRL Instance Document	Filed herewith.
101.SCH	XBRL Taxonomy Extension Schema Document	Filed herewith.
101.CAL	XBRL Taxonomy Extension Calculation Linkbase Document	Filed herewith.
101.DEF	XBRL Taxonomy Extension Definition Linkbase Document	Filed herewith.
101.LAB	XBRL Taxonomy Extension Label Linkbase Document	Filed herewith.
101.PRE	XBRL Taxonomy Extension Presentation Linkbase Document	Filed herewith.

* Schedules and exhibits have been omitted pursuant to Item 601(b)(2) of Regulation S-K. Copies of any omitted schedule or exhibit will be furnished supplementally to the SEC upon request.

- 116 -

** Other instruments defining the rights of holders of long-term debt of the registrant and its consolidated subsidiaries may be omitted from Exhibit 4 in accordance with Item 601(b)(4)(iii)(A) of Regulation S-K. Copies of any such agreements will be furnished supplementally to the SEC upon request.

Item 16. Form 10-K summary

None.

SIGNATURES

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

WALGREENS BOOTS ALLIANCE, INC.

October 28, 2019

By: <u>/s/ James Kehoe</u> James Kehoe Executive Vice President and Global Chief Financial Officer

- 118 -

Table of Contents

Pursuant to the requirements of the Securities and Exchange Act of 1934 this report has been signed below by the following persons on behalf of the Registrant and in the capacities and on the dates indicated.

Name	Title	Date
/s/ Stefano Pessina Stefano Pessina	Executive Vice Chairman and Chief Executive Officer (Principal Executive Officer) and Director	October 28, 2019
/s/ James Kehoe James Kehoe	Executive Vice President and Global Chief Financial Officer (Principal Financial Officer)	October 28, 2019
/s/ Heather Dixon Heather Dixon	Senior Vice President, Global Controller and Chief Accounting Officer (Principal Accounting Officer)	October 28, 2019
/s/ James A. Skinner James A. Skinner	Executive Chairman	October 28, 2019
/s/ José E. Almeida José E. Almeida	Director	October 28, 2019
/s/ Janice M. Babiak Janice M. Babiak	Director	October 28, 2019
/s/ David J. Brailer David J. Brailer	Director	October 28, 2019
/s/ William C. Foote William C. Foote	Director	October 28, 2019
/s/ Ginger L. Graham Ginger L. Graham	Director	October 28, 2019
/s/ John A. Lederer John A. Lederer	Director	October 28, 2019
/s/ Dominic P. Murphy Dominic P. Murphy	Director	October 28, 2019
/s/ Nancy M. Schlichting Nancy M. Schlichting	Director	October 28, 2019

- 119 -

DESCRIPTION OF THE REGISTRANT'S SECURITIES REGISTERED PURSUANT TO SECTION 12 OF THE SECURITIES EXCHANGE ACT OF 1934

As of August 31, 2019, Walgreens Boots Alliance, Inc. ("Walgreens Boots Alliance," the "Company," "we," "our" and "us" refer solely to Walgreens Boots Alliance, Inc. and not its subsidiaries and any person that succeeds thereto) has four classes of securities registered under Section 12 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"): (1) our common stock; (2) our \notin 750,000,000 of 2.125% notes due 2026 (the "notes due 2026" or the "euro notes"), (3) £400,000,000 of 2.875% notes due 2020 (the "notes due 2020"), and (4) £300,000,000 of 3.600% notes due 2025 (the "notes due 2025," together with the notes due 2020, the "sterling notes," and the sterling notes together with the euro notes, the "notes").

Description of Common Stock

The following description of our common stock is a summary and does not purport to be complete. It is subject to and qualified in its entirety by reference to our Amended and Restated Certificate of Incorporation (the "Certificate of Incorporation") and our Amended and Restated By-Laws (the "By-Laws"), each of which are incorporated by reference as exhibits to the Annual Report on Form 10-K. We encourage you to read our Certificate of Incorporation, our By-Laws and the applicable provisions of the Delaware General Corporation Law, for additional information.

General

The authorized capital stock of Walgreens Boots Alliance consists of 3,200,000,000 shares of common stock, par value \$0.01 per share, which is registered under Section 12 of the Exchange Act. All of the shares issued and outstanding are fully paid and nonassessable. As of August 31, 2019, the total number of outstanding shares of Walgreens Boots Alliance is 895,387,502.

Dividends; Liquidation. Subject to the preferences of any outstanding shares of preferred stock, holders of common stock of Walgreens Boots Alliance have equal ratable rights to dividends (payable in cash, stock or otherwise) out of funds legally available for that purpose, when and if dividends are declared by the Board of Walgreens Boots Alliance (the "Board of Directors"). Holders of common stock are entitled to share ratably, as a single class, in all of Walgreens Boots Alliance's assets available for distribution to holders of shares of common stock upon Walgreens Boots Alliance's liquidation or dissolution or the winding up of Walgreens Boots Alliance's affairs, after payment of Walgreens Boots Alliance's liabilities and any amounts to holders of outstanding shares of preferred stock.

Voting Rights. Generally, holders of Walgreens Boots Alliance's common stock vote together as a single class on every matter acted upon by the shareholders. Holders of Walgreens Boots Alliance common stock are entitled to one vote per share on all matters submitted to a vote of shareholders, except that in all elections of directors, holders of Walgreens Boots Alliance common stock will be entitled to cumulate their votes for one or more than one candidate. A majority of the outstanding shares entitled to vote on a matter, represented in person or by proxy, will constitute a quorum at any meeting of shareholders. If a quorum is present, the affirmative vote of the majority of shares represented at the meeting and entitled to vote on a matter will be the act of the shareholders, unless the vote of a greater or different number or voting by classes is required by the DGCL, the Certificate of Incorporation or the By-Laws. Except as otherwise provided by law, or by the resolution or resolutions adopted by the Board of Directors designating the rights, powers and preferences of any series and/or class of preferred stock, the holders of Walgreens Boots Alliance common stock have the exclusive right to vote for the election of directors and for all other purposes, and holders of preferred stock are not entitled to receive notice of any meeting of stockholders at which they are not entitled to vote.

Other. The holders of Walgreens Boots Alliance common stock have no preferences or rights of conversion, exchange, pre-emption or other subscription rights. There are no redemption or sinking fund provisions applicable to the common stock. The rights and preferences of holders of common stock are subject to the rights of any series of preferred stock that Walgreens Boots Alliance may issue.

Delaware Anti-Takeover Statute

Delaware corporations may elect not to be governed by Section 203 of the DGCL, *i.e.*, Delaware's anti-takeover law. Walgreens Boots Alliance has not made this election. Delaware's anti-takeover law provides that an "interested stockholder," defined as a person who owns 15% or more of the outstanding voting stock of a corporation or a person who is an associate or affiliate of the corporation and, within the preceding three-year period, owned 15% or more of the outstanding voting stock, may not engage in specified business combinations with the corporation for a period of three years after the date on which the person became an interested stockholder. The law defines the term "business combination" to encompass a wide variety of transactions with or caused by an interested stockholder, including mergers, asset sales and transactions in which the interested stockholder receives or could receive a benefit on other than a pro rata basis with other shareholders. With the affirmative vote of the holders of a majority of the voting power of all then-outstanding shares of Walgreens Boots Alliance capital stock entitled to vote in the election of directors, voting together as a single class, Walgreens Boots Alliance may amend its Certificate of Incorporation in the future to no longer be governed by the anti-takeover law. This amendment would have the effect of allowing any person who owns at least 15% of our outstanding voting stock to pursue a takeover transaction that was not approved by the Board of Directors, the provision might discourage takeover attempts that might result in a premium over the market price for shares of Walgreens Boots Alliance's common stock.

Limitations of Director Liability and Indemnification

The Certificate of Incorporation provides that directors shall not be personally liable to the corporation or to its shareholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL.

Delaware law currently provides that this waiver may not apply to liability:

- for any breach of the director's duty of loyalty to us or our shareholders;
- for acts or omissions not in good faith or that involve intentional misconduct or a knowing violation of law;
- under Section 174 of the DGCL (governing distributions to shareholders); or
- for any transaction from which the director derived any improper personal benefit.

In the event the DGCL is amended to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of our directors will be eliminated or limited to the fullest extent permitted by the DGCL, as so amended. The bylaws of Walgreens Boots Alliance further provide that we will indemnify each of our directors and officers, trustees, fiduciaries, employees and agents to the fullest extent permitted by Delaware law.

Transfer Agent

EQ Shareowner Services serves as the transfer agent and registrar for Walgreens Boots Alliance's common stock.

Description of the Notes

The following description of our notes is a summary and does not purport to be complete. The summary is subject to and qualified in its entirety by reference to the indenture between the Walgreens Boots Alliance and Wells Fargo Bank, National Association, as trustee, dated as of November 18, 2014 and the forms of the notes due 2020, the notes due 2025 and the notes due 2026, each of which are incorporated by reference as exhibits to the Annual Report on Form 10-K. We encourage you to read the above referenced indenture for additional information.

General

The notes due 2020 were initially issued in an aggregate principal amount of £400,000,000. The notes due 2025 were

initially issued in an aggregate principal amount of £300,000,000. The notes due 2026 were initially issued in an aggregate principal amount of €750,000,000 aggregate principal amount. As of August 31, 2019, no such additional notes have been issued.

The notes do not have the benefit of any sinking fund. The notes will not be convertible or exchangeable.

The provisions of the indenture relating to defeasance and covenant defeasance as described in the indenture apply to the notes.

The euro notes were issued in minimum denominations of \pounds 100,000 and integral multiples of \pounds 1,000 in excess thereof. The sterling notes were issued in minimum denominations of \pounds 100,000 and integral multiples of \pounds 1,000 in excess thereof.

The notes due 2020, the notes due 2025 and the notes due 2026 are each traded on the Nasdaq Stock Market under the bond trading symbols of "WBA20," "WBA25," and "WBA26" respectively.

Ranking

Each series of notes are our unsecured, unsubordinated debt obligations and rank equally in right of payment with all of our other unsecured and unsubordinated debt from time to time outstanding.

Interest Payments and Maturity

The notes due 2020 bear interest at a rate of 2.875%, the notes due 2025 bear interest at a rate of 3.600% and the notes due 2026 bear interest at a rate of 2.125%, each accruing from November 20, 2014 or the most recent interest payment date to which interest has been paid or provided for.

We will pay, or cause the paying agent to pay, interest on the notes annually in arrears on November 20 of each year, in each case, to persons in whose names the notes are registered at the close of business on the preceding November 6 (whether or not a Business Day), as the case may be. We will calculate the amount of interest payable on the notes on the basis of the actual number of days in the period for which interest is being calculated and the actual number of days from and including the last date on which interest was paid on the notes, to but excluding the next scheduled interest payment date. This payment convention is referred to as ACTUAL/ACTUAL (ICMA) as defined in the rulebook of the International Capital Market Association. If the date on which a payment of interest or principal on the notes is scheduled to be paid is not a Business Day, then that interest or principal will be paid on the next succeeding Business Day but no further interest will be paid in respect of the delay in such payment.

"Business Day" means any Monday, Tuesday, Wednesday, Thursday or Friday which is not a day when banking institutions are authorized or obligated by law or executive order to be closed in New York City or London and, for any place of payment outside of New York City or London, in such place of payment, and on which the Trans-European Automated Real-time Gross Settlement Express Transfer system (the TARGET2 system), or any successor thereto, operates.

Payments of principal, interest and Additional Amounts (as defined below), if any, in respect of (i) the euro notes are payable in euro and (ii) the sterling notes, are payable in sterling. If euro or sterling, as the case may be, is unavailable to us due to the imposition of exchange controls or other circumstances beyond our control (including the dissolution of the euro) or if the euro is no longer being used by the then member states of the European Monetary Union that have adopted the euro as their currency or for the settlement of transactions by public institutions of or within the international banking community, then all payments in respect of the applicable notes will be made in U.S. dollars until the euro or sterling, as applicable, is again available to us or so used. The amount payable on any date in euro or sterling, as applicable, will be converted into U.S. dollars at the rate mandated by the U.S. Federal Reserve Board as of the close of business on the second business day prior to the relevant payment date or, in the event the U.S. Federal Reserve Board has not mandated a rate of conversion, on the basis of the then most recent U.S. dollar/euro exchange rate or U.S. dollar/sterling exchange rate, as applicable, available on or prior to the second business day prior to the relevant payment date as determined by us in our sole discretion. Any payment in respect of the notes so made in U.S. dollars will not constitute an event of default under the notes or the indenture governing the notes. Neither the trustee nor the paying agent shall have any responsibility for any calculation or conversion in connection with the foregoing.

The notes will cease to bear interest upon maturity unless, upon due presentation, payment of the amount due is

improperly withheld or refused, in which case the notes will continue to bear interest (before as well as after judgment) until the day on which all sums due in respect of such notes up to that day are received by or on behalf of the relevant holder of such notes.

Investors are subject to foreign exchange risks as to payments of principal, interest and Additional Amounts, if any, that may have important economic and tax consequences to them.

Optional Redemption

We may redeem (i) the notes due 2020, at any time prior to October 20, 2020 (one month prior to the maturity date of the notes due 2020) in whole or from time to time prior to October 20, 2020 in part, (ii) the notes due 2025, at any time prior to August 20, 2025 (three months prior to the maturity date of the notes due 2025) in part and (iii) the notes due 2026, at any time prior to August 20, 2026 (three months prior to the maturity date of the notes due 2026) in whole or from time to time prior to August 20, 2026 in part, in each case, at our option at a redemption price equal to the greater of (the "Applicable Premium"):

(1) 100% of the principal amount of the notes to be redeemed; or

(2) the sum of the present values of the remaining scheduled payments of principal and interest thereon (not including any portion of such payments of interest accrued as of the redemption date), discounted to the redemption date on an annual basis (ACTUAL/ACTUAL (ICMA)) at the applicable Comparable Government Bond Rate, plus 20 basis points for the euro notes, plus 20 basis points for the notes due 2020 and plus 20 basis points for the notes due 2025, plus, in every case, accrued and unpaid interest on the notes to be redeemed to, but excluding, the redemption date.

In addition, at any time on or after October 20, 2020 (one month prior to the maturity date of the notes due 2020), with respect to the notes due 2020, or August 20, 2025 (three months prior to the maturity date of the notes due 2025), with respect to the notes due 2025, or August 20, 2026 (three months prior to the maturity date of the notes due 2026, we may redeem some or all of the applicable series of notes at our option, at a redemption price equal to 100% of the principal amount of the applicable notes to be redeemed, plus, in every case, accrued and unpaid interest on the notes to be redeemed to, but excluding, the redemption date.

In any case, the principal amount of a note remaining outstanding after a redemption in part shall be \pounds 100,000 or \pounds 100,000 or an integral multiple of \pounds 1,000 or \pounds 1,000 in excess thereof.

Further, installments of interest on notes to be redeemed that are due and payable on interest payment dates falling on or prior to a redemption date will be payable on the applicable interest payment date to the registered holders as of the close of business on the relevant record date according to such notes and the indenture.

For purposes of the optional redemption provisions of the notes, the following terms are applicable:

"Comparable Government Bond" means, in relation to any Comparable Government Bond Rate calculation, at the discretion of an independent investment bank selected by us, (i) with respect to any series of euro notes, a German federal government bond whose maturity is closest to the maturity of the euro notes to be redeemed, or if such independent investment bank in its discretion determines that such similar bond is not in issue, such other German federal government bonds selected by us, determine to be appropriate for determining the Comparable Government Bond Rate and (ii) with respect to the sterling notes, the United Kingdom government security or securities whose maturity is closest to the maturity of the sterling notes to be redeemed, or if such independent investment bank in its discretion determines that such other United Kingdom government security or securities as such independent investment bank may, with the advice of three brokers of, and/or market makers in, German security or securities as such independent investment bank may, with the advice of three brokers of, and/or united Kingdom government security or securities used independent investment bank may, with the advice of three brokers of, and/or market makers in, United Kingdom government securities selected by us, determine to be appropriate for determines that such similar bond is not in issue, such other United Kingdom government securities selected by us, determine to be appropriate for determines that such similar bond is not in issue, such other United Kingdom government securities selected by us, determine to be appropriate for determines that such similar bond is not in issue, such other United Kingdom government securities selected by us, determine to be appropriate for determining the Comparable Government Bond Rate.

"Comparable Government Bond Rate" means, with respect to any redemption date, the price, expressed as a percentage (rounded to three decimal places, with 0.0005 being rounded upwards), at which the gross redemption yield on the notes to be redeemed, if they were to be purchased at such price on the third business day prior to the redemption date, would be equal to the gross redemption yield on such Business Day of the Comparable Government Bond (as defined above) on the basis of the middle market price of the Comparable Government Bond prevailing at 11:00 a.m. (London time) on such Business Day as determined by an independent investment bank selected by us.

Notice of any redemption will be mailed, or delivered electronically if held by any depositary in accordance with such depositary's customary procedures, at least 30 days but not more than 60 days before the redemption date to each registered holder of the notes to be redeemed. Unless we default in payment of the redemption price, on and after the redemption date, interest will cease to accrue on the notes or portions thereof called for redemption. If less than all of the notes of any series are to be redeemed, the notes to be redeemed shall be selected by the securities registrar in accordance with applicable procedures of Clearstream or Euroclear.

The notes are also subject to redemption if certain events occur involving United States taxation.

Additional Amounts

All payments of principal and interest in respect of the notes by us or a paying agent on our behalf will be made free and clear of, and without deduction or withholding for or on account of, any present or future taxes, duties, assessments or other similar governmental charges imposed or levied by the United States or any political subdivision or taxing authority of or in the United States (collectively, "Taxes"), unless such withholding or deduction is required by law.

In the event such withholding or deduction for Taxes is required by law, subject to the limitations described below, we will pay to any beneficial owner of a note that is neither a U.S. Holder (as defined below) nor a partnership for U.S. federal income tax purposes such additional amounts ("Additional Amounts") as may be necessary to ensure that the net amount received by such person, after withholding or deduction for such Taxes, will be equal to the amount such person would have received in the absence of such withholding or deduction.

However, no Additional Amounts shall be payable with respect to any Taxes if such Taxes are imposed or levied for reasons unrelated to the holder's or beneficial owner's ownership or disposition of notes, nor shall Additional Amounts be payable for or on account of:

(a) any Taxes which would not have been so imposed, withheld or deducted but for:

(1) the existence of any present or former connection between the holder or beneficial owner (or between a fiduciary, settlor, beneficiary, member or shareholder or other equity owner of, or a person having a power over, such holder or beneficial owner, if such holder or beneficial owner is an estate, a trust, a limited liability company, a partnership, a corporation or other entity) and the United States, including, without limitation, such holder or beneficial owner (or such fiduciary, settlor, beneficiary, member, shareholder or other equity owner or person having such a power) being or having been a citizen or resident or treated as a resident of the United States, being or having been engaged in a trade or business in the United States, being or having been present in the United States, or having or having had a permanent establishment in the United States;

(2) the failure of the holder or beneficial owner to comply with any applicable certification, information, documentation or other reporting requirement, if compliance is required under the tax laws and regulations of the United States or any political subdivision or taxing authority of or in the United States to establish entitlement to a partial or complete exemption from such Taxes (including, but not limited to, the requirement to provide Internal Revenue Service Form W-8BEN,

Form W-8BEN-E, Form W-8ECI, or any subsequent versions thereof or successor thereto); or

(3) the holder's or beneficial owner's present or former status as a personal holding company or a foreign personal holding company with respect to the United States, as a controlled foreign corporation with respect to the United States, as a passive foreign investment company with respect to the United States, as a foreign tax exempt organization with respect to the United States or as a corporation that accumulates earnings to avoid United States federal income tax;

(b) any Taxes which would not have been imposed, withheld or deducted but for the failure of the holder or beneficial owner to meet the requirements (including the certification requirements) of Section 871(h) or Section 881(c) of the Internal Revenue Code of 1986, as amended (the "Code");

(c) any Taxes which would not have been imposed, withheld or deducted but for the presentation by the holder or beneficial owner of such note for payment on a date more than 30 days after the date on which such payment became due and payable or the date on which payment of the note is duly provided for and notice is given to holders, whichever occurs later, except to the extent that the holder or beneficial owner would have been entitled to such Additional Amounts on presenting such note on any date during such 30-day period;

(d) any estate, inheritance, gift, sales, excise, transfer, personal property, wealth or similar Taxes;

(e) any Taxes which are payable otherwise than by withholding or deduction from a payment on such note;

(f) any Taxes which are imposed, withheld or deducted with respect to, or payable by, a holder that is not the beneficial owner of the note, or a portion of the note, or that is a fiduciary, partnership, limited liability company or other similar entity, but only to the extent that a beneficial owner, a beneficiary or settlor with respect to such fiduciary or member of such partnership, limited liability company or similar entity would not have been entitled to the payment of an Additional Amount had such beneficial owner, settlor, beneficiary or member received directly its beneficial or distributive share of the payment;

(g) any Taxes required to be withheld or deducted by any paying agent from any payment on any note, if such payment can be made without such withholding or deduction by at least one other paying agent;

(h) any Taxes required to be withheld or deducted where such withholding or deduction is imposed pursuant to European Council Directive 2003/48/EC on the taxation of savings income, or any law implementing or complying with, or introduced in order to conform to, such European Council Directive;

(i) any Taxes imposed, withheld or deducted under Sections 1471 through 1474 of the Code (or any amended or successor provisions), any current or future regulations or official interpretations thereof, any agreement entered into pursuant to Section 1471(b) of the Code or any fiscal or regulatory legislation, rules or practices adopted pursuant to any intergovernmental agreement entered into in connection with the implementation of such sections of the Code;

(j) any Taxes that would not have been imposed, withheld or deducted but for a change in any law, treaty, regulation, or administrative or judicial interpretation that becomes effective more than 15 days after the applicable payment becomes due or is duly provided for, whichever occurs later; or

(k) any combination of items (a), (b), (c), (d), (e), (f), (g), (h), (i) and (j).

Any Additional Amounts paid on the euro notes will be paid in euro and any Additional Amounts paid on the sterling notes will be paid in sterling.

For purposes of this section, the acquisition, ownership, enforcement, or holding of or the receipt of any payment with respect to a note will not constitute a connection (1) between the holder or beneficial owner and the United States or (2) between a fiduciary, settlor, beneficiary, member or shareholder or other equity owner of, or a person having a power over, such holder or beneficial owner if such holder or beneficial owner is an estate, a trust, a limited liability company, a partnership, a corporation or other entity and the United States.

Except as specifically provided under this section "Additional Amounts," we will not be required to make any payment with respect to any tax, duty, assessment or other governmental charge imposed by any government or any political subdivision or taxing authority.

If we are required to pay Additional Amounts with respect to the notes, we will notify the trustee and paying agent pursuant to an officers' certificate that specifies the Additional Amounts payable and when the Additional Amounts are payable. If the trustee and the paying agent do not receive such an officers' certificate from us, the trustee and paying agent may rely on the absence of such an officers' certificate in assuming that no such Additional Amounts are payable.

In addition, we undertake that, to the extent permitted by law, we will maintain a paying agent that will not require withholding or deduction of tax pursuant to European Council Directive 2003/48/EC on the taxation of savings income or any law implementing or complying with, or introduced in order to conform to, such European Council Directive.

- As used herein, a "U.S. Holder" means a beneficial owner of a note that is, for U.S. federal income tax purposes:
- an individual who is a citizen or resident of the United States;
- a corporation created or organized in or under the laws of the United States, any state within the United States, or the District of Columbia;
- an estate the income of which is subject to U.S. federal income tax regardless of its source; or
- a trust if (i) a court within the United States is able to exercise primary supervision over the administration of the trust and one or more U.S. persons have the authority to control all substantial decisions of the trust, or (ii) the trust validly elected to be treated as a U.S. person under applicable Treasury regulations.

Redemption for Tax Reasons

We may redeem each series of the notes at our option, in whole but not in part, at a redemption price equal to 100% of the principal amount of the notes to be redeemed, together with any accrued and unpaid interest on the notes to be redeemed to, but excluding, the redemption date, at any time, if:

(i) we have or will become obliged to pay Additional Amounts with respect to such series of notes as a result of any change in, or amendment to, the laws, regulations, treaties, or rulings of the United States or any political subdivision of or in the United States or any taxing authority thereof or therein affecting taxation, or any change in, or amendment to, the application, official interpretation, administration or enforcement of such laws, regulations, treaties or rulings (including a holding by a court of competent jurisdiction in the United States), which change or amendment is enacted, adopted, announced or becomes effective on or after November 10, 2014; or

(ii) on or after November 10, 2014, any action is taken by a taxing authority of, or any action has been brought in a court of competent jurisdiction in, the United States or any political subdivision of or in the United States or any taxing authority thereof or therein, including any of those actions specified in clause (i) above, whether or not such action was taken or brought with respect to us, or there is any change, amendment, clarification, application or interpretation of such laws, regulations, treaties or rulings, which in any such case, will result in a material probability that we will be required to pay Additional Amounts with respect to such notes (it being understood that such material probability will be deemed to result if the written opinion of independent tax counsel described in clause (b) below to such effect is delivered to the trustee and the paying agent).

Notice of any redemption will be mailed, or delivered electronically if held by any depositary in accordance with such depositary's customary procedures, at least 30 days but not more than 60 days before the redemption date to each registered holder of the notes to be redeemed; provided, however, that the notice of redemption shall not be given earlier than 90 days before the earliest date on which we would be obligated to pay such Additional Amounts if a payment in respect of the notes was then due.

Prior to the mailing or delivery of any notice of redemption pursuant to this section, in case of a redemption for the reasons specified in clause (i) or (ii) above, we will deliver to the trustee and the paying agent:

(a) a certificate signed by one of our officers stating that we are entitled to effect such redemption and setting forth a statement of facts showing that the conditions precedent to our rights to so redeem have occurred, and

(b) a written opinion of independent tax counsel of nationally recognized standing to the effect that we have or will become obligated to pay such Additional Amounts as a result of such change or amendment or that there is a material probability that we will be required to pay Additional Amounts as a result of such action, change, amendment, clarification, application or interpretation, as the case may be.

Such notice, once delivered by us will be irrevocable.

Change of Control

If a change of control triggering event occurs with respect to the notes, unless we have exercised our option to redeem the notes as described above or have defeased the notes as described in the indenture, we will be required to make an offer (a "change of control offer") to each holder of the notes to repurchase all or any part (equal, in respect of the euro notes, to \pounds 100,000 or an integral multiple of \pounds 1,000 in excess thereof) of that holder's notes on the terms set forth in such notes. In a change of control offer, we will be required to offer payment in cash equal to 101% of the aggregate principal amount of notes repurchased, plus accrued and unpaid interest, if any, on the notes repurchased to, but excluding, the date of repurchase (a "change of control payment"). Within 30 days following any change of control triggering event or, at our option, prior to any change of control, but after public announcement of the transaction that constitutes or may constitute the change of control triggering event and offering to repurchase the notes on the date specified in the applicable notice, which date will be no earlier than 30 days and no later than 60 days from the date such notice is mailed (or delivered electronically) (a "change of control payment date"). The notice will, if mailed (or delivered electronically) prior to the date of consummation of the change of control, state that the change of control offer is conditioned on the change of control triggering event occurring on or prior to the applicable change of control, state that the change of control offer is conditioned on the change of control payment date.

On each change of control payment date, we will, to the extent lawful:

• accept for payment all notes or portions of notes properly tendered pursuant to the applicable change of control

offer;

•

- deposit with the paying agent an amount equal to the change of control payment in respect of all notes or portions of notes properly tendered; and
- deliver or cause to be delivered to the trustee the notes properly accepted together with an officers' certificate (with a copy to the paying agent) stating the aggregate principal amount of notes or portions of notes being repurchased.

We will not be required to make a change of control offer upon the occurrence of a change of control triggering event if a third party makes such an offer in the manner, at the times and otherwise in compliance with the requirements for an offer made by us and such third party purchases all notes properly tendered and not withdrawn under its offer. In addition, we will not repurchase any notes if there has occurred and is continuing on the change of control payment date an event of default under the indenture, other than a default in the payment of the change of control payment upon a change of control triggering event.

We will be required to comply with the requirements of Rule 14e-1 under the Exchange Act, and any other securities laws and regulations thereunder to the extent those laws and regulations are applicable in connection with the repurchase of the notes as a result of a change of control triggering event. To the extent that the provisions of any securities laws or regulations conflict with the change of control offer provisions of the notes, we will be required to comply with such securities laws and regulations and will not be deemed to have breached our obligations under the change of control offer provisions of the notes by virtue of any such conflict and compliance.

For purposes of the change of control offer provisions of the notes, the following terms are applicable:

"Board of Directors" means our board of directors or any authorized committee thereof.

"Change of control" means the occurrence of any of the following: (1) the direct or indirect sale, lease, transfer, conveyance or other disposition (other than by way of merger or consolidation), in one or more series of related transactions, of all or substantially all of our assets and the assets of our subsidiaries, taken as a whole, to any person, other than us or one of our subsidiaries; (2) the consummation of any transaction (including, without limitation, any merger or consolidation) the result of which is that any person becomes the beneficial owner (as defined in Rules 13d-3 and 13d-5 under the Exchange Act), directly or indirectly, of more than 50% of our outstanding voting stock or other voting stock into which our voting stock is reclassified, consolidated, exchanged or changed, measured by voting power rather than number of shares; (3) we consolidate with, or merge with or into, any person, or any person consolidates with, or merges with or into, us, in any such event pursuant to a transaction in which any of our outstanding voting stock or the voting stock of such other person is converted into or exchanged for cash, securities or other property, other than any such transaction where the shares of our voting stock outstanding immediately prior to such transaction constitute, or are converted into or exchanged for, a majority of the voting stock of the surviving person or any direct or indirect parent company of the surviving person immediately after giving effect to such transaction, measured by voting power rather than number of shares; (4) the first day on which a majority of the members of our Board of Directors are not continuing directors; or (5) the adoption of a plan relating to our liquidation or dissolution. Notwithstanding the foregoing, (i) the reorganization (and each transaction in connection therewith or related thereto) shall not constitute a change of control, (ii) the Walgreens Merger (and each transaction in connection therewith or related thereto) shall not constitute a change of control and (iii) a transaction will not be deemed to involve a change of control under clause (2) above if (1) we become a direct or indirect wholly owned subsidiary of a holding company and (2)(A) the direct or indirect holders of the voting stock of such holding company immediately following that transaction are substantially the same as the holders of our voting stock immediately prior to that transaction or (B) immediately following that transaction no person (other than a holding company satisfying the requirements of this sentence) is the beneficial owner, directly or indirectly, of more than 50% of the voting stock of such holding company. The term "person," as used in this definition, has the meaning given thereto in Section 13(d)(3) of the Exchange Act.

The definition of change of control includes a phrase relating to the direct or indirect sale, lease, transfer, conveyance or other disposition of "all or substantially all" of our assets and those of our subsidiaries taken as a whole. Although there is a limited body of case law interpreting the phrase "substantially all" there is no precise established definition of the phrase under applicable law.

Accordingly, the ability of a holder of notes to require us to repurchase its notes as a result of a sale, lease, transfer, conveyance or other disposition of less than all of our assets and of those of our subsidiaries taken as a whole to another person or group may be uncertain.

"Change of control triggering event" means the occurrence of both a change of control and a rating event.

"Continuing directors" means, as of any date of determination, any member of our Board of Directors who (1) was a member of such Board of Directors on the date the notes were issued or

(2) was nominated for election, elected or appointed to such Board of Directors with the approval of a majority of the continuing directors who were members of such Board of Directors at the time of such nomination, election or appointment (either by a specific vote or by approval of our proxy statement in which such member was named as a nominee for election as a director, without objection to such nomination).

"Investment grade rating" means a rating equal to or higher than Baa3 (or the equivalent) by Moody's and BBB- (or the equivalent) by S&P, and the equivalent investment grade credit rating from any replacement rating agency or rating agencies selected by us.

"Moody's" means Moody's Investors Service, Inc. and its successors.

"Rating agencies" means (1) each of Moody's and S&P; and (2) if either of Moody's or S&P ceases to rate the notes or fails to make a rating of the notes publicly available for reasons outside of our control, a "nationally recognized statistical rating organization" as defined under Section 3(a)(62) of the Exchange Act selected by us (as certified by a resolution of our Board of Directors) as a replacement agency for Moody's or S&P, or both of them, as the case may be.

"Rating event" means the rating on the notes is lowered by both rating agencies and the notes are rated below an investment grade rating by both rating agencies, in any case on any day during the period (which period will be extended so long as the rating of the notes is under publicly announced consideration for a possible downgrade by any of the rating agencies) commencing upon the first public notice of the occurrence of a change of control or our intention to effect a change of control and ending 60 days following the consummation of the change of control.

"Reorganization" means the reorganization of Walgreen Co. into a holding company structure under which Ontario Merger Sub, Inc., a direct wholly owned subsidiary of Walgreens Boots Alliance, Inc. will merge with and into Walgreen Co. (subject to the satisfaction or waiver of specified closing conditions) and Walgreen Co. will survive such merger as a direct wholly owned subsidiary of Walgreens Boots Alliance, Inc.

"S&P" means Standard & Poor's Rating Services, a division of The McGraw-Hill Companies, Inc., and its successors.

"Second Step Transaction" means the acquisition by Walgreen Co., Walgreens Boots Alliance, Inc. or any of their respective consolidated subsidiaries of the remaining 55% of the issued and outstanding share capital of Alliance Boots GmbH in exchange for £3.133 billion in cash, payable in British pounds sterling, and 144,333,468 shares of Walgreen Co. common stock (or, if the Reorganization is consummated, 144,333,468 shares of Walgreens Boots Alliance Inc.'s common stock rather than Walgreen Co. common stock), subject to certain specified adjustments.

"Voting stock" means, with respect to any specified "person" (as that term is used in

Section 13(d)(3) of the Exchange Act) as of any date, the capital stock of such person that is at the time entitled to vote generally in the election of the board of directors of such person.

"Walgreens Merger" is the merger of Walgreens Boots Alliance, Inc. into Walgreen Co., with Walgreen Co. surviving such merger, if after the consummation of the Second Step Transaction, the Reorganization is not consummated on or prior to the date of the consummation of the Second Step Transaction.

Certain Covenants

Limitation on Liens

We agree that we will not, and will not permit any Restricted Subsidiary (as defined below) to, create, incur, issue, assume or guarantee any indebtedness for borrowed money ("Debt"), secured by a Mortgage (as defined below) upon any Operating Property (as defined below) owned by, or leased to, us or any of our Restricted Subsidiaries, or upon shares of capital stock or Debt issued by any Restricted Subsidiary and owned by us or any Restricted Subsidiary, at the issue date of each applicable series of outstanding debt securities or thereafter acquired, without effectively providing concurrently that such outstanding debt securities authenticated and delivered under the indenture (together with, if we so determine, any other Debt of ours or any Restricted Subsidiary then existing or thereafter created which is not subordinate in right of payment to such outstanding debt securities) are secured equally and ratably with, or at our option, prior to such Debt so long as such Debt is so

secured. The foregoing restrictions will not apply to Debt secured by the following, and the Debt so secured will be excluded from any computation under the next succeeding paragraph below:

1. Mortgages on property existing at the time of the acquisition thereof;

2. Mortgages on property of a corporation or other entity existing at the time such corporation or other entity is merged into or consolidated with us or a Restricted Subsidiary or at the time of a sale, lease or other disposition of the properties of such corporation or other entity (or a division of such corporation 7 or other entity) as an entirety or substantially as an entirety to us or a Restricted Subsidiary, provided that any such Mortgage does not extend to any property owned by us or any Restricted Subsidiary immediately prior to such merger, consolidation, sale, lease or disposition;

3. Mortgages on property of a corporation or other entity existing at the time such corporation or other entity becomes a Restricted Subsidiary;

4. Mortgages in favor of us or a Restricted Subsidiary;

5. Mortgages to secure all or part of the cost of acquisition, construction, development or improvement of the underlying property, or to secure Debt incurred to provide funds for any such purpose, provided that the commitment of the creditor to extend the credit secured by any such Mortgage shall have been obtained not later than 365 days after the later of (a) the completion of the acquisition, construction, development or improvement of such property or (b) the placing in operation of such property;

6. Mortgages in favor of the United States or any state thereof, or any department, agency or instrumentality or political subdivision of the United States or any state thereof, or in favor of any other country, or any department, agency or instrumentality or any political subdivision thereof, to secure partial, progress, advance or other payments; and

7. Mortgages existing on the issue date of the applicable series of outstanding debt securities or any extension, renewal, replacement or refunding of any Debt secured by a Mortgage existing on the issue date of the applicable series of outstanding debt securities or referred to in clauses (1) to (3) or (5), provided that the principal amount of Debt secured thereby and not otherwise authorized by clauses (1) to (3) or (5) shall not exceed the principal amount of Debt, plus any premium or fee payable in connection with any such extension, renewal, replacement or refunding, so secured at the time of such extension, renewal, replacement or refunding.

Notwithstanding the restrictions described above, we and our Restricted Subsidiaries may create, incur, issue, assume or guarantee Debt secured by Mortgages without equally and ratably securing the outstanding debt securities authenticated and delivered under the indenture if, at the time of such creation, incurrence, issuance, assumption or guarantee, after giving effect thereto and to the retirement of any Debt which is concurrently being retired, the aggregate amount of all such Debt secured by Mortgages (other than (i) any Debt secured by Mortgages permitted as described in clauses (1) through (7) of the immediately preceding paragraph and (ii) any Debt secured in compliance with the first paragraph of this covenant) that would otherwise be subject to these restrictions, together with all Attributable Debt (as defined below) with respect to Sale and Leaseback Transactions (as defined below) (other than with respect to certain Sale and Leaseback Transactions that are permitted as described in the second full paragraph under the caption "-Limitation on Sale and Leaseback Transactions" below) does not exceed 15% of Consolidated Net Tangible Assets (as defined below).

"Board of Directors" means our board of directors or any authorized committee thereof.

"Consolidated Net Tangible Assets" means, at any date, the total amount, as shown on or reflected in our (or, if applicable at such date, our predecessor's) most recent consolidated balance sheet as at the end of our fiscal quarter ending not more than 135 days prior to such date, of all assets of the Company and our consolidated subsidiaries on a consolidated basis in accordance with United States generally accepted accounting principles (giving pro forma effect to any acquisition or disposition of assets of the Company or any of our subsidiaries with fair value in excess of \$100,000,000 that has occurred since the end of such fiscal quarter as if such acquisition or disposition had occurred on the last day of such fiscal quarter), less (i) all current liabilities (due within one year) as shown on such balance sheet, except for current maturities of long-term debt and of obligations under capital leases, (ii) investments in and advances to Unrestricted Subsidiaries and (iii) Intangible Assets.

"Domestic Subsidiary" means any Subsidiary of ours that is not a Foreign Subsidiary.

"Foreign Subsidiary" means any Subsidiary of ours that is not organized under the laws of the United States or any jurisdiction within the United States and any direct or indirect Subsidiary thereof.

"Intangible Assets" means, at any date, the value, as shown on or reflected in our (or, if applicable at such date, our predecessor's) most recent consolidated balance sheet as at the end of our fiscal quarter ending not more than 135 days prior to such date, of all trade names, trademarks, licenses, patents, copyrights, service marks, goodwill and other like intangibles of the Company and our consolidated subsidiaries on a consolidated basis in accordance with United States generally accepted accounting principles (and giving pro forma effect to any acquisition or disposition of assets of the Company or any of our subsidiaries with fair value in excess of \$100,000,000 that has occurred since the end of such fiscal quarter as if such acquisition or disposition had occurred on the last day of such fiscal quarter).

"Mortgage" means, with respect to any property or assets, any mortgage, deed of trust, pledge, hypothecation, assignment, security interest, lien, encumbrance, or other security arrangement of any kind or nature whatsoever on or with respect to such property or assets (including any conditional sale or other title retention agreement having substantially the same economic effect as any of the foregoing).

"Operating Property" means any real property or equipment located within the United States and owned by, or leased to, us or any of our Subsidiaries that has a net book value (after deduction of accumulated depreciation) in excess of 1.0% of Consolidated Net Tangible Assets.

"Restricted Subsidiary" means any Domestic Subsidiary other than an Unrestricted Subsidiary; provided, however, the Board of Directors of the Company may declare any such Unrestricted Subsidiary to be a Restricted Subsidiary effective as of the date such resolution is adopted.

"Subsidiary" means any corporation or other entity of which at least a majority of the outstanding capital stock or other equity interests having by the terms thereof ordinary voting power to elect a majority of the directors, managers or trustees of such corporation or other entity, irrespective of whether or not at the time capital stock or other equity securities of any other class or classes of such corporation or other entity shall have or might have voting power by reason of the happening of any contingency, is at the time, directly or indirectly, owned or controlled by us or by one or more of our Subsidiaries, or by us and one or more of our Subsidiaries.

"Unrestricted Subsidiary" means any Domestic Subsidiary designated as an Unrestricted Subsidiary from time to time by our Board of Directors; provided, however, that our Board of Directors (i) will not designate as an Unrestricted Subsidiary any of our Domestic Subsidiaries that owns any Operating Property or any capital stock of a Restricted Subsidiary, (ii) will not continue the designation of any of our Domestic Subsidiaries as an Unrestricted Subsidiary at any time that such Domestic Subsidiary owns any Operating Property, and (iii) will not, nor will it cause or permit any Restricted Subsidiary to, transfer or otherwise dispose of any Operating Property to any Unrestricted Subsidiary (unless such Unrestricted Subsidiary will in connection therewith be redesignated as a Restricted Subsidiary and any pledge, mortgage, security interest or other lien arising in connection with any Debt of such Unrestricted Subsidiary so redesignated does not extend to such Operating Property (unless the existence of such pledge, mortgage, security interest or other lien would otherwise be permitted under the indenture)).

Limitation on Sale and Leaseback Transactions

We agree that we will not, and will not permit any Restricted Subsidiary to, enter into any arrangement with any person providing for the leasing by us or any Restricted Subsidiary of any Operating Property that has been or is to be sold or transferred by us or such Restricted Subsidiary to such person with the intention of taking back a lease of such property (a "Sale and Leaseback Transaction"), unless the terms of such sale or transfer have been determined by our Board of Directors to be fair and arm's-length and either:

within 180 days after the receipt of the proceeds of the sale or transfer, we or any Restricted Subsidiary apply an amount equal to the greater of the net proceeds of the sale or transfer or the fair value of such Operating Property at the time of such sale or transfer to either (or a combination of) (i) the prepayment 9 or retirement (other than any mandatory prepayment or retirement of unsecured Debt) of Senior Funded Debt (as defined below) or (ii) the purchase, construction or development of other comparable property; or

we or such Restricted Subsidiary would be entitled, at the effective date of the sale or transfer, to incur Debt secured by a Mortgage on such Operating Property, in an amount at least equal to the Attributable Debt in respect of the Sale and Leaseback Transaction, without equally and ratably securing the debt securities pursuant to the covenant described under "-Limitation on Liens" above. The foregoing restriction in the paragraph above will not apply to any Sale and Leaseback Transaction (i) for a term of not more than three years including renewals or (ii) between us and a Restricted Subsidiary or between Restricted Subsidiaries, provided that the lessor is us or a wholly owned Restricted Subsidiary.

"Attributable Debt" in respect of a Sale and Leaseback Transaction means, at the time of determination, the amount of future minimum operating lease payments required to be disclosed by United States generally accepted accounting principles, less any amounts required to be paid on account of maintenance and repairs, insurance, taxes, assessments, water rates and similar charges, discounted using the methodology used to calculate the present value of operating lease payments in our (or, if applicable at such date, our predecessor's) most recent Annual Report on Form 10-K preceding the date of determination reflecting that calculation.

"Funded Debt" means Debt which matures more than one year from the date of creation, or which is extendable or renewable at the sole option of the obligor so that it may become payable more than one year from such date or which is classified, in accordance with United States generally accepted accounting principles, as long-term debt on the consolidated balance sheet for the most-recently ended fiscal quarter (or if incurred subsequent to the date of such balance sheet, would have been so classified) of the person for which the determination is being made. Funded Debt does not include (1) obligations created pursuant to leases, (2) any Debt or portion thereof maturing by its terms within one year from the time of any computation of the amount of outstanding Funded Debt unless such debt shall be extendable or renewable at the sole option of the obligor in such manner that it may become payable more than one year from such time, or (3) any Debt for which money in the amount necessary for the payment or redemption of such Debt is deposited in trust either at or before the maturity date thereof.

"Senior Funded Debt" means all Funded Debt of ours or any person (except Funded Debt, the payment of which is subordinated to the payment of the debt securities authenticated and delivered under the indenture).

Merger, Consolidation or Sale of Assets

We covenant not to (1) consolidate or amalgamate with or merge into any other person (whether or not affiliated with us) or convey, transfer or lease our properties and assets as an entirety or substantially as an entirety to any other person (whether or not affiliated with us) or (2) permit any other person (whether or not affiliated with us) to consolidate or amalgamate with or merge into us, or convey, transfer or lease its properties and assets as an entirety or substantially as an entirety to us, unless (a) in the case of (1) above, the person formed by such consolidation or amalgamation or into which we are merged or the person which acquires by conveyance or transfer, or which leases, our properties and assets as an entirety or substantially as an entirety is a person organized and existing under the laws of the United States, any state thereof or the District of Columbia, and shall expressly assume, by supplemental indenture satisfactory in form to the trustee, executed by the successor person and delivered to the trustee, the due and punctual payment of the principal of, and premium, if any, and interest on, and additional amounts, if any, with respect to all of the debt securities authenticated and delivered under the indenture, and the performance of our obligations under the indenture and the outstanding debt securities authenticated and delivered thereunder and shall provide for conversion or exchange rights in accordance with the provisions of the debt securities authenticated and delivered under the indenture of any series that are convertible or exchangeable into common stock or other securities; (b) immediately after giving 10 effect to such transaction and treating any indebtedness which becomes an obligation of ours or a Subsidiary as a result of such transaction as having been incurred by us or such Subsidiary at the time of such transaction, no event of default, and no event which, after notice or lapse of time, or both, would become an event of default, has occurred and is continuing; and (c) we or the successor person have delivered to the trustee an officers' certificate and an opinion of counsel, each satisfactory to the trustee and stating that such transaction and, if a supplemental indenture is required in connection with such transaction, such supplemental indenture, comply with this covenant and that all conditions precedent in the indenture provided for relating to such transaction have been complied with.

Notwithstanding the foregoing, any conveyance, transfer or lease of assets between or among the Company, Walgreens Co. and their respective subsidiaries shall not be prohibited under the indenture.

Defeasance; Satisfaction and Discharge

The notes are subject to defeasance and discharge, as set forth in the indenture, provided, that (i) upon any redemption that requires the payment of the Applicable Premium, the amount deposited shall be sufficient for purposes of the indenture to the extent that an amount is deposited with the trustee or the paying agent, as applicable, equal to the Applicable Premium calculated as of the date of the notice of redemption (and calculated as though the redemption date were the date of such notice of redemption), with any deficit as of the redemption date only required to be deposited with the trustee or the paying agent, as applicable, on or prior to the redemption date and (ii) any reference to "Government Obligations" in respect of the

euro notes shall refer to "Federal Republic of Germany Obligations" and "Federal Republic of Germany Obligations" shall mean (1) direct obligations of the Federal Republic of Germany, where the payment or payment thereunder are supported by the full faith and credit of the Federal Republic of Germany or (2) obligations of a person controlled or supervised by and acting as an agency or instrumentality of the Federal Republic of Germany, where the timely payment or payments thereunder are unconditionally guaranteed as a full faith and credit obligation by the Federal Republic of Germany, which, in either case under clauses (1) or (2) are not callable or redeemable at the option of the issuer thereof, and shall also include a depositary receipt issued by a bank or trust company as custodian with respect to any such Federal Republic of Germany Obligations or a specific payment of interest on or principal of or other amount with respect to any such Federal Republic of Germany Obligations held by such custodian for the account of the holder of a depositary receipt, provided that (except as required by law) such custodian is not authorized to make any deduction from the amount payable to the holder of such depositary receipt from any amount received by the custodian in respect of the Federal Republic of Germany Obligations or the specific payment of interest on or principal of or other amount with respect to the Federal Republic of Germany Obligations evidenced by such depositary receipt.

Events of Default

Each of the following events will constitute an event of default under the indenture with respect to the notes issued:

- default in the payment of any interest on any debt security of such series, or any additional amounts payable with respect thereto, when interest or additional amounts become due and payable, and continuance of such default for a period of 30 days;
- default in the payment of the principal of or any premium on any debt security of such series, or any additional amounts payable with respect thereto, when such principal, premium or such additional amounts become due and payable at their maturity, upon any redemption, upon declaration of acceleration or otherwise;
- default in the deposit of any sinking fund payment when and as due by the terms of any debt security of such series; or
- default in the performance, or breach, of any covenant or warranty of ours contained in the indenture for the benefit of such series or in the debt securities of such series (other than a covenant or warranty a default in the performance or the breach of which is dealt with elsewhere in the indenture or which is expressly included in the indenture solely for the benefit of a series of debt securities other than such series), and continuance of such default or breach for a period of 60 days after written notice as provided in the indenture;
- if any event of default as defined in any mortgage, indenture or instrument under which there may be issued, or by which there may be secured or evidenced, any of our debt (including any event of default under any other series of debt securities), whether such debt now exists or is created or incurred, shall happen and shall consist of default in the payment of more than \$200 million of such debt at its maturity (after giving effect to any applicable grace period) or shall result in more than \$200 million in principal amount of such debt becoming or being declared due and payable prior to the date on which it would otherwise become due and payable; provided, however, that, if such default under such mortgage, indenture or instrument is cured by us, or waived by the holders of such debt, in each case as may be permitted by such mortgage, indenture or instrument, then the event of default under the indenture caused by such default will be deemed likewise to be cured or waived;
- · particular events in bankruptcy, insolvency or reorganization; or
- any other event of default provided in or pursuant to the indenture with respect to debt securities of such series.

No event of default with respect to a particular series of debt securities issued under the indenture necessarily constitutes an event of default with respect to any other series of debt securities issued thereunder. Any modifications to the foregoing events of default will be described in any prospectus supplement.

The indenture provides that if an event of default with respect to the debt securities of any series at the time outstanding (other than an event of default described in the sixth bullet above) occurs and is continuing, either Wells Fargo Bank, National Association (the "Trustee") or the holders of not less than 25% in principal amount of the outstanding debt securities of such series may declare the principal amount of all outstanding debt securities of such series, or such lesser amount as may be provided for in the debt securities of such series, to be due and payable immediately, by a notice in writing to us (and to the Trustee if given by the holders), and upon any such declaration such principal or such lesser amount shall become immediately due and payable.

If an event of default described in the sixth bullet above (relating to events in bankruptcy, insolvency or reorganization of us) occurs, all unpaid principal of and accrued interest on the outstanding debt securities of that series (or such lesser amount as may be provided for in the debt securities of such series) shall ipso facto become and be immediately due and payable without any declaration or other act on the part of the Trustee or any holder of any debt security of that series.

At any time after a declaration of acceleration with respect to the debt securities of any series is made and before a judgment or decree for payment of the money due is obtained by the Trustee, and subject to particular other provisions of the indenture, the holders of not less than a majority in principal amount of the outstanding debt securities of such series, by written notice to us and the Trustee, may, under some circumstances, rescind and annul such declaration and its consequences.

Within 90 days after the occurrence of any default under the indenture with respect to the debt securities of any series, the Trustee shall deliver to all holders of debt securities of such series notice of such default hereunder actually known to a responsible officer of the Trustee, unless such default shall have been cured or waived; provided, however, that, except in the case of a default in the payment of the principal of (or premium, if any), or interest, if any, on, or additional amounts or any sinking fund or purchase fund installment with respect to, any debt security of such series, the Trustee shall be protected in withholding such notice if and so long as the Trustee in good faith determines that the withholding of such notice is in the best interest of the holders of debt securities of such series; and provided, further, that in the case of any default of the character specified in the fourth bullet of the first paragraph above with respect to debt securities of such series, no such notice to holders shall be given until at least 60 days after the occurrence thereof. For the purpose of this paragraph, the term "default" means any event which is, or after notice or lapse of time or both would become, an event of default with respect to debt securities of such series.

Concerning the Trustee and Paying Agent

Wells Fargo Bank, National Association is the trustee. Deutsche Bank Trust Company Americas is the paying agent and authenticating agent for the notes. Deutsche Bank Luxembourg S.A. is the securities registrar for the notes. We entered into a registrar and paying agent agreement in relation to the notes between us, Deutsche Bank Trust Company Americas, as paying agent, and Deutsche Bank Luxembourg S.A., as securities registrar. Payment of principal of and interest on the notes are made through the office of the paying agent. Each of Wells Fargo Bank, National Association, Deutsche Bank Trust Company Americas and Deutsche Bank Luxembourg S.A., each in each of its capacities, including without limitation as trustee, paying agent and securities registrar, as applicable, assumes no responsibility for the accuracy or completeness of the information contained in this document or the related documents or for any failure by us or any other party to disclose events that may have occurred and may affect the significance or accuracy of such information. We maintain banking relationships in the ordinary course of business with the trustee and its affiliates, the paying agent and its affiliates and the securities registrar and its affiliates.

Book-entry System

Global Notes

We issued the notes in the form of one or more global notes (the "global notes") in definitive, fully registered, book-entry form without coupons. The global notes were deposited with a common depositary (and registered in the name of its nominee) for, and in respect of interests held through, Clearstream Banking, *société anonyme*, which we refer to as "Clearstream," or Euroclear Bank S.A./ N.V., which we refer to as "Euroclear."

Except as set forth below, the global notes may be transferred, in whole and not in part, only to a common depository for Clearstream and Euroclear or its nominee. No link is expected to be established among The Depository Trust Company and Clearstream or Euroclear in connection with the issuance of the notes.

Clearstream and Euroclear

Beneficial interests in the global notes are represented through book-entry accounts of financial institutions acting on behalf of beneficial owners as direct and indirect participants in Clearstream or Euroclear. Those beneficial interests are in denominations of \pounds 100,000 and integral multiples of \pounds 1,000 in excess thereof with respect to the euro notes and in denominations of \pounds 100,000 and integral multiples of \pounds 1,000 in excess thereof with respect to the sterling notes. Should certificates be issued to individual holders of the notes, a holder of notes who, as a result of trading or otherwise, holds a principal amount of notes of a specified series that is less than the minimum denomination of notes specified for such series would be required to purchase an additional principal amount of notes such that its holding of notes of such series amounts to the minimum specified denomination. Investors may hold interests in the global notes through Clearstream or Euroclear

either directly if they are participants in such systems or indirectly through organizations that are participants in such systems.

Except as set forth in the indenture, owners of beneficial interests in the global notes will not be entitled to have notes registered in their names, and will not receive or be entitled to receive physical delivery of notes in definitive form. Except as provided below, beneficial owners will not be considered the owners or holders of the notes under the indenture. Accordingly, each beneficial owner must rely on the procedures of the clearing systems and, if such person is not a participant of the clearing systems, on the procedures of the participant through which such person owns its interest, to exercise any rights of a holder under the indenture. Under existing industry practices, if we request any action of holders or a beneficial owner desires to give or take any action which a holder is entitled to give or take under the indenture, the clearing systems would authorize their participants holding the relevant beneficial interests to give or take action and the participants would authorize beneficial owners owning through the participants to give or take such action or would otherwise act upon the instructions of beneficial owners. Conveyance of notices and other communications by the clearing systems to their participants, by the participants to indirect participants and by the participants and indirect participants to beneficial owners will be governed by arrangements among them, subject to any statutory or regulatory requirements as may be in effect from time to time. Persons who are not Euroclear or Clearstream participants may beneficially own notes held by the common depositary for Euroclear and Clearstream only through direct or indirect participants in Euroclear and Clearstream.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

PERFORMANCE SHARE AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of your company's securities or financial instruments. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

PERFORMANCE SHARE AWARD AGREEMENT

Participant Name:

Participant ID:

Grant Date:

Performance Period: Fiscal Years - 2020 - 2022 (the "Performance Period")

Shares Granted:

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the Performance Share Award (the "Award") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Performance Shares</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you the target number of Performance Shares specified above (the "Performance Shares"), subject to the terms and conditions of the Plan and this Agreement. This "target" number of shares is computed by dividing the target award dollar amount for your position by the average closing stock price of the Company's common stock, par value US\$.01 per share ("Stock") for the last 30 trading days of the fiscal year preceding the Grant Date.

2. <u>Performance Measure</u>. The number of Performance Shares earned at the end of the three-year Performance Period will vary depending on the degree to which cumulative adjusted earnings per share performance goals for the Performance Period, as established by the Committee, are met.

3. <u>Determination of Performance Shares Earned</u>. At the target levels, 100% of the Performance Shares will be earned. At the threshold levels, 50% of the Performance Shares will be earned. Below the threshold levels of performance, no Performance Shares are earned. At the maximum levels or more, 150% of the Performance Shares will be earned. Performance between minimum and target, and between target and maximum, will earn Performance Shares on a pro-rated basis between 50% and 100%, and 150%, respectively.

The amount earned will be calculated according to the following:

Percent of

Performance	=	Target	Х	Target
Shares Awarded		Performance Shares		Performance Shares Earned

4. <u>Disability or Death</u>. If during the Performance Period you have a Termination of Service by reason of Disability or death, then the number of Performance Shares earned (based on performance as of the end of the Performance Period) shall become vested at the end of the Performance Period. Any Performance Shares becoming vested by reason of your Termination of Service by reason of Disability or death shall be paid at the same time Performance Shares are paid to other Participants.

5. <u>Retirement</u>. If during the Performance Period you have a Termination of Service by reason of Retirement, as reasonably determined and approved by the Committee or its delegates, then, subject to such approval, the number of Performance Shares earned (based on performance as of the end of the Performance Period) will be prorated to reflect the portion of the Performance Period during which you remained employed by the Company. Such prorated portion shall equal the full number of earned Performance Shares, multiplied by a fraction equal to the number of full months of the Performance Period completed as of your Termination of Service, divided by 36. Any Performance Shares becoming vested by reason of your Retirement shall be paid at the same time Performance Shares are paid to other Participants.

6. <u>Termination of Service Following a Change in Control</u>. If during the Performance Period there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 7), then your earned Award shall equal your target number of Performance Shares, prorated to reflect the portion of the Performance Period during which you remained employed by the Company. Such prorated portion shall equal your target number of Performance Shares, multiplied by a fraction equal to the number of full months of the Performance Period completed as of your Termination of Service, divided by the number of months in the Performance Period. This prorated award will be settled in cash (subject to required tax withholdings) in accordance with Section 9.01(b) of the Plan within 45 days after your Termination of Service. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such Plan.

7. <u>Other Termination of Service</u>. If during the Performance Period you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 4, 5 or 6 above, as determined by the Committee, then all of your Performance Shares shall be forfeited. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

8. <u>Settlement of Earned Performance Shares</u>. At the end of the Performance Period actual performance for the entire Performance Period shall be reviewed, and the amount of the earned Award shall be determined based on this performance and communicated to you. Subject to the requirements of Section 12 below, the Company shall transfer to you one (1) share of Stock for each Performance Share earned at that time, net of any applicable tax withholding requirements in accordance with Section 9 below. The Performance Shares payable under this Agreement are intended to be exempt from Code Section 409A under the exemption for short-term deferrals. Accordingly, the Performance Shares will be settled in shares of Stock no later than the 15th day of the third month following the end of the fiscal year of the Company (or if later, the calendar year) in which the Performance Shares are earned.

Notwithstanding the foregoing, if you are resident or employed outside of the U.S., the Company, in its sole discretion, may provide for the settlement of the Performance Shares in the form of:

(a) a cash payment (in an amount equal to the Fair Market Value of the shares of Stock that corresponds with the number of earned Performance Shares) to the extent that settlement in shares of Stock (i) is prohibited under local law, (ii) would require you, the Company or an Affiliate to obtain the approval of any governmental or regulatory body in your country of residence (or country of employment, if different), (iii) would result in adverse tax consequences for you, the Company or an Affiliate or (iv) is administratively burdensome; or

(b) shares of Stock, but require you to sell such shares of Stock immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

9. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or your Employer, the ultimate liability for all income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("Tax-Related Items"), is and remains your responsibility and may exceed the amount actually withheld by the Company or your Employer, if any. You further acknowledge that the Company and/or your Employer (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Award, including, but not limited to, the grant, vesting or settlement of the Award, the subsequent sale of shares of Stock acquired pursuant to such settlement and the receipt of any dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Award to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or your Employer to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon settlement of the Award that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Performance Shares and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, your Employer or its agent to cover the Tax-Related Items that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company applicable tax withholding event, the Company and/or your Employer may

(in its sole discretion) allow you to make a cash payment to the Company, your Employer or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon settlement of the Award, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the earned Award, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock (or cash payment) or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

10. <u>Nontransferability</u>. During the Performance Period and thereafter until shares of Stock are transferred to you in settlement thereof, you may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Performance Shares, whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, or by will or by the laws of intestacy.

11. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the Performance Shares until such time as a certificate of stock for the shares of Stock issued in settlement of the Performance Shares has been issued to you or such shares of Stock have been recorded in your name in book entry form. Except as provided in Section 17 below, no adjustment shall be made for dividends or distributions or other rights with respect to such shares of Stock for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

12. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares of Stock except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

13. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Performance Shares is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Performance Shares is not subject to the supervision of the local securities authorities.

14. <u>Insider Trading/Market Abuse Laws</u>. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock, or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need

to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

15. <u>Repatriation; Compliance with Law</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of residence to comply with your personal obligations under local laws, rules and/or regulations in your country of residence (and country of employment, if different).

16. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Performance Shares. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

17. <u>Change in Stock</u>. In the event of any change in the shares of Stock, by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of the Performance Shares subject to this Award Agreement shall be equitably adjusted by the Committee.

18. <u>Nature of the Award</u>. In accepting the Award, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Award is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of Performance Shares, or benefits in lieu of Performance Shares, even if Performance Shares have been granted in the past;

(c) all decisions with respect to future Awards or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of the Award, the number of shares of Stock subject to the Award, and the earning provisions applicable to the Award;

(d) the Award and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Award and the shares of Stock subject to the Award are not intended to replace any pension rights or compensation;

(g) the Award, the shares of Stock subject to the Award and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare

benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company, your Employer or any Affiliate;

(h) the future value of the shares of Stock underlying the Award is unknown, indeterminable and cannot be predicted with certainty;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Award resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Award to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Award and the benefits evidenced by this Agreement do not create any entitlement to have the Award or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Award or of any amounts due to you pursuant to the settlement of the Award or the subsequent sale of any shares of Stock acquired upon settlement of the Award.

19. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

20. <u>Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Performance Shares and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and your Employer hold certain personal information about you, specifically: your name, home address, email address and telephone number, date of birth, social security, passport or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and your Employer will process

Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and your Employer will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or your Employer may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of Data, (iv) oppose, for legal reasons, the collection, processing or transfer of Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, your Performance Shares will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

21. Addendum to Agreement. Notwithstanding any provision of this Agreement to the contrary, the Performance Shares shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to this Agreement, attached hereto as Exhibit A (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Performance Shares and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

22. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Performance Shares, any shares of Stock acquired pursuant to the Performance Shares and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Performance Shares and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

23. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

24. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Award or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

25. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdiction.

26. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Award, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English so as to allow you to understand the terms and conditions of the Agreement, the Plan or any other documents related to the Award. If you have received this Agreement, the Plan or any other documents related into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

27. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

28. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibit A. Once you have read and understood this Agreement and Exhibit A, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibit A and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Award granted hereunder.

EXHIBIT A

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

PERFORMANCE SHARE AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Award is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 21 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Performance Shares and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

EUROPEAN UNION ("EU") / EUROPEAN ECONOMIC AREA ("EEA")

Personal Data. The following provision replaces Section 20 of the Agreement in its entirety:

The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all Performance Shares, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Performance Shares under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. As such, by accepting the Performance Shares, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the

Performance Shares. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) International Data Transfers. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) Data Subject Rights. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources Department in writing. Your provision of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Performance Shares, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources Department in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

CHILE

Private Placement. The following provision shall replace Section 13 of the Agreement:

The grant of the Performance Shares hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;
- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and

- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;
- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y

d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente. FRANCE

1. <u>Nature of Grant</u>. The Performance Shares are not granted under the French specific regime provided by Articles L. 225-197-1 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Form of Payment</u>. Notwithstanding any provision in the Agreement or Plan to the contrary, the Performance Shares shall be settled only in Shares (and not in cash).

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement, this Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Performance Shares and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Performance Shares and shares of Stock subject to the Performance Shares do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

<u>Plan Acknowledgment</u>. In accepting the Performance Shares, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 3: Determination of Performance Shares Earned (threshold levels for earning Performance Shares); Section 4: Disability or Death (terms of payment of Performance Shares upon a Termination of Service by reason of Disability or death); Section 5: Retirement (terms of payment of Performance Shares upon a Termination of Service by reason of retirement); Section 6: Termination of Service Following a Change in Control (terms of payment of Performance Shares in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (forfeiture of Performance Shares in other cases of Termination of Service); Section 9(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Performance Shares and legally applicable to the participant); Section 10: Nontransferability (Performance Shares shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 17: Change in Stock (right of the Company to equitably adjust the number of Performance Shares subject to this Agreement in the event of any change in Stock); Section 18(j): Nature of the Award (waive any claim or entitlement to compensation or damages arising from forfeiture of the Performance Shares resulting from a Termination of Service); Section 18(I): Nature of the Award (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Performance Shares); Section 19: Committee Authority; Recoupment (right of the Committee to administer, construe, and make all determinations necessary or appropriate for the administration of the Performance Shares and this Agreement, including the enforcement of any recoupment policy); Section 21: Addendum to Agreement (the Performance Shares are subject to the terms of the Addendum); Section 22: Additional Requirements (Company right to impose additional requirements on the Performance Shares in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Performance Shares and the Plan); Section 24: Electronic Delivery (Company may deliver documents related to the Award or Plan electronically); Section 25: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 26: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Performance Shares does not constitute an employment relationship between you and the Company. You have been granted the Performance Shares as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you ("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan are not part of the employment conditions and/or benefits provided by WBA Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. <u>Extraordinary Item of Compensation</u>. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Award, the shares of Stock subject to the Award and the income and value of the same is an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses,

long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Performance Shares, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Performance Shares. Upon the grant of Performance Shares, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

SPAIN

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Award, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted Performance Shares under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Performance Shares are granted on the assumption and condition that the Performance Shares and the shares of Stock acquired upon settlement of the Performance Shares shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Award shall be null and void. Further, you understand and agree that the earning of the Performance Shares is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Performance Shares may be forfeited effective on the date of your Termination of Service (unless otherwise specifically provided in Section 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to the Performance Shares as of the date of your Termination of Service, as described in the Plan and Agreement.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on your Award.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 7 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Performance Shares. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Performance Shares have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

<u>Securities Law Notification</u>. The Performance Shares are not considered a public offering in Switzerland; therefore, the offer of Performance Shares is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Performance Shares constitute a prospectus as such term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Performance Shares may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Performance Shares have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. <u>Indemnification for Tax-Related Items</u>. Without limitation to Section 9 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required

to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 10 of the Agreement.

2. Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Performance Shares, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Performance Shares. Upon the grant of the Performance Shares, you shall be deemed irrevocably to have waived any such entitlement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Performance Share Award Agreement to which this Addendum is attached as Exhibit A, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of your company's securities or financial instruments. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

Participant Name:

Participant ID:

Grant Date:

Grant Price:

Shares Granted:

Vesting: One third of the Shares Granted vest on each of the first, second and third anniversaries of the Grant Date (the "Vesting Dates")

Expiration Date:

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the stock option (the "Option") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as in effect on the date of this Agreement and as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Option</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you an Option to purchase all or any part of the number of Shares Granted set forth above of common stock of the Company, par value US\$.01 ("Stock"), at the per-share exercise price, which is 100% of the fair market value of a share of Stock on the Grant Date (the "Exercise Price"), subject to the terms and conditions of the Plan and this Agreement. The Option is intended to be a "non-qualified stock option" and shall not be treated as an incentive stock option within the meaning of Section 422 of the Code.

2. <u>Vesting/Exercise/Expiration</u>. You may not exercise the Option prior to the Vesting Date or Dates set forth above absent action by the Committee to waive or alter such restrictions or as may be permitted under the below paragraphs. Thereafter, except as hereinafter provided, you may exercise the Option, to the extent it is vested, at any time and from time to time until the close of business on the Expiration Date set forth above. The Option may be exercised to purchase any number of whole shares of Stock, except that no purchase shall be for less than ten (10) full shares of Stock, or the remaining unexercised shares, if less. The Option is deemed to be "outstanding" until it has been exercised in full or expired pursuant to the terms of this Agreement.

3. <u>Disability</u>. If, without having fully exercised the Option, you have a Termination of Service due to Disability, then any Shares Granted under the Option that are not yet vested at that time shall thereupon become vested and (a) you may exercise the Option for the full number of Shares Granted (less any shares for which the Option was previously exercised), but (b) your right to exercise the Option shall terminate upon the earlier of the Expiration Date or a date which is one (1) year following the date of your Termination of Service.

4. <u>Death</u>. If, without having fully exercised the Option, you have a Termination of Service due to your death, then any Shares Granted under the Option that are not yet vested at that time shall thereupon become fully vested and (a) the Option may be exercised by the executor or administrator of your estate or by such person or persons who shall have acquired your rights hereunder by bequest or inheritance or by designation as your beneficiary for the full number of Shares Granted (less any shares for which the Option was previously exercised), but (b), such person's right to exercise the Option shall terminate upon the earlier of the Expiration Date or a date which is one (1) year after the date of your death.

5. <u>Retirement</u>. If without having fully exercised the Option you have a Termination of Service by reason of Retirement, as reasonably determined and approved by the Committee or its delegates, then, subject to such approval (a) any Shares Granted under the Option that are not vested at that time shall thereupon become vested, and (b) you may exercise the Option for the full number of Shares Granted (less any shares for which the Option was previously exercised) until the Expiration Date set forth above.

6. <u>Termination of Service Following a Change in Control</u>. If there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 8), then any Shares Granted under the Option that are not yet vested at that time shall thereupon become vested, and you may exercise the Option for the full number of Shares Granted (less any shares for which the Option under this Section 6 shall be forfeited. The foregoing is also subject to the Committee's exercise of its discretion under Section 9.01 of the Plan. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such plan.

7. <u>Other Termination of Service</u>. If without having fully exercised the Option you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 3, 4, 5 or 6 above, as determined by the Committee, then (a) for any Shares Granted with respect to which such Termination of Service is prior to the applicable Vesting Date, the Option shall be forfeited, and (b) for any Shares Granted with respect to which such Terminate upon the earlier of the Expiration Date or a date which is ninety (90) days after the date of your Termination of Service. The foregoing is subject to the right of the Committee to extend the exercise period of the Option, including any extension granted by the Committee or its delegate as needed to allow your right to exercise to extend beyond a period during which you are restricted from exercising the Option due to a Company-designated trading blackout period, and is subject to earlier expiration as provided in Section 8 below.

8. <u>Forfeiture of Outstanding Options Upon Termination for Cause or Upon Other Violations</u>. Notwithstanding any provision of this Agreement to the contrary, the Option (whether vested or unvested) shall immediately terminate if you are terminated for Cause or if and when you violate any obligation that you may have to the Company during or post-employment, including but not limited to a violation of any applicable provision of the NNCA Agreement (defined in Section 22 below) or any other non-competition, non-solicitation, confidentiality, non-disparagement or other restrictive covenant. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

9. <u>Exercise Process</u>. The Option may be exercised by giving notice to Fidelity Stock Plan Services, LLC ("Fidelity"), the third party administrator administering the Option exercise process or any other third party administrator the Company may engage in the future. The exercise notice (a) shall be signed by you or (in the event of your death) your legal representative, (b) shall specify the number of full shares of Stock then elected to be purchased, and (c) shall be accompanied by payment in full of the Exercise Price of the shares of Stock to be purchased. Payment may be made in cash or by check payable to the order of the Company, and such payment shall include any income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan that are required to be withheld ("Tax-Related Items"), as set forth in Section 10 below. Alternatively, the Committee may allow for one or more of the following methods of exercising the Option:

(a) Payment for shares of Stock as to which the Option is being exercised and/or payment of any Tax-Related Items may be made by transfer to the Company of shares of Stock you already own, or any combination of such shares of Stock and cash, having a fair market value determined at the time of exercise of the Option equal to, but not exceeding, the Exercise Price and/or any Tax-Related Items, as the case may be.

(b) A "same day sale" transaction pursuant to which a third party (engaged by you or the Company) loans funds to you to enable you to purchase the shares of Stock and pay any Tax-Related Items, and then sells a sufficient number of the exercised shares of Stock on your behalf to enable you to repay the loan and any fees. The remaining shares of Stock and/or cash are then delivered by the third party to you.

(c) A "net exercise" transaction, pursuant to which the Company delivers to you the net number of whole shares of Stock remaining from the portion of the Option being exercised after deduction of a number of shares of Stock with a fair market value equal to the Exercise Price and the amount of any Tax-Related Items.

As promptly as practicable after receipt of such notice of exercise and payment (including payment with respect to any Tax-Related Items), subject to Section 13 below, the Company shall cause to be issued and delivered to you (or in the event of your death to your legal representative, as the case may be), certificates for the shares of Stock so purchased. Alternatively, such shares of Stock may be issued and held in book entry form.

Notwithstanding any provision within this Agreement to the contrary, if you are resident or employed outside of the U.S., the Committee may require that you (or in the event of your death, your legal representative, as the case may be) exercise the Option in a method other than as specified above, may

require you to exercise the Option only by means of a "same day sale" transaction (either a "sell-all" transaction or a "sell-tocover" transaction) as it shall determine in its sole discretion, or may require you to sell any shares of Stock you acquire under the Plan immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

10. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or your Employer, the ultimate liability for all Tax-Related Items related to your participation in the Plan and legally applicable to you is and remains your responsibility and may exceed the amount actually withheld by the Company or your Employer, if any. You further acknowledge that the Company and/or your Employer (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Option, including, but not limited to, the grant, vesting or exercise of the Option, the subsequent sale of shares of Stock acquired pursuant to such exercise and the receipt of any dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Option to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or your Employer to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon exercise of the Option that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Option and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company, your Employer or its agent to cover the Tax-Related Items that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or your Employer may (in its sole discretion) allow you to make a cash payment to the Company, your Employer or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon exercise of the Option, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the exercised Option, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

11. <u>Limited Transferability</u>. You may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Option, whether voluntarily or involuntarily or by operation of law, other than by beneficiary

designation effective upon your death, by will or by the laws of intestacy. During your lifetime, the Option and all rights granted hereunder shall be exercisable only by you. Notwithstanding the foregoing, you may transfer the Option, in whole or in part, by gift to a Permitted Transferee in accordance with rules and subject to any conditions specified by the Committee under the Plan.

12. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the shares of Stock subject to the Option until such time as the Exercise Price has been paid and a certificate of stock for such shares has been issued to you or such shares of Stock have been recorded in your name in book entry form. Except as provided in Section 18 below, no adjustment shall be made for dividends or distributions or other rights with respect to such shares of Stock for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

13. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares of Stock except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

14. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Option is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Option is not subject to the supervision of the local securities authorities.

15. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock (e.g., the Option) or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

16. <u>Repatriation; Compliance with Law; Method of Exercise</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of employment, if different). Finally, you agree to take any and all actions as may be required to comply with your personal obligations under local laws, rules and/or regulations in your country of residence (and country of employment, if different).

17. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Option. Investment in shares of Stock involves a degree of risk. Before deciding to purchase shares of Stock pursuant to the Option, you should carefully consider all risk factors relevant to the acquisition of shares of Stock under the Plan and you should carefully review all of the materials related to the Option and the Plan. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

18. <u>Change in Stock</u>. In the event of any change in the shares of Stock by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of shares of Stock subject to the Option and the Exercise Price shall be equitably adjusted by the Committee.

19. <u>Nature of the Option</u>. In accepting the Option, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Option is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of stock options, or benefits in lieu of stock options, even if stock options have been granted in the past;

(c) all decisions with respect to future grants of stock options or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of any grant, the number of shares of Stock subject to the stock options, vesting provisions, and the exercise price applicable to the stock option;

(d) the Option and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Option and the shares of Stock subject to the Option are not intended to replace any pension rights or compensation;

(g) the Option, the shares of Stock subject to the Option and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company, your Employer or any Affiliate;

(h) the future value of the shares of Stock underlying the Option is unknown, indeterminable and cannot be predicted with certainty and if you exercise a vested Option, the value of the underlying shares of Stock may increase or decrease, even below the Exercise Price;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Option resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Option to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Option and the benefits evidenced by this Agreement do not create any entitlement to have the Option or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Option or of any amounts due to you pursuant to the settlement of the Option or the subsequent sale of any shares of Stock acquired upon settlement of the Option.

20. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

21. <u>Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Option and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and your Employer hold certain personal information about you, specifically: your name, home address, email address and telephone number, date of birth, social security, passport or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and your Employer will process Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and your Employer will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or your Employer may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of Data, (iv) oppose, for legal reasons, the collection, processing or transfer of the Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, the Option will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

22. <u>Non-Competition, Non-Solicitation and Confidentiality</u>. As a condition to the receipt of the Option, you must agree to the Non-Competition, Non-Solicitation and Confidentiality Agreement (the "NNCA Agreement") attached hereto as Exhibit A. By clicking the acceptance box for this Agreement, you also agree to the terms and conditions expressed in the NNCA Agreement. Failure to accept the terms of this Agreement and NNCA Agreement within 120 days of the Grant Date shall constitute your decision to decline to accept this Award.

23. Addendum to Agreement. Notwithstanding any provision of this Agreement to the contrary, the Option shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to this Agreement, attached hereto as Exhibit B (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

24. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Option, any shares of Stock acquired pursuant to the Option and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

25. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

26. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Option or your future participation in the Plan. You hereby consent to receive

such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

27. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdiction.

28. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Option, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English, so as to allow you to understand the terms and conditions of this Agreement, the Plan or any other document related to the Option. If you have received this Agreement, the Plan or any other document related into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

29. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

30. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibits A and B. Once you have read and understood this Agreement and Exhibits A and B, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibits A and B, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Option granted hereunder.

EXHIBIT A

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

This Exhibit (the "Non-Compete Agreement") forms a part of the Stock Option Award Agreement covering Options awarded to an employee ("Employee" or "I") of Walgreens Boots Alliance, Inc. or an affiliate thereof, on behalf of itself, its affiliates, subsidiaries, and successors (collectively referred to as the "Company").

WHEREAS, the Company develops and/or uses valuable business, technical, proprietary, customer and patient information it protects by limiting its disclosure and by keeping it secret or confidential;

WHEREAS, Employee acknowledges that during the course of employment, he or she has or will receive, contribute, or develop such Confidential Information and Trade Secrets (as defined below); and

WHEREAS, the Company desires to protect from its competitors such Confidential Information and Trade Secrets and also desires to protect its legitimate business interests and goodwill in maintaining its employee and customer relationships.

NOW THEREFORE, in consideration of the Stock Option issued to Employee pursuant to the Agreement to which this is attached as Exhibit A and for other good and valuable consideration, including but not limited to employment or continued employment, the specialized knowledge, skill and training that the Company provides Employee, the goodwill that Employee develops with customers on behalf of the Company, Employee agrees to be bound by the terms of this Non-Compete Agreement as follows:

1. <u>Confidentiality</u>.

(a) At all times during and after the termination of my employment with the Company, I will not, without the Company's prior written permission, directly or indirectly for any purpose other than performance of my duties for the Company, utilize or disclose to anyone outside of the Company any Trade Secrets (defined in subparagraph 1(a)(i)) or other Confidential Information (defined in subparagraph 1(a)(ii)) or any information received by the Company in confidence from or about third parties, as long as such matters remain Trade Secrets or otherwise confidential.

(i) For purposes of this Non-Compete Agreement, "**Trade Secrets**" means a form of intellectual property that are protectable under applicable state and/or Federal law, including the Uniform Trade Secrets Act (as amended and adapted by the states) and the Federal Defend Trade Secrets Act of 2016 (the "DTSA"). They include all tangible and intangible (*e.g.*, electronic) forms and types of information that is held and kept confidential by the Company and is not generally known outside of the Company, including but not limited to information about: the Company's financial, business, scientific, technical, economic, or engineering information, including patterns, plans, compilations, program devices, formulas, designs, prototypes, methods, techniques, processes, procedures, programs or codes, and may in particular include such things as pricing information, business records, software programs, algorithms, inventions, patent applications, and designs and processes not known outside the Company.

(ii) For purposes of this Non-Compete Agreement, "**Confidential Information**" means Trade Secrets and, more broadly, any other tangible and intangible (*e.g.*, electronic) forms and types of information that are held and kept confidential by the Company and are not generally known outside the Company, and which relates to the actual or anticipated business of the Company or the Company's actual or prospective vendors or clients. Confidential Information shall not be considered generally known to the public if is revealed improperly to the public by me or others without the Company's express written consent and/or in violation of an obligation of confidentiality to the Company. Examples of Confidential Information include, but are not limited to: customer, referral source, supplier and contractor identification and contacts; special contract terms; pricing and margins; business, marketing and customer plans and strategies; financial data; company created (or licensed) techniques; technical know-how; research, development and production information; processes, prototypes, software, patent applications and plans, projections, proposals, discussion guides, and/or personal or performance information about employees. (b) I understand that this obligation of non-disclosure shall last so long as the information remains confidential. I, however, understand that, if I live and work primarily in Wisconsin, Virginia, or any other state requiring a temporal limit on non-disclosure clauses, Confidential Information shall be protected for two (2) years following termination of my employment (for any reason). I also understand that Trade Secrets are protected by statute and are not subject to any time limits. I also agree to contact the Company before using, disclosing, or distributing any Confidential Information or Trade Secrets if I have any questions about whether such information is protected information.

(C) The restrictions set forth in this paragraph are in addition to and not in lieu of any obligations I have by law with respect to the Company's Confidential Information, including any obligations I may owe under the DTSA and any applicable state statutes. Nothing herein shall prohibit me from divulging evidence of criminal wrongdoing to law enforcement or prohibit me from disclosing Confidential Information or Trade Secrets if compelled by order of court or an agency of competent jurisdiction or as required by law; however, I shall promptly inform the Company of any such situations and shall take reasonable steps to prevent disclosure of Confidential Information or Trade Secrets until the Company has been informed of such required disclosure and has had a reasonable opportunity to seek a protective order. Pursuant to the DTSA, I understand that an individual may not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a Trade Secret that: (A) is made (i) in confidence to a federal, state or local government official, either directly or indirectly, or to an attorney, and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. Additionally, I understand that an individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the Trade Secret to his or her attorney and use the Trade Secret information in the court proceeding, so long as any document containing the Trade Secret is filed under seal and the individual does not disclose the Trade Secret, except pursuant to court order. Nothing in this Non-Compete Agreement is intended to conflict with the DTSA or create liability for disclosures of Trade Secrets that are expressly allowed by DTSA.

2. Non-Competition. I agree that during my employment with the Company and for twelve (12) months after the termination of my employment (for any reason), I will not, directly or indirectly have Responsibilities with respect to any Competing Business Line. As set forth in paragraph 9(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. These restrictions shall not apply to passive investments of less than five percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "Responsibilities" means the same or similar material responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company and within the same geographic scope, or portion thereof, where I performed those responsibilities for the Company. For purposes of this Non-Compete Agreement, "Competing Business Line" means any business that is in competition with any business engaged in by the Company and for which I had Responsibilities during the two (2) years prior to my last day of employment with the Company. Competing Business Line shall also include businesses or business lines that may not be directly competitive with the Company in most respects (such as pharmacy benefit managers), but only to the extent I am engaged by any such business in a role: (a) that involves my performing Responsibilities for Competing Products or Services; or (b) where I would be called upon to inevitably rely upon or disclose Confidential Information and such reliance or disclosure would competitively harm the Company. For purposes of this Non-Compete Agreement, "Competing Products or Services" means products or services that are competitive with products or services offered by, developed by, designed by or distributed by the Company during the two (2) years prior to my last day of employment with the Company.

3. <u>Non-Solicitation</u>. I agree that during my employment with the Company and for two (2) years after the termination of my employment from the Company (for any reason):

(a) I will not directly or indirectly, solicit any Restricted Customer for purposes of providing Competing Products or Services, or offer, provide or sell Competing Products or Services to any Restricted Customer. For purposes of this Non-Compete Agreement, "Restricted Customer" means any person, company or entity that was a customer, vendor, supplier or referral source of the Company and with which I had direct contact for purposes of performing responsibilities for the Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company. To the extent permitted by applicable law, "Restricted Customer" also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company; and

(b) I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I had direct contact work during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company; (ii) interfere with the performance by any such employee of his/her duties for the Company; and/or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b).

4. <u>Non-Inducement</u>. I will not directly or indirectly assist or encourage any person or entity in carrying out or conducting any activity that would be prohibited by this Non-Compete Agreement if such activity were carried out or conducted by me.

5. <u>Non-Disparagement</u>. During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees, shareholders, members, agents or products. The foregoing shall not be violated by truthful statements in response to legal process, required governmental testimony or filings, or administrative or arbitral proceedings (including, without limitation, depositions in connection with such proceedings); and the foregoing shall not apply to any claims for harassment or discrimination to the extent so restricted by applicable state law.

6. Intellectual Property. The term "Intellectual Property" shall mean all trade secrets, ideas, inventions, designs, developments, devices, software, computer programs, methods and processes (whether or not patented or patentable, reduced to practice or included in the Confidential Information) and all patents and patent applications related thereto, all copyrights, copyrightable works and mask works (whether or not included in the Confidential Information) and all patents and applications for registration related thereto, all Confidential Information, and all other proprietary rights contributed to, or conceived or created by, or reduced to practice by Employee or anyone acting on his/her behalf (whether alone or jointly with others) at any time from the beginning of Employee's employment with the Company to the termination of that employment plus ninety (90) days, that (i) relate to the business or to the actual or anticipated research or development of the Company; (ii) result from any services that Employee or anyone acting on its behalf perform for the Company; or (iii) are created using the equipment, supplies or facilities of the Company or any Confidential Information.

a. <u>Ownership</u>. All Intellectual Property is, shall be and shall remain the exclusive property of the Company. Employee hereby assigns to the Company all right, title and interest, if any, in and to the Intellectual Property; provided, however, that, when applicable, the Company shall own the copyrights in all copyrightable works included in the Intellectual Property pursuant to the "work-made-for-hire" doctrine (rather than by assignment), as such term is defined in the 1976 Copyright Act. All Intellectual Property shall be owned by the Company irrespective of any copyright notices or confidentiality legends to the contrary that may be placed on such works by Employee or by others. Employee shall ensure that all copyright notices and confidentiality legends on all work product authored by Employee or anyone acting on his/her behalf shall conform to the Company's practices and shall specify the Company as the owner of the work. The Company hereby provides notice to

Employee that the obligation to assign does not apply to an invention for which no equipment, supplies, facility, or Trade Secrets of the Company was used and which was developed entirely on the Employee's own time, unless (i) the invention relates (1) to the business of the Company, or (2) to the Company's actual or demonstrably anticipated research or development, or (ii) the invention results from any work performed by Employee for the Company.

b. <u>Keep Records</u>. Employee shall keep and maintain, or cause to be kept and maintained by anyone acting on his/her behalf, adequate and current written records of all Intellectual Property in the form of notes, sketches, drawings, computer files, reports or other documents relating thereto. Such records shall be and shall remain the exclusive property of the Company and shall be available to the Company at all times during my employment with the Company.

c. <u>Assistance</u>. Employee shall supply all assistance requested in securing for the Company's benefit any patent, copyright, trademark, service mark, license, right or other evidence of ownership of any such Intellectual Property, and will provide full information regarding any such item and execute all appropriate documentation prepared by Company in applying or otherwise registering, in the Company's name, all rights to any such item or the defense and protection of such Intellectual Property.

d. <u>Prior Inventions</u>. Employee has disclosed to the Company any continuing obligations to any third party with respect to Intellectual Property. Employee claims no rights to any inventions created prior to his/her employment for which a patent application has not previously been filed, unless he/she has described them in detail on a schedule attached to this Non-Compete Agreement.

e. <u>Trade Secret Provisions</u>. The provisions in paragraph 1 of this Non-Compete Agreement with regard to Trade Secrets and the DTSA shall apply as well in the context of the parties' Intellectual Property rights and obligations.

7. <u>Return of Company Property</u>. I agree that all documents and data accessible to me during my employment with the Company, including Confidential Information and Trade Secrets, regardless of format (electronic or hard copy), including but not limited to any Company computer, monitor, printer equipment, external drives, wireless access equipment, telecom equipment and systems ("Company Equipment"), are and remain the sole and exclusive property of the Company and/or its clients, and must be returned to the Company upon separation or upon demand by the Company. I further agree that I will provide passwords to access such Company Equipment and I will not print, retain, copy, destroy, modify or erase Company U.S. data on Company Equipment or otherwise wipe Company Equipment prior to returning the Company Equipment.

8. <u>Consideration and Acknowledgments</u>. I acknowledge and agree that the covenants described in this Non-Compete Agreement are essential terms, and the underlying Stock Option Award would not be provided by the Company in the absence of these covenants. I further acknowledge that these covenants are supported by adequate consideration as set forth in this Non-Compete Agreement and are not in conflict with any public interest. I further acknowledge and agree that I fully understand these covenants, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms. I further acknowledge and agree that these covenants are reasonable and enforceable in all respects.

9. Enforceability; General Provisions.

(a) I agree that the restrictions contained in this Non-Compete Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Non-Compete Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place undue restraint on me. I

further understand that the restrictions in this Non-Compete Agreement apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary.

(b) Because the Company's current base of operations is in Illinois and my connections thereto, (i) this Non-Compete Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Non-Compete Agreement is entered into, without giving effect to any conflict of law provisions, and (ii) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute or declaration arising out of this Non-Compete Agreement.

(c) In the event of a breach or a threatened breach of this Non-Compete Agreement, I acknowledge that the Company will face irreparable injury which may be difficult to calculate in dollar terms and that the Company shall be entitled, in addition to all remedies otherwise available in law or in equity, to temporary restraining orders and preliminary and final injunctions enjoining such breach or threatened breach in any court of competent jurisdiction without the necessity of posting a surety bond, as well as to obtain an equitable accounting of all profits or benefits arising out of any violation of this Non-Compete Agreement.

(d) I agree that if a court determines that any of the provisions in this Non-Compete Agreement is unenforceable or unreasonable in duration, territory, or scope, then that court shall modify those provisions so they are reasonable and enforceable, and enforce those provisions as modified.

(e) If any one or more provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement or its application is determined to be invalid, illegal, or unenforceable to any extent or for any reason by a court of competent jurisdiction, I agree that the remaining provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement will still be valid and the provision declared to be invalid or illegal or unenforceable will be considered to be severed and deleted from the rest of this Non-Compete Agreement. I further agree that if any court of competent jurisdiction finds any of the restrictions set forth in this Non-Compete Agreement to be overly broad and unenforceable, the restriction shall be interpreted to extend only over the maximum time period, geographic area, or range of activities or clients that such court deems enforceable

(f) Notwithstanding the foregoing provisions of this Non-Compete Agreement, the non-competition provisions of paragraph 2 above shall not restrict Employee from performing legal services as a licensed attorney for a Competing Business to the extent that the attorney licensure requirements in the applicable jurisdiction do not permit Employee to agree to the otherwise applicable restrictions of paragraph 2.

(g) Waiver of any of the provisions of this Non-Compete Agreement by the Company in any particular instance shall not be deemed to be a waiver of any provision in any other instance and/or of the Company's other rights at law or under this Non-Compete Agreement.

(h) I agree that the Company may assign this Non-Compete Agreement to its successors and assigns and that any such successor or assign may stand in the Company's stead for purposes of enforcing this Non-Compete Agreement.

(i) I agree to reimburse the Company for all attorneys' fees, costs, and expenses that it reasonably incurs in connection with enforcing its rights and remedies under this Non-Compete Agreement, but only to the extent the Company is ultimately the prevailing party in the applicable legal proceedings.

(j) I understand and agree that, where allowed by applicable law, the time for my obligations set out in paragraphs 2 - 6 shall be extended for period of non-compliance up to an additional two (2) years following my last day of employment with the Company (for any reason).

(k) I fully understand my obligations in this Non-Compete Agreement, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms.

(I) I agree that all non-competition, non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Non-Compete Agreement shall survive any termination of this Non-Compete Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Non-Compete Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations.

(m) I understand that nothing in this Non-Compete Agreement, including the non-disclosure and nondisparagement provisions, limit my ability to file a charge or complaint with the Equal Employment Opportunity Commission, Department of Labor, National Labor Relations Board, Occupational Safety and Health Administration, Securities and Exchange Commission or any other federal, state or local governmental agency or commission. I also understand that this Non-Compete Agreement does not limit my ability to communicate with any government agencies or otherwise participate in any investigation or proceeding that may be conducted by any government agency, including providing documents or other information, without notice to the Company. Finally, I understand that nothing in this Non-Compete Agreement is intended to restrict my legally-protected right to discuss wages, hours or other working condition with co-workers, or in any way limit my rights under the National Labor Relations Act or any whistleblower act.

10. <u>Relationship of Parties</u>. I acknowledge that my relationship with the Company is "terminable at will" by either party and that the Company or I can terminate the relationship with or without cause and without following any specific procedures. Nothing contained in this Non-Compete Agreement is intended to or shall be relied upon to alter the "terminable at will" relationship between the parties. I agree that my obligations in this Non-Compete Agreement shall survive the termination of my employment from the Company for any reason and shall be binding upon my successors, heirs, executors and representatives.

11. <u>Modifications and Other Agreements</u>. I agree that the terms of this Non-Compete Agreement may not be modified except by a written agreement signed by both me and the Company. This Non-Compete Agreement shall not supersede any other restrictive covenants to which I may be subject under an employment contract, benefit program or otherwise, such that the Company may enforce the terms of any and all restrictive covenants to which I am subject. The obligations herein are in addition to and do not limit any obligations arising under applicable statutes and common law.

12. <u>State and Commonwealth Law Modifications</u>. I agree that if I primarily reside and work in California, Massachusetts, Puerto Rico, South Carolina, Washington or Wisconsin, I am subject to the modifications to this Non-Compete Agreement set forth in Exhibit A-1 applying to such state and to the extent such state law applies.

13. <u>Notification</u>. I agree that in the event I am offered employment at any time in the future with any entity that may be considered a Competing Business Line, I shall immediately notify such Competing Business of the existence and terms of this Non-Compete Agreement. I also understand and agree that the Company may notify anyone attempting to or later employing me of the existence and provisions of this Non-Compete Agreement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Stock Option Award Agreement to which this Non-Compete Agreement is attached as Exhibit A, and I agree to the terms and conditions expressed in this Non-Compete Agreement, including the modifications set forth in Exhibit A-1, as applicable.

EXHIBIT A-1

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

State and Commonwealth Law Modifications

This Exhibit A-1 to the Walgreens Boots Alliance, Inc. Non-Competition, Non-Solicitation and Confidentiality Agreement (the "Non-Compete Agreement") includes jurisdiction-specific "Addenda," which modify the Non-Compete Agreement as applied to individuals who primarily reside and work in one of the applicable jurisdictions, but only to the extent the laws of such jurisdiction are applicable to the Non-Compete Agreement. The Addenda of this Exhibit A-1 should be read in conjunction with the rest of the Non-Compete Agreement and enforced to the fullest extent permissible to protect the Company's legitimate business interests.

<u>No. 1</u>:

CALIFORNIA ADDENDUM

The covenants in **Paragraph 2 "Non-Competition"** apply during my employment with the Company, but do not apply post-employment, during such time that my base location is in California.

<u>No. 2</u>:

The covenants in **Paragraph 3 "Non-Solicitation"** apply during my employment with the Company, but do not apply postemployment, during such time that my base location is in California.

<u>No. 3</u>:

Paragraph 5 "Non-Disparagement" is replaced with the following:

During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees, shareholders, members, agents or products, except as otherwise allowed by law, including California Government Code Section 12964.5.

<u>No. 4</u>:

Paragraph 6 "Intellectual Property" is supplemented with the following language:

The terms of this Agreement requiring disclosure and assignment of inventions to the Company do not apply to any invention that qualifies fully under California Labor Code Section 2870, which reads:

(a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer's

equipment, supplies, facilities, or trade secret information except for those inventions that either:

(1) Relate at the time of conception or reduction to practice of the invention to the employer's business, or actual or demonstrably anticipated research or development of the employer; or

(2) Result from any work performed by the employee for the employer.

(b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

While employed, I will advise the Company promptly in writing of any inventions that I believe meet the criteria in California Labor Code Section 2870 for a confidential ownership determination.

<u>No. 5</u>:

Paragraph 9 "Enforceability; General Provisions", Subparagraph (b) by this Addendum shall substitute "California" for "Illinois" with respect to the choice of law and forum, during such time that I primarily reside and work in California.

MASSACHUSETTS ADDENDUM

<u>No. 1</u>:

Paragraph 2 "Non-Competition" is stricken and replaced with the following:

2. <u>Non-Competition.</u> In exchange for Company providing me the consideration set forth in the Agreement, I agree that during my employment and for a period of one (1) year from the Termination Date - *i.e.*, the date of my voluntary termination of employment, or of the involuntary termination of my employment with Cause (as defined below) - I will not, directly or indirectly, engage in "Competition" (as defined below) within the "Geographic Region" (as defined below).

(a) **"Cause"** means misconduct, violation of any policy of the Company, including any rule of conduct or standard of ethics of the Company, breach of the Agreement (including this Addendum) or the breach of any confidentiality, nondisclosure, non-solicitation or assignment of inventions obligations to the Company, failure to meet the Company's reasonable performance expectations, or other grounds directly and reasonably related to the legitimate business needs of the Company.

(b) **"Competing Business"** means a business that is in competition with any business engaged in by the Company.

(c) **"Competition**" means to provide the same or substantially similar services to a Competing Business as those that I provided to the Company during the last two (2) years of my employment with the Company. "Competition" does

not include passive investments of less than five percent (5%) ownership interest in any entity.

(d) **"Geographic Region"** means the geographic area in which you, during any time within the last two years of your employment with the Company, provided services or had a material presence or influence.

(e) If the Company enforces the restrictions in this Paragraph 2 for a period of time after the Termination Date (the "Restraint Period"), it will pay me, during the Restraint Period, an amount equal to fifty percent (50%) of my annual base salary. My annual base salary, for the purposes of this Paragraph 2(e), will be calculated based on my average annual salary for my last two (2) years of employment, less any applicable deductions, and excluding any incentive compensation, bonuses, benefits, or other compensation, less any applicable deductions (the "Restraint Payment"). The Restraint Payment will be paid on a pro-rata basis during the Restraint Period in the same manner that I would have received wages from the Company had I been employed during the Restraint Period.

(f) The Restraint Period shall be extended from one (1) year following the Termination Date to two (2) years following the Termination Date if I (i) breached Employee's fiduciary duty(ies) to the Company, or (ii) unlawfully took, physically or electronically, property belonging to the Company. In the event that the Restraint Period is extended due to my breach of my fiduciary duty(ies) to the Company, or due to my having unlawfully taken, physically or electronically, property belonging to the Company, the Company shall not be required to provide payments to me during the extension of the Restraint Period.

(g) I understand that if the Company elects to waive the non-competition provisions set forth herein, I will not receive any compensation or consideration described in Paragraph 2(e). I further understands that at the time of my separation from employment, the Company shall elect whether to waive its enforcement of the non-competition provisions in the Agreement (including this Massachusetts Addendum). You will be notified by the Company of its election or waiver by letter, in a form of the following notice:

Walgreens Boots Alliance, Inc. (the "Company"), pursuant to Paragraph 2 of the Massachusetts Addendum (dated ________, 20___) to the Walgreens Boots Alliance, Inc. Non-Competition, Non-Solicitation, and Confidentiality Agreement (the "Agreement"), in its sole discretion, elects to:

Enforce the one year Restraint Period according to Paragraph 2 of the Addendum. As agreed to by the parties, the Company agrees to pay the employee the amounts described in subparagraph 2(e) of the Addendum.

Waive enforcement of the Restraint Period. The undersigned employee shall not receive any compensation or consideration pursuant to subparagraph 2(e) of the Addendum.

Regardless of the election or waiver, the undersigned employee remains bound by all other terms of the Agreement, and also remains bound by the terms of any and all other agreements between the undersigned employee and the Company.

(h) If I was already employed by the Company on the date of my signature on the Agreement, I acknowledge that the Agreement, including this Massachusetts Addendum, was delivered to me at least ten (10) business days before the date that this Addendum was executed by both of the parties (the "Effective Date"). If I was not already employed by the Company on the date of the my signature on the Agreement, I acknowledge that the Agreement, including this Massachusetts Addendum, was delivered to me (i) before a formal offer of employment was made by the Company, or (ii) ten (10) business days before the commencement of my employment with the Company, whichever was earlier.

(i) I acknowledge that I have been advised of my right to consult with counsel of my own choosing prior to signing the Agreement and this Massachusetts Addendum. By signing the Agreement and this Addendum, I acknowledge that I had time to read and understand the terms of the Agreement and this Addendum, and to consult with my own legal counsel, not including counsel for the Company, regarding the Agreement and the Addendum prior to their execution. I agree that I have actually read and understand the Agreement and this Addendum and all of their terms, and that I am entering into and signing the Agreement and this Addendum knowingly and voluntarily, and that in doing so I am not relying upon any statements or representations by the Company or its agents.

(j) I acknowledge that (i) the Non-Competition covenant contained in this Paragraph 2 is no broader than necessary to protect the Company's trade secrets, Confidential Information, and goodwill, and (ii) the business interests identified in the Agreement cannot be adequately protected through restrictive covenants other than the Non-Competition covenant contained in this Paragraph 2, including without limitation the non-solicitation and non-disclosure restrictions set forth in the Agreement.

<u>No. 2</u>:

Paragraph 9(a) "Enforceability; General Restrictions" is stricken and replaced with the following:

(a) I agree that the restrictions contained in this Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place an undue restraint on me. I further understand that the restrictions in this Agreement - other than the non-competition restrictions set forth in Paragraph 2 - apply no matter whether my employment is terminated by the Company or me and no matter whether that termination is voluntary or involuntary. I understand that the non-competition provisions in Paragraph 2 apply following the voluntary termination of my employment or the involuntary termination of my employment for Cause, as defined in Paragraph 2, unless the Employer elects to waive the non-competition provisions of Paragraph 2 as set forth in subparagraph 2(g).

<u>No. 3</u>:

Paragraph 9(b) "Enforceability; General Restrictions" is stricken and replaced with the following:

(b)(i) Because the Company's current base of operations is in Illinois and my connections thereto, (1) except with respect to the non-competition provisions of Paragraph 2, this Agreement shall be governed by and construed in accordance

with the laws of the State of Illinois, where this Agreement is entered into, without giving effect to any conflict of law provisions, and (2) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute, or declaration - other than a claim, dispute, or declaration arising out of Paragraph 2 - that arises out of this Agreement.

(b)(ii) The interpretation, validity, and enforcement of the non-competition provisions set forth in Paragraph 2 of this Agreement and Massachusetts Addendum will be governed by the laws of the Commonwealth of Massachusetts, without regard to any conflicts of laws principles that would require the application of the law of another jurisdiction. The parties agree that any action relating to or arising out of the non-competition provisions shall be brought in (1) the United States District Court for the District of Massachusetts, Eastern Division, if that Court has subject matter jurisdiction over the dispute; or, if it does not, in (2) the Business Litigation Session of the Suffolk County Superior Court, or, if the Business Litigation Session does not accept the case for whatever reason whatsoever, the Suffolk County Superior Courts of Massachusetts for resolution of any disputes or litigation arising under or in connection with the Non-Competition provisions set forth in Paragraph 2 of this Agreement and Massachusetts Addendum, and waive any objections or defenses to personal jurisdiction or venue in any such proceeding before any such court.

<u>No. 4</u>:

Paragraph 9(I) "Enforceability; General Restrictions" is stricken and replaced with the following:

(a) I agree that all non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Agreement shall survive any termination of this Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations. I further agree that all non-competition obligations in this Agreement shall survive the voluntary termination of my employment or the involuntary termination of my employment for Cause, as defined in Paragraph 2, unless the Employer elects to waive the non-competition provisions of Paragraph 2 as set forth in subparagraph 2(g), and no dispute regarding any other provisions of this Agreement or regarding my employment shall prevent the operation and enforcement of my employment shall prevent the operation and enforcement or regarding my employment shall prevent the operation provisions of Paragraph 2 as set forth in subparagraph 2(g), and no dispute regarding any other provisions of this Agreement or regarding my employment or the termination of my employment or the termination of my employment shall prevent the operation and enforcement of these obligations.

<u>No. 1</u>:

PUERTO RICO ADDENDUM

Paragraphs 2 and 3 are replaced by the following covenants and definitions:

"Similar Business" means the same or substantially the same business activity or activities performed or engaged by me for, or on behalf, of the Business of the Company or one of its subsidiaries or affiliated companies.

"Engage" means participate in, consult with, be employed by, or assist with the organization, policy making, ownership, financing, management, operation or control of any Similar Business in any capacity (*i.e.*, as an

independent contractor, consultant, employee, shareholder, member-owner, or business partner).

"Goodwill" means any tendency of customers, distributors, representatives, employees, vendors, suppliers, or federal, state, local or foreign governmental entities to continue or renew any valuable business relationship with the Company or any Similar Business with which I may be associated, based in whole or in part on past successful relationships with the Company or the lawful efforts of the Company to foster such relationships, and in which I actively participated at any time during the most recent twelve (12) months of my employment.

"Competing Business" means any individual (including me), corporation, limited liability company, partnership, joint venture, association, or other entity, regardless of form, that is directly engaged in whole or in relevant part in any business or enterprise that is the same as, or substantially the same as, that part of the Company for which I provided services during the last two (2) years of my employment, or that is taking material steps to engage in such business.

"**Customers**" means those individuals, companies, or other entities for which the Company has provided or does provide products or services in connection with the business of the Company, or those individuals, companies, or other entities to which the Company has provided written proposals concerning the business of the Company in the two (2) year period preceding the termination of my employment.

"Restricted Territory" means those municipalities within the Commonwealth of Puerto Rico in which I performed the Competing Business.

Non-Competition. I acknowledge and agree that the Company would be irreparably damaged if I - in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member, owner or business partner) - were to provide services to any person directly or indirectly competing with the Company or any of its affiliates or Engaged in a Competing Business and that such competition by me would result in a significant loss of Goodwill by the Company. Therefore, I agree that the following are reasonable restrictions and agree to be bound by such restrictions:

(a) During my employment, and for a period of twelve (12) months immediately following the termination of such employment for any reason, I shall not, directly or indirectly - in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member, owner or business partner) - Engage in Competing Business services or activities within the Restricted Territory; provided, that nothing herein shall prohibit me from being a passive owner of not more than five percent (5%) of the outstanding stock of any class of a corporation which is publicly traded so long as I do not have any active participation in the business of such corporation.

(b) I warrant and represent that the nature and extent of this non-competition clause has been fully explained to me by the Company and that my decision to accept the same is made voluntarily, knowingly, intelligently and free from any undue pressure or coercion. I further warrant and represent that I have agreed to this non-competition clause in consideration of the Stock Option I will be receiving under this Agreement.

Non-Solicitation of Customers. I agree that for a period of twelve (12) months following the voluntary or involuntary termination of my employment for any reason, I will not, either on my own behalf or for any Competing Business, directly or indirectly solicit, divert, or appropriate (or attempt to solicit, divert, or appropriate) any Customer with which I had material business contact in the six (6) month period preceding the termination of my employment, for providing products or services that are the same as or substantially similar to those provided by the Company.

Non-Solicitation of Employees. I recognize and admit that the Company has a legitimate business interest in retaining its employees, representatives, agents and/or consultants and of protecting its business from previous employees, representatives, agents and/or consultants, which makes necessary the establishment of a non-solicitation clause in the Agreement. I agree that for a period of twelve (12) months following the voluntary or involuntary termination of my employment for any reason, I shall not, directly or indirectly, (a) induce or attempt to induce any employee, representative, agent or consultant of the Company or any of its affiliates or subsidiaries to leave the employ or services of the Company or any of its affiliates or subsidiaries and any employee, representative, agent or consultant thereof or (b) hire any person who was an employee, representative, agent or consultant of the Company or any of its affiliates or subsidiaries at any time during the twelve (12) month period immediately prior to the date on which such hiring would take place. No action by another person or entity shall be deemed to be a breach of this provision unless I directly or indirectly assisted, encouraged or otherwise counseled such person or entity to engage in such activity.

<u>No. 2</u>:

Subparagraph 9(b) "Enforceability; General Restrictions" is stricken and replaced with the following:

(b)(i) Because the Company's current base of operations is in Illinois, as are my connections thereto (except with respect to the non-competition provisions of Paragraph 2), (1) I agree this Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Agreement is entered into, without giving effect to any conflict of law provisions, and (2) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute, or declaration - other than a claim, dispute, or declaration arising out of Paragraph 2 - that arises out of this Agreement.

(b)(ii) The laws of Puerto Rico will govern the interpretation, validity, and enforcement of the non-competition provisions set forth in Paragraph 2 of this Agreement and Puerto Rico Addendum.

<u>No. 3</u>:

Subparagraph 9(d), subparagraph 9(e), and subparagraph 9(j) "Enforceability; General Restrictions" are stricken.

SOUTH CAROLINA ADDENDUM

<u>No. 1</u>:

The definition of **"Confidential Information"** in **Paragraph 1** is further limited to that Confidential Information I learn about or am exposed to through my employment with the Company.

<u>No. 2</u>:

Paragraphs 2 and 3 of the Agreement are replaced by the following covenants and definitions:

"Competing Business" means any individual (including me), corporation, limited liability company, partnership, joint venture, association, or other entity, regardless of form, that is directly engaged in whole or in relevant part in any business or enterprise that is the same as, or substantially the same as, that part of the Company for which I provided services during the last two (2) years of my employment, or that is taking material steps to engage in such business.

"Customers" means those individuals, companies, or other entities for which the Company has provided or does provide products or services in connection with the business of the Company, or those individuals, companies, or other entities to which the Company has provided written proposals concerning the business of the Company in the two (2) year period preceding the termination of my employment.

"Restricted Territory" means:

1) the counties or areas where I worked for the Company or had material business contact with the Customers in the two (2) year period preceding the termination of my employment with the Company: and/or

2) the geographic territory in which I worked for the Company, represented the Company, or had material business contact with the Customers in the two (2) year period preceding the termination of my employment with the Company.

I agree that subsections 1) and 2) above are separate and severable covenants.

Non-Competition. I agree that for a period of one (1) year following the voluntary or involuntary termination of my employment for any reason, I will not, directly or indirectly, own, manage, operate, join, control, be employed by or with, or participate in any manner with a Competing Business anywhere in the Restricted Territory where doing so will require me to provide the same or substantially similar services to any such Competing Business as those that I provided to the Company during the last two (2) years of my employment.

Non-Solicitation of Customers. I agree that for a period of two (2) years following the voluntary or involuntary termination of my employment for any reason, I will not, either on my own behalf or for any Competing Business, directly or indirectly solicit, divert, or appropriate, or attempt to solicit, divert, or appropriate any Customer with which I had material business contact in the two (2) year period preceding the termination of my employment, for the purposes of providing products or services that are the same as or substantially similar to those provided by the Company.

Paragraph 2 is replaced in its entirety as follows:

Non-Competition.

(a) The non-competition provisions of this Paragraph 2 shall <u>only</u> apply when my annualized salaried exceeds the compensation requirements of the Restrictions on Noncompetition Covenants Bill 5478 as codified in RCW 49.

(b) I agree that during my employment with the Company and for one (1) year after the termination of my employment for any reason, I will not, directly or indirectly, engage in Competing Services with respect to any Competing Business Line. As set forth in Paragraph 10(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. The above restrictions shall not apply to passive investments of less than five percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "Competing Business Line" means any business that is in competition with any business engaged in by the Company and for which I performed Competing services during the two (2) years prior to my last day of employment with the Company. For purposes of this Non-Compete Agreement, "Competing Services" means the same or similar responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company and within the same geographic scope, or portion thereof, with respect to which I performed those responsibilities for the Company.

I agree that, if and after my employment with the Company ends because of or in connection with (C) a layoff or reduction-in-force, the non-competition provisions of Paragraph 2(a) above will not be enforced by the Company unless and to the extent that it pays me an amount that is equal to or greater than my base salary rate that is in effect on the last day of my employment with the Company. Such payments will be made to me at regular payroll intervals for the duration of the one (1) year post-employment non-competition period or such shorter period during which the Company enforces these non-competition provisions. I agree that I must promptly inform the Company of the date on which I begin any other employment or engagement by, with or for the benefit of any other individual or entity, at which time I agree the Company may and will terminate all such payments to me. Although such payments by the Company will terminate when I commence employment or any other engagement by, with or for the benefit of another individual, entity or employer, I agree that the Paragraph 2(a) non-competition restrictions will remain in effect until one (1) year after my Company employment ends. I also agree that if I fail to timely notify the Company of any other employment or engagement, and if the Company's payments to me therefore continue after I have commenced any such employment or engagement, then any such payments to me will be deemed to be placed by me in constructive trust for the benefit of the Company, and I agree that I must and will promptly return all such payments to the Company.

<u>No. 2</u>:

Subparagraph 9(b) of the Agreement is replaced in its entirety as follows:

(a) This Agreement shall be governed by and construed in accordance with the laws of the State of Washington without giving effect to any conflict of law provisions. Any claim, dispute or declaration arising out of or in connection with this Agreement will be resolved exclusively in the state or federal courts in the State of Washington.

WISCONSIN ADDENDUM

Paragraph 1 "Confidentiality" is amended by adding the following at the end of Paragraph 1(b):

To the extent the above obligation of non-use and non-disclosure of Confidential Information applies after the termination of my employment and to Confidential Information that does not meet the definition of a trade secret under applicable law, it shall apply only for two years after the termination of my employment and only in geographic areas in which the unauthorized use or disclosure of such Confidential Information would be competitively damaging to the Company.

<u>No. 2</u>:

Paragraph 2 "Non-Competition" is amended by replacing the definition of "Responsibilities" with the following:

"**Responsibilities**" means the same or similar responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company in which the Confidential Information I have would be competitively valuable and within the same geographic scope, or portion thereof, with respect to which I performed those responsibilities for the Company.

<u>No. 3</u>:

Paragraph 3 "Non-Solicitation" is amended by replacing the definition of "Restricted Customer" in paragraph 3(a) with the following:

"Restricted Customer" means any person, company or entity that was a customer of the Company and with which I had direct contact for purposes of performing responsibilities for the Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company.

Paragraph 3(a) is further amended by striking the following sentence:

To the extent permitted by applicable law, Restricted Customer also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company;

Paragraph 3(b) is amended by replacing it with the following:

I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I worked at any point during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company and join a competitor; (ii) interfere with the performance by any such employee of his/her duties for the Company; or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b). This restriction shall apply in all geographic areas in which the Company does business. **Paragraph 9 "Enforceability, General Provisions"** is amended as follows: **Paragraph 9(e)** is amended by adding the following text to the end of the paragraph:

"The restrictive covenants in this agreement are intended to be divisible and interpreted and applied independent of each other."

Paragraph 9(j) is stricken and shall not be applied or referred to.

<u>EXHIBIT B</u>

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Option is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 23 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

EUROPEAN UNION ("EU") / EUROPEAN ECONOMIC AREA ("EEA")

Personal Data. The following provision replaces Section 21 of the Agreement in its entirety:

The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all stock options, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Option under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. As such, by accepting the Option, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the Option. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) <u>International Data Transfers</u>. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) Data Subject Rights. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources Department in writing. Your provision of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Option, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources pepartment in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

CHILE

Private Placement. The following provision shall replace Section 14 of the Agreement:

The grant of the Option hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;
- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and
- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;
- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y
- d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente. **FRANCE**

1. <u>Nature of Grant</u>. The Option is not granted under the French specific regime provided by Articles L.225-177 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Sale of Shares of Stock</u>. Shares of Stock purchased upon exercise of the Option are accepted as a personal investment. In the event that shares of Stock are issued in respect of the Option within six (6) months after the Grant Date, you agree that the shares of Stock may not be offered to the public or otherwise disposed of prior to the sixmonth anniversary of the Grant Date.

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement the Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Option and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Option and shares of Stock subject to the Option do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

1. <u>Plan Document Acknowledgment</u>. In accepting the Option, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 2: Vesting/Exercise/Expiration (expiration of the right to exercise the Option after the Expiration Date); Section 3: Disability (term for exercising the Option prior to the Vesting Dates in the case of a Termination of Service due to Disability); Section 4: Death (term for exercising the option prior to the Vesting Dates in the case of a Termination of Service due to death); Section 5: Retirement (term for exercising the Option prior to the Vesting Dates in the case of a Termination of Service by reason of Retirement); Section 6: Termination of Service Following a Change in Control (term for exercising the Option in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (term to exercise the vested Option and forfeiture of the unvested Option in other cases of Termination of Service); Section 8: Forfeiture of Outstanding Options Upon Termination for Cause or Following Termination of Service; Section 10(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Option and legally applicable to the participant); Section 11: Limited Transferability (Option shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 18: Change in Stock (right of the Company to equitably adjust the Option and the Exercise Price in the event of any change in the Stock); Section 19(j): Nature of the Option (waive any claim or entitlement to compensation or damages arising from forfeiture of the Option resulting from a Termination of Service); Section 19(I): Nature of the Option (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Option); Section 22: Non-Competition, Non-Solicitation and Confidentiality (the receipt of the Option is conditioned upon agreement of the Non-Competition, Non-Solicitation and Confidentiality Agreement attached hereto as Exhibit A); Section 23: Addendum to Agreement (the Option is subject to the terms of the Addendum); Section 24: Additional Requirements (Company right to impose additional requirements on the Option in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Option and the Plan): Section 26: Electronic Delivery (Company may deliver documents related to the Option or Plan electronically); Section 27: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 28: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Option does not constitute an employment relationship between you and the Company. You have been granted the Option as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you ("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan are not part of the employment conditions and/or benefits provided by WBA Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. Extraordinary Item of Compensation. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Option, the shares of Stock subject to the Option and the value of the same are an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

<u>Exclusion of Claim</u>. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Option, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Option. Upon the grant of the Option, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

SPAIN

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Option, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted an Option under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Option is granted on the assumption and condition that the Option and the shares of Stock acquired upon exercise of the Option shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Option shall be null and void.

Further, you understand and agree that the vesting of the Option is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Option may cease vesting immediately, in whole or in part, effective on the date of your Termination of Service (unless otherwise specifically provided in Section 3, 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to the Option that were not vested on the date of your Termination of Service, as described in the Plan and Agreement. In addition, you understand and agree that the post-Termination of Service exercise period specified in the Agreement shall run from the date of your Termination of Service, as determined by the Committee, in its sole discretion.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on the Option.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 8 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Option. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Option have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

<u>Securities Law Notification</u>. The Option is not considered a public offering in Switzerland; therefore, it is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Option constitute a prospectus as such term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Option may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Option have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. Indemnification for Tax-Related Items. Without limitation to Section 10 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 10 of the Agreement.

2. <u>Exclusion of Claim</u>. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Option, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Option. Upon the grant of the Option, you shall be deemed irrevocably to have waived any such entitlement.

3. <u>Post-Termination Restrictions</u>. To the extent that you are employed by your Employer pursuant to an employment agreement governed by the laws of England, Wales, Scotland and/or Northern Ireland, Paragraphs 2 and 3 of the NNCA Agreement attached to the Agreement as Exhibit A shall not apply to you.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Stock Option Award Agreement to which this Addendum is attached as Exhibit B, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

PERFORMANCE SHARE AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of your company's securities or financial instruments. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

PERFORMANCE SHARE AWARD AGREEMENT

Participant Name: Stefano Pessina

Participant ID:

Grant Date:

Performance Period: Fiscal Years - 2020 - 2022 (the "Performance Period")

Shares Granted:

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the Performance Share Award (the "Award") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Performance Shares</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you the target number of Performance Shares specified above (the "Performance Shares"), subject to the terms and conditions of the Plan and this Agreement. This "target" number of shares is computed by dividing a target award dollar amount approved for you by the Committee by the average closing stock price of the Company's common stock, par value US\$.01 per share ("Stock") for the last 30 trading days of the fiscal year preceding the Grant Date.

2. <u>Performance Measure</u>. The number of Performance Shares earned at the end of the three-year Performance Period will vary depending on the degree to which cumulative adjusted earnings per share performance goals for the Performance Period, as established by the Committee, are met.

3. <u>Determination of Performance Shares Earned</u>. At the target levels, 100% of the Performance Shares will be earned. At the threshold levels, 50% of the Performance Shares will be earned. Below the threshold levels of performance, no Performance Shares are earned. At the maximum levels or more, 150% of the Performance Shares will be earned. Performance between minimum and target, and between target and maximum, will earn Performance Shares on a pro-rated basis between 50% and 100%, and 150%, respectively.

The amount earned will be calculated according to the following:

Percent of

Performance	=	Target	Х	Target
Shares Awarded		Performance Shares		Performance Shares Earned

4. <u>Disability or Death</u>. If during the Performance Period you have a Termination of Service by reason of Disability or death, then the number of Performance Shares earned (based on performance as of the end of the Performance Period) shall become vested at the end of the Performance Period. Any Performance Shares becoming vested by reason of your Termination of Service by reason of Disability or death shall be paid at the same time Performance Shares are paid to other Participants.

5. Retirement. If prior to the 12-month anniversary of the Grant Date you have a Termination of Service by reason of retirement from the Company's Board of Directors, as reasonably determined and approved by the Committee, then, subject to such approval, the number of Performance Shares earned (based on performance as of the end of the Performance Period) will be prorated to reflect the portion of the Performance Period during which you remained employed by the Company. Such prorated portion shall equal the number of Performance Shares that you would otherwise have earned, multiplied by a fraction equal to the number of full months of the Performance Period completed as of your Termination of Service, divided by the number of months in the Performance Period. If on or after the 12-month anniversary of the Grant Date, you have a Termination of Service by reason of retirement from the Company's Board of Directors, as reasonably determined and approved by the Committee, then, subject to such approval, the full number of Performance Shares earned (based on performance as of the end of the Performance Period) shall become vested at the end of the Performance Period. Any Performance Shares becoming vested by reason of your retirement shall be paid at the same time Performance Shares are paid to other Participants.

6. <u>Termination of Service Following a Change in Control</u>. If during the Performance Period there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 7), then your earned Award shall equal your target number of Performance Shares, prorated to reflect the portion of the Performance Period during which you remained employed by the Company. Such prorated portion shall equal your target number of Performance Shares, multiplied by a fraction equal to the number of full months of the Performance Period award will be settled in cash (subject to required tax withholdings) in accordance with Section 9.01(b) of the Plan within 45 days after your Termination of Service. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such Plan.

7. <u>Other Termination of Service</u>. If during the Performance Period you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 4, 5 or 6 above, as determined by the Committee, then all of your Performance Shares shall be forfeited. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

8. <u>Settlement of Earned Performance Shares</u>. At the end of the Performance Period actual performance for the entire Performance Period shall be reviewed, and the amount of the earned Award shall be determined based on this performance and communicated to you. Subject to the requirements of Section 12 below, the Company shall transfer to you one (1) share of Stock for each Performance Share earned at that time, net of any applicable tax withholding requirements in accordance with Section 9 below. The Performance Shares payable under this Agreement are intended to be exempt from Code Section 409A under the exemption for short-term deferrals. Accordingly, the Performance Shares will be settled in shares of Stock no later than the 15th day of the third month following the end of the fiscal year of the Company (or if later, the calendar year) in which the Performance Shares are earned.

Notwithstanding the foregoing, if you are resident or employed outside of the U.S., the Company, in its sole discretion, may provide for the settlement of the Performance Shares in the form of:

(a) a cash payment (in an amount equal to the Fair Market Value of the shares of Stock that corresponds with the number of earned Performance Shares) to the extent that settlement in shares of Stock (i) is prohibited under local law, (ii) would require you, the Company or an Affiliate to obtain the approval of any governmental or regulatory body in your country of residence (or country of employment, if different), (iii) would result in adverse tax consequences for you, the Company or an Affiliate or (iv) is administratively burdensome; or

(b) shares of Stock, but require you to sell such shares of Stock immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

9. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or any Affiliate, the ultimate liability for all income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("Tax-Related Items"), is and remains your responsibility and may exceed the amount actually withheld by the Company or any Affiliates, if any. You further acknowledge that the Company and/or any Affiliate (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Award, including, but not limited to, the grant, vesting or settlement of the Award, the subsequent sale of shares of Stock acquired pursuant to such settlement and the receipt of any dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Award to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or any Affiliate may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or any Affiliate to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon settlement of the Award that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Performance Shares and the issuance of shares of Stock hereunder.

Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, your Employer or its agent to cover the Tax-Related Items that the Company or any Affiliate may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or any Affiliate may (in its sole discretion) allow you to make a cash payment to the Company, your Employer or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon settlement of the Award, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the earned Award, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock (or cash payment) or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

10. <u>Nontransferability</u>. During the Performance Period and thereafter until shares of Stock are transferred to you in settlement thereof, you may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Performance Shares, whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, or by will or by the laws of intestacy.

11. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the Performance Shares until such time as a certificate of stock for the shares of Stock issued in settlement of the Performance Shares has been issued to you or such shares of Stock have been recorded in your name in book entry form. Except as provided in Section 17 below, no adjustment shall be made for dividends or distributions or other rights with respect to such shares of Stock for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

12. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares of Stock except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

13. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Performance Shares is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Performance Shares is not subject to the supervision of the local securities authorities.

14. <u>Insider Trading/Market Abuse Laws</u>. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept,

acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock, or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

15. <u>Repatriation; Compliance with Law</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of residence to comply with your personal obligations under local laws, rules and/or regulations in your country of employment, if different).

16. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Performance Shares. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

17. <u>Change in Stock</u>. In the event of any change in the shares of Stock, by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of the Performance Shares subject to this Award Agreement shall be equitably adjusted by the Committee.

18. <u>Nature of the Award</u>. In accepting the Award, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Award is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of Performance Shares, or benefits in lieu of Performance Shares, even if Performance Shares have been granted in the past;

(c) all decisions with respect to future Awards or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of the Award, the number of shares of Stock subject to the Award, and the earning provisions applicable to the Award;

(d) the Award and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Award and the shares of Stock subject to the Award are not intended to replace any pension rights or compensation;

(g) the Award, the shares of Stock subject to the Award and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or any Affiliate;

(h) the future value of the shares of Stock underlying the Award is unknown, indeterminable and cannot be predicted with certainty;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Award resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Award to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Award and the benefits evidenced by this Agreement do not create any entitlement to have the Award or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Award or of any amounts due to you pursuant to the settlement of the Award or the subsequent sale of any shares of Stock acquired upon settlement of the Award.

19. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

20. <u>Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Performance Shares and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and its Affiliates hold certain personal information about you, specifically: your name, home address, email address and telephone number, date of birth, social

security, passport or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and/or its Affiliates will process Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and its Affiliates will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or any Affiliate may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of Data, (iv) oppose, for legal reasons, the collection, processing or transfer of Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, your Performance Shares will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

21. <u>Addendum to Agreement</u>. Notwithstanding any provision of this Agreement to the contrary, the Performance Shares shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to this Agreement, attached hereto as Exhibit A (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Performance Shares and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

22. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Performance Shares, any shares of Stock acquired pursuant to the Performance Shares and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Performance Shares and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

23. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

24. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Award or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

25. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdiction.

26. <u>English Language</u>. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Award, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English so as to allow you to understand the terms and conditions of the Agreement, the Plan or any other documents related to the Award. If you have received this Agreement, the Plan or any other documents related into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

27. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

28. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibit A. Once you have read and understood this Agreement and Exhibit A, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibit A and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Award granted hereunder.

EXHIBIT A

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

PERFORMANCE SHARE AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Award is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 21 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Performance Shares and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

EUROPEAN UNION ("EU") / EUROPEAN ECONOMIC AREA ("EEA")

Personal Data. The following provision replaces Section 20 of the Agreement in its entirety:

The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all Performance Shares, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Performance Shares under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. As such, by accepting the Performance Shares, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the Performance Shares. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU

Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) <u>International Data Transfers</u>. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) <u>Data Subject Rights</u>. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources Department in writing. Your provision of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Performance Shares, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources Department in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

CHILE

Private Placement. The following provision shall replace Section 13 of the Agreement:

The grant of the Performance Shares hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;
- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and
- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;

- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y
- d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente.

FRANCE

1. <u>Nature of Grant</u>. The Performance Shares are not granted under the French specific regime provided by Articles L. 225-197-1 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Form of Payment</u>. Notwithstanding any provision in the Agreement or Plan to the contrary, the Performance Shares shall be settled only in Shares (and not in cash).

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement, this Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Performance Shares and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Performance Shares and shares of Stock subject to the Performance Shares do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

<u>Plan Acknowledgment</u>. In accepting the Performance Shares, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 3: Determination of Performance Shares Earned (threshold levels for earning

Performance Shares); Section 4: Disability or Death (terms of payment of Performance Shares upon a Termination of Service by reason of Disability or death); Section 5: Retirement (terms of payment of Performance Shares upon a Termination of Service by reason of retirement); Section 6: Termination of Service Following a Change in Control (terms of payment of Performance Shares in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (forfeiture of Performance Shares in other cases of Termination of Service); Section 9(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Performance Shares and legally applicable to the participant); Section 10: Nontransferability (Performance Shares shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 17: Change in Stock (right of the Company to equitably adjust the number of Performance Shares subject to this Agreement in the event of any change in Stock); Section 18(j): Nature of the Award (waive any claim or entitlement to compensation or damages arising from forfeiture of the Performance Shares resulting from a Termination of Service); Section 18(I): Nature of the Award (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Performance Shares); Section 19: Committee Authority; Recoupment (right of the Committee to administer, construe, and make all determinations necessary or appropriate for the administration of the Performance Shares and this Agreement, including the enforcement of any recoupment policy); Section 21: Addendum to Agreement (the Performance Shares are subject to the terms of the Addendum); Section 22: Additional Requirements (Company right to impose additional requirements on the Performance Shares in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Performance Shares and the Plan); Section 24: Electronic Delivery (Company may deliver documents related to the Award or Plan electronically); Section 25: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 26: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Performance Shares does not constitute an employment relationship between you and the Company. You have been granted the Performance Shares as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you ("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan do not establish of the employment conditions and/or benefits provided by WBA Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. <u>Extraordinary Item of Compensation</u>. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Award, the shares of Stock subject to the Award and the income and value of the same is an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Performance Shares, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Performance Shares. Upon the grant of Performance Shares, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

SPAIN

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Award, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted Performance Shares under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Performance Shares are granted on the assumption and condition that the Performance Shares and the shares of Stock acquired upon settlement of the Performance Shares shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Award shall be null and void.

Further, you understand and agree that the earning of the Performance Shares is expressly conditioned on your continued and active rendering of service, such that upon a Termination of

Service, the Performance Shares may be forfeited effective on the date of your Termination of Service (unless otherwise specifically provided in Section 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to the Performance Shares as of the date of your Termination of Service, as described in the Plan and Agreement.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on your Award.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 7 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Performance Shares. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Performance Shares have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

Securities Law Notification. The Performance Shares are not considered a public offering in Switzerland; therefore, the offer of Performance Shares is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Performance Shares constitute a prospectus as such term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Performance Shares may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Performance Shares have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. Indemnification for Tax-Related Items. Without limitation to Section 9 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national

insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 10 of the Agreement.

2. Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Performance Shares, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Performance Shares. Upon the grant of the Performance Shares, you shall be deemed irrevocably to have waived any such entitlement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Performance Share Award Agreement to which this Addendum is attached as Exhibit A, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of your company's securities or financial instruments. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

Participant Name: Stefano Pessina

Participant ID:

Grant Date:

Grant Price:

Shares Granted:

Vesting: One third of the Shares Granted vest on each of the first, second and third anniversaries of the Grant Date (the "Vesting Dates")

Expiration Date:

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the stock option (the "Option") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as in effect on the date of this Agreement and as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Option</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you an Option to purchase all or any part of the number of Shares Granted set forth above of common stock of the Company, par value US\$.01 ("Stock"), at the per-share exercise price, which is 100% of the fair market value of a share of Stock on the Grant Date (the "Exercise Price"), subject to the terms and conditions of the Plan and this Agreement. The Option is intended to be a "non-qualified stock option" and shall not be treated as an incentive stock option within the meaning of Section 422 of the Code.

2. <u>Vesting/Exercise/Expiration</u>. You may not exercise the Option prior to the Vesting Date or Dates set forth above absent action by the Committee to waive or alter such restrictions or as may be permitted under the below paragraphs. Thereafter, except as hereinafter provided, you may exercise the Option, to the extent it is vested, at any time and from time to time until the close of business on the Expiration Date set forth above. The Option may be exercised to purchase any number of whole shares of Stock, except that no purchase shall be for less than ten (10) full shares of Stock, or the remaining unexercised shares, if less. The Option is deemed to be "outstanding" until it has been exercised in full or expired pursuant to the terms of this Agreement.

3. <u>Disability</u>. If, without having fully exercised the Option, you have a Termination of Service due to Disability, then any Shares Granted under the Option that are not yet vested at that time shall thereupon become vested and (a) you may exercise the Option for the full number of Shares Granted (less any shares for which the Option was previously exercised), but (b) your right to exercise the Option shall terminate upon the earlier of the Expiration Date or a date which is one (1) year following the date of your Termination of Service.

4. <u>Death</u>. If, without having fully exercised the Option, you have a Termination of Service due to your death, then any Shares Granted under the Option that are not yet vested at that time shall thereupon become fully vested and (a) the Option may be exercised by the executor or administrator of your estate or by such person or persons who shall have acquired your rights hereunder by bequest or inheritance or by designation as your beneficiary for the full number of Shares Granted (less any shares for which the Option was previously exercised), but (b), such person's right to exercise the Option shall terminate upon the earlier of the Expiration Date or a date which is one (1) year after the date of your death.

5. <u>Retirement</u>. If without having fully exercised the Option you have a Termination of Service by reason of retirement from the Company's Board of Directors as reasonably determined and approved by the Committee, then, subject to such approval (a) any Shares Granted under the Option that are not vested at that time shall thereupon become vested, and (b) you may exercise the Option for the full number of Shares Granted (less any shares for which the Option was previously exercised) until the Expiration Date set forth above.

6. Termination of Service Following a Change in Control. If there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 8), then any Shares Granted under the Option that are not yet vested at that time shall thereupon become vested, and you may exercise the Option for the full number of Shares Granted (less any shares for which the Option under this Section 6 shall be forfeited. The foregoing is also subject to the Committee's exercise of its discretion under Section 9.01 of the Plan. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such plan.

7. <u>Other Termination of Service</u>. If without having fully exercised the Option you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 3, 4, 5 or 6 above, as determined by the Committee, then (a) for any Shares Granted with respect to which such Termination of Service is prior to the applicable Vesting Date, the Option shall be forfeited, and (b) for any Shares Granted with respect to which such Terminate upon the earlier of the Expiration Date or a date which is ninety (90) days after the date of your Termination of Service. The foregoing is subject to the right of the Committee to extend the exercise period of the Option, including any extension granted by the Committee or its delegate as needed to allow your right to exercise to extend beyond a period during which you are restricted from exercising the Option due to a Company-designated trading blackout period, and is subject to earlier expiration as provided in Section 8 below.

8. <u>Forfeiture of Outstanding Options Upon Termination for Cause or Upon Other Violations</u>. Notwithstanding any provision of this Agreement to the contrary, the Option (whether vested or unvested) shall immediately terminate if you are terminated for Cause or if and when you violate any obligation that you may have to the Company during or post-employment, including but not limited to a violation of any applicable provision of the NNCA Agreement (defined in Section 22 below) or any other non-competition, non-solicitation, confidentiality, non-disparagement or other restrictive covenant. For purposes of this

Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

9. Exercise Process. The Option may be exercised by giving notice to Fidelity Stock Plan Services, LLC ("Fidelity"), the third party administrator administering the Option exercise process or any other third party administrator the Company may engage in the future. The exercise notice (a) shall be signed by you or (in the event of your death) your legal representative, (b) shall specify the number of full shares of Stock then elected to be purchased, and (c) shall be accompanied by payment in full of the Exercise Price of the shares of Stock to be purchased. Payment may be made in cash or by check payable to the order of the Company, and such payment shall include any income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan that are required to be withheld ("Tax-Related Items"), as set forth in Section 10 below. Alternatively, the Committee may allow for one or more of the following methods of exercising the Option:

(a) Payment for shares of Stock as to which the Option is being exercised and/or payment of any Tax-Related Items may be made by transfer to the Company of shares of Stock you already own, or any combination of such shares of Stock and cash, having a fair market value determined at the time of exercise of the Option equal to, but not exceeding, the Exercise Price and/or any Tax-Related Items, as the case may be.

(b) A "same day sale" transaction pursuant to which a third party (engaged by you or the Company) loans funds to you to enable you to purchase the shares of Stock and pay any Tax-Related Items, and then sells a sufficient number of the exercised shares of Stock on your behalf to enable you to repay the loan and any fees. The remaining shares of Stock and/or cash are then delivered by the third party to you.

(c) A "net exercise" transaction, pursuant to which the Company delivers to you the net number of whole shares of Stock remaining from the portion of the Option being exercised after deduction of a number of shares of Stock with a fair market value equal to the Exercise Price and the amount of any Tax-Related Items.

As promptly as practicable after receipt of such notice of exercise and payment (including payment with respect to any Tax-Related Items), subject to Section 13 below, the Company shall cause to be issued and delivered to you (or in the event of your death to your legal representative, as the case may be), certificates for the shares of Stock so purchased. Alternatively, such shares of Stock may be issued and held in book entry form. Notwithstanding any provision within this Agreement to the contrary, if you are resident or employed outside of the U.S., the Committee may require that you (or in the event of your death, your legal representative, as the case may be) exercise the Option in a method other than as specified above, may require you to exercise the Option only by means of a "same day sale" transaction (either a "sell-all" transaction or a "sell-to-cover" transaction) as it shall determine in its sole discretion, or may require you to sell any shares of Stock you acquire under the Plan immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

10. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or any Affiliate, the ultimate liability for all Tax-Related Items related to your participation in the Plan and legally applicable to you is and remains your responsibility and may exceed the amount actually withheld by the Company or any Affiliate, if any. You further acknowledge that the Company and/or any Affiliate (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Option, including, but not limited to, the grant, vesting or exercise of the Option, the subsequent sale of shares of Stock acquired pursuant to such exercise and the receipt of any dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Option to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or any Affiliate may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or any Affiliate to satisfy all Tax-Related Items. In this regard, except as provided below, the Company or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon exercise of the Option that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Option and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, any Affiliate or its agent to cover the Tax-Related Items that the Company or any Affiliate may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or any Affiliate may (in its sole discretion) allow you to make a cash payment to the Company, any Affiliate or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon exercise of the Option, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the exercised Option, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

11. <u>Limited Transferability</u>. You may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Option, whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, by will or by the laws of intestacy. During your lifetime, the Option and all rights granted hereunder shall be exercisable only by you. Notwithstanding the foregoing, you may transfer the Option, in whole or in part, by gift to a Permitted Transferee in accordance with rules and subject to any conditions specified by the Committee under the Plan.

12. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the shares of Stock subject to the Option until such time as the Exercise Price has been paid and a certificate of stock for such shares has been issued to you or such shares of Stock have been recorded in your name in book entry form. Except as provided in Section 18 below, no adjustment shall be made for dividends or distributions or other rights with respect to such shares of Stock for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

13. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares of Stock except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

14. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Option is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Option is not subject to the supervision of the local securities authorities.

15. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock (e.g., the Option) or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

16. <u>Repatriation; Compliance with Law; Method of Exercise</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of employment, if different). Finally, you agree to take any and all actions as may be required

to comply with your personal obligations under local laws, rules and/or regulations in your country of residence (and country of employment, if different).

17. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Option. Investment in shares of Stock involves a degree of risk. Before deciding to purchase shares of Stock pursuant to the Option, you should carefully consider all risk factors relevant to the acquisition of shares of Stock under the Plan and you should carefully review all of the materials related to the Option and the Plan. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

18. <u>Change in Stock</u>. In the event of any change in the shares of Stock by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of shares of Stock subject to the Option and the Exercise Price shall be equitably adjusted by the Committee.

19. <u>Nature of the Option</u>. In accepting the Option, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Option is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of stock options, or benefits in lieu of stock options, even if stock options have been granted in the past;

(c) all decisions with respect to future grants of stock options or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of any grant, the number of shares of Stock subject to the stock options, vesting provisions, and the exercise price applicable to the stock option;

(d) the Option and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Option and the shares of Stock subject to the Option are not intended to replace any pension rights or compensation;

(g) the Option, the shares of Stock subject to the Option and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or any Affiliate;

(h) the future value of the shares of Stock underlying the Option is unknown, indeterminable and cannot be predicted with certainty and if you exercise a vested Option, the value of the underlying shares of Stock may increase or decrease, even below the Exercise Price;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Option resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Option to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Option and the benefits evidenced by this Agreement do not create any entitlement to have the Option or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Option or of any amounts due to you pursuant to the settlement of the Option or the subsequent sale of any shares of Stock acquired upon settlement of the Option.

20. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

21. <u>Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Option and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and its Affiliates hold certain personal information about you, specifically: your name, home address, email address and telephone number, date of birth, social security, passport or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and/or its Affiliates will process Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and its Affiliates may transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or its Affiliates may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of Data, (iv) oppose, for legal reasons, the collection, processing or transfer of the Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, the Option will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

22. <u>Non-Competition, Non-Solicitation and Confidentiality</u>. As a condition to the receipt of the Option, you must agree to the Non-Competition, Non-Solicitation and Confidentiality Agreement (the "NNCA Agreement") attached hereto as Exhibit A. By clicking the acceptance box for this Agreement, you also agree to the terms and conditions expressed in the NNCA Agreement. Failure to accept the terms of this Agreement and NNCA Agreement within 120 days of the Grant Date shall constitute your decision to decline to accept this Award.

23. <u>Addendum to Agreement</u>. Notwithstanding any provision of this Agreement to the contrary, the Option shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to this Agreement, attached hereto as Exhibit B (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

24. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Option, any shares of Stock acquired pursuant to the Option and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

25. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

26. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Option or your future participation in the Plan. You hereby consent to receive

such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

27. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdiction.

28. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Option, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English, so as to allow you to understand the terms and conditions of this Agreement, the Plan or any other document related to the Option. If you have received this Agreement, the Plan or any other document related into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

29. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

30. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibits A and B. Once you have read and understood this Agreement and Exhibits A and B, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibits A and B, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Option granted hereunder.

EXHIBIT A

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

This Exhibit (the "Non-Compete Agreement") forms a part of the Stock Option Award Agreement covering Options awarded to an employee ("Employee" or "I") of Walgreens Boots Alliance, Inc. or an affiliate thereof, on behalf of itself, its affiliates, subsidiaries, and successors (collectively referred to as the "Company").

WHEREAS, the Company develops and/or uses valuable business, technical, proprietary, customer and patient information it protects by limiting its disclosure and by keeping it secret or confidential;

WHEREAS, Employee acknowledges that during the course of employment, he or she has or will receive, contribute, or develop such Confidential Information and Trade Secrets (as defined below); and

WHEREAS, the Company desires to protect from its competitors such Confidential Information and Trade Secrets and also desires to protect its legitimate business interests and goodwill in maintaining its employee and customer relationships.

NOW THEREFORE, in consideration of the Stock Option issued to Employee pursuant to the Agreement to which this is attached as Exhibit A and for other good and valuable consideration, including but not limited to employment or continued employment, the specialized knowledge, skill and training that the Company provides Employee, the goodwill that Employee develops with customers on behalf of the Company, Employee agrees to be bound by the terms of this Non-Compete Agreement as follows:

1. <u>Confidentiality</u>.

(a) At all times during and after the termination of my employment with the Company, I will not, without the Company's prior written permission, directly or indirectly for any purpose other than performance of my duties for the Company, utilize or disclose to anyone outside of the Company any Trade Secrets (defined in subparagraph 1(a)(i)) or other Confidential Information (defined in subparagraph 1(a)(ii)) or any information received by the Company in confidence from or about third parties, as long as such matters remain Trade Secrets or otherwise confidential.

(i) For purposes of this Non-Compete Agreement, "**Trade Secrets**" means a form of intellectual property that are protectable under applicable state and/or Federal law, including the Uniform Trade Secrets Act (as amended and adapted by the states) and the Federal Defend Trade Secrets Act of 2016 (the "DTSA"). They include all tangible and intangible (*e.g.*, electronic) forms and types of information that is held and kept confidential by the Company and is not generally known outside of the Company, including but not limited to information about: the Company's financial, business, scientific, technical, economic, or engineering information, including patterns, plans, compilations, program devices, formulas, designs, prototypes, methods, techniques, processes, procedures, programs or codes, and may in particular include such things as pricing information, business records, software programs, algorithms, inventions, patent applications, and designs and processes not known outside the Company.

(ii) For purposes of this Non-Compete Agreement, "**Confidential Information**" means Trade Secrets and, more broadly, any other tangible and intangible (*e.g.*, electronic) forms and types of information that are held and kept confidential by the Company and are not generally known outside the Company, and which relates to the actual or anticipated business of the Company or the Company's actual or prospective vendors or clients. Confidential Information shall not be considered generally known to the public if is revealed improperly to the public by me or others without the Company's express written consent and/or in violation of an obligation of confidentiality to the Company. Examples of Confidential Information include, but are not limited to: customer, referral source, supplier and contractor identification and contacts; special contract terms; pricing and margins; business, marketing and customer plans and strategies; financial data; company created (or licensed) techniques; technical know-how; research, development and production information; processes, prototypes, software, patent applications and plans, projections, proposals, discussion guides, and/or personal or performance information about employees.

(b) I understand that this obligation of non-disclosure shall last so long as the information remains confidential. I, however, understand that, if I live and work primarily in Wisconsin, Virginia, or any other state requiring a temporal limit on non-disclosure clauses, Confidential Information shall be protected for two (2) years following termination of my employment (for any reason). I also understand that Trade Secrets are protected by statute and are not subject to any time limits. I also

agree to contact the Company before using, disclosing, or distributing any Confidential Information or Trade Secrets if I have any questions about whether such information is protected information.

The restrictions set forth in this paragraph are in addition to and not in lieu of any obligations I have by (C) law with respect to the Company's Confidential Information, including any obligations I may owe under the DTSA and any applicable state statutes. Nothing herein shall prohibit me from divulging evidence of criminal wrongdoing to law enforcement or prohibit me from disclosing Confidential Information or Trade Secrets if compelled by order of court or an agency of competent jurisdiction or as required by law; however, I shall promptly inform the Company of any such situations and shall take reasonable steps to prevent disclosure of Confidential Information or Trade Secrets until the Company has been informed of such required disclosure and has had a reasonable opportunity to seek a protective order. Pursuant to the DTSA, I understand that an individual may not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a Trade Secret that: (A) is made (i) in confidence to a federal, state or local government official, either directly or indirectly, or to an attorney, and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. Additionally, I understand that an individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the Trade Secret to his or her attorney and use the Trade Secret information in the court proceeding, so long as any document containing the Trade Secret is filed under seal and the individual does not disclose the Trade Secret, except pursuant to court order. Nothing in this Non-Compete Agreement is intended to conflict with the DTSA or create liability for disclosures of Trade Secrets that are expressly allowed by DTSA.

2. Non-Competition. I agree that during my employment with the Company and for twelve (12) months after the termination of my employment (for any reason), I will not, directly or indirectly have Responsibilities with respect to any Competing Business Line. As set forth in paragraph 9(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. These restrictions shall not apply to passive investments of less than five percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "Responsibilities" means the same or similar material responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company and within the same geographic scope, or portion thereof, where I performed those responsibilities for the Company. For purposes of this Non-Compete Agreement, "Competing Business Line" means any business that is in competition with any business engaged in by the Company and for which I had Responsibilities during the two (2) years prior to my last day of employment with the Company. Competing Business Line shall also include businesses or business lines that may not be directly competitive with the Company in most respects (such as pharmacy benefit managers), but only to the extent I am engaged by any such business in a role: (a) that involves my performing Responsibilities for Competing Products or Services; or (b) where I would be called upon to inevitably rely upon or disclose Confidential Information and such reliance or disclosure would competitively harm the Company. For purposes of this Non-Compete Agreement, "Competing Products or Services" means products or services that are competitive with products or services offered by, developed by, designed by or distributed by the Company during the two (2) years prior to my last day of employment with the Company.

3. <u>Non-Solicitation</u>. I agree that during my employment with the Company and for two (2) years after the termination of my employment from the Company (for any reason):

(a) I will not directly or indirectly, solicit any Restricted Customer for purposes of providing Competing Products or Services, or offer, provide or sell Competing Products or Services to any Restricted Customer. For purposes of this Non-Compete Agreement, "**Restricted Customer**" means any person, company or entity that was a customer, vendor, supplier or referral source of the Company and with which I had direct contact for purposes of performing responsibilities for the

Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company. To the extent permitted by applicable law, "**Restricted Customer**" also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company; and

(b) I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I had direct contact work during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company; (ii) interfere with the performance by any such employee of his/her duties for the Company; and/or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b).

4. <u>Non-Inducement</u>. I will not directly or indirectly assist or encourage any person or entity in carrying out or conducting any activity that would be prohibited by this Non-Compete Agreement if such activity were carried out or conducted by me.

5. <u>Non-Disparagement</u>. During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees, shareholders, members, agents or products. The foregoing shall not be violated by truthful statements in response to legal process, required governmental testimony or filings, or administrative or arbitral proceedings (including, without limitation, depositions in connection with such proceedings); and the foregoing shall not apply to any claims for harassment or discrimination to the extent so restricted by applicable state law.

6. Intellectual Property. The term "Intellectual Property" shall mean all trade secrets, ideas, inventions, designs, developments, devices, software, computer programs, methods and processes (whether or not patented or patentable, reduced to practice or included in the Confidential Information) and all patents and patent applications related thereto, all copyrights, copyrightable works and mask works (whether or not included in the Confidential Information) and all patents and patent applications related thereto, all copyrights, copyrightable works and mask works (whether or not included in the Confidential Information) and all registrations and applications for registration related thereto, all Confidential Information, and all other proprietary rights contributed to, or conceived or created by, or reduced to practice by Employee or anyone acting on his/her behalf (whether alone or jointly with others) at any time from the beginning of Employee's employment with the Company to the termination of that employment plus ninety (90) days, that (i) relate to the business or to the actual or anticipated research or development of the Company; (ii) result from any services that Employee or anyone acting on its behalf perform for the Company; or (iii) are created using the equipment, supplies or facilities of the Company or any Confidential Information.

a. <u>Ownership</u>. All Intellectual Property is, shall be and shall remain the exclusive property of the Company. Employee hereby assigns to the Company all right, title and interest, if any, in and to the Intellectual Property; provided, however, that, when applicable, the Company shall own the copyrights in all copyrightable works included in the Intellectual Property pursuant to the "work-made-for-hire" doctrine (rather than by assignment), as such term is defined in the 1976 Copyright Act. All Intellectual Property shall be owned by the Company irrespective of any copyright notices or confidentiality legends to the contrary that may be placed on such works by Employee or by others. Employee shall ensure that all copyright notices and confidentiality legends on all work product authored by Employee or anyone acting on his/her behalf shall conform to the Company's practices and shall specify the Company as the owner of the work. The Company hereby provides notice to Employee that the obligation to assign does not apply to an invention for which no equipment, supplies, facility, or Trade Secrets of the Company was used and which was developed entirely on the Employee's own time, unless (i) the invention relates (1) to the business of the Company, or (2) to the Company's actual or demonstrably anticipated research or development, or (ii) the invention results from any work performed by Employee for the Company.

b. <u>Keep Records</u>. Employee shall keep and maintain, or cause to be kept and maintained by anyone acting on his/her behalf, adequate and current written records of all Intellectual Property in the form of notes, sketches, drawings, computer files, reports or other documents relating thereto. Such records shall be and shall remain the exclusive property of the Company and shall be available to the Company at all times during my employment with the Company.

c. <u>Assistance</u>. Employee shall supply all assistance requested in securing for the Company's benefit any patent, copyright, trademark, service mark, license, right or other evidence of ownership of any such Intellectual Property, and will provide full information regarding any such item and execute all appropriate documentation prepared by Company in applying or otherwise registering, in the Company's name, all rights to any such item or the defense and protection of such Intellectual Property.

d. <u>Prior Inventions</u>. Employee has disclosed to the Company any continuing obligations to any third party with respect to Intellectual Property. Employee claims no rights to any inventions created prior to his/her employment for which a patent application has not previously been filed, unless he/she has described them in detail on a schedule attached to this Non-Compete Agreement.

e. <u>Trade Secret Provisions</u>. The provisions in paragraph 1 of this Non-Compete Agreement with regard to Trade Secrets and the DTSA shall apply as well in the context of the parties' Intellectual Property rights and obligations.

7. <u>Return of Company Property</u>. I agree that all documents and data accessible to me during my employment with the Company, including Confidential Information and Trade Secrets, regardless of format (electronic or hard copy), including but not limited to any Company computer, monitor, printer equipment, external drives, wireless access equipment, telecom equipment and systems ("Company Equipment"), are and remain the sole and exclusive property of the Company and/or its clients, and must be returned to the Company upon separation or upon demand by the Company. I further agree that I will provide passwords to access such Company Equipment and I will not print, retain, copy, destroy, modify or erase Company U.S. data on Company Equipment or otherwise wipe Company Equipment prior to returning the Company Equipment.

8. <u>Consideration and Acknowledgments</u>. I acknowledge and agree that the covenants described in this Non-Compete Agreement are essential terms, and the underlying Stock Option Award would not be provided by the Company in the absence of these covenants. I further acknowledge that these covenants are supported by adequate consideration as set forth in this Non-Compete Agreement and are not in conflict with any public interest. I further acknowledge and agree that I fully understand these covenants, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms. I further acknowledge and agree that these covenants are reasonable and enforceable in all respects.

9. Enforceability; General Provisions.

(a) I agree that the restrictions contained in this Non-Compete Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Non-Compete Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place undue restraint on me. I further understand that the restrictions in this Non-Compete Agreement apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary.

(b) Because the Company's current base of operations is in Illinois and my connections thereto, (i) this Non-Compete Agreement shall be governed by and construed in accordance with

the laws of the State of Illinois, where this Non-Compete Agreement is entered into, without giving effect to any conflict of law provisions, and (ii) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute or declaration arising out of this Non-Compete Agreement.

(c) In the event of a breach or a threatened breach of this Non-Compete Agreement, I acknowledge that the Company will face irreparable injury which may be difficult to calculate in dollar terms and that the Company shall be entitled, in addition to all remedies otherwise available in law or in equity, to temporary restraining orders and preliminary and final injunctions enjoining such breach or threatened breach in any court of competent jurisdiction without the necessity of posting a surety bond, as well as to obtain an equitable accounting of all profits or benefits arising out of any violation of this Non-Compete Agreement.

(d) I agree that if a court determines that any of the provisions in this Non-Compete Agreement is unenforceable or unreasonable in duration, territory, or scope, then that court shall modify those provisions so they are reasonable and enforceable, and enforce those provisions as modified.

(e) If any one or more provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement or its application is determined to be invalid, illegal, or unenforceable to any extent or for any reason by a court of competent jurisdiction, I agree that the remaining provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement will still be valid and the provision declared to be invalid or illegal or unenforceable will be considered to be severed and deleted from the rest of this Non-Compete Agreement. I further agree that if any court of competent jurisdiction finds any of the restrictions set forth in this Non-Compete Agreement to be overly broad and unenforceable, the restriction shall be interpreted to extend only over the maximum time period, geographic area, or range of activities or clients that such court deems enforceable

(f) Notwithstanding the foregoing provisions of this Non-Compete Agreement, the non-competition provisions of paragraph 2 above shall not restrict Employee from performing legal services as a licensed attorney for a Competing Business to the extent that the attorney licensure requirements in the applicable jurisdiction do not permit Employee to agree to the otherwise applicable restrictions of paragraph 2.

(g) Waiver of any of the provisions of this Non-Compete Agreement by the Company in any particular instance shall not be deemed to be a waiver of any provision in any other instance and/or of the Company's other rights at law or under this Non-Compete Agreement.

(h) I agree that the Company may assign this Non-Compete Agreement to its successors and assigns and that any such successor or assign may stand in the Company's stead for purposes of enforcing this Non-Compete Agreement.

(i) I agree to reimburse the Company for all attorneys' fees, costs, and expenses that it reasonably incurs in connection with enforcing its rights and remedies under this Non-Compete Agreement, but only to the extent the Company is ultimately the prevailing party in the applicable legal proceedings.

(j) I understand and agree that, where allowed by applicable law, the time for my obligations set out in paragraphs 2 - 6 shall be extended for period of non-compliance up to an additional two (2) years following my last day of employment with the Company (for any reason).

(k) I fully understand my obligations in this Non-Compete Agreement, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these

covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms.

(I) I agree that all non-competition, non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Non-Compete Agreement shall survive any termination of this Non-Compete Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Non-Compete Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations.

(m) I understand that nothing in this Non-Compete Agreement, including the non-disclosure and nondisparagement provisions, limit my ability to file a charge or complaint with the Equal Employment Opportunity Commission, Department of Labor, National Labor Relations Board, Occupational Safety and Health Administration, Securities and Exchange Commission or any other federal, state or local governmental agency or commission. I also understand that this Non-Compete Agreement does not limit my ability to communicate with any government agencies or otherwise participate in any investigation or proceeding that may be conducted by any government agency, including providing documents or other information, without notice to the Company. Finally, I understand that nothing in this Non-Compete Agreement is intended to restrict my legally-protected right to discuss wages, hours or other working condition with co-workers, or in any way limit my rights under the National Labor Relations Act or any whistleblower act.

10. <u>Relationship of Parties</u>. I acknowledge that my relationship with the Company is "terminable at will" by either party and that the Company or I can terminate the relationship with or without cause and without following any specific procedures. Nothing contained in this Non-Compete Agreement is intended to or shall be relied upon to alter the "terminable at will" relationship between the parties. I agree that my obligations in this Non-Compete Agreement shall survive the termination of my employment from the Company for any reason and shall be binding upon my successors, heirs, executors and representatives.

11. <u>Modifications and Other Agreements</u>. I agree that the terms of this Non-Compete Agreement may not be modified except by a written agreement signed by both me and the Company. This Non-Compete Agreement shall not supersede any other restrictive covenants to which I may be subject under an employment contract, benefit program or otherwise, such that the Company may enforce the terms of any and all restrictive covenants to which I am subject. The obligations herein are in addition to and do not limit any obligations arising under applicable statutes and common law.

12. <u>State and Commonwealth Law Modifications</u>. I agree that if I primarily reside and work in California, Massachusetts, Puerto Rico, South Carolina, Washington or Wisconsin, I am subject to the modifications to this Non-Compete Agreement set forth in Exhibit A-1 applying to such state and to the extent such state law applies.

13. <u>Notification</u>. I agree that in the event I am offered employment at any time in the future with any entity that may be considered a Competing Business Line, I shall immediately notify such Competing Business of the existence and terms of this Non-Compete Agreement. I also understand and agree that the Company may notify anyone attempting to or later employing me of the existence and provisions of this Non-Compete Agreement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Stock Option Award Agreement to which this Non-Compete Agreement is attached as Exhibit A, and I agree to the terms and conditions expressed in this Non-Compete Agreement, including the modifications set forth in Exhibit A-1, as applicable.

EXHIBIT A-1

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

State and Commonwealth Law Modifications

This Exhibit A-1 to the Walgreens Boots Alliance, Inc. Non-Competition, Non-Solicitation and Confidentiality Agreement (the "Non-Compete Agreement") includes jurisdiction-specific "Addenda," which modify the Non-Compete Agreement as applied to individuals who primarily reside and work in one of the applicable jurisdictions, but only to the extent the laws of such jurisdiction are applicable to the Non-Compete Agreement. The Addenda of this Exhibit A-1 should be read in conjunction with the rest of the Non-Compete Agreement and enforced to the fullest extent permissible to protect the Company's legitimate business interests.

<u>No. 1</u>:

CALIFORNIA ADDENDUM

The covenants in **Paragraph 2 "Non-Competition"** apply during my employment with the Company, but do not apply post-employment, during such time that my base location is in California.

<u>No. 2</u>:

The covenants in **Paragraph 3 "Non-Solicitation"** apply during my employment with the Company, but do not apply postemployment, during such time that my base location is in California.

<u>No. 3</u>:

Paragraph 5 "Non-Disparagement" is replaced with the following:

During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees, shareholders, members, agents or products, except as otherwise allowed by law, including California Government Code Section 12964.5.

<u>No. 4</u>:

Paragraph 6 "Intellectual Property" is supplemented with the following language:

The terms of this Agreement requiring disclosure and assignment of inventions to the Company do not apply to any invention that qualifies fully under California Labor Code Section 2870, which reads:

(a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer's

equipment, supplies, facilities, or trade secret information except for those inventions that either:

(1) Relate at the time of conception or reduction to practice of the invention to the employer's business, or actual or demonstrably anticipated research or development of the employer; or

(2) Result from any work performed by the employee for the employer.

(b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

While employed, I will advise the Company promptly in writing of any inventions that I believe meet the criteria in California Labor Code Section 2870 for a confidential ownership determination.

<u>No. 5</u>:

Paragraph 9 "Enforceability; General Provisions", Subparagraph (b) by this Addendum shall substitute "California" for "Illinois" with respect to the choice of law and forum, during such time that I primarily reside and work in California.

MASSACHUSETTS ADDENDUM

<u>No. 1</u>:

Paragraph 2 "Non-Competition" is stricken and replaced with the following:

2. <u>Non-Competition.</u> In exchange for Company providing me the consideration set forth in the Agreement, I agree that during my employment and for a period of one (1) year from the Termination Date - *i.e.*, the date of my voluntary termination of employment, or of the involuntary termination of my employment with Cause (as defined below) - I will not, directly or indirectly, engage in "Competition" (as defined below) within the "Geographic Region" (as defined below).

(a) **"Cause"** means misconduct, violation of any policy of the Company, including any rule of conduct or standard of ethics of the Company, breach of the Agreement (including this Addendum) or the breach of any confidentiality, nondisclosure, non-solicitation or assignment of inventions obligations to the Company, failure to meet the Company's reasonable performance expectations, or other grounds directly and reasonably related to the legitimate business needs of the Company.

(b) **"Competing Business"** means a business that is in competition with any business engaged in by the Company.

(c) **"Competition**" means to provide the same or substantially similar services to a Competing Business as those that I provided to the Company during the last two (2) years of my employment with the Company. "Competition" does

not include passive investments of less than five percent (5%) ownership interest in any entity.

(d) "Geographic Region" means the geographic area in which you, during any time within the last two years of your employment with the Company, provided services or had a material presence or influence.

(e) If the Company enforces the restrictions in this Paragraph 2 for a period of time after the Termination Date (the "Restraint Period"), it will pay me, during the Restraint Period, an amount equal to fifty percent (50%) of my annual base salary. My annual base salary, for the purposes of this Paragraph 2(e), will be calculated based on my average annual salary for my last two (2) years of employment, less any applicable deductions, and excluding any incentive compensation, bonuses, benefits, or other compensation, less any applicable deductions (the "Restraint Payment"). The Restraint Payment will be paid on a pro-rata basis during the Restraint Period in the same manner that I would have received wages from the Company had I been employed during the Restraint Period.

(f) The Restraint Period shall be extended from one (1) year following the Termination Date to two (2) years following the Termination Date if I (i) breached Employee's fiduciary duty(ies) to the Company, or (ii) unlawfully took, physically or electronically, property belonging to the Company. In the event that the Restraint Period is extended due to my breach of my fiduciary duty(ies) to the Company, or due to my having unlawfully taken, physically or electronically, property belonging to the Company, the Company shall not be required to provide payments to me during the extension of the Restraint Period.

(g) I understand that if the Company elects to waive the non-competition provisions set forth herein, I will not receive any compensation or consideration described in Paragraph 2(e). I further understands that at the time of my separation from employment, the Company shall elect whether to waive its enforcement of the non-competition provisions in the Agreement (including this Massachusetts Addendum). You will be notified by the Company of its election or waiver by letter, in a form of the following notice:

Walgreens Boots Alliance, Inc. (the "Company"), pursuant to Paragraph 2 of the Massachusetts Addendum (dated ________, 20___) to the Walgreens Boots Alliance, Inc. Non-Competition, Non-Solicitation, and Confidentiality Agreement (the "Agreement"), in its sole discretion, elects to:

Enforce the one year Restraint Period according to Paragraph 2 of the Addendum. As agreed to by the parties, the Company agrees to pay the employee the amounts described in subparagraph 2(e) of the Addendum.

Waive enforcement of the Restraint Period. The undersigned employee shall not receive any compensation or consideration pursuant to subparagraph 2(e) of the Addendum.

Regardless of the election or waiver, the undersigned employee remains bound by all other terms of the Agreement, and also remains bound by the terms of any and all other agreements between the undersigned employee and the Company.

(h) If I was already employed by the Company on the date of my signature on the Agreement, I acknowledge that the Agreement, including this Massachusetts Addendum, was delivered to me at least ten (10) business days before the date that this Addendum was executed by both of the parties (the "Effective Date"). If I was not already employed by the Company on the date of the my signature on the Agreement, I acknowledge that the Agreement, including this Massachusetts Addendum, was delivered to me (i) before a formal offer of employment was made by the Company, or (ii) ten (10) business days before the commencement of my employment with the Company, whichever was earlier.

(i) I acknowledge that I have been advised of my right to consult with counsel of my own choosing prior to signing the Agreement and this Massachusetts Addendum. By signing the Agreement and this Addendum, I acknowledge that I had time to read and understand the terms of the Agreement and this Addendum, and to consult with my own legal counsel, not including counsel for the Company, regarding the Agreement and the Addendum prior to their execution. I agree that I have actually read and understand the Agreement and this Addendum and all of their terms, and that I am entering into and signing the Agreement and this Addendum knowingly and voluntarily, and that in doing so I am not relying upon any statements or representations by the Company or its agents.

(j) I acknowledge that (i) the Non-Competition covenant contained in this Paragraph 2 is no broader than necessary to protect the Company's trade secrets, Confidential Information, and goodwill, and (ii) the business interests identified in the Agreement cannot be adequately protected through restrictive covenants other than the Non-Competition covenant contained in this Paragraph 2, including without limitation the non-solicitation and non-disclosure restrictions set forth in the Agreement.

<u>No. 2</u>:

Paragraph 9(a) "Enforceability; General Restrictions" is stricken and replaced with the following:

(a) I agree that the restrictions contained in this Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place an undue restraint on me. I further understand that the restrictions in this Agreement - other than the non-competition restrictions set forth in Paragraph 2 - apply no matter whether my employment is terminated by the Company or me and no matter whether that termination is voluntary or involuntary. I understand that the non-competition provisions in Paragraph 2 apply following the voluntary termination of my employment or the involuntary termination of my employment for Cause, as defined in Paragraph 2, unless the Employer elects to waive the non-competition provisions of Paragraph 2 as set forth in subparagraph 2(g).

<u>No. 3</u>:

Paragraph 9(b) "Enforceability; General Restrictions" is stricken and replaced with the following:

(b)(i) Because the Company's current base of operations is in Illinois and my connections thereto, (1) except with respect to the non-competition provisions of Paragraph 2, this Agreement shall be governed by and construed in accordance

with the laws of the State of Illinois, where this Agreement is entered into, without giving effect to any conflict of law provisions, and (2) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute, or declaration - other than a claim, dispute, or declaration arising out of Paragraph 2 - that arises out of this Agreement.

(b)(ii) The interpretation, validity, and enforcement of the non-competition provisions set forth in Paragraph 2 of this Agreement and Massachusetts Addendum will be governed by the laws of the Commonwealth of Massachusetts, without regard to any conflicts of laws principles that would require the application of the law of another jurisdiction. The parties agree that any action relating to or arising out of the non-competition provisions shall be brought in (1) the United States District Court for the District of Massachusetts, Eastern Division, if that Court has subject matter jurisdiction over the dispute; or, if it does not, in (2) the Business Litigation Session of the Suffolk County Superior Court, or, if the Business Litigation Session does not accept the case for whatever reason whatsoever, the Suffolk County Superior Courts of Massachusetts for resolution of any disputes or litigation arising under or in connection with the Non-Competition provisions set forth in Paragraph 2 of this Agreement and Massachusetts Addendum, and waive any objections or defenses to personal jurisdiction or venue in any such proceeding before any such court.

<u>No. 4</u>:

Paragraph 9(I) "Enforceability; General Restrictions" is stricken and replaced with the following:

(a) I agree that all non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Agreement shall survive any termination of this Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations. I further agree that all non-competition obligations in this Agreement shall survive the voluntary termination of my employment or the involuntary termination of my employment for Cause, as defined in Paragraph 2, unless the Employer elects to waive the non-competition provisions of Paragraph 2 as set forth in subparagraph 2(g), and no dispute regarding any other provisions of this Agreement or regarding my employment shall prevent the operation and enforcement of my employment shall prevent the operation and enforcement or regarding my employment shall prevent the operation provisions of this Agreement or regarding my employment or the termination of these obligations.

<u>No. 1</u>:

PUERTO RICO ADDENDUM

Paragraphs 2 and 3 are replaced by the following covenants and definitions:

"Similar Business" means the same or substantially the same business activity or activities performed or engaged by me for, or on behalf, of the Business of the Company or one of its subsidiaries or affiliated companies.

"Engage" means participate in, consult with, be employed by, or assist with the organization, policy making, ownership, financing, management, operation or control of any Similar Business in any capacity (*i.e.*, as an

independent contractor, consultant, employee, shareholder, member-owner, or business partner).

"Goodwill" means any tendency of customers, distributors, representatives, employees, vendors, suppliers, or federal, state, local or foreign governmental entities to continue or renew any valuable business relationship with the Company or any Similar Business with which I may be associated, based in whole or in part on past successful relationships with the Company or the lawful efforts of the Company to foster such relationships, and in which I actively participated at any time during the most recent twelve (12) months of my employment.

"Competing Business" means any individual (including me), corporation, limited liability company, partnership, joint venture, association, or other entity, regardless of form, that is directly engaged in whole or in relevant part in any business or enterprise that is the same as, or substantially the same as, that part of the Company for which I provided services during the last two (2) years of my employment, or that is taking material steps to engage in such business.

"**Customers**" means those individuals, companies, or other entities for which the Company has provided or does provide products or services in connection with the business of the Company, or those individuals, companies, or other entities to which the Company has provided written proposals concerning the business of the Company in the two (2) year period preceding the termination of my employment.

"Restricted Territory" means those municipalities within the Commonwealth of Puerto Rico in which I performed the Competing Business.

Non-Competition. I acknowledge and agree that the Company would be irreparably damaged if I - in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member, owner or business partner) - were to provide services to any person directly or indirectly competing with the Company or any of its affiliates or Engaged in a Competing Business and that such competition by me would result in a significant loss of Goodwill by the Company. Therefore, I agree that the following are reasonable restrictions and agree to be bound by such restrictions:

(a) During my employment, and for a period of twelve (12) months immediately following the termination of such employment for any reason, I shall not, directly or indirectly - in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member, owner or business partner) - Engage in Competing Business services or activities within the Restricted Territory; provided, that nothing herein shall prohibit me from being a passive owner of not more than five percent (5%) of the outstanding stock of any class of a corporation which is publicly traded so long as I do not have any active participation in the business of such corporation.

(b) I warrant and represent that the nature and extent of this non-competition clause has been fully explained to me by the Company and that my decision to accept the same is made voluntarily, knowingly, intelligently and free from any undue pressure or coercion. I further warrant and represent that I have agreed to this non-competition clause in consideration of the Stock Option I will be receiving under this Agreement.

Non-Solicitation of Customers. I agree that for a period of twelve (12) months following the voluntary or involuntary termination of my employment for any reason, I will not, either on my own behalf or for any Competing Business, directly or indirectly solicit, divert, or appropriate (or attempt to solicit, divert, or appropriate) any Customer with which I had material business contact in the six (6) month period preceding the termination of my employment, for providing products or services that are the same as or substantially similar to those provided by the Company.

Non-Solicitation of Employees. I recognize and admit that the Company has a legitimate business interest in retaining its employees, representatives, agents and/or consultants and of protecting its business from previous employees, representatives, agents and/or consultants, which makes necessary the establishment of a non-solicitation clause in the Agreement. I agree that for a period of twelve (12) months following the voluntary or involuntary termination of my employment for any reason, I shall not, directly or indirectly, (a) induce or attempt to induce any employee, representative, agent or consultant of the Company or any of its affiliates or subsidiaries to leave the employ or services of the Company or any of its affiliates or subsidiaries and any employee, representative, agent or consultant thereof or (b) hire any person who was an employee, representative, agent or consultant of the Company or any of its affiliates or subsidiaries at any time during the twelve (12) month period immediately prior to the date on which such hiring would take place. No action by another person or entity shall be deemed to be a breach of this provision unless I directly or indirectly assisted, encouraged or otherwise counseled such person or entity to engage in such activity.

<u>No. 2</u>:

Subparagraph 9(b) "Enforceability; General Restrictions" is stricken and replaced with the following:

(b)(i) Because the Company's current base of operations is in Illinois, as are my connections thereto (except with respect to the non-competition provisions of Paragraph 2), (1) I agree this Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Agreement is entered into, without giving effect to any conflict of law provisions, and (2) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute, or declaration - other than a claim, dispute, or declaration arising out of Paragraph 2 - that arises out of this Agreement.

(b)(ii) The laws of Puerto Rico will govern the interpretation, validity, and enforcement of the non-competition provisions set forth in Paragraph 2 of this Agreement and Puerto Rico Addendum.

<u>No. 3</u>:

Subparagraph 9(d), subparagraph 9(e), and subparagraph 9(j) "Enforceability; General Restrictions" are stricken.

SOUTH CAROLINA ADDENDUM

<u>No. 1</u>:

The definition of **"Confidential Information"** in **Paragraph 1** is further limited to that Confidential Information I learn about or am exposed to through my employment with the Company.

<u>No. 2</u>:

Paragraphs 2 and 3 of the Agreement are replaced by the following covenants and definitions:

"Competing Business" means any individual (including me), corporation, limited liability company, partnership, joint venture, association, or other entity, regardless of form, that is directly engaged in whole or in relevant part in any business or enterprise that is the same as, or substantially the same as, that part of the Company for which I provided services during the last two (2) years of my employment, or that is taking material steps to engage in such business.

"Customers" means those individuals, companies, or other entities for which the Company has provided or does provide products or services in connection with the business of the Company, or those individuals, companies, or other entities to which the Company has provided written proposals concerning the business of the Company in the two (2) year period preceding the termination of my employment.

"Restricted Territory" means:

1) the counties or areas where I worked for the Company or had material business contact with the Customers in the two (2) year period preceding the termination of my employment with the Company: and/or

2) the geographic territory in which I worked for the Company, represented the Company, or had material business contact with the Customers in the two (2) year period preceding the termination of my employment with the Company.

I agree that subsections 1) and 2) above are separate and severable covenants.

Non-Competition. I agree that for a period of one (1) year following the voluntary or involuntary termination of my employment for any reason, I will not, directly or indirectly, own, manage, operate, join, control, be employed by or with, or participate in any manner with a Competing Business anywhere in the Restricted Territory where doing so will require me to provide the same or substantially similar services to any such Competing Business as those that I provided to the Company during the last two (2) years of my employment.

Non-Solicitation of Customers. I agree that for a period of two (2) years following the voluntary or involuntary termination of my employment for any reason, I will not, either on my own behalf or for any Competing Business, directly or indirectly solicit, divert, or appropriate, or attempt to solicit, divert, or appropriate any Customer with which I had material business contact in the two (2) year period preceding the termination of my employment, for the purposes of providing products or services that are the same as or substantially similar to those provided by the Company.

Paragraph 2 is replaced in its entirety as follows:

Non-Competition.

(a) The non-competition provisions of this Paragraph 2 shall <u>only</u> apply when my annualized salaried exceeds the compensation requirements of the Restrictions on Noncompetition Covenants Bill 5478 as codified in RCW 49.

(b) I agree that during my employment with the Company and for one (1) year after the termination of my employment for any reason, I will not, directly or indirectly, engage in Competing Services with respect to any Competing Business Line. As set forth in Paragraph 10(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. The above restrictions shall not apply to passive investments of less than five percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "Competing Business Line" means any business that is in competition with any business engaged in by the Company and for which I performed Competing services during the two (2) years prior to my last day of employment with the Company. For purposes of this Non-Compete Agreement, "Competing Services" means the same or similar responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company and within the same geographic scope, or portion thereof, with respect to which I performed those responsibilities for the Company.

(C) I agree that, if and after my employment with the Company ends because of or in connection with a layoff or reduction-in-force, the non-competition provisions of Paragraph 2(a) above will not be enforced by the Company unless and to the extent that it pays me an amount that is equal to or greater than my base salary rate that is in effect on the last day of my employment with the Company. Such payments will be made to me at regular payroll intervals for the duration of the one (1) year post-employment non-competition period or such shorter period during which the Company enforces these non-competition provisions. I agree that I must promptly inform the Company of the date on which I begin any other employment or engagement by, with or for the benefit of any other individual or entity, at which time I agree the Company may and will terminate all such payments to me. Although such payments by the Company will terminate when I commence employment or any other engagement by, with or for the benefit of another individual, entity or employer, I agree that the Paragraph 2(a) non-competition restrictions will remain in effect until one (1) year after my Company employment ends. I also agree that if I fail to timely notify the Company of any other employment or engagement, and if the Company's payments to me therefore continue after I have commenced any such employment or engagement, then any such payments to me will be deemed to be placed by me in constructive trust for the benefit of the Company, and I agree that I must and will promptly return all such payments to the Company.

<u>No. 2</u>:

Subparagraph 9(b) of the Agreement is replaced in its entirety as follows:

(a) This Agreement shall be governed by and construed in accordance with the laws of the State of Washington without giving effect to any conflict of law provisions. Any claim, dispute or declaration arising out of or in connection with this Agreement will be resolved exclusively in the state or federal courts in the State of Washington.

<u>No. 1</u>:

WISCONSIN ADDENDUM

Paragraph 1 "Confidentiality" is amended by adding the following at the end of Paragraph 1(b):

To the extent the above obligation of non-use and non-disclosure of Confidential Information applies after the termination of my employment and to Confidential Information that does not meet the definition of a trade secret under applicable law, it shall apply only for two years after the termination of my employment and only in geographic areas in which the unauthorized use or disclosure of such Confidential Information would be competitively damaging to the Company.

<u>No. 2</u>:

Paragraph 2 "Non-Competition" is amended by replacing the definition of "Responsibilities" with the following:

"**Responsibilities**" means the same or similar responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company in which the Confidential Information I have would be competitively valuable and within the same geographic scope, or portion thereof, with respect to which I performed those responsibilities for the Company.

<u>No. 3</u>:

Paragraph 3 "Non-Solicitation" is amended by replacing the definition of "Restricted Customer" in paragraph 3(a) with the following:

"Restricted Customer" means any person, company or entity that was a customer of the Company and with which I had direct contact for purposes of performing responsibilities for the Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company.

Paragraph 3(a) is further amended by striking the following sentence:

To the extent permitted by applicable law, Restricted Customer also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company;

Paragraph 3(b) is amended by replacing it with the following:

I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I worked at any point during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company and join a competitor; (ii) interfere with the performance by any such employee of his/her duties for the Company; or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b). This restriction shall apply in all geographic areas in which the Company does business.

<u>No. 4</u>:

Paragraph 9 "Enforceability, General Provisions" is amended as follows: **Paragraph 9(e)** is amended by adding the following text to the end of the paragraph:

"The restrictive covenants in this agreement are intended to be divisible and interpreted and applied independent of each other."

Paragraph 9(j) is stricken and shall not be applied or referred to.

EXHIBIT B

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Option is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 23 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

EUROPEAN UNION ("EU") / EUROPEAN ECONOMIC AREA ("EEA")

Personal Data. The following provision replaces Section 21 of the Agreement in its entirety:

The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all stock options, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Option under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual

obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. As such, by accepting the Option, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the Option. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) International Data Transfers. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) <u>Data Subject Rights</u>. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources Department in writing. Your provision of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Option, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources Department in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

CHILE

Private Placement. The following provision shall replace Section 14 of the Agreement:

The grant of the Option hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;
- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and
- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;
- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y
- d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente.

FRANCE

1. <u>Nature of Grant</u>. The Option is not granted under the French specific regime provided by Articles L.225-177 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Sale of Shares of Stock</u>. Shares of Stock purchased upon exercise of the Option are accepted as a personal investment. In the event that shares of Stock are issued in respect of the Option within six (6) months after the Grant Date, you agree that the shares of Stock may not be offered to the public or otherwise disposed of prior to the sixmonth anniversary of the Grant Date.

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement the Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Option and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Option and shares of Stock subject to the Option do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

1. <u>Plan Document Acknowledgment</u>. In accepting the Option, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 2: Vesting/Exercise/Expiration (expiration of the right to exercise the Option after the Expiration Date); Section 3: Disability (term for exercising the Option prior to the Vesting Dates in the case of a Termination of Service due to Disability); Section 4: Death (term for exercising the option prior to the Vesting Dates in the case of a Termination of Service due to death); Section 5: Retirement (term for exercising the Option prior to the Vesting Dates in the case of a Termination of Service by reason of Retirement); Section 6: Termination of Service Following a Change in Control (term for exercising the Option in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (term to exercise the vested Option and forfeiture of the unvested Option in other cases of Termination of Service); Section 8: Forfeiture of Outstanding Options Upon Termination for Cause or Following Termination of Service; Section 10(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Option and legally applicable to the participant); Section 11: Limited Transferability (Option shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 18: Change in Stock (right of the Company to equitably adjust the Option and the Exercise Price in the event of any change in the Stock); Section 19(j): Nature of the Option (waive any claim or entitlement to compensation or damages arising from forfeiture of the Option resulting from a Termination of Service); Section 19(I): Nature of the Option (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Option); Section 22: Non-Competition, Non-Solicitation and Confidentiality (the receipt of the Option is conditioned upon agreement of the Non-Competition, Non-Solicitation and Confidentiality Agreement attached hereto as Exhibit A); Section 23: Addendum to Agreement (the Option is subject to the terms of the Addendum); Section 24: Additional Requirements (Company right to impose additional requirements on the Option in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Option and the Plan); Section 26: Electronic Delivery (Company may deliver documents related to the Option or Plan electronically); Section 27: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 28: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Option does not constitute an employment relationship between you and the Company. You have been granted the Option as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you ("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan are not part of the employment conditions and/or benefits provided by WBA

Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. Extraordinary Item of Compensation. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Option, the shares of Stock subject to the Option and the value of the same are an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Option, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Option. Upon the grant of the Option, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

SPAIN

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Option, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted an Option under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Option is granted on the assumption and condition that the Option and the shares of Stock acquired upon exercise of the Option shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Option shall be null and void.

Further, you understand and agree that the vesting of the Option is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Option may cease vesting immediately, in whole or in part, effective on the date of your Termination of Service (unless otherwise specifically provided in Section 3, 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to the Option that were not vested on the date of your Termination of Service, as described in the Plan and Agreement. In addition, you understand and agree that the post-Termination of Service exercise period specified in the Agreement shall run from the date of your Termination of Service, as determined by the Committee, in its sole discretion.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on the Option.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 8 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Option. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Option have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

<u>Securities Law Notification</u>. The Option is not considered a public offering in Switzerland; therefore, it is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Option constitute a prospectus as such term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Option may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Option have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. Indemnification for Tax-Related Items. Without limitation to Section 10 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 10 of the Agreement.

2. <u>Exclusion of Claim</u>. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Option, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Option. Upon the grant of the Option, you shall be deemed irrevocably to have waived any such entitlement.

3. <u>Post-Termination Restrictions</u>. To the extent that you are employed by your Employer pursuant to an employment agreement governed by the laws of England, Wales, Scotland and/or Northern Ireland, Paragraphs 2 and 3 of the NNCA Agreement attached to the Agreement as Exhibit A shall not apply to you.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Stock Option Award Agreement to which this Addendum is attached as Exhibit B, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of securities or financial instruments of Walgreens Boots Alliance, Inc. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN RESTRICTED STOCK UNIT AWARD AGREEMENT

Participant Name: Stefano Pessina

Participant ID:

Grant Date: Units Granted:

Vesting: One third of the Shares Granted vest on each of the first, second and third anniversaries of the Grant Date (the "Vesting Dates")

Dates")

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the Restricted Stock Unit Award (the "Award") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Restricted Stock Units</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you the number of Restricted Stock Units specified above (the "Restricted Stock Units"), subject to the terms and conditions of the Plan and this Agreement.

2. <u>Restricted Stock Unit Account and Dividend Equivalents</u>. The Company will maintain an account (the "Account") on its books in your name to reflect the number of Restricted Stock Units awarded to you as well as any additional Restricted Stock Units credited as a result of Dividend Equivalents. The Account will be administered as follows:

(a) The Account is for recordkeeping purposes only, and no assets or other amounts shall be set aside from the Company's general assets with respect to such Account.

(b) As of each record date with respect to which a cash dividend is to be paid with respect to shares of Company common stock par value US\$.01 per share ("Stock"), the Company will credit your Account with an equivalent amount of Restricted Stock Units determined by dividing the value of the cash dividend that would have been paid on your Restricted Stock Units if they had been shares of Stock, divided by the value of Stock on such date.

(c) If dividends are paid in the form of shares of Stock rather than cash, then your Account will be credited with one additional Restricted Stock Unit for each share of Stock that would have been received as a dividend had your outstanding Restricted Stock Units been shares of Stock.

(d) Additional Restricted Stock Units credited via Dividend Equivalents shall vest or be forfeited at the same time as the Restricted Stock Units to which they relate.

3. <u>Restricted Period</u>. The period prior to the vesting date with respect each Restricted Stock Unit is referred to as the "Restricted Period." Subject to the provisions of the Plan and this Agreement, unless vested or forfeited earlier as described in Section 4, 5, 6 or 7 of this Agreement, as applicable, your Restricted Stock Units will become vested and be settled as described in Section 8 below, as of the vesting date or dates indicated in the introduction to this Agreement.

4. <u>Disability or Death</u>. If during the Restricted Period you have a Termination of Service by reason of Disability or death, then the Restricted Stock Units will become fully vested as of the date of your Termination of Service and the Vesting Date shall become the date of your Termination of Service. Any Restricted Stock Units becoming vested by reason of your Termination of Service by reason of Disability or death shall be settled as provided in Section 8.

5. <u>Retirement</u>. If prior to the end of the first 12 months of the Restricted Period you have a Termination of Service by reason of retirement from the Company's Board of Directors, as reasonably determined and approved by the Committee, then, subject to such approval, the Restricted Stock Units will become vested on a prorated basis as of the date of your Termination of Service, with such pro-ration based on the number of full months of service completed during the Restricted Period, divided by 36 months. If on or after the end of the first 12 months of the Restricted Period you have a Termination of Service by reason of retirement from the Company's Board of Directors, as reasonably determined and approved by the Committee, then, subject to such approval, the Restricted Stock Units will become fully vested as of the date of your Termination of Service. Any Restricted Stock Units becoming vested by reason of your retirement shall be settled as provided in Section 8.

6. <u>Termination of Service Following a Change in Control</u>. If during the Restricted Period there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 7), then your Restricted Stock Units shall become fully vested, and they shall be settled in accordance with Section 9. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such Plan.

7. <u>Other Termination of Service</u>. If during the Restricted Period you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 4, 5 or 6 above or Section 9 below, as determined by the Committee, then you shall thereupon forfeit any Restricted Stock Units that are still in a Restricted Period on your termination date. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

8. <u>Settlement of Vested Restricted Stock Units</u>. Subject to the requirements of Section 13 below, as promptly as practicable after the applicable Vesting Date, whether occurring upon your Separation from Service or otherwise, but in no event later than 75 days after the Vesting Date, the Company shall transfer to you one share of Stock for each Restricted Stock Unit becoming vested at such time, net of any applicable tax withholding requirements in accordance with Section 10 below; provided, however, that, if you are a Specified Employee at the time of Separation from Service, then to the extent your Restricted Stock Units are deferred compensation subject to Section 409A of the Code, settlement of which is triggered by your Separation from Service (other than for death), payment shall not be made until the date which is six months after your Separation from Service.

Notwithstanding the foregoing, if you are resident or employed outside of the U.S., the Company, in its sole discretion, may provide for the settlement of the Restricted Stock Units in the form of:

(a) a cash payment (in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units) to the extent that settlement in shares of Stock (i) is prohibited under local law, (ii) would require you, the Company or an Affiliate to obtain the approval of any governmental or regulatory body in your country of residence (or country of employment, if different), (iii) would result in adverse tax consequences for you, the Company or an Affiliate or (iv) is administratively burdensome; or

(b) shares of Stock, but require you to sell such shares of Stock immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

9. <u>Settlement Following Change in Control</u>. Notwithstanding any provision of this Agreement to the contrary, the Company may, in its sole discretion, fulfill its obligation with respect to all or any portion of the Restricted Stock Units that become vested in accordance with Section 6 above, by:

(a) delivery of (i) the number of shares of Stock that corresponds with the number of Restricted Stock Units that have become vested or (ii) such other ownership interest as such shares of Stock that correspond with the vested Restricted Stock Units may be converted into by virtue of the Change in Control transaction;

(b) payment of cash in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units at that time; or

(c) delivery of any combination of shares of Stock (or other converted ownership interest) and cash having an aggregate Fair Market Value equal to the Fair Market Value of the Stock that corresponds with the number of Restricted Stock Units that have become vested at that time.

Settlement shall be made as soon as practical after the Restricted Stock Units become fully vested under Section 6, but in no event later than 30 days after such date.

10. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or any Affiliate, the ultimate liability for all income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("Tax-Related Items"), is and remains your responsibility and may exceed the amount actually withheld by the Company or any Affiliate, if any. You further acknowledge that the Company and/or

any Affiliate (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Award, including, but not limited to, the grant, vesting or settlement of the Award, the subsequent sale of shares of Stock acquired pursuant to such settlement and the receipt of any Dividend Equivalents and/or dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Award to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or any Affiliate may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or any Affiliates to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon settlement of the Award that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Restricted Stock Units and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, any Affiliate or its agent to cover the Tax-Related Items that the Company or any Affiliate may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or any Affiliate may (in its sole discretion) allow you to make a cash payment to the Company, any Affiliate or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon settlement of the Award, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the earned Award, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock (or cash payment) or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

11. <u>Nontransferability</u>. During the Restricted Period and thereafter until Stock is transferred to you in settlement thereof, you may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Restricted Stock Units whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, or by will or by the laws of intestacy.

12. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the Restricted Stock Units until such time as a certificate of stock for the Stock issued in settlement of such Restricted Stock Units has been issued to you or such shares of Stock have been recorded in your name in book entry form. Until that time, you shall not have any voting rights with respect to the Restricted Stock Units. Except as provided in Section 9 above, no adjustment shall be made for dividends or distributions or other rights with respect to such shares for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the

Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

13. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

14. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Restricted Stock Units is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Restricted Stock Units is not subject to the supervision of the local securities authorities.

15. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock, or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

16. <u>Repatriation; Compliance with Law</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of residence to comply with your personal obligations under local laws, rules and/or regulations in your country of residence (and country of employment, if different).

17. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Restricted Stock Units. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

18. <u>Change in Stock</u>. In the event of any change in Stock, by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of Restricted Stock Units subject to this Agreement shall be equitably adjusted by the Committee.

19. <u>Nature of the Award</u>. In accepting the Award, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Award is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of Restricted Stock Units, or benefits in lieu of Restricted Stock Units, even if Restricted Stock Units have been granted in the past;

(c) all decisions with respect to future Awards or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of the Award, the number of shares subject to the Award, and the vesting provisions applicable to the Award;

(d) the Award and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Award and the shares of Stock subject to the Award are not intended to replace any pension rights or compensation;

(g) the Award, the shares of Stock subject to the Award and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or any Affiliate;

(h) the future value of the shares of Stock underlying the Award is unknown, indeterminable and cannot be predicted with certainty;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Award resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Award to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Award and the benefits evidenced by this Agreement do not create any entitlement to have the Award or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Award or of any amounts due to you pursuant to the settlement of the Award or the subsequent sale of any shares of Stock acquired upon settlement of the Award.

20. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

21. Personal Data. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Restricted Stock Units and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and its Affiliates hold certain personal information about you, specifically: your name, home address and telephone number, date of birth, social security number or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). The Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and any of its Affiliates will process the Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. The Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and any of its Affiliates will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or any Affiliate may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer the Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of the Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of the Data, (iv) oppose, for legal reasons, the collection, processing or transfer of the Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder

(in which case, your Restricted Stock Units will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

22. <u>Addendum to Agreement</u>. Notwithstanding any provision of this Agreement to the contrary, the Restricted Stock Units shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to the Agreement, attached hereto as Exhibit A (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

23. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Restricted Stock Units, any shares of Stock acquired pursuant to the Restricted Stock Units and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

24. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

25. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Award or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

26. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdictions.

27. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Award, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English, so as to allow you to understand the terms and conditions of this Agreement, the Plan and any other documents related to the Award. If you have received this Agreement, the Plan or any other documents related to the Award translated into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

28. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this

Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

29. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibit A. Once you have read and understood this Agreement and Exhibit A, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibit A, as applicable, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Award granted hereunder.

EXHIBIT A

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Award is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 23 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

EUROPEAN UNION ("EU") / EUROPEAN ECONOMIC AREA ("EEA")

Personal Data. The following provision replaces Section 19 of the Agreement in its entirety:

The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all Restricted Stock Units, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Restricted Stock Units under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business

interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. As such, by accepting the Restricted Stock Units, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the Restricted Stock Units. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) International Data Transfers. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) <u>Data Subject Rights</u>. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources Department in writing. Your provision of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Restricted Stock Units, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources Department in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

CHILE

<u>Private Placement</u>. The following provision shall replace Section 14 of the Agreement:

The grant of the Restricted Stock Units hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;
- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and
- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;
- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y
- d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente

FRANCE

1. <u>Nature of Grant</u>. The Restricted Stock Units are not granted under the French specific regime provided by Articles L. 225-197-1 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Form of Payment</u>. Notwithstanding any provision in the Agreement or Plan to the contrary, the Restricted Stock Units shall be settled only in Shares (and not in cash).

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement, this Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Restricted Stock Units and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have

any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Restricted Stock Units and shares of Stock subject to the Restricted Stock Units do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

<u>Plan Acknowledgment</u>. In accepting the Restricted Stock Units, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 3: Restricted Period (terms of lapse of restrictions on Restricted Stock Units); Section 4: Disability or Death (terms of payment of Restricted Stock Units upon a Termination of Service by reason of Disability or death); Section 5: Retirement (terms of payment of Restricted Stock Units upon a Termination of Service by reason of retirement); Section 6: Termination of Service Following a Change in Control (terms of payment of Restricted Stock Units in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (forfeiture of Restricted Stock Units in other cases of Termination of Service); Section 10(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Restricted Stock Units and legally applicable to the participant); Section 11: Nontransferability (Restricted Stock Units shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 18: Change in Stock (right of the Company to equitably adjust the number of Restricted Stock Units subject to this Agreement in the event of any change in Stock); Section 19(j): Nature of the Award (waive any claim or entitlement to compensation or damages arising from forfeiture of the Restricted Stock Units resulting from a Termination of Service); Section 19(I): Nature of the Award (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Restricted Stock Units); Section 20: Committee Authority; Recoupment (right of the Committee to administer, construe, and make all determinations necessary or appropriate for the administration of the Restricted Stock Units and this Agreement, including the enforcement of any recoupment policy); Section 21: Non-Competition, Non-Solicitation and Confidentiality (the receipt of the Award is conditioned upon agreement to the Non-Competition, Non-Solicitation and Confidentiality Agreement attached hereto as Exhibit A); Section 23: Addendum to Agreement (the Restricted Stock Units are subject to the terms of the Addendum); Section 24: Additional Requirements (Company right to impose additional requirements on the Restricted Stock Units in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Restricted Stock Units and the Plan); Section 26: Electronic Delivery (Company may deliver documents related to the Award or Plan electronically); Section 27: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 28: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Restricted Stock Units does not constitute an employment relationship between you and the Company. You have been granted the Restricted Stock Units as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you

("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan are not part of the employment conditions and/or benefits provided by WBA Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. Extraordinary Item of Compensation. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Award, the shares of Stock subject to the Award and the income and value of the same is an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Restricted Stock Units, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Restricted Stock Units. Upon the grant of Restricted Stock Units, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Award, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted Restricted Stock Units under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Restricted Stock Units are granted on the assumption and condition that the Restricted Stock Units and the shares of Stock acquired upon settlement of the Restricted Stock Units shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Award shall be null and void.

Further, you understand and agree that the vesting of the Restricted Stock Units is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Restricted Stock Units may cease vesting immediately, in whole or in part, effective on the date of your Termination of Service (unless otherwise specifically provided in Section 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to Restricted Stock Units that were not vested as of the date of your Termination of Service, as described in the Plan and Agreement.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on your Award.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 7 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Restricted Stock Units. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Restricted Stock Units have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

<u>Securities Law Notification</u>. The Restricted Stock Units are not considered a public offering in Switzerland; therefore, the offer of Restricted Stock Units is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Restricted Stock Units constitute a prospectus as such

term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Restricted Stock Units may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Restricted Stock Units have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. Indemnification for Tax-Related Items. Without limitation to Section 10 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 10 of the Agreement.

2. <u>Exclusion of Claim</u>. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Restricted Stock Units, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Restricted Stock Units. Upon the grant of the Restricted Stock Units, you shall be deemed irrevocably to have waived any such entitlement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Restricted Stock Unit Award Agreement to which this Addendum is attached as Exhibit A, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of securities or financial instruments of Walgreens Boots Alliance, Inc. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN RESTRICTED STOCK UNIT AWARD AGREEMENT

Participant Name: James Skinner

Participant ID:

Grant Date:

Units Granted:

Vesting: Three years from Grant Date (the "Vesting Date")

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the Restricted Stock Unit Award (the "Award") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Restricted Stock Units</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you the number of Restricted Stock Units specified above (the "Restricted Stock Units"), subject to the terms and conditions of the Plan and this Agreement.

2. <u>Restricted Stock Unit Account and Dividend Equivalents</u>. The Company will maintain an account (the "Account") on its books in your name to reflect the number of Restricted Stock Units awarded to you as well as any additional Restricted Stock Units credited as a result of Dividend Equivalents. The Account will be administered as follows:

(a) The Account is for recordkeeping purposes only, and no assets or other amounts shall be set aside from the Company's general assets with respect to such Account.

(b) As of each record date with respect to which a cash dividend is to be paid with respect to shares of Company common stock par value US\$.01 per share ("Stock"), the Company will credit your Account with an equivalent amount of Restricted Stock Units determined by dividing the value of the cash dividend that would have been paid on your Restricted Stock Units if they had been shares of Stock, divided by the value of Stock on such date.

(c) If dividends are paid in the form of shares of Stock rather than cash, then your Account will be credited with one additional Restricted Stock Unit for each share of Stock that would have been received as a dividend had your outstanding Restricted Stock Units been shares of Stock.

(d) Additional Restricted Stock Units credited via Dividend Equivalents shall vest or be forfeited at the same time as the Restricted Stock Units to which they relate.

3. <u>Restricted Period</u>. The period prior to the vesting date with respect each Restricted Stock Unit is referred to as the "Restricted Period." Subject to the provisions of the Plan and this Agreement,

unless vested or forfeited earlier as described in Section 4, 5, 6 or 7 of this Agreement, as applicable, your Restricted Stock Units will become vested and be settled as described in Section 8 below, as of the vesting date or dates indicated in the introduction to this Agreement, provided the performance goal in this Section 3 ("Performance Goal") is satisfied as of the end of the applicable performance period. The Performance Goal will be established and certified by the Committee and cover one or more Company performance goals over the course of the Company's 2020 fiscal year. If the Performance Goal is not attained as of the end of this performance period, the Restricted Stock Units awarded hereunder shall be thereupon forfeited.

4. <u>Disability or Death</u>. If during the Restricted Period you have a Termination of Service by reason of Disability or death, then the Restricted Stock Units will become fully vested as of the date of your Termination of Service and the Vesting Date shall become the date of your Termination of Service. Any Restricted Stock Units becoming vested by reason of your Termination of Service by reason of Disability or death shall be settled as provided in Section 8.

5. <u>Retirement</u>. If prior to the end of the first 12 months of the Restricted Period you have a Termination of Service by reason of retirement from the Company's Board of Directors, as reasonably determined and approved by the Committee, then, subject to such approval and subject to satisfaction of the Performance Goal, the Restricted Stock Units will become vested on a prorated basis as of the later of the end of the performance period for the Performance Goal and the date of your Termination of Service, with such pro-ration based on the number of full months of service completed during the Restricted Period, divided by 36 months. If on or after the end of the first 12 months of the Restricted Period you have a Termination of Service by reason of retirement from the Company's Board of Directors, as reasonably determined and approved by the Committee, then, subject to such approval and subject to satisfaction of the Performance Goal, the Restricted Stock Units will become fully vested as of the date of your Termination of Service. Any Restricted Stock Units becoming vested by reason of your retirement shall be settled as provided in Section 8.

6. <u>Termination of Service Following a Change in Control</u>. If during the Restricted Period there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 7), then your Restricted Stock Units shall become fully vested, and they shall be settled in accordance with Section 9. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such Plan.

7. <u>Other Termination of Service</u>. If during the Restricted Period you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 4, 5 or 6 above or Section 9 below, as determined by the Committee, then you shall thereupon forfeit any Restricted Stock Units that are still in a Restricted Period on your termination date. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

8. <u>Settlement of Vested Restricted Stock Units</u>. Subject to the requirements of Section 13 below, as promptly as practicable after the applicable Vesting Date, whether occurring upon your Separation from Service or otherwise, but in no event later than 75 days after the Vesting Date, the Company shall transfer to you one share of Stock for each Restricted Stock Unit becoming vested at such time, net of any applicable tax withholding requirements in accordance with Section 10 below; provided, however, that, if you are a Specified Employee at the time of Separation from Service, then to the extent your Restricted Stock Units are deferred compensation subject to Section 409A of the Code, settlement of which is triggered by your Separation from Service (other than for death), payment shall not be made until the date which is six months after your Separation from Service.

Notwithstanding the foregoing, if you are resident or employed outside of the U.S., the Company, in its sole discretion, may provide for the settlement of the Restricted Stock Units in the form of:

(a) a cash payment (in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units) to the extent that settlement in shares of Stock (i) is prohibited under local law, (ii) would require you, the Company or an Affiliate to obtain the approval of any governmental or regulatory body in your country of residence (or country of employment, if different), (iii) would result in adverse tax consequences for you, the Company or an Affiliate or (iv) is administratively burdensome; or

(b) shares of Stock, but require you to sell such shares of Stock immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

9. <u>Settlement Following Change in Control</u>. Notwithstanding any provision of this Agreement to the contrary, the Company may, in its sole discretion, fulfill its obligation with respect to all or any portion of the Restricted Stock Units that become vested in accordance with Section 6 above, by:

(a) delivery of (i) the number of shares of Stock that corresponds with the number of Restricted Stock Units that have become vested or (ii) such other ownership interest as such shares of Stock that correspond with the vested Restricted Stock Units may be converted into by virtue of the Change in Control transaction;

(b) payment of cash in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units at that time; or

(c) delivery of any combination of shares of Stock (or other converted ownership interest) and cash having an aggregate Fair Market Value equal to the Fair Market Value of the Stock that corresponds with the number of Restricted Stock Units that have become vested at that time.

Settlement shall be made as soon as practical after the Restricted Stock Units become fully vested under Section 6, but in no event later than 30 days after such date.

10. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or any Affiliate, the ultimate liability for all income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to

you ("Tax-Related Items"), is and remains your responsibility and may exceed the amount actually withheld by the Company or any Affiliate, if any. You further acknowledge that the Company and/or any Affiliate (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Award, including, but not limited to, the grant, vesting or settlement of the Award, the subsequent sale of shares of Stock acquired pursuant to such settlement and the receipt of any Dividend Equivalents and/or dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Award to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or any Affiliate may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or any Affiliates to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon settlement of the Award that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Restricted Stock Units and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, any Affiliate or its agent to cover the Tax-Related Items that the Company or any Affiliate may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or any Affiliate may (in its sole discretion) allow you to make a cash payment to the Company, any Affiliate or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon settlement of the Award, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the earned Award, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock (or cash payment) or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

11. <u>Nontransferability</u>. During the Restricted Period and thereafter until Stock is transferred to you in settlement thereof, you may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Restricted Stock Units whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, or by will or by the laws of intestacy.

12. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the Restricted Stock Units until such time as a certificate of stock for the Stock issued in settlement of such Restricted Stock Units has been issued to you or such shares of Stock have been recorded in your name in book entry form. Until that time, you shall not have any voting rights with respect to the Restricted Stock Units. Except as provided in Section 9 above, no adjustment shall be made for dividends or distributions or other rights with respect to such shares for which the record date is prior to the date on which you become

the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

13. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

14. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Restricted Stock Units is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Restricted Stock Units is not subject to the supervision of the local securities authorities.

15. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock, or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

16. <u>Repatriation; Compliance with Law</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of residence to comply with your personal obligations under local laws, rules and/or regulations in your country of employment, if different).

17. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Restricted Stock Units. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

18. <u>Change in Stock</u>. In the event of any change in Stock, by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of Restricted Stock Units subject to this Agreement shall be equitably adjusted by the Committee.

19. <u>Nature of the Award</u>. In accepting the Award, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Award is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of Restricted Stock Units, or benefits in lieu of Restricted Stock Units, even if Restricted Stock Units have been granted in the past;

(c) all decisions with respect to future Awards or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of the Award, the number of shares subject to the Award, and the vesting provisions applicable to the Award;

(d) the Award and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Award and the shares of Stock subject to the Award are not intended to replace any pension rights or compensation;

(g) the Award, the shares of Stock subject to the Award and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company or any Affiliate;

(h) the future value of the shares of Stock underlying the Award is unknown, indeterminable and cannot be predicted with certainty;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Award resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Award to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Award and the benefits evidenced by this Agreement do not create any entitlement to have the Award or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed

out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Award or of any amounts due to you pursuant to the settlement of the Award or the subsequent sale of any shares of Stock acquired upon settlement of the Award.

20. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

21. <u>Non-Competition, Non-Solicitation and Confidentiality</u>. As a condition to the receipt of this Award, you must agree to the Non-Competition, Non-Solicitation and Confidentiality Agreement attached hereto as Exhibit A by executing that Agreement. Failure to execute and return the Non-Competition, Non-Solicitation and Confidentiality Agreement within 120 days of the Grant Date shall constitute your decision to decline to accept this Award.

22. <u>Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Restricted Stock Units and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and its Affiliates hold certain personal information about you, specifically: your name, home address and telephone number, date of birth, social security number or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). The Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and any of its Affiliates will process the Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. The Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and any of its Affiliates will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or any Affiliate may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer the Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of the Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of the Data, (iv) oppose, for legal reasons, the collection, processing or transfer of the Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, your Restricted Stock Units will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

23. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Restricted Stock Units, any shares of Stock acquired pursuant to the Restricted Stock Units and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

24. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

25. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Award or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

26. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdictions.

27. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Award, be drawn up in English. You further acknowledge that you are sufficiently proficient in English or have consulted with an advisor who is sufficiently proficient in English so as to allow you to understand the terms and conditions of this Agreement, the Plan and any other documents related to the Award. If you have received this Agreement, the Plan or any other documents related to the Award translated into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

28. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

29. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibit A. Once you have read and understood this Agreement and Exhibit A, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibit A, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Award granted hereunder.

EXHIBIT A

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

This Exhibit (the "Non-Compete Agreement") forms a part of the Restricted Stock Unit Award Agreement covering Restricted Stock Units awarded to an employee ("Employee" or "I") of Walgreens Boots Alliance, Inc. or an affiliate thereof, on behalf of itself, its affiliates, subsidiaries, and successors (collectively referred to as the "Company").

WHEREAS, the Company develops and/or uses valuable business, technical, proprietary, customer and patient information it protects by limiting its disclosure and by keeping it secret or confidential;

WHEREAS, Employee acknowledges that during the course of employment, he or she has or will receive, contribute, or develop such Confidential Information and Trade Secrets (as defined below); and

WHEREAS, the Company desires to protect from its competitors such Confidential Information and Trade Secrets and also desires to protect its legitimate business interests and goodwill in maintaining its employee and customer relationships.

NOW THEREFORE, in consideration of the Restricted Stock Units issued to Employee pursuant to the Agreement to which this is attached as Exhibit A and for other good and valuable consideration, including but not limited to employment or continued employment, the specialized knowledge, skill and training that the Company provides Employee, the goodwill that Employee develops with customers on behalf of the Company, Employee agrees to be bound by the terms of this Non-Compete Agreement as follows:

1. <u>Confidentiality</u>.

(a) At all times during and after the termination of my employment with the Company, I will not, without the Company's prior written permission, directly or indirectly for any purpose other than performance of my duties for the Company, utilize or disclose to anyone outside of the Company any Trade Secrets (defined in subparagraph 1(a)(i)) or other Confidential Information (defined in subparagraph 1(a)(ii)) or any information received by the Company in confidence from or about third parties, as long as such matters remain Trade Secrets or otherwise confidential.

(i) For purposes of this Non-Compete Agreement, "**Trade Secrets**" means a form of intellectual property that are protectable under applicable state and/or Federal law, including the Uniform Trade Secrets Act (as amended and adapted by the states) and the Federal Defend Trade Secrets Act of 2016 (the "DTSA"). They include all tangible and intangible (*e.g.*, electronic) forms and types of information that is held and kept confidential by the Company and is not generally known outside of the Company, including but not limited to information about: the Company's financial, business, scientific, technical, economic, or engineering information, including patterns, plans, compilations, program devices, formulas, designs, prototypes, methods, techniques, processes, procedures, programs or codes, and may in particular include such things as pricing information, business records, software programs, algorithms, inventions, patent applications, and designs and processes not known outside the Company.

(ii) For purposes of this Non-Compete Agreement, "**Confidential Information**" means Trade Secrets and, more broadly, any other tangible and intangible (*e.g.*, electronic) forms and types of information that are held and kept confidential by the Company and are not generally known outside the Company, and which relates to the actual or anticipated business of the Company or the Company's actual or prospective vendors or clients. Confidential Information shall not be considered generally known to the public if is revealed improperly to the public by me or others without the Company's express written consent and/or in violation of an obligation of confidentiality to the Company. Examples of Confidential Information include, but are not limited to: customer, referral source, supplier and contractor identification and contacts; special contract terms; pricing and margins; business, marketing and customer plans and strategies; financial data; company created (or licensed) techniques; technical know-how; research, development and production information; processes, prototypes, software, patent applications and plans, projections, proposals, discussion guides, and/or personal or performance information about employees.

(b) I understand that this obligation of non-disclosure shall last so long as the information remains confidential. I, however, understand that, if I live and work primarily in Wisconsin, Virginia, or any other state requiring a temporal limit on non-disclosure clauses, Confidential Information shall be protected for two (2) years following termination of my employment (for any reason). I also understand that Trade Secrets are protected by statute and are not subject to any time limits. I also agree to contact the Company before using, disclosing, or distributing any Confidential Information or Trade Secrets if I have any questions about whether such information is protected information.

The restrictions set forth in this paragraph are in addition to and not in lieu of any obligations I have by (c) law with respect to the Company's Confidential Information, including any obligations I may owe under the DTSA and any applicable state statutes. Nothing herein shall prohibit me from divulging evidence of criminal wrongdoing to law enforcement or prohibit me from disclosing Confidential Information or Trade Secrets if compelled by order of court or an agency of competent jurisdiction or as required by law; however, I shall promptly inform the Company of any such situations and shall take reasonable steps to prevent disclosure of Confidential Information or Trade Secrets until the Company has been informed of such required disclosure and has had a reasonable opportunity to seek a protective order. Pursuant to the DTSA, I understand that an individual may not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a Trade Secret that: (A) is made (i) in confidence to a federal, state or local government official, either directly or indirectly, or to an attorney, and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. Additionally, I understand that an individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the Trade Secret to his or her attorney and use the Trade Secret information in the court proceeding, so long as any document containing the Trade Secret is filed under seal and the individual does not disclose the Trade Secret, except pursuant to court order. Nothing in this Non-Compete Agreement

is intended to conflict with the DTSA or create liability for disclosures of Trade Secrets that are expressly allowed by DTSA.

Non-Competition. I agree that during my employment with the Company and for twelve (12) months after 2. the termination of my employment (for any reason), I will not, directly or indirectly have Responsibilities with respect to any Competing Business Line. As set forth in paragraph 9(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. These restrictions shall not apply to passive investments of less than five percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "Responsibilities" means the same or similar material responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company and within the same geographic scope, or portion thereof, where I performed those responsibilities for the Company. For purposes of this Non-Compete Agreement, "Competing Business Line" means any business that is in competition with any business engaged in by the Company and for which I had Responsibilities during the two (2) years prior to my last day of employment with the Company. Competing Business Line shall also include businesses or business lines that may not be directly competitive with the Company in most respects (such as pharmacy benefit managers), but only to the extent I am engaged by any such business in a role: (a) that involves my performing Responsibilities for Competing Products or Services; or (b) where I would be called upon to inevitably rely upon or disclose Confidential Information and such reliance or disclosure would competitively harm the Company. For purposes of this Non-Compete Agreement, "Competing Products or Services" means products or services that are competitive with products or services offered by, developed by, designed by or distributed by the Company during the two (2) years prior to my last day of employment with the Company.

3. <u>Non-Solicitation</u>. I agree that during my employment with the Company and for two (2) years after the termination of my employment from the Company (for any reason):

(a) I will not directly or indirectly, solicit any Restricted Customer for purposes of providing Competing Products or Services, or offer, provide or sell Competing Products or Services to any Restricted Customer. For purposes of this Non-Compete Agreement, "Restricted Customer" means any person, company or entity that was a customer, vendor, supplier or referral source of the Company and with which I had direct contact for purposes of performing responsibilities for the Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company. To the extent permitted by applicable law, "Restricted Customer" also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company; and

(b) I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I had direct contact work during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company; (ii) interfere with the performance by any such employee of his/her duties for the Company; and/or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b).

4. <u>Non-Inducement</u>. I will not directly or indirectly assist or encourage any person or entity in carrying out or conducting any activity that would be prohibited by this Non-Compete Agreement if such activity were carried out or conducted by me.

5. <u>Non-Disparagement</u>. During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees,

shareholders, members, agents or products. The foregoing shall not be violated by truthful statements in response to legal process, required governmental testimony or filings, or administrative or arbitral proceedings (including, without limitation, depositions in connection with such proceedings); and the foregoing shall not apply to any claims for harassment or discrimination to the extent so restricted by applicable state law.

6. <u>Intellectual Property.</u> The term **"Intellectual Property"** shall mean all trade secrets, ideas, inventions, designs, developments, devices, software, computer programs, methods and processes (whether or not patented or patentable, reduced to practice or included in the Confidential Information) and all patents and patent applications related thereto, all copyrights, copyrightable works and mask works (whether or not included in the Confidential Information) and all patents and all other proprietary rights contributed to, or conceived or created by, or reduced to practice by Employee or anyone acting on his/her behalf (whether alone or jointly with others) at any time from the beginning of Employee's employment with the Company to the termination of that employment plus ninety (90) days, that (i) relate to the business or to the actual or anticipated research or development of the Company; (ii) result from any services that Employee or anyone acting on its behalf perform for the Company; or (iii) are created using the equipment, supplies or facilities of the Company or any Confidential Information.

a. <u>Ownership</u>. All Intellectual Property is, shall be and shall remain the exclusive property of the Company. Employee hereby assigns to the Company all right, title and interest, if any, in and to the Intellectual Property; provided, however, that, when applicable, the Company shall own the copyrights in all copyrightable works included in the Intellectual Property pursuant to the "work-made-for-hire" doctrine (rather than by assignment), as such term is defined in the 1976 Copyright Act. All Intellectual Property shall be owned by the Company irrespective of any copyright notices or confidentiality legends to the contrary that may be placed on such works by Employee or by others. Employee shall ensure that all copyright notices and confidentiality legends on all work product authored by Employee or anyone acting on his/her behalf shall conform to the Company's practices and shall specify the Company as the owner of the work. The Company hereby provides notice to Employee that the obligation to assign does not apply to an invention for which no equipment, supplies, facility, or Trade Secrets of the Company was used and which was developed entirely on the Employee's own time, unless (i) the invention relates (1) to the business of the Company, or (2) to the Company's actual or demonstrably anticipated research or development, or (ii) the invention results from any work performed by Employee for the Company.

b. <u>Keep Records</u>. Employee shall keep and maintain, or cause to be kept and maintained by anyone acting on his/her behalf, adequate and current written records of all Intellectual Property in the form of notes, sketches, drawings, computer files, reports or other documents relating thereto. Such records shall be and shall remain the exclusive property of the Company and shall be available to the Company at all times during my employment with the Company.

c. <u>Assistance</u>. Employee shall supply all assistance requested in securing for the Company's benefit any patent, copyright, trademark, service mark, license, right or other evidence of ownership of any such Intellectual Property, and will provide full information regarding any such item and execute all appropriate documentation prepared by Company in applying or otherwise registering, in the Company's name, all rights to any such item or the defense and protection of such Intellectual Property.

d. <u>Prior Inventions</u>. Employee has disclosed to the Company any continuing obligations to any third party with respect to Intellectual Property. Employee claims no rights to any inventions created prior to his/her employment for which a patent application has not previously been filed, unless he/she has described them in detail on a schedule attached to this Non-Compete Agreement.

e. <u>Trade Secret Provisions</u>. The provisions in paragraph 1 of this Non-Compete Agreement with regard to Trade Secrets and the DTSA shall apply as well in the context of the parties' Intellectual Property rights and obligations.

7. <u>Return of Company Property</u>. I agree that all documents and data accessible to me during my employment with the Company, including Confidential Information and Trade Secrets, regardless of format (electronic or hard copy), including but not limited to any Company computer, monitor, printer equipment, external drives, wireless access equipment, telecom equipment and systems ("Company Equipment"), are and remain the sole and exclusive property of the Company and/or its clients, and must be returned to the Company upon separation or upon demand by the Company. I further agree that I will provide passwords to access such Company Equipment and I will not print, retain, copy, destroy, modify or erase Company U.S. data on Company Equipment or otherwise wipe Company Equipment prior to returning the Company Equipment.

8. <u>Consideration and Acknowledgments</u>. I acknowledge and agree that the covenants described in this Non-Compete Agreement are essential terms, and the underlying Restricted Stock Units would not be provided by the Company in the absence of these covenants. I further acknowledge that these covenants are supported by adequate consideration as set forth in this Non-Compete Agreement and are not in conflict with any public interest. I further acknowledge and agree that I fully understand these covenants, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms. I further acknowledge and agree that these covenants are reasonable and enforceable in all respects.

9. Enforceability; General Provisions.

(a) I agree that the restrictions contained in this Non-Compete Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Non-Compete Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place undue restraint on me. I further understand that the restrictions in this Non-Compete Agreement apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary.

(b) Because the Company's current base of operations is in Illinois and my connections thereto, (i) this Non-Compete Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Non-Compete Agreement is entered into, without giving effect to any conflict of law provisions, and (ii) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute or declaration arising out of this Non-Compete Agreement.

(c) In the event of a breach or a threatened breach of this Non-Compete Agreement, I acknowledge that the Company will face irreparable injury which may be difficult to calculate in dollar terms and that the Company shall be entitled, in addition to all remedies otherwise available in law or in equity, to temporary restraining orders and preliminary and final injunctions enjoining such breach or threatened breach in any court of competent jurisdiction without the necessity of posting a surety bond, as well as to obtain an equitable accounting of all profits or benefits arising out of any violation of this Non-Compete Agreement.

(d) I agree that if a court determines that any of the provisions in this Non-Compete Agreement is unenforceable or unreasonable in duration, territory, or scope, then that court shall modify those provisions so they are reasonable and enforceable, and enforce those provisions as modified.

(e) If any one or more provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement or its application is determined to be invalid, illegal, or unenforceable to any extent or for any reason by a court of competent jurisdiction, I agree that the remaining provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement will still be valid and the provision declared to be invalid or illegal or unenforceable will be considered to be severed and deleted from the rest of this Non-Compete Agreement. I further agree that if any court of competent jurisdiction finds any of the restrictions set forth in this Non-Compete Agreement to be overly broad and unenforceable, the restriction shall be interpreted to extend only over the maximum time period, geographic area, or range of activities or clients that such court deems enforceable

(f) Notwithstanding the foregoing provisions of this Non-Compete Agreement, the non-competition provisions of paragraph 2 above shall not restrict Employee from performing legal services as a licensed attorney for a Competing Business to the extent that the attorney licensure requirements in the applicable jurisdiction do not permit Employee to agree to the otherwise applicable restrictions of paragraph 2.

(g) Waiver of any of the provisions of this Non-Compete Agreement by the Company in any particular instance shall not be deemed to be a waiver of any provision in any other instance and/or of the Company's other rights at law or under this Non-Compete Agreement.

(h) I agree that the Company may assign this Non-Compete Agreement to its successors and assigns and that any such successor or assign may stand in the Company's stead for purposes of enforcing this Non-Compete Agreement.

(i) I agree to reimburse the Company for all attorneys' fees, costs, and expenses that it reasonably incurs in connection with enforcing its rights and remedies under this Non-Compete Agreement, but only to the extent the Company is ultimately the prevailing party in the applicable legal proceedings.

(j) I understand and agree that, where allowed by applicable law, the time for my obligations set out in paragraphs 2 - 6 shall be extended for period of non-compliance up to an additional two (2) years following my last day of employment with the Company (for any reason).

(k) I fully understand my obligations in this Non-Compete Agreement, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms.

(I) I agree that all non-competition, non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Non-Compete Agreement shall survive any termination of this Non-Compete Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Non-Compete Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations.

(m) I understand that nothing in this Non-Compete Agreement, including the non-disclosure and nondisparagement provisions, limit my ability to file a charge or complaint with the Equal Employment Opportunity Commission, Department of Labor, National Labor Relations Board, Occupational Safety and Health Administration, Securities and Exchange Commission or any other federal, state or local governmental agency or commission. I also understand that this Non-Compete Agreement does not limit my ability to communicate with any government agencies or otherwise participate in any investigation or proceeding that may be conducted by any government agency, including providing documents or other information, without notice to the Company. Finally, I understand that nothing in this Non-Compete Agreement is intended to restrict my legally-protected right to discuss wages, hours or other working condition with co-workers, or in any way limit my rights under the National Labor Relations Act or any whistleblower act.

10. <u>Relationship of Parties</u>. I acknowledge that my relationship with the Company is "terminable at will" by either party and that the Company or I can terminate the relationship with or without cause and without following any specific procedures. Nothing contained in this Non-Compete Agreement is intended to or shall be relied upon to alter the "terminable at will" relationship between the parties. I agree that my obligations in this Non-Compete Agreement shall survive the termination of my employment from the Company for any reason and shall be binding upon my successors, heirs, executors and representatives.

11. <u>Modifications and Other Agreements</u>. I agree that the terms of this Non-Compete Agreement may not be modified except by a written agreement signed by both me and the Company. This Non-Compete Agreement shall not supersede any other restrictive covenants to which I may be subject under an employment contract, benefit program or otherwise, such that the Company may enforce the terms of any and all restrictive covenants to which I am subject. The obligations herein are in addition to and do not limit any obligations arising under applicable statutes and common law.

12. <u>State and Commonwealth Law Modifications</u>. I agree that if I primarily reside and work in California, Massachusetts, Puerto Rico, South Carolina, Washington or Wisconsin, I am subject to the modifications to this Non-Compete Agreement set forth in Exhibit A-1 applying to such state and to the extent such state law applies.

13. <u>Notification</u>. I agree that in the event I am offered employment at any time in the future with any entity that may be considered a Competing Business Line, I shall immediately notify such Competing Business of the existence and terms of this Non-Compete Agreement. I also understand and agree that the Company may notify anyone attempting to or later employing me of the existence and provisions of this Non-Compete Agreement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Restricted Stock Unit Award Agreement to which this Non-Compete Agreement is attached as Exhibit A, and I agree to the terms and conditions expressed in this Non-Compete Agreement, including the modifications set forth in Exhibit A-1, as applicable.

EXHIBIT A-1

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

State and Commonwealth Law Modifications

This Exhibit A-1 to the Walgreens Boots Alliance, Inc. Non-Competition, Non-Solicitation and Confidentiality Agreement (the "Non-Compete Agreement") includes jurisdiction-specific "Addenda," which modify the Non-Compete Agreement as applied to individuals who primarily reside and work in one of the applicable jurisdictions, but only to the extent the laws of such jurisdiction are applicable to the Non-Compete Agreement. The Addenda of this Exhibit A-1 should be read in conjunction with the rest of the Non-Compete Agreement and enforced to the fullest extent permissible to protect the Company's legitimate business interests.

CALIFORNIA ADDENDUM

<u>No. 1</u>:

The covenants in **Paragraph 2 "Non-Competition"** apply during my employment with the Company, but do not apply post-employment, during such time that my base location is in California.

<u>No. 2</u>:

The covenants in **Paragraph 3 "Non-Solicitation"** apply during my employment with the Company, but do not apply postemployment, during such time that my base location is in California.

<u>No. 3</u>:

Paragraph 5 "Non-Disparagement" is replaced with the following:

During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees, shareholders, members, agents or products, except as otherwise allowed by law, including California Government Code Section 12964.5.

<u>No. 4</u>:

Paragraph 6 "Intellectual Property" is supplemented with the following language:

The terms of this Agreement requiring disclosure and assignment of inventions to the Company do not apply to any invention that qualifies fully under California Labor Code Section 2870, which reads:

(a) Any provision in an employment agreement which provides that an employee shall assign, or offer to assign, any of his or her rights in an invention to his or her employer shall not apply to an invention that the employee developed entirely on his or her own time without using the employer's equipment, supplies, facilities, or trade secret information except for those inventions that either:

(1) Relate at the time of conception or reduction to practice of the invention to the employer's business, or actual or demonstrably anticipated research or development of the employer; or

(2) Result from any work performed by the employee for the employer.

(b) To the extent a provision in an employment agreement purports to require an employee to assign an invention otherwise excluded from being required to be assigned under subdivision (a), the provision is against the public policy of this state and is unenforceable.

While employed, I will advise the Company promptly in writing of any inventions that I believe meet the criteria in California Labor Code Section 2870 for a confidential ownership determination.

<u>No. 5</u>:

Paragraph 9 "Enforceability; General Provisions", Subparagraph (b) by this Addendum shall substitute "California" for "Illinois" with respect to the choice of law and forum, during such time that I primarily reside and work in California.

MASSACHUSETTS ADDENDUM

Paragraph 2 "Non-Competition" is stricken and replaced with the following:

2. <u>Non-Competition.</u> In exchange for Company providing me the consideration set forth in the Agreement, I agree that during my employment and for a period of one (1) year from the Termination Date - *i.e.*, the date of my voluntary termination of employment, or of the involuntary termination of my employment with Cause (as defined below) - I will not, directly or indirectly, engage in "Competition" (as defined below) within the "Geographic Region" (as defined below).

(a) **"Cause"** means misconduct, violation of any policy of the Company, including any rule of conduct or standard of ethics of the Company, breach of the Agreement (including this Addendum) or the breach of any confidentiality, non-disclosure, non-solicitation or assignment of inventions obligations to the Company, failure to meet the Company's reasonable performance expectations, or other grounds directly and reasonably related to the legitimate business needs of the Company.

(b) **"Competing Business**" means a business that is in competition with any business engaged in by the Company.

(c) **"Competition"** means to provide the same or substantially similar services to a Competing Business as those that I provided to the Company during the last two (2) years of my employment with the Company. "Competition" does not include passive investments of less than five percent (5%) ownership interest in any entity.

(d) "Geographic Region" means the geographic area in which you, during any time within the last two years of your employment with the Company, provided services or had a material presence or influence.

(e) If the Company enforces the restrictions in this Paragraph 2 for a period of time after the Termination Date (the "Restraint Period"), it will pay me, during the Restraint Period, an amount equal to fifty percent (50%) of my annual base salary. My annual base salary, for the purposes of this Paragraph 2(e), will be calculated based on my average annual salary for my last two (2) years of employment, less any applicable deductions, and excluding any incentive compensation, bonuses, benefits, or other compensation, less any applicable deductions (the "Restraint Payment"). The Restraint Payment will be paid on a pro-rata basis during the Restraint Period in the same manner that I would have received wages from the Company had I been employed during the Restraint Period.

(f) The Restraint Period shall be extended from one (1) year following the Termination Date to two (2) years following the Termination Date if I (i) breached Employee's fiduciary duty(ies) to the Company, or (ii) unlawfully took, physically or electronically, property belonging to the Company. In the event that the Restraint Period is extended due to my breach of my fiduciary duty(ies) to the Company, or due to my having unlawfully taken, physically or electronically, property

belonging to the Company, the Company shall not be required to provide payments to me during the extension of the Restraint Period.

(g) I understand that if the Company elects to waive the non-competition provisions set forth herein, I will not receive any compensation or consideration described in Paragraph 2(e). I further understands that at the time of my separation from employment, the Company shall elect whether to waive its enforcement of the non-competition provisions in the Agreement (including this Massachusetts Addendum). You will be notified by the Company of its election or waiver by letter, in a form of the following notice:

Walgreens Boots Alliance, Inc. (the "Company"), pursuant to Paragraph 2 of the Massachusetts Addendum (dated ________, 20___) to the Walgreens Boots Alliance, Inc. Non-Competition, Non-Solicitation, and Confidentiality Agreement (the "Agreement"), in its sole discretion, elects to:

Enforce the one year Restraint Period according to Paragraph 2 of the Addendum. As agreed to by the parties, the Company agrees to pay the employee the amounts described in subparagraph 2(e) of the Addendum.

Waive enforcement of the Restraint Period. The undersigned employee shall not receive any compensation or consideration pursuant to subparagraph 2(e) of the Addendum.

Regardless of the election or waiver, the undersigned employee remains bound by all other terms of the Agreement, and also remains bound by the terms of any and all other agreements between the undersigned employee and the Company.

(h) If I was already employed by the Company on the date of my signature on the Agreement, I acknowledge that the Agreement, including this Massachusetts Addendum, was delivered to me at least ten (10) business days before the date that this Addendum was executed by both of the parties (the "Effective Date"). If I was not already employed by the Company on the date of the my signature on the Agreement, I acknowledge that the Agreement, including this Massachusetts Addendum, was delivered to me (i) before a formal offer of employment was made by the Company, or (ii) ten (10) business days before the commencement of my employment with the Company, whichever was earlier.

(i) I acknowledge that I have been advised of my right to consult with counsel of my own choosing prior to signing the Agreement and this Massachusetts Addendum. By signing the Agreement and this Addendum, I acknowledge that I had time to read and understand the terms of the Agreement and this Addendum, and to consult with my own legal counsel, not including counsel for the Company, regarding the Agreement and the Addendum prior to their execution. I agree that I have actually read and understand the Agreement and this Addendum and all of their terms, and that I am entering into and signing the Agreement and this Addendum knowingly and voluntarily, and that in doing so I am not relying upon any statements or representations by the Company or its agents.

(j) I acknowledge that (i) the Non-Competition covenant contained in this Paragraph 2 is no broader than necessary to protect the Company's trade secrets, Confidential Information, and goodwill, and (ii) the business interests identified in

the Agreement cannot be adequately protected through restrictive covenants other than the Non-Competition covenant contained in this Paragraph 2, including without limitation the non-solicitation and non-disclosure restrictions set forth in the Agreement.

<u>No. 2</u>:

Paragraph 9(a) "Enforceability; General Restrictions" is stricken and replaced with the following:

(a) I agree that the restrictions contained in this Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place an undue restraint on me. I further understand that the restrictions in this Agreement - other than the non-competition restrictions set forth in Paragraph 2 - apply no matter whether my employment is terminated by the Company or me and no matter whether that termination is voluntary or involuntary. I understand that the non-competition provisions in Paragraph 2 apply following the voluntary termination of my employment or the involuntary termination of my employment for Cause, as defined in Paragraph 2, unless the Employer elects to waive the non-competition provisions of Paragraph 2 as set forth in subparagraph 2(g).

<u>No. 3</u>:

Paragraph 9(b) "Enforceability; General Restrictions" is stricken and replaced with the following:

(b)(i) Because the Company's current base of operations is in Illinois and my connections thereto, (1) except with respect to the non-competition provisions of Paragraph 2, this Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Agreement is entered into, without giving effect to any conflict of law provisions, and (2) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute, or declaration - other than a claim, dispute, or declaration arising out of Paragraph 2 - that arises out of this Agreement.

(b)(ii) The interpretation, validity, and enforcement of the non-competition provisions set forth in Paragraph 2 of this Agreement and Massachusetts Addendum will be governed by the laws of the Commonwealth of Massachusetts, without regard to any conflicts of laws principles that would require the application of the law of another jurisdiction. The parties agree that any action relating to or arising out of the non-competition provisions shall be brought in (1) the United States District Court for the District of Massachusetts, Eastern Division, if that Court has subject matter jurisdiction over the dispute; or, if it does not, in (2) the Business Litigation Session of the Suffolk County Superior Court, or, if the Business Litigation Session does not accept the case for whatever reason whatsoever, the Suffolk County Superior Courts of Massachusetts for resolution of any disputes or litigation arising under or in connection with the Non-Competition provisions set forth in Paragraph 2 of this Agreement and Massachusetts Addendum, and waive any objections or defenses to personal jurisdiction or venue in any such proceeding before any such court.

<u>No. 4</u>:

Paragraph 9(I) "Enforceability; General Restrictions" is stricken and replaced with the following:

(a) I agree that all non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Agreement shall survive any termination of this Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations. I further agree that all non-competition obligations in this Agreement shall survive the voluntary termination of my employment or the involuntary termination of my employment for Cause, as defined in Paragraph 2, unless the Employer elects to waive the non-competition provisions of Paragraph 2 as set forth in subparagraph 2(g), and no dispute regarding any other provisions of this Agreement or regarding my employment shall prevent the operation and enforcement of my employment shall prevent the operation and enforcement or regarding my employment shall prevent the operation provisions of this Agreement or regarding my employment or the termination of these obligations.

<u>No. 1</u>:

PUERTO RICO ADDENDUM

Paragraphs 2 and 3 are replaced by the following covenants and definitions:

"Similar Business" means the same or substantially the same business activity or activities performed or engaged by me for, or on behalf, of the Business of the Company or one of its subsidiaries or affiliated companies.

"Engage" means participate in, consult with, be employed by, or assist with the organization, policy making, ownership, financing, management, operation or control of any Similar Business in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member-owner, or business partner).

"Goodwill" means any tendency of customers, distributors, representatives, employees, vendors, suppliers, or federal, state, local or foreign governmental entities to continue or renew any valuable business relationship with the Company or any Similar Business with which I may be associated, based in whole or in part on past successful relationships with the Company or the lawful efforts of the Company to foster such relationships, and in which I actively participated at any time during the most recent twelve (12) months of my employment.

"Competing Business" means any individual (including me), corporation, limited liability company, partnership, joint venture, association, or other entity, regardless of form, that is directly engaged in whole or in relevant part in any business or enterprise that is the same as, or substantially the same as, that part of the Company for which I provided services during the last two (2) years of my employment, or that is taking material steps to engage in such business.

"Customers" means those individuals, companies, or other entities for which the Company has provided or does provide products or services in connection with the business of the Company, or those individuals, companies, or other entities to which the Company has provided written proposals concerning the

business of the Company in the two (2) year period preceding the termination of my employment.

"**Restricted Territory**" means those municipalities within the Commonwealth of Puerto Rico in which I performed the Competing Business.

Non-Competition. I acknowledge and agree that the Company would be irreparably damaged if I - in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member, owner or business partner) - were to provide services to any person directly or indirectly competing with the Company or any of its affiliates or Engaged in a Competing Business and that such competition by me would result in a significant loss of Goodwill by the Company. Therefore, I agree that the following are reasonable restrictions and agree to be bound by such restrictions:

(a) During my employment, and for a period of twelve (12) months immediately following the termination of such employment for any reason, I shall not, directly or indirectly - in any capacity (*i.e.*, as an independent contractor, consultant, employee, shareholder, member, owner or business partner) - Engage in Competing Business services or activities within the Restricted Territory; provided, that nothing herein shall prohibit me from being a passive owner of not more than five percent (5%) of the outstanding stock of any class of a corporation which is publicly traded so long as I do not have any active participation in the business of such corporation.

(b) I warrant and represent that the nature and extent of this non-competition clause has been fully explained to me by the Company and that my decision to accept the same is made voluntarily, knowingly, intelligently and free from any undue pressure or coercion. I further warrant and represent that I have agreed to this non-competition clause in consideration of the Stock Option I will be receiving under this Agreement.

Non-Solicitation of Customers. I agree that for a period of twelve (12) months following the voluntary or involuntary termination of my employment for any reason, I will not, either on my own behalf or for any Competing Business, directly or indirectly solicit, divert, or appropriate (or attempt to solicit, divert, or appropriate) any Customer with which I had material business contact in the six (6) month period preceding the termination of my employment, for providing products or services that are the same as or substantially similar to those provided by the Company.

Non-Solicitation of Employees. I recognize and admit that the Company has a legitimate business interest in retaining its employees, representatives, agents and/or consultants and of protecting its business from previous employees, representatives, agents and/or consultants, which makes necessary the establishment of a non-solicitation clause in the Agreement. I agree that for a period of twelve (12) months following the voluntary or involuntary termination of my employment for any reason, I shall not, directly or indirectly, (a) induce or attempt to induce any employee, representative, agent or consultant of the Company or any of its affiliates or subsidiaries to leave the employ or services of the Company or any of its affiliates or subsidiaries and any employee, representative, agent or consultant thereof or (b) hire any person who was an employee, representative, agent or consultant of the Company or any of its affiliates or subsidiaries at any time during the twelve (12) month period immediately prior to the date on which such hiring would take place. No action by another person or entity shall be deemed to be a breach of this provision unless I

directly or indirectly assisted, encouraged or otherwise counseled such person or entity to engage in such activity.

No. 2:

Subparagraph 9(b) "Enforceability; General Restrictions" is stricken and replaced with the following:

(b)(i) Because the Company's current base of operations is in Illinois, as are my connections thereto (except with respect to the non-competition provisions of Paragraph 2), (1) I agree this Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Agreement is entered into, without giving effect to any conflict of law provisions, and (2) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute, or declaration - other than a claim, dispute, or declaration arising out of Paragraph 2 - that arises out of this Agreement.

(b)(ii) The laws of Puerto Rico will govern the interpretation, validity, and enforcement of the non-competition provisions set forth in Paragraph 2 of this Agreement and Puerto Rico Addendum.

<u>No. 3</u>:

Subparagraph 9(d), subparagraph 9(e), and subparagraph 9(j) "Enforceability; General Restrictions" are stricken.

<u>No. 1</u>:

The definition of **"Confidential Information"** in **Paragraph 1** is further limited to that Confidential Information I learn about or am exposed to through my employment with the Company.

SOUTH CAROLINA ADDENDUM

<u>No. 2</u>:

Paragraphs 2 and 3 of the Agreement are replaced by the following covenants and definitions:

"Competing Business" means any individual (including me), corporation, limited liability company, partnership, joint venture, association, or other entity, regardless of form, that is directly engaged in whole or in relevant part in any business or enterprise that is the same as, or substantially the same as, that part of the Company for which I provided services during the last two (2) years of my employment, or that is taking material steps to engage in such business.

"Customers" means those individuals, companies, or other entities for which the Company has provided or does provide products or services in connection with the business of the Company, or those individuals, companies, or other entities to which the Company has provided written proposals concerning the business of the Company in the two (2) year period preceding the termination of my employment.

"Restricted Territory" means:

1) the counties or areas where I worked for the Company or had material business contact with the Customers in the two (2) year period preceding the termination of my employment with the Company: and/or

2) the geographic territory in which I worked for the Company, represented the Company, or had material business contact with the Customers in the two (2) year period preceding the termination of my employment with the Company.

I agree that subsections 1) and 2) above are separate and severable covenants.

Non-Competition. I agree that for a period of one (1) year following the voluntary or involuntary termination of my employment for any reason, I will not, directly or indirectly, own, manage, operate, join, control, be employed by or with, or participate in any manner with a Competing Business anywhere in the Restricted Territory where doing so will require me to provide the same or substantially similar services to any such Competing Business as those that I provided to the Company during the last two (2) years of my employment.

Non-Solicitation of Customers. I agree that for a period of two (2) years following the voluntary or involuntary termination of my employment for any reason, I will not, either on my own behalf or for any Competing Business, directly or indirectly solicit, divert, or appropriate, or attempt to solicit, divert, or appropriate any Customer with which I had material business contact in the two (2) year period preceding the termination of my employment, for the purposes of providing products or services that are the same as or substantially similar to those provided by the Company.

WASHINGTON ADDENDUM

<u>No. 1</u>:

Paragraph 2 is replaced in its entirety as follows:

Non-Competition.

(a) The non-competition provisions of this Paragraph 2 shall <u>only</u> apply when my annualized salaried exceeds the compensation requirements of the Restrictions on Noncompetition Covenants Bill 5478 as codified in RCW 49.

(b) I agree that during my employment with the Company and for one (1) year after the termination of my employment for any reason, I will not, directly or indirectly, engage in Competing Services with respect to any Competing Business Line. As set forth in Paragraph 10(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. The above restrictions shall not apply to passive investments of less than five percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "Competing Business Line" means any business that is in competition with any business engaged in by the Company and for which I performed Competing services during the two (2) years prior to my last day of employment with the Company. For purposes of this Non-Compete Agreement, "Competing Services" means the same or similar responsibilities I performed for the Company during the two (2) years prior to my last day of

employment with the Company and within the same geographic scope, or portion thereof, with respect to which I performed those responsibilities for the Company.

I agree that, if and after my employment with the Company ends because of or in connection with (C) a layoff or reduction-in-force, the non-competition provisions of Paragraph 2(a) above will not be enforced by the Company unless and to the extent that it pays me an amount that is equal to or greater than my base salary rate that is in effect on the last day of my employment with the Company. Such payments will be made to me at regular payroll intervals for the duration of the one (1) year post-employment non-competition period or such shorter period during which the Company enforces these non-competition provisions. I agree that I must promptly inform the Company of the date on which I begin any other employment or engagement by, with or for the benefit of any other individual or entity, at which time I agree the Company may and will terminate all such payments to me. Although such payments by the Company will terminate when I commence employment or any other engagement by, with or for the benefit of another individual, entity or employer, I agree that the Paragraph 2(a) non-competition restrictions will remain in effect until one (1) year after my Company employment ends. I also agree that if I fail to timely notify the Company of any other employment or engagement, and if the Company's payments to me therefore continue after I have commenced any such employment or engagement, then any such payments to me will be deemed to be placed by me in constructive trust for the benefit of the Company, and I agree that I must and will promptly return all such payments to the Company.

<u>No. 2</u>:

Subparagraph 9(b) of the Agreement is replaced in its entirety as follows:

(a) This Agreement shall be governed by and construed in accordance with the laws of the State of Washington without giving effect to any conflict of law provisions. Any claim, dispute or declaration arising out of or in connection with this Agreement will be resolved exclusively in the state or federal courts in the State of Washington.

WISCONSIN ADDENDUM

<u>No. 1</u>:

Paragraph 1 "Confidentiality" is amended by adding the following at the end of Paragraph 1(b):

To the extent the above obligation of non-use and non-disclosure of Confidential Information applies after the termination of my employment and to Confidential Information that does not meet the definition of a trade secret under applicable law, it shall apply only for two years after the termination of my employment and only in geographic areas in which the unauthorized use or disclosure of such Confidential Information would be competitively damaging to the Company.

<u>No. 2</u>:

Paragraph 2 "Non-Competition" is amended by replacing the definition of "Responsibilities" with the following:

"**Responsibilities**" means the same or similar responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company in which the Confidential Information I have would be competitively valuable and within the same geographic scope, or portion

thereof, with respect to which I performed those responsibilities for the Company.

<u>No. 3</u>:

Paragraph 3 "Non-Solicitation" is amended by replacing the definition of "Restricted Customer" in paragraph 3(a) with the following:

"Restricted Customer" means any person, company or entity that was a customer of the Company and with which I had direct contact for purposes of performing responsibilities for the Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company.

Paragraph 3(a) is further amended by striking the following sentence:

To the extent permitted by applicable law, Restricted Customer also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company;

Paragraph 3(b) is amended by replacing it with the following:

I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I worked at any point during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company and join a competitor; (ii) interfere with the performance by any such employee of his/her duties for the Company; or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b). This restriction shall apply in all geographic areas in which the Company does business.

<u>No. 4</u>:

Paragraph 9 "Enforceability, General Provisions" is amended as follows: **Paragraph 9(e)** is amended by adding the following text to the end of the paragraph:

"The restrictive covenants in this agreement are intended to be divisible and interpreted and applied independent of each other."

Paragraph 9(j) is stricken and shall not be applied or referred to.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of securities or financial instruments of Walgreens Boots Alliance, Inc. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN RESTRICTED STOCK UNIT AWARD AGREEMENT

Participant Name:

Participant ID:

Grant Date:

Vesting: One third of the Shares Granted vest on each of the first, second and third anniversaries of the Grant Date (the "Vesting Dates")

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the Restricted Stock Unit Award (the "Award") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Restricted Stock Units</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you the number of Restricted Stock Units specified above (the "Restricted Stock Units"), subject to the terms and conditions of the Plan and this Agreement.

2. <u>Restricted Stock Unit Account and Dividend Equivalents</u>. The Company will maintain an account (the "Account") on its books in your name to reflect the number of Restricted Stock Units awarded to you as well as any additional Restricted Stock Units credited as a result of Dividend Equivalents. The Account will be administered as follows:

(a) The Account is for recordkeeping purposes only, and no assets or other amounts shall be set aside from the Company's general assets with respect to such Account.

(b) As of each record date with respect to which a cash dividend is to be paid with respect to shares of Company common stock par value US\$.01 per share ("Stock"), the Company will credit your Account with an equivalent amount of Restricted Stock Units determined by dividing the value of the cash dividend that would have been paid on your Restricted Stock Units if they had been shares of Stock, divided by the value of Stock on such date.

(c) If dividends are paid in the form of shares of Stock rather than cash, then your Account will be credited with one additional Restricted Stock Unit for each share of Stock that would have been received as a dividend had your outstanding Restricted Stock Units been shares of Stock.

(d) Additional Restricted Stock Units credited via Dividend Equivalents shall vest or be forfeited at the same time as the Restricted Stock Units to which they relate.

3. <u>Restricted Period</u>. The period prior to the vesting date with respect each Restricted Stock Unit is referred to as the "Restricted Period." Subject to the provisions of the Plan and this Agreement,

unless vested or forfeited earlier as described in Section 4, 5, 6 or 7 of this Agreement, as applicable, your Restricted Stock Units will become vested and be settled as described in Section 8 below, as of the vesting date or dates indicated in the introduction to this Agreement.

4. <u>Disability or Death</u>. If during the Restricted Period you have a Termination of Service by reason of Disability or death, then the Restricted Stock Units will become fully vested as of the date of your Termination of Service and the Vesting Date shall become the date of your Termination of Service. Any Restricted Stock Units becoming vested by reason of your Termination of Service by reason of Disability or death shall be settled as provided in Section 8.

5. <u>Retirement</u>. If during the Restricted Period you have a Termination of Service by reason of Retirement, as reasonably determined and approved by the Committee or its delegates, then, subject to such approval, the number of Restricted Stock Units that become vested by reason of your Retirement will be prorated to reflect the portion of the vesting period during which you remained employed by the Company. For each Vesting Date occurring after such Termination of Service, such prorated portion shall equal the number of Restricted Stock Units scheduled to vest as of that Vesting Date, multiplied by a fraction equal to the number of full months of that portion of the vesting period completed as of your Termination of Service, divided by 12. Any Restricted Stock Units becoming vested by reason of your Retirement shall be settled as provided in Section 8.

6. <u>Termination of Service Following a Change in Control</u>. If during the Restricted Period there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 7), then your Restricted Stock Units shall become fully vested, and they shall be settled in accordance with Section 9. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such Plan.

7. <u>Other Termination of Service</u>. If during the Restricted Period you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 4, 5 or 6 above or Section 9 below, as determined by the Committee, then you shall thereupon forfeit any Restricted Stock Units that are still in a Restricted Period on your termination date. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

8. <u>Settlement of Vested Restricted Stock Units</u>. Subject to the requirements of Section 13 below, as promptly as practicable after the applicable Vesting Date, whether occurring upon your

Separation from Service or otherwise, but in no event later than 75 days after the Vesting Date, the Company shall transfer to you one share of Stock for each Restricted Stock Unit becoming vested at such time, net of any applicable tax withholding requirements in accordance with Section 10 below; provided, however, that, if you are a Specified Employee at the time of Separation from Service, then to the extent your Restricted Stock Units are deferred compensation subject to Section 409A of the Code, settlement of which is triggered by your Separation from Service (other than for death), payment shall not be made until the date which is six months after your Separation from Service.

Notwithstanding the foregoing, if you are resident or employed outside of the U.S., the Company, in its sole discretion, may provide for the settlement of the Restricted Stock Units in the form of:

(a) a cash payment (in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units) to the extent that settlement in shares of Stock (i) is prohibited under local law, (ii) would require you, the Company or an Affiliate to obtain the approval of any governmental or regulatory body in your country of residence (or country of employment, if different), (iii) would result in adverse tax consequences for you, the Company or an Affiliate or (iv) is administratively burdensome; or

(b) shares of Stock, but require you to sell such shares of Stock immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

9. <u>Settlement Following Change in Control</u>. Notwithstanding any provision of this Agreement to the contrary, the Company may, in its sole discretion, fulfill its obligation with respect to all or any portion of the Restricted Stock Units that become vested in accordance with Section 6 above, by:

(a) delivery of (i) the number of shares of Stock that corresponds with the number of Restricted Stock Units that have become vested or (ii) such other ownership interest as such shares of Stock that correspond with the vested Restricted Stock Units may be converted into by virtue of the Change in Control transaction;

(b) payment of cash in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units at that time; or

(c) delivery of any combination of shares of Stock (or other converted ownership interest) and cash having an aggregate Fair Market Value equal to the Fair Market Value of the Stock that corresponds with the number of Restricted Stock Units that have become vested at that time.

Settlement shall be made as soon as practical after the Restricted Stock Units become fully vested under Section 6, but in no event later than 30 days after such date.

10. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or your Employer, the ultimate liability for all income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("Tax-Related Items"), is and remains your responsibility and may exceed the amount actually withheld by the Company or your Employer, if any. You further acknowledge that the Company and/or your Employer (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Award, including, but not limited to, the grant, vesting or settlement of the Award, the subsequent sale of shares of Stock acquired pursuant to such settlement and the receipt of any Dividend Equivalents and/or dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect

of the Award to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or your Employer to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon settlement of the Award that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Restricted Stock Units and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, your Employer or its agent to cover the Tax-Related Items that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or your Employer may (in its sole discretion) allow you to make a cash payment to the Company, your Employer or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon settlement of the Award, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the earned Award, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock (or cash payment) or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

11. <u>Nontransferability</u>. During the Restricted Period and thereafter until Stock is transferred to you in settlement thereof, you may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Restricted Stock Units whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, or by will or by the laws of intestacy.

12. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the Restricted Stock Units until such time as a certificate of stock for the Stock issued in settlement of such Restricted Stock Units has been issued to you or such shares of Stock have been recorded in your name in book entry form. Until that time, you shall not have any voting rights with respect to the Restricted Stock Units. Except as provided in Section 9 above, no adjustment shall be made for dividends or distributions or other rights with respect to such shares for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

13. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement,

you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

14. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Restricted Stock Units is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Restricted Stock Units is not subject to the supervision of the local securities authorities.

15. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock, or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

16. <u>Repatriation; Compliance with Law</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of residence to comply with your personal obligations under local laws, rules and/or regulations in your country of residence (and country of employment, if different).

17. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Restricted Stock Units. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

18. <u>Change in Stock</u>. In the event of any change in Stock, by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of Restricted Stock Units subject to this Agreement shall be equitably adjusted by the Committee.

19. <u>Nature of the Award</u>. In accepting the Award, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Award is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of Restricted Stock Units, or benefits in lieu of Restricted Stock Units, even if Restricted Stock Units have been granted in the past;

(c) all decisions with respect to future Awards or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of the Award, the number of shares subject to the Award, and the vesting provisions applicable to the Award;

(d) the Award and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Award and the shares of Stock subject to the Award are not intended to replace any pension rights or compensation;

(g) the Award, the shares of Stock subject to the Award and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company, your Employer or any Affiliate;

(h) the future value of the shares of Stock underlying the Award is unknown, indeterminable and cannot be predicted with certainty;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Award resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Award to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Award and the benefits evidenced by this Agreement do not create any entitlement to have the Award or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Award

or of any amounts due to you pursuant to the settlement of the Award or the subsequent sale of any shares of Stock acquired upon settlement of the Award.

20. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

21.

21. <u>Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Restricted Stock Units and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and your Employer hold certain personal information about you, specifically: your name, home address and telephone number, date of birth, social security number or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). The Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and your Employer will process the Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. The Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for your participation in the Plan.

(b) The Company and your Employer will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or your Employer may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer the Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of the Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of the Data, (iv) oppose, for legal reasons, the collection, processing or transfer of the Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, your Restricted Stock Units will become null and void). You may seek to exercise

these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

22. <u>Addendum to Agreement</u>. Notwithstanding any provision of this Agreement to the contrary, the Restricted Stock Units shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to the Agreement, attached hereto as Exhibit A (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

23. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Restricted Stock Units, any shares of Stock acquired pursuant to the Restricted Stock Units and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

24. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

25. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Award or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

26. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdictions.

27. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Award, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English, so as to allow you to understand the terms and conditions of this Agreement, the Plan and any other documents related to the Award. If you have received this Agreement, the Plan or any other documents related to the Award translated into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

28. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this

Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

29. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibit A. Once you have read and understood this Agreement and Exhibit A, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibit A, as applicable, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Award granted hereunder.

EXHIBIT A

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Award is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 23 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

EUROPEAN UNION ("EU") / EUROPEAN ECONOMIC AREA ("EEA")

Personal Data. The following provision replaces Section 19 of the Agreement in its entirety:

The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all Restricted Stock Units, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Restricted Stock Units under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business

interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to participate in the Plan. As such, by accepting the Restricted Stock Units, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the Restricted Stock Units. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) International Data Transfers. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) <u>Data Subject Rights</u>. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources Department in writing. Your provision of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Restricted Stock Units, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources Department in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

CHILE

<u>Private Placement</u>. The following provision shall replace Section 14 of the Agreement:

The grant of the Restricted Stock Units hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;
- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and
- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;
- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y
- d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente

FRANCE

1. <u>Nature of Grant</u>. The Restricted Stock Units are not granted under the French specific regime provided by Articles L. 225-197-1 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Form of Payment</u>. Notwithstanding any provision in the Agreement or Plan to the contrary, the Restricted Stock Units shall be settled only in Shares (and not in cash).

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement, this Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Restricted Stock Units and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have

any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Restricted Stock Units and shares of Stock subject to the Restricted Stock Units do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

<u>Plan Acknowledgment</u>. In accepting the Restricted Stock Units, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 3: Restricted Period (terms of lapse of restrictions on Restricted Stock Units); Section 4: Disability or Death (terms of payment of Restricted Stock Units upon a Termination of Service by reason of Disability or death); Section 5: Retirement (terms of payment of Restricted Stock Units upon a Termination of Service by reason of retirement); Section 6: Termination of Service Following a Change in Control (terms of payment of Restricted Stock Units in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (forfeiture of Restricted Stock Units in other cases of Termination of Service); Section 10(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Restricted Stock Units and legally applicable to the participant); Section 11: Nontransferability (Restricted Stock Units shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 18: Change in Stock (right of the Company to equitably adjust the number of Restricted Stock Units subject to this Agreement in the event of any change in Stock); Section 19(j): Nature of the Award (waive any claim or entitlement to compensation or damages arising from forfeiture of the Restricted Stock Units resulting from a Termination of Service); Section 19(I): Nature of the Award (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Restricted Stock Units); Section 20: Committee Authority; Recoupment (right of the Committee to administer, construe, and make all determinations necessary or appropriate for the administration of the Restricted Stock Units and this Agreement, including the enforcement of any recoupment policy); Section 21: Non-Competition, Non-Solicitation and Confidentiality (the receipt of the Award is conditioned upon agreement to the Non-Competition, Non-Solicitation and Confidentiality Agreement attached hereto as Exhibit A); Section 23: Addendum to Agreement (the Restricted Stock Units are subject to the terms of the Addendum); Section 24: Additional Requirements (Company right to impose additional requirements on the Restricted Stock Units in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Restricted Stock Units and the Plan); Section 26: Electronic Delivery (Company may deliver documents related to the Award or Plan electronically); Section 27: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 28: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Restricted Stock Units does not constitute an employment relationship between you and the Company. You have been granted the Restricted Stock Units as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you

("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan are not part of the employment conditions and/or benefits provided by WBA Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. Extraordinary Item of Compensation. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Award, the shares of Stock subject to the Award and the income and value of the same is an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Restricted Stock Units, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Restricted Stock Units. Upon the grant of Restricted Stock Units, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Award, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted Restricted Stock Units under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Restricted Stock Units are granted on the assumption and condition that the Restricted Stock Units and the shares of Stock acquired upon settlement of the Restricted Stock Units shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Award shall be null and void.

Further, you understand and agree that the vesting of the Restricted Stock Units is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Restricted Stock Units may cease vesting immediately, in whole or in part, effective on the date of your Termination of Service (unless otherwise specifically provided in Section 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to Restricted Stock Units that were not vested as of the date of your Termination of Service, as described in the Plan and Agreement.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on your Award.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 7 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Restricted Stock Units. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Restricted Stock Units have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

<u>Securities Law Notification</u>. The Restricted Stock Units are not considered a public offering in Switzerland; therefore, the offer of Restricted Stock Units is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Restricted Stock Units constitute a prospectus as such

term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Restricted Stock Units may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Restricted Stock Units have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. Indemnification for Tax-Related Items. Without limitation to Section 10 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 10 of the Agreement.

2. <u>Exclusion of Claim</u>. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Restricted Stock Units, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Restricted Stock Units. Upon the grant of the Restricted Stock Units, you shall be deemed irrevocably to have waived any such entitlement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Restricted Stock Unit Award Agreement to which this Addendum is attached as Exhibit A, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of securities or financial instruments of Walgreens Boots Alliance, Inc. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN RESTRICTED STOCK UNIT AWARD AGREEMENT

Participant Name:

Participant ID: Grant Date:

Units Granted:

Vesting: One half of the Shares Granted vest on each of the first and second anniversaries of the Grant Date (the "Vesting Dates")

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the Restricted Stock Unit Award (the "Award") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the Plan. For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as it may be amended from time to time, is incorporated into this Agreement by this reference.

You and the Company agree as follows:

1. <u>Grant of Restricted Stock Units</u>. Pursuant to the approval and direction of the Compensation Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you the number of Restricted Stock Units specified above (the "Restricted Stock Units"), subject to the terms and conditions of the Plan and this Agreement.

2. <u>Restricted Stock Unit Account and Dividend Equivalents</u>. The Company will maintain an account (the "Account") on its books in your name to reflect the number of Restricted Stock Units awarded to you as well as any additional Restricted Stock Units credited as a result of Dividend Equivalents. The Account will be administered as follows:

(a) The Account is for recordkeeping purposes only, and no assets or other amounts shall be set aside from the Company's general assets with respect to such Account.

(b) As of each record date with respect to which a cash dividend is to be paid with respect to shares of Company common stock par value US\$.01 per share ("Stock"), the Company will credit your Account with an equivalent amount of Restricted Stock Units determined by dividing the value of the cash dividend that would have been paid on your Restricted Stock Units if they had been shares of Stock, divided by the value of Stock on such date.

(c) If dividends are paid in the form of shares of Stock rather than cash, then your Account will be credited with one additional Restricted Stock Unit for each share of Stock that would have been received as a dividend had your outstanding Restricted Stock Units been shares of Stock.

(d) Additional Restricted Stock Units credited via Dividend Equivalents shall vest or be forfeited at the same time as the Restricted Stock Units to which they relate.

3. <u>Restricted Period</u>. The period prior to the vesting date with respect each Restricted Stock Unit is referred to as the "Restricted Period." Subject to the provisions of the Plan and this Agreement, unless vested or forfeited earlier as described in Section 4 or 5 of this Agreement, as applicable, your Restricted Stock Units will become vested and be settled as described in Section 6 or 7 below, as of the vesting date or dates indicated in the introduction to this Agreement.

4. <u>Termination of Service Following a Change in Control</u>. If during the Restricted Period there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined below), then your Restricted Stock Units shall become fully vested, and they shall be settled in accordance with Section 7. For purposes of this Section 4, a Termination of Service initiated by your Employer shall include a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such Plan. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

5. <u>Other Termination of Service</u>. If during the Restricted Period you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 4 above or Section 7 below, as determined by the Committee, then you shall thereupon forfeit any Restricted Stock Units that are still in a Restricted Period on your termination date.

6. <u>Settlement of Vested Restricted Stock Units</u>. Subject to the requirements of Section 11 below, as promptly as practicable after the applicable Vesting Date, whether occurring upon your Separation from Service or otherwise, but in no event later than 75 days after the Vesting Date, the Company shall transfer to you one share of Stock for each Restricted Stock Unit becoming vested at such time, net of any applicable tax withholding requirements in accordance with Section 8 below; provided, however, that, if you are a Specified Employee at the time of Separation from Service, then to the extent your Restricted Stock Units are deferred compensation subject to Section 409A of the Code, settlement of which is triggered by your Separation from Service (other than for death), payment shall not be made until the date which is six months after your Separation from Service.

Notwithstanding the foregoing, if you are resident or employed outside of the U.S., the Company, in its sole discretion, may provide for the settlement of the Restricted Stock Units in the form of:

(a) a cash payment (in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units) to the extent that settlement in shares of Stock (i) is prohibited under local law, (ii) would require you, the Company or an Affiliate to obtain the approval of any governmental or regulatory body in your country of residence (or country of employment, if different), (iii) would result in adverse tax consequences for you, the Company or an Affiliate or (iv) is administratively burdensome; or

(b) shares of Stock, but require you to sell such shares of Stock immediately or within a specified period following your Termination of Service (in which case, you hereby agree that the Company shall have the authority to issue sale instructions in relation to such shares of Stock on your behalf).

7. <u>Settlement Following Change in Control</u>. Notwithstanding any provision of this Agreement to the contrary, the Company may, in its sole discretion, fulfill its obligation with respect to all or any portion of the Restricted Stock Units that become vested in accordance with Section 4 above, by:

(a) delivery of (i) the number of shares of Stock that corresponds with the number of Restricted Stock Units that have become vested or (ii) such other ownership interest as such shares of Stock that correspond with the vested Restricted Stock Units may be converted into by virtue of the Change in Control transaction;

(b) payment of cash in an amount equal to the Fair Market Value of the Stock that corresponds with the number of vested Restricted Stock Units at that time; or

(c) delivery of any combination of shares of Stock (or other converted ownership interest) and cash having an aggregate Fair Market Value equal to the Fair Market Value of the Stock that corresponds with the number of Restricted Stock Units that have become vested at that time.

Settlement shall be made as soon as practical after the Restricted Stock Units become fully vested under Section 4, but in no event later than 30 days after such date.

8. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or your Employer, the ultimate liability for all income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to your participation in the Plan and legally applicable to you ("Tax-Related Items"), is and remains your responsibility and may exceed the amount actually withheld by the Company or your Employer, if any. You further acknowledge that the Company and/or your Employer (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Award, including, but not limited to, the grant, vesting or settlement of the Award, the subsequent sale of shares of Stock acquired pursuant to such settlement and the receipt of any Dividend Equivalents and/or dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Award to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or your Employer to satisfy all Tax-Related Items. In this regard, except as provided below, the Company, your Employer or its agent

shall satisfy the obligations with regard to all Tax-Related Items by withholding from the shares of Stock to be delivered upon settlement of the Award that number of shares of Stock having a Fair Market Value equal to the amount required by law to be withheld. For purposes of the foregoing, no fractional shares of Stock will be withheld or issued pursuant to the grant of the Restricted Stock Units and the issuance of shares of Stock hereunder. Notwithstanding the foregoing, if you are a Section 16 officer of the Company under the Exchange Act at the time of any applicable tax withholding event, you may make a cash payment to the Company, your Employer or its agent to cover the Tax-Related Items that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan. If you are not a Section 16 officer of the Company at the time of any applicable tax withholding event, the Company and/or your Employer may (in its sole discretion) allow you to make a cash payment to the Company, your Employer to the Company, your Employer or its agent to cover such Tax-Related Items.

The Company may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates, including maximum applicable rates, in which case you will receive a refund of any over-withheld amount in cash and will have no entitlement to the share equivalent. If the obligation for Tax-Related Items is satisfied by withholding from the shares of Stock to be delivered upon settlement of the Award, for tax purposes, you will be deemed to have been issued the full number of shares of Stock subject to the earned Award, notwithstanding that a number of the shares of Stock are held back solely for the purpose of paying the Tax-Related Items.

The Company may refuse to issue or deliver the shares of Stock (or cash payment) or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

9. <u>Nontransferability</u>. During the Restricted Period and thereafter until Stock is transferred to you in settlement thereof, you may not sell, transfer, pledge, assign or otherwise alienate or hypothecate the Restricted Stock Units whether voluntarily or involuntarily or by operation of law, other than by beneficiary designation effective upon your death, or by will or by the laws of intestacy.

10. <u>Rights as Shareholder</u>. You shall have no rights as a shareholder of the Company with respect to the Restricted Stock Units until such time as a certificate of stock for the Stock issued in settlement of such Restricted Stock Units has been issued to you or such shares of Stock have been recorded in your name in book entry form. Until that time, you shall not have any voting rights with respect to the Restricted Stock Units. Except as provided in Section 7 above, no adjustment shall be made for dividends or distributions or other rights with respect to such shares for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

11. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

12. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Restricted Stock Units is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other

filings with the local securities authorities (unless otherwise required under local law), and the grant of the Restricted Stock Units is not subject to the supervision of the local securities authorities.

13. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock, or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

14. <u>Repatriation; Compliance with Law</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of regulations in your country of residence (and country of regulations in your country of residence). Finally, you agree to take any and all actions as may be required to comply with your personal obligations under local laws, rules and/or regulations in your country of residence and country of employment, if different).

15. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Restricted Stock Units. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

16. <u>Change in Stock</u>. In the event of any change in Stock, by reason of any stock dividend, recapitalization, reorganization, split-up, merger, consolidation, exchange of shares, or of any similar change affecting the shares of Stock, the number of Restricted Stock Units subject to this Agreement shall be equitably adjusted by the Committee.

17. <u>Nature of the Award</u>. In accepting the Award, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Award is voluntary and occasional and does not create any contractual or other right to receive future grants of Restricted Stock Units, or benefits in lieu of Restricted Stock Units, even if Restricted Stock Units have been granted in the past;

(c) all decisions with respect to future Awards or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of the Award, the number of shares subject to the Award, and the vesting provisions applicable to the Award;

(d) the Award and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Award and the shares of Stock subject to the Award are not intended to replace any pension rights or compensation;

(g) the Award, the shares of Stock subject to the Award and the value of same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments;

(h) the future value of the shares of Stock underlying the Award is unknown, indeterminable and cannot be predicted with certainty;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Award resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Award to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) unless otherwise provided herein, in the Plan or by the Company in its discretion, the Award and the benefits evidenced by this Agreement do not create any entitlement to have the Award or any such benefits transferred to, or assumed by, another company nor to be exchanged, cashed out or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Award or of any amounts due to you pursuant to the settlement of the Award or the subsequent sale of any shares of Stock acquired upon settlement of the Award.

18. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy, all of

which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan.

19. <u>Consent to Collection/Processing/Transfer of Personal Data</u>. Pursuant to applicable personal data protection laws, the Company hereby notifies you of the following in relation to your personal data and the collection, processing and transfer of such data in relation to the Company's grant of the Restricted Stock Units and your participation in the Plan. The collection, processing and transfer of personal data is necessary for the Company's administration of the Plan and your participation in the Plan, and your denial and/or objection to the collection, processing and transfer of personal data may affect your participation in the Plan. As such, you voluntarily acknowledge and consent (where required under applicable law) to the collection, use, processing and transfer of personal data as described herein:

(a) The Company and your Employer hold certain personal information about you, including (but not limited to) your name, home address and telephone number, date of birth, social security number or other employee identification number, salary, nationality, job title, any shares of Stock or directorships held in the Company, details of all entitlements to shares of Stock awarded, canceled, purchased, vested, unvested or outstanding in your favor, for the purpose of managing and administering the Plan ("Data"). The Data may be provided by you or collected, where lawful, from the Company, its Affiliates and/or third parties, and the Company and your Employer will process the Data for the exclusive purpose of implementing, administering and managing your participation in the Plan. The Data processing will take place through electronic and non-electronic means according to logics and procedures strictly correlated to the purposes for which Data are collected and with confidentiality and security provisions as set forth by applicable laws and regulations in your country of residence (or country of employment, if different). Data processing operations will be performed minimizing the use of personal and identification data when such operations are unnecessary for the processing purposes sought. Data will be accessible within the organization only by those persons requiring access for purposes of the implementation, administration and operation of the Plan and for the participation in the Plan.

(b) The Company and your Employer will transfer Data internally as necessary for the purpose of implementation, administration and management of your participation in the Plan, and the Company and/or your Employer may further transfer Data to any third parties assisting the Company in the implementation, administration and management of the Plan. These recipients may be located in the European Economic Area, or elsewhere throughout the world, such as the United States. You hereby authorize (where required under applicable law) the recipients to receive, possess, use, retain and transfer the Data, in electronic or other form, as may be required for the administration of the Plan and/or the subsequent holding of the shares of Stock on your behalf, to a broker or other third party with whom you may elect to deposit any shares of Stock acquired pursuant to the Plan.

(c) You may, at any time, exercise your rights provided under applicable personal data protection laws, which may include the right to (i) obtain confirmation as to the existence of Data, (ii) verify the content, origin and accuracy of the Data, (iii) request the integration, update, amendment, deletion or blockage (for breach of applicable laws) of the Data, (iv) oppose, for legal reasons, the collection, processing or transfer of the Data which is not necessary or required for the implementation, administration and/or operation of the Plan and your participation in the Plan, and (v) withdraw your consent to the collection, processing or transfer of Data as provided hereunder (in which case, your Restricted Stock Units will become null and void). You may seek to exercise these rights by contacting your Human Resources manager or the Company's Human Resources Department, who may direct the matter to the applicable Company privacy official.

(d) Finally, you understand that the Company may rely on a different legal basis for the processing and/or transfer of Data in the future and/or request you to provide another data privacy

consent. If applicable and upon request of the Company, you agree to provide an executed acknowledgment or data privacy consent (or any other acknowledgments, agreements or consents) to the Company or the Employer that the Company and/or the Employer may deem necessary to obtain under the data privacy laws in your country, either now or in the future. You understand that you will not be able to participate in the Plan if you fail to execute any such acknowledgment or consent requested by the Company and/or the Employer.

20. <u>Addendum to Agreement</u>. Notwithstanding any provision of this Agreement to the contrary, the Restricted Stock Units shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to the Agreement, attached hereto as Exhibit A (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

21. <u>Additional Requirements</u>. The Company reserves the right to impose other requirements on the Restricted Stock Units, any shares of Stock acquired pursuant to the Restricted Stock Units and your participation in the Plan to the extent the Company determines, in its sole discretion, that such other requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan. Such requirements may include (but are not limited to) requiring you to sign any agreements or undertakings that may be necessary to accomplish the foregoing.

22. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived unless the amendment or waiver is agreed to in writing, signed by you and by a duly authorized officer of the Company. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

23. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Award or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

24. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdictions.

25. <u>English Language</u>. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Award, be drawn up in English. If you have received this Agreement, the Plan or any other documents related to the Award translated into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

26. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this

Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

27. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibit A. Once you have read and understood this Agreement and Exhibit A, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibit A, as applicable, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Award granted hereunder.

EXHIBIT A

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

RESTRICTED STOCK UNIT AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Award is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 20 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Restricted Stock Units and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

FRANCE

1. <u>Nature of Grant</u>. The Restricted Stock Units are not granted under the French specific regime provided by Articles L. 225-197-1 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

ITALY

<u>Plan Acknowledgment</u>. In accepting the Restricted Stock Units, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 3: Restricted Period (terms of lapse of restrictions on Restricted Stock Units); Section 4: Termination of Service Following a Change in Control (terms of payment of Restricted Stock Units in the event of a Termination of Service following a Change in Control); Section 5: Other Termination of Service (forfeiture of Restricted Stock Units in other cases of Termination of Service); Section 8(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Restricted Stock Units and legally applicable to the participant; Section 9: Nontransferability (Restricted Stock Units shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 16: Change in Stock (right of the Company to equitably adjust the number of Restricted Stock Units subject to this Agreement in the event of any change in Stock); Section 17(j): Nature of the Award (waive any claim or entitlement to compensation or damages arising from forfeiture of the Restricted Stock Units resulting from a Termination of Service); Section 17(I): Nature of the Award (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Restricted Stock Units); Section 18: Committee Authority; Recoupment (right of the Committee to administer, construe, and make all determinations necessary or appropriate for the administration of the Restricted Stock Units and this Agreement, including the enforcement of any recoupment policy); Section 20: Addendum to Agreement (the Restricted Stock Units are subject to the terms of the Addendum); Section 21: Additional Requirements (Company right to impose additional requirements on the Restricted Stock Units in case such requirements are necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate operation and administration of the Restricted Stock Units and the Plan); Section 23: Electronic Delivery (Company may deliver documents related to the Award or Plan electronically): Section 24: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); Section 25: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Restricted Stock Units, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Restricted Stock Units. Upon the grant of Restricted Stock Units, you shall be deemed irrevocably to have waived any such entitlement.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

1. <u>No Offering of Securities in Russia</u>. The grant of the Restricted Stock Units is not intended to be an offering of securities within the territory of the Russian Federation, and you acknowledge and agree that you will be unable to make any subsequent sale of the shares of Stock acquired pursuant to the Restricted Stock Units in the Russian Federation.

2. <u>Repatriation Requirements</u>. You agree to promptly repatriate the proceeds resulting from the sale of shares of Stock acquired under the Plan to a foreign currency account at an authorized bank in Russia if legally required at the time shares of Stock are sold and to comply with all applicable local foreign exchange rules and regulations. Neither the Company nor any of its Affiliates shall be liable for any fines or penalties resulting from your failure to comply with applicable laws.

SPAIN

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Award, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted Restricted Stock Units under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Restricted Stock Units are granted on the assumption and condition that the Restricted Stock Units and the shares of Stock acquired upon settlement of the Restricted Stock Units shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Award shall be null and void.

Further, you understand and agree that the vesting of the Restricted Stock Units is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Restricted Stock Units may cease vesting immediately, in whole or in part, effective on the date of your Termination of Service (unless otherwise specifically provided in Section 4 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service for any of the above reasons, you may automatically lose any rights to Restricted Stock Units that were not vested as of the date of your Termination of Service, as described in the Plan and Agreement.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on your Award.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 4 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Restricted Stock Units. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Restricted Stock Units have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

Securities Law Notification. The Restricted Stock Units are not considered a public offering in Switzerland; therefore, the offer of Restricted Stock Units is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Restricted Stock Units constitute a prospectus as such term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Restricted Stock Units may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Restricted Stock Units have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

UNITED KINGDOM

1. Indemnification for Tax-Related Items. Without limitation to Section 8 of the Agreement, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of the immediately foregoing provision will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or your Employer (as applicable) for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in Section 8 of the Agreement.

2. Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Restricted Stock Units, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Restricted Stock Units. Upon the grant of the Restricted Stock Units, you shall be deemed irrevocably to have waived any such entitlement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Restricted Stock Unit Award Agreement to which this Addendum is attached as Exhibit A, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

UK SUB-PLAN

STOCK OPTION AWARD AGREEMENT

These materials, which may include descriptions of company stock plans, prospectuses and other information and documents, and the information they contain, are provided by Walgreens Boots Alliance, Inc., not by Fidelity, and are not an offer or solicitation by Fidelity for the purchase of any securities or financial instruments. These materials were prepared by Walgreens Boots Alliance, Inc., which is solely responsible for their contents and for compliance with legal and regulatory requirements. Fidelity is not connected with any offering or acting as an underwriter in connection with any offering of your company's securities or financial instruments. Fidelity does not review, approve or endorse the contents of these materials and is not responsible for their content.

WALGREENS BOOTS ALLIANCE, INC.

2013 OMNIBUS INCENTIVE PLAN

UK SUB-PLAN

STOCK OPTION AWARD AGREEMENT

Participant Name:

Participant ID:

Grant Date:

Grant Price:

No. of Shares under Option Granted:

Vesting: One third of the Shares Granted vest on each of the first, second and third anniversaries of the Grant Date (the "Vesting Dates")

Expiration Date:

Acceptance Date:

Electronic Signature:

This document (referred to below as this "Agreement") spells out the terms and conditions of the stock option (the "Option") granted to you by Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), pursuant to the UK Sub-Plan of the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan (the "UK Sub-Plan") on and as of the Grant Date designated above. Except as otherwise defined herein, capitalized terms used in this Agreement have the respective meanings set forth in the UK Sub-Plan (which incorporates and modifies the terms of the Walgreens Boots Alliance, Inc. 2013 Omnibus Incentive Plan" and, together with the UK Sub-Plan, the "Plan"). For purposes of this Agreement, "Employer" means the entity (the Company or the Affiliate) that employs you on the applicable date. The Plan, as in effect on the date of this Agreement and as it may be amended from time to time, is incorporated into this Agreement by this reference.

The terms of the Option, including any restrictions on the Shares underlying the Option, the times at which the Option may be exercised (in whole or in part), the circumstances under which the Option will lapse or be cancelled (in whole or in part), any conditions to which the exercise of the Option is subject (in whole or in part) and any mechanism for varying the terms of the Option, are set out in this Agreement and the rules of the Plan. The restrictions on the Shares underlying the Option include but are not limited to Section 20 (Recoupment) of this Agreement. The rules of the UK Sub-Plan, the rules of the US Plan and applicable Company policies can be accessed at MyHR or from your HR representative.

You and the Company agree as follows:

1. <u>Grant of Option</u>. Pursuant to the approval and direction of the Compensation and Leadership Performance Committee of the Company's Board of Directors (the "Committee"), the Company hereby grants you an Option to purchase all or any part of the number of Shares Granted set forth above of common stock of the Company, par value US\$.01 ("Stock"), at the per-share exercise price, which is 100% of the Fair Market Value of a share of Stock on the Grant Date (the "Exercise Price"), subject to the terms and conditions of the Plan and this Agreement. The Option shall not be treated as an incentive stock option within the meaning of Section 422 of the Code.

2. <u>Vesting/Exercise/Expiration</u>. You may not exercise the Option prior to the Vesting Date or Dates set forth above absent action by the Committee to waive or alter such restrictions as may be permitted by the Plan or as may be permitted under the below paragraphs. Thereafter, except as hereinafter provided, you may exercise the Option, to the extent it is vested, at any time and from time to time until the close of business on the Expiration Date set forth above. The Option may be exercised to purchase any number of whole shares of Stock, except that no purchase shall be for less than ten (10) full shares of Stock, or the remaining unexercised shares, if less. The Option is deemed to be "outstanding" until it has been exercised in full or expired pursuant to the terms of this Agreement.

3. <u>Disability</u>. If, without having fully exercised the Option, you have a Termination of Service due to disability or injury (as evidenced to the satisfaction of the Committee or its delegate), then any Shares Granted under the Option that are not yet vested at that time shall thereupon become vested and (a) you may exercise the Option for the full number of Shares Granted (less any shares for which the Option was previously exercised), but (b) your right to exercise the Option shall terminate upon the earlier of the Expiration Date or a date which is one (1) year following the date of your Termination of Service.

4. <u>Death</u>. If you die without having fully exercised the Option, then any Shares Granted under the Option that are not yet vested (but which have not lapsed or been forfeited) at that time shall thereupon become fully vested and (a) the Option may be exercised, subject to the UK Sub-Plan, by your personal representative(s) for the full number of Shares Granted (less any shares for which the Option was previously exercised), but (b), such person's right to exercise the Option shall terminate not later than one (1) year after the date of your death.

5. <u>Retirement</u>. If without having fully exercised the Option you have a Termination of Service by reason of retirement, as reasonably determined and approved by the Committee or its delegates, then, subject to such approval (a) any Shares Granted under the Option that are not vested at that time shall thereupon become vested, and (b) you may exercise the Option for the full number of Shares Granted (less any shares for which the Option was previously exercised) until the Expiration Date set forth above.

6. Termination of Service Following a Change in Control. If there is a Change in Control of the Company and within the one-year period thereafter you have a Termination of Service initiated by your Employer other than for Cause (as defined in Section 8), then any Shares Granted under the Option that are not yet vested at that time shall thereupon become vested, and you may exercise the Option for the full number of Shares Granted (less any shares for which the Option under this Section 6 shall be forfeited. The foregoing is also subject to the Committee's exercise of its discretion under Section 9.01 of the Plan. For purposes of this Section 6, a Termination of Service initiated by your Employer shall include (without limitation) a Termination of Employment for Good Reason under - and pursuant to the terms and conditions of - the Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, but only to the extent applicable to you as an eligible participant in such plan, and (without limitation) any Termination of Service other than for Cause which is initiated by your Employer and which falls within any of the circumstances listed in section 524(2B) of the UK Income Tax (Earnings and Pensions) Act 2003.

7. <u>Other Termination of Service</u>. If without having fully exercised the Option you have a voluntary or involuntary Termination of Service for any reason other than as set forth in Section 3, 4, 5 or 6 above, as determined by the Committee (including, without limitation, by reason of redundancy (within the meaning of the Employment Rights Act 1996 or the Employment Rights (Northern Ireland) Order 1996), a relevant transfer within the meaning of the Transfer of Undertakings (Protection of Employment) Regulations 2006 or the company with which you hold office or employment ceasing to be controlled by the Company), then (a) for any Shares Granted with respect to which such Termination of Service is prior to the applicable Vesting Date, the Option shall be forfeited, and (b) for any Shares Granted with respect to which such

Termination of Service is on or after the applicable Vesting Date, then your right to exercise the Option shall terminate upon the earlier of the Expiration Date or a date which is ninety (90) days after the date of your Termination of Service.

8. <u>Forfeiture of Outstanding Options Upon Termination for Cause or Upon Other Violations</u>. Notwithstanding any provision of this Agreement to the contrary, the Option (whether vested or unvested) shall immediately terminate if you are terminated for Cause or if and when you violate any obligation that you may have to the Company during or post-employment, including but not limited to a violation of any applicable provision of the NNCA Agreement (defined in Section 22 below) or any other non-competition, non-solicitation, confidentiality, non-disparagement or other restrictive covenant. For purposes of this Agreement, "Cause" means any one or more of the following, as determined by the Committee in its sole discretion:

(a) your commission of a felony or any crime of moral turpitude;

(b) your dishonesty or material violation of standards of integrity in the course of fulfilling your duties to the Company or any Affiliate;

(c) your material violation of a material written policy of the Company or any Affiliate violation of which is grounds for immediate termination;

(d) your willful and deliberate failure to perform your duties to the Company or any Affiliate in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) your failure to comply in any material respect with the United States ("U.S.") Foreign Corrupt Practices Act, the U.S. Securities Act of 1933, the U.S. Securities Exchange Act of 1934, the U.S. Sarbanes-Oxley Act of 2002, the U.S. Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the U.S. Truth in Negotiations Act, or any rules or regulations thereunder.

9. Exercise Process. The Option may be exercised by giving notice to Fidelity Stock Plan Services, LLC ("Fidelity"), the third party administrator administering the Option exercise process or any other third party administrator the Company may engage in the future. The exercise notice (a) shall be signed by you or (in the event of your death) your personal representative(s), (b) shall specify the number of full shares of Stock then elected to be purchased, and (c) shall be accompanied by payment in full of the Exercise Price of the shares of Stock to be purchased. Payment may be made in cash or by check payable to the order of the Company, and such payment shall include any income tax, social insurance, payroll tax, fringe benefits tax, payment on account or other tax-related items related to the Option that are required to be withheld ("Tax-Related Items"), as set forth in Section 10 below. Alternatively, the Committee may allow for a "same day sale" transaction pursuant to which a third party (engaged by you or the Company) loans funds to you to enable you to purchase the shares of Stock and pay any Tax-Related Items, and then sells a sufficient number of the exercised shares of Stock on your behalf to enable you to repay the loan and any fees. The remaining shares of Stock are then delivered by the third party to you.

As promptly as practicable after receipt of such notice of exercise and payment (including payment with respect to any Tax-Related Items), subject to Section 13 below, the Company shall in accordance with the UK Sub-Plan cause to be issued and delivered to you (or in the event of your death to your personal representative(s), as the case may be), certificates for the shares of Stock so purchased. Alternatively, such shares of Stock may be issued and held in book entry form.

10. <u>Responsibility for Taxes; Tax Withholding</u>.

(a) You acknowledge that, regardless of any action taken by the Company or your Employer, the ultimate liability for all Tax-Related Items related to your participation in the Plan and legally applicable to you is and remains your responsibility and may exceed the amount actually withheld by the Company or your Employer. You further acknowledge that the Company and/or your Employer (1) make no representations or undertakings regarding the treatment of any Tax-Related Items in connection with any aspect of the Option, including, but not limited to, the grant, vesting or

exercise of the Option, the subsequent sale of shares of Stock acquired pursuant to such exercise and the receipt of any dividends; and (2) do not commit to and are under no obligation to structure the terms of the grant or any aspect of the Option to reduce or eliminate your liability for Tax-Related Items or achieve any particular tax result. Further, if you are subject to Tax-Related Items in more than one jurisdiction between the Grant Date and the date of any relevant taxable or tax withholding event, as applicable, you acknowledge that the Company and/or your Employer (or former employer, as applicable) may be required to withhold or account for Tax-Related Items in more than one jurisdiction.

(b) Prior to any relevant taxable or tax withholding event, as applicable, you agree to make adequate arrangements satisfactory to the Company and/or your Employer to satisfy all Tax-Related Items. In this regard, if your notice to exercise your Option is not accompanied by a sufficient amount to cover the Tax-Related Items arising on exercise and you do not otherwise provide the Company or your Employer with a sufficient amount to cover the Tax-Related Items arising on exercise and you do not otherwise provide the Company or your Employer or its agent to satisfy the obligations with regard to all Tax-Related Items by one or a combination of the following: (i) withholding from your wages or other cash compensation paid to you by the Company and/or your Employer; or (ii) withholding from proceeds of the sale of sufficient shares of Stock acquired upon exercise of the Option either through a voluntary sale or through a mandatory sale arranged by the Company (on your behalf pursuant to this authorization without further consent). For purposes of the foregoing, no fractional shares of Stock will be issued pursuant to the grant of the Option and the issuance of shares of Stock hereunder.

Depending on the withholding method, the Company or your Employer may withhold or account for Tax-Related Items by considering applicable statutory withholding rates (as determined by the Company in good faith and in its sole discretion) or other applicable withholding rates. You agree to pay to the Company or your Employer any amount of Tax-Related Items that the Company or your Employer may be required to withhold or account for as a result of your participation in the Plan that cannot be satisfied by the means previously described. The Company may refuse to issue or deliver the shares of Stock or the proceeds from the sale of shares of Stock if you fail to comply with your obligations in connection with the Tax-Related Items.

(c) Without limitation to paragraph (a) and (b) above, you hereby agree that you are liable for all Tax-Related Items and hereby covenant to pay all such Tax-Related Items, as and when requested by the Company or, if different, your Employer or by Her Majesty's Revenue & Customs ("HMRC") (or any other tax authority or any other relevant authority). You also hereby agree to indemnify and keep indemnified the Company and, if different, your Employer against any Tax-Related Items that they are required to pay or withhold or have paid or will pay on your behalf to HMRC (or any other tax authority or any other relevant authority).

Notwithstanding the foregoing, if you are a director or executive officer (as within the meaning of Section 13(k) of the Exchange Act), the terms of paragraph (c) above will not apply. In the event that you are a director or executive officer and income tax due is not collected from or paid by you within 90 days after the U.K. tax year in which an event giving rise to the indemnification described above occurs, the amount of any uncollected tax may constitute a benefit to you on which additional income tax and national insurance contributions may be payable. You acknowledge that you ultimately will be responsible for reporting and paying any income tax due on this additional benefit directly to HMRC under the self-assessment regime and for reimbursing the Company or the Employer (as applicable for the value of any employee national insurance contributions due on this additional benefit, which the Company and/or your Employer may recover from you at any time thereafter by any of the means referred to in paragraph (b) above.

11. <u>Limited Transferability</u>. You may not sell, transfer, pledge, assign or otherwise alienate, dispose of, encumber or hypothecate the Option, whether voluntarily or involuntarily or by operation of law. During your lifetime, the Option and all rights granted hereunder shall be exercisable only by you. After your death, the Option may only be exercised by your personal representative(s) in accordance with Section 4 of the Agreement.

12. <u>Rights as Stockholder</u>. You shall have no rights as a stockholder of the Company with respect to the shares of Stock subject to the Option until such time as the Exercise Price has been paid and a certificate of stock for such shares has been issued to you or such shares of Stock have been recorded in your name in book entry form. Except as provided in Section 18 below, no adjustment shall be made for dividends or distributions or other rights with respect to such shares of Stock for which the record date is prior to the date on which you become the holder of record thereof. Anything herein to the contrary notwithstanding, if a law or any regulation of the U.S. Securities and Exchange Commission or of any other regulatory body having jurisdiction shall require the Company or you to take any action before shares of Stock can be delivered to you hereunder, then the date of delivery of such shares may be delayed accordingly.

13. <u>Securities Laws</u>. If a Registration Statement under the U.S. Securities Act of 1933, as amended, is not in effect with respect to the shares of Stock to be delivered pursuant to this Agreement, you hereby represent that you are acquiring the shares of Stock for investment and with no present intention of selling or transferring them and that you will not sell or otherwise transfer the shares of Stock except in compliance with all applicable securities laws and requirements of any stock exchange on which the shares of Stock may then be listed.

14. <u>Not a Public Offering</u>. If you are resident outside the U.S., the grant of the Option is not intended to be a public offering of securities in your country of residence (or country of employment, if different). The Company has not submitted any registration statement, prospectus or other filings with the local securities authorities (unless otherwise required under local law), and the grant of the Option is not subject to the supervision of the local securities authorities.

15. Insider Trading/Market Abuse Laws. By participating in the Plan, you agree to comply with the Company's policy on insider trading, to the extent that it is applicable to you. You further acknowledge that, depending on your or your broker's country of residence or where the shares of Stock are listed, you may be subject to insider trading restrictions and/or market abuse laws that may affect your ability to accept, acquire, sell or otherwise dispose of shares of Stock, rights to shares of Stock (e.g., the Option) or rights linked to the value of shares of Stock during such times you are considered to have "inside information" regarding the Company as defined by the laws or regulations in your country. Local insider trading laws and regulations may prohibit the cancellation or amendment of orders you place before you possessed inside information. Furthermore, you could be prohibited from (i) disclosing the inside information to any third party (other than on a "need to know" basis) and (ii) "tipping" third parties or causing them otherwise to buy or sell securities. You understand that third parties include fellow employees and/or service providers. Any restrictions under these laws and regulations are separate from and in addition to any restrictions that may be imposed under any applicable Company insider trading policy. You acknowledge that it is your responsibility to comply with any applicable restrictions and, therefore, you should consult your personal advisor on this matter.

16. <u>Repatriation; Compliance with Law; Method of Exercise</u>. If you are resident or employed outside the U.S., you agree to repatriate all payments attributable to the shares of Stock and/or cash acquired under the Plan in accordance with applicable foreign exchange rules and regulations in your country of residence (and country of employment, if different). In addition, you agree to take any and all actions, and consent to any and all actions taken by the Company and its Affiliates, as may be required to allow the Company and its Affiliates to comply with local laws, rules and/or regulations in your country of residence (and country of employment, if different). Finally, you agree to take any and all actions as may be required

to comply with your personal obligations under local laws, rules and/or regulations in your country of residence (and country of employment, if different).

17. <u>No Advice Regarding Grant</u>. No employee of the Company is permitted to advise you regarding your participation in the Plan or your acquisition or sale of the shares of Stock underlying the Option. Investment in shares of Stock involves a degree of risk. Before deciding to purchase shares of Stock pursuant to the Option, you should carefully consider all risk factors relevant to the acquisition of shares of Stock under the Plan and you should carefully review all of the materials related to the Option and the Plan. You are hereby advised to consult with your own personal tax, legal and financial advisors before taking any action related to the Plan.

18. <u>Change in Stock</u>. The Option may be adjusted by the Committee in accordance with Rule 4.2 of the UK Sub-Plan.

19. <u>Nature of the Option</u>. In accepting the Option, you acknowledge, understand and agree that:

(a) the Plan is established voluntarily by the Company, it is discretionary in nature and limited in duration, and it may be modified, amended, suspended or terminated by the Company, in its sole discretion, at any time;

(b) the grant of the Option is exceptional, voluntary and occasional and does not create any contractual or other right to receive future grants of stock options, or benefits in lieu of stock options, even if stock options have been granted in the past;

(c) all decisions with respect to future grants of stock options or other grants, if any, will be at the sole discretion of the Company, including, but not limited to, the form and timing of any grant, the number of shares of Stock subject to the stock options, vesting provisions, and the exercise price applicable to the stock option;

(d) the Option and your participation in the Plan shall not create a right to employment or be interpreted as forming an employment or service contract with the Company or any Affiliate and shall not interfere with the ability of the Company, your Employer or an Affiliate, as applicable, to terminate your employment or service relationship;

(e) you are voluntarily participating in the Plan;

(f) the Option and the shares of Stock subject to the Option are not intended to replace any pension rights or compensation;

(g) the Option, the shares of Stock subject to the Option and the income and value of the same, is an extraordinary item of compensation outside the scope of your employment (and employment contract, if any) and is not part of normal or expected compensation for any purpose, including, without limitation, calculating any severance, resignation, termination, redundancy, dismissal, end-of-service payments, bonuses, long-service awards, pension or retirement or welfare benefits or similar payments and in no event should be considered as compensation for, or relating in any way to, past services for the Company, your Employer or any Affiliate;

(h) the future value of the shares of Stock underlying the Option is unknown, indeterminable and cannot be predicted with certainty and if you exercise the vested Option, the value of the underlying shares of Stock may increase or decrease, even below the Exercise Price;

(i) unless otherwise determined by the Committee in its sole discretion, a Termination of Service shall be effective from the date on which active employment or service ends and shall not be extended by any statutory or common law notice of termination period;

(j) no claim or entitlement to compensation or damages shall arise from forfeiture of the Option resulting from a Termination of Service (for any reason whatsoever, whether or not later found to be invalid or in breach of employment laws in the jurisdiction where you are employed or the terms of your employment agreement, if any), and in consideration of the grant of the Option to which you are otherwise not entitled, you irrevocably agree never to institute any claim against the Company, your Employer or any Affiliate, waive your ability, if any, to bring any such claim, and release the Company, the Employer and all Affiliates from any such claim; if, notwithstanding the foregoing, any such claim is allowed by a court of competent jurisdiction, then, by participating in the Plan, you shall be deemed irrevocably to have agreed not to pursue such claim and agree to execute any and all documents necessary to request dismissal or withdrawal of such claim;

(k) you do not have any entitlement to have the Option cashed out and, unless otherwise provided herein, in the Plan or by the Company in its discretion in accordance with the Plan, the Option and the benefits evidenced by this Agreement do not create any entitlement to have the Option or any such benefits transferred to, or assumed by, another company nor to be exchanged, or substituted for, in connection with any corporate transaction affecting the shares of Stock of the Company; and

(I) neither the Company nor any Affiliate shall be liable for any foreign exchange rate fluctuation between your local currency and the U.S. dollar that may affect the value of the Option or of any amounts due to you pursuant to the settlement of the Option or the subsequent sale of any shares of Stock acquired upon settlement of the Option.

20. <u>Committee Authority; Recoupment</u>. It is expressly understood that the Committee is authorized to administer, construe, and make all determinations necessary or appropriate for the administration of the Plan and this Agreement, including the enforcement of any recoupment policy which applies in respect of the Option and any shares you acquire pursuant to the Option, all of which shall be binding upon you and any claimant. Any inconsistency between this Agreement and the Plan shall be resolved in favor of the Plan. If there is any conflict between the terms of the UK Sub-Plan and the US Plan, the terms of the UK Sub-Plan shall take precedence.

The Option and any shares of Stock you acquire pursuant to the Option shall be subject to the terms and conditions of any recoupment policy adopted by the Company in effect at the time of the grant of the Option as amended from time to time as required by law or any regulatory body, or any other recoupment policy adopted by the Company from time to time as may be required by law or any regulatory body.

21. <u>Personal Data</u>. The Company, with its registered address at 108 Wilmot Road, Deerfield, Illinois 60015, U.S.A. is the controller responsible for the processing of your personal data by the Company and the third parties noted below.

(a) <u>Data Collection and Usage</u>. Pursuant to applicable data protection laws, you are hereby notified that the Company collects, processes and uses certain personal information about you for the legitimate purpose of implementing, administering and managing the Plan and generally administering awards; specifically: your name, home address, email address and telephone number, date of birth, social insurance number or other identification number, salary, citizenship, job title, any shares or directorships held in the Company, and details of all stock options, any entitlement to shares of Stock awarded, canceled, exercised, vested, or outstanding in your favor, which the Company receives from you or the Employer ("Personal Data"). In granting the Option under the Plan, the Company will collect, process, use, disclose and transfer (collectively, "Processing") Personal Data for purposes of implementing, administering and managing the Plan. The Company's legal basis for the Processing of Personal Data is the Company's legitimate business interests of managing the Plan, administering employee awards and complying with its contractual and statutory obligations, as well as the necessity of the Processing for the Company to perform its contractual obligations under the Agreement and the Plan. Your refusal to provide Personal Data would make it impossible for the Company to perform its contractual obligations and may affect your ability to

participate in the Plan. As such, by accepting the Option, you voluntarily acknowledge the Processing of your Personal Data as described herein.

(b) <u>Stock Plan Administration Service Provider</u>. The Company may transfer Personal Data to Fidelity Stock Plan Services, LLC ("Fidelity"), an independent service provider based, in relevant part, in the United States, which may assist the Company with the implementation, administration and management of the Plan. In the future, the Company may select a different service provider and share Personal Data with another company that serves in a similar manner. The Company's service provider will open an account for you to receive and trade shares of Stock pursuant to the Option. The Processing of Personal Data will take place through both electronic and non-electronic means. Personal Data will only be accessible by those individuals requiring access to it for purposes of implementing, administering and operating the Plan. When receiving your Personal Data, if applicable, Fidelity provides appropriate safeguards in accordance with the EU Standard Contractual Clauses or other appropriate cross-border transfer solutions. By participating in the Plan, you understand that the service provider will Process your Personal Data for the purposes of implementing, administering and managing your participation in the Plan.

(c) <u>International Data Transfers</u>. The Company is based in the United States, which means it will be necessary for Personal Data to be transferred to, and Processed in the United States. When transferring your Personal Data to the United States, the Company provides appropriate safeguards in accordance with the EU Standard Contractual Clauses, and other appropriate cross-border transfer solutions. You may request a copy of the appropriate safeguards with Fidelity or the Company by contacting your Human Resources manager or the Company's Human Resources Department.

(d) <u>Data Retention</u>. The Company will use Personal Data only as long as is necessary to implement, administer and manage your participation in the Plan or as required to comply with legal or regulatory obligations, including tax and securities laws. When the Company no longer needs Personal Data related to the Plan, the Company will remove it from its systems. If the Company keeps Personal Data longer, it would be to satisfy legal or regulatory obligations and the Company's legal basis would be for compliance with relevant laws or regulations.

(e) <u>Data Subject Rights</u>. To the extent provided by law, you have the right to (i) subject to certain exceptions, request access or copies of Personal Data the Company Processes, (ii) request rectification of incorrect Personal Data, (iii) request deletion of Personal Data, (iv) place restrictions on Processing of Personal Data, (v) lodge complaints with competent authorities in your country, and/or (vi) request a list with the names and addresses of any potential recipients of Personal Data. To receive clarification regarding your rights or to exercise your rights, you may contact your Human Resources manager or the Company's Human Resources Department. You also have the right to object, on grounds related to a particular situation, to the Processing of Personal Data, as well as opt-out of the Plan herein, in any case without cost, by contacting your Human Resources manager or the Company's Human Resources manager or the Company's Human Resources of Personal Data is a contractual requirement. You understand, however, that the only consequence of refusing to provide Personal Data is that the Company may not be able to administer the Option, or grant other awards or administer or maintain such awards. For more information on the consequences of the refusal to provide Personal Data, you may contact your Human Resources manager or the Company's Human Resources Department in writing. You may also have the right to lodge a complaint with the relevant data protection supervisory authority.

22. <u>Non-Competition, Non-Solicitation and Confidentiality</u>. Subject to the last sentence of this Section 22, as a condition to the receipt of the Option, you must agree to the Non-Competition, Non-Solicitation and Confidentiality Agreement (the "NNCA Agreement") attached hereto as Exhibit A. By clicking the acceptance box for this Agreement, you also agree to the terms and conditions expressed in the NNCA

Agreement. Failure to accept the terms of this Agreement and NNCA Agreement within 120 days of the Grant Date shall constitute your decision to decline to accept this Award. Notwithstanding the foregoing, to the extent that you are employed by your Employer pursuant to an employment agreement governed by the laws of England, Wales, Scotland and/or Northern Ireland, Paragraphs 2 and 3 of the NNCA Agreement shall not apply to you.

23. Addendum to Agreement. Notwithstanding any provision of this Agreement to the contrary, the Option shall be subject to any special terms and conditions for your country of residence (and country of employment, if different) as set forth in the addendum to this Agreement, attached hereto as Exhibit B (the "Addendum"). Further, if you transfer your residence and/or employment to another country reflected in the Addendum, the special terms and conditions for such country will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan (or the Company may establish alternative terms and conditions as may be necessary or advisable to accommodate your transfer). The Addendum shall constitute part of this Agreement.

24. <u>Amendment or Modification, Waiver</u>. Except as set forth in the Plan, no provision of this Agreement may be amended or waived. No waiver of any condition or provision of this Agreement shall be deemed a waiver of a similar or dissimilar condition or provision at the same time, any prior time or any subsequent time.

25. <u>Electronic Delivery</u>. The Company may, in its sole discretion, deliver by electronic means any documents related to the Option or your future participation in the Plan. You hereby consent to receive such documents by electronic delivery and agree to participate in the Plan through an on-line or electronic system established and maintained by the Company or a third party designated by the Company.

26. <u>Governing Law and Jurisdiction</u>. This Agreement is governed by the substantive and procedural laws of the state of Illinois. You and the Company shall submit to the exclusive jurisdiction of, and venue in, the courts in Illinois in any dispute relating to this Agreement without regard to any choice of law rules thereof which might apply the laws of any other jurisdiction.

27. English Language. If you are resident in a country where English is not an official language, you acknowledge and agree that it is your express intent that this Agreement, the Plan and all other documents, notices and legal proceedings entered into, given or instituted pursuant to the Option, be drawn up in English. You further acknowledge that you are sufficiently proficient in English, or have consulted with an advisor who is sufficiently proficient in English, so as to allow you to understand the terms and conditions of this Agreement, the Plan or any other document related to the Option. If you have received this Agreement, the Plan or any other document related into a language other than English, and if the meaning of the translated version is different than the English version, the English version will control.

28. <u>Conformity with Applicable Law</u>. If any provision of this Agreement is determined to be invalid, illegal or unenforceable in any respect under any applicable law or rule in any jurisdiction, such invalidity, illegality or unenforceability shall not affect the validity, legality or enforceability of any other provision of this Agreement or the validity, legality or enforceability of such provision in any other jurisdiction, but this Agreement shall be reformed, construed and enforced in such jurisdiction as if such invalid, illegal or unenforceable provision had never been contained herein.

29. <u>Successors</u>. This Agreement shall be binding upon and inure to the benefit of any successor or successors of the Company and any person or persons who shall, upon your death, acquire any rights hereunder.

This Agreement contains highly sensitive and confidential information. Please handle it accordingly.

Please read the attached Exhibits A and B. Once you have read and understood this Agreement and Exhibits A and B, please click the acceptance box to certify and confirm your agreement to be bound by the terms and conditions of this Agreement and Exhibits A and B, and to acknowledge your receipt of the Prospectus, the Plan and this Agreement and your acceptance of the terms and conditions of the Option granted hereunder.

<u>EXHIBIT A</u>

WALGREENS BOOTS ALLIANCE, INC. NON-COMPETITION, NON-SOLICITATION

AND CONFIDENTIALITY AGREEMENT

This Exhibit (the "Non-Compete Agreement") forms a part of the Stock Option Award Agreement covering Options awarded to an employee ("Employee" or "I") of Walgreens Boots Alliance, Inc. or an affiliate thereof, on behalf of itself, its affiliates, subsidiaries, and successors (collectively referred to as the "Company").

WHEREAS, the Company develops and/or uses valuable business, technical, proprietary, customer and patient information it protects by limiting its disclosure and by keeping it secret or confidential;

WHEREAS, Employee acknowledges that during the course of employment, he or she has or will receive, contribute, or develop such Confidential Information and Trade Secrets (as defined below); and

WHEREAS, the Company desires to protect from its competitors such Confidential Information and Trade Secrets and also desires to protect its legitimate business interests and goodwill in maintaining its employee and customer relationships.

NOW THEREFORE, in consideration of the Stock Option issued to Employee pursuant to the Agreement to which this is attached as Exhibit A and for other good and valuable consideration, including but not limited to employment or continued employment, the specialized knowledge, skill and training that the Company provides Employee, the goodwill that Employee develops with customers on behalf of the Company, Employee agrees to be bound by the terms of this Non-Compete Agreement as follows:

1. <u>Confidentiality</u>.

(a) At all times during and after the termination of my employment with the Company, I will not, without the Company's prior written permission, directly or indirectly for any purpose other than performance of my duties for the Company, utilize or disclose to anyone outside of the Company any Trade Secrets (defined in subparagraph 1(a)(i)) or other Confidential Information (defined in subparagraph 1(a)(ii)) or any information received by the Company in confidence from or about third parties, as long as such matters remain Trade Secrets or otherwise confidential.

(i) For purposes of this Non-Compete Agreement, "**Trade Secrets**" means a form of intellectual property that are protectable under applicable state and/or Federal law, including the Uniform Trade Secrets Act (as amended and adapted by the states) and the Federal Defend Trade Secrets Act of 2016 (the "DTSA"). They include all tangible and intangible (*e.g.*, electronic) forms and types of information that is held and kept confidential by the Company and is not generally known outside of the Company, including but not limited to information about: the Company's financial, business, scientific, technical, economic, or engineering information, including patterns, plans, compilations, program devices, formulas, designs, prototypes, methods, techniques, processes, procedures, programs or codes, and may in particular include such things as pricing information, business records, software

programs, algorithms, inventions, patent applications, and designs and processes not known outside the Company.

(ii) For purposes of this Non-Compete Agreement, "**Confidential Information**" means Trade Secrets and, more broadly, any other tangible and intangible (e.g., electronic) forms and types of information that are held and kept confidential by the Company and are not generally known outside the Company, and which relates to the actual or anticipated business of the Company or the Company's actual or prospective vendors or clients. Confidential Information shall not be considered generally known to the public if is revealed improperly to the public by me or others without the Company's express written consent and/or in violation of an obligation of confidentiality to the Company. Examples of Confidential Information include, but are not limited to: customer, referral source, supplier and contractor identification and contacts; special contract terms; pricing and margins; business, marketing and customer plans and strategies; financial data; company created (or licensed) techniques; technical know-how; research, development and production information; processes, prototypes, software, patent applications and plans, projections, proposals, discussion guides, and/or personal or performance information about employees.

(b) I understand that this obligation of non-disclosure shall last so long as the information remains confidential. I, however, understand that, if I live and work primarily in Wisconsin, Virginia, or any other state requiring a temporal limit on non-disclosure clauses, Confidential Information shall be protected for two (2) years following termination of my employment (for any reason). I also understand that Trade Secrets are protected by statute and are not subject to any time limits. I also agree to contact the Company before using, disclosing, or distributing any Confidential Information or Trade Secrets if I have any questions about whether such information is protected information.

The restrictions set forth in this paragraph are in addition to and not in lieu of any obligations I have by (C) law with respect to the Company's Confidential Information, including any obligations I may owe under the DTSA and any applicable state statutes. Nothing herein shall prohibit me from divulging evidence of criminal wrongdoing to law enforcement or prohibit me from disclosing Confidential Information or Trade Secrets if compelled by order of court or an agency of competent jurisdiction or as required by law; however, I shall promptly inform the Company of any such situations and shall take reasonable steps to prevent disclosure of Confidential Information or Trade Secrets until the Company has been informed of such required disclosure and has had a reasonable opportunity to seek a protective order. Pursuant to the DTSA, I understand that an individual may not be held criminally or civilly liable under any federal or state trade secret law for the disclosure of a Trade Secret that: (A) is made (i) in confidence to a federal, state or local government official, either directly or indirectly, or to an attorney, and (ii) solely for the purpose of reporting or investigating a suspected violation of law; or (B) is made in a complaint or other document filed in a lawsuit or other proceeding, if such filing is made under seal. Additionally, I understand that an individual who files a lawsuit for retaliation by an employer for reporting a suspected violation of law may disclose the Trade Secret to his or her attorney and use the Trade Secret information in the court proceeding, so long as any document containing the Trade Secret is filed under seal and the individual does not disclose the Trade Secret, except pursuant to court order. Nothing in this Non-Compete Agreement is intended to conflict with the DTSA or create liability for disclosures of Trade Secrets that are expressly allowed by DTSA.

2. <u>Non-Competition</u>. I agree that during my employment with the Company and for twelve (12) months after the termination of my employment (for any reason), I will not, directly or indirectly have Responsibilities with respect to any Competing Business Line. As set forth in paragraph 9(a) below, I understand that the restrictions in this paragraph apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary. These restrictions shall not apply to passive investments of less than five

percent (5%) ownership interest in any entity. For purposes of this Non-Compete Agreement, "**Responsibilities**" means the same or similar material responsibilities I performed for the Company during the two (2) years prior to my last day of employment with the Company and within the same geographic scope, or portion thereof, where I performed those responsibilities for the Company. For purposes of this Non-Compete Agreement, "**Competing Business Line**" means any business that is in competition with any business engaged in by the Company and for which I had Responsibilities during the two (2) years prior to my last day of employment with the Company. Competing Business Line shall also include businesses or business lines that may not be directly competitive with the Company in most respects (such as pharmacy benefit managers), but only to the extent I am engaged by any such business in a role: (a) that involves my performing Responsibilities for Competing Products or Services; or (b) where I would be called upon to inevitably rely upon or disclose Confidential Information and such reliance or disclosure would competitively harm the Company. For purposes of this Non-Compete Agreement, "**Competing Products or Services**" means products or services that are competitive with products or services offered by, developed by, designed by or distributed by the Company during the two (2) years prior to my last day of employment with the Company.

3. <u>Non-Solicitation</u>. I agree that during my employment with the Company and for two (2) years after the termination of my employment from the Company (for any reason):

(a) I will not directly or indirectly, solicit any Restricted Customer for purposes of providing Competing Products or Services, or offer, provide or sell Competing Products or Services to any Restricted Customer. For purposes of this Non-Compete Agreement, "Restricted Customer" means any person, company or entity that was a customer, vendor, supplier or referral source of the Company and with which I had direct contact for purposes of performing responsibilities for the Company or for which I had supervisory responsibilities on behalf of the Company, in either case at any time during the two (2) years prior to my last day of employment with the Company. To the extent permitted by applicable law, "Restricted Customer" also means any prospective customer(s), vendor(s), supplier(s) or referral source(s) with which I had business contact on behalf of the Company in the twelve (12) months prior to my last day of employment with the Company; and

(b) I will not, nor will I assist any third party to, directly or indirectly (i) raid, solicit, or attempt to persuade any thencurrent employee of the Company with whom I currently work or with whom I had direct contact work during the two years prior to my last day of employment with the Company, and who possesses or had access to Confidential Information of the Company, to leave the employ of the Company; (ii) interfere with the performance by any such employee of his/her duties for the Company; and/or (iii) communicate with any such employee for the purposes described in items (i) and (ii) in this subparagraph 3(b).

4. <u>Non-Inducement</u>. I will not directly or indirectly assist or encourage any person or entity in carrying out or conducting any activity that would be prohibited by this Non-Compete Agreement if such activity were carried out or conducted by me.

5. <u>Non-Disparagement</u>. During my employment with the Company and thereafter, I agree not to make negative comments or otherwise disparage the Company or any of its officers, directors, employees, shareholders, members, agents or products. The foregoing shall not be violated by truthful statements in response to legal process, required governmental testimony or filings, or administrative or arbitral proceedings (including, without limitation, depositions in connection with such proceedings); and the foregoing shall not apply to any claims for harassment or discrimination to the extent so restricted by applicable state law.

6. <u>Intellectual Property.</u> The term **"Intellectual Property"** shall mean all trade secrets, ideas, inventions, designs, developments, devices, software, computer programs, methods and processes (whether or not patented or patentable, reduced to practice or included in the Confidential Information) and

all patents and patent applications related thereto, all copyrights, copyrightable works and mask works (whether or not included in the Confidential Information) and all registrations and applications for registration related thereto, all Confidential Information, and all other proprietary rights contributed to, or conceived or created by, or reduced to practice by Employee or anyone acting on his/her behalf (whether alone or jointly with others) at any time from the beginning of Employee's employment with the Company to the termination of that employment plus ninety (90) days, that (i) relate to the business or to the actual or anticipated research or development of the Company; (ii) result from any services that Employee or anyone acting on its behalf perform for the Company; or (iii) are created using the equipment, supplies or facilities of the Company or any Confidential Information.

a. <u>Ownership</u>. All Intellectual Property is, shall be and shall remain the exclusive property of the Company. Employee hereby assigns to the Company all right, title and interest, if any, in and to the Intellectual Property; provided, however, that, when applicable, the Company shall own the copyrights in all copyrightable works included in the Intellectual Property pursuant to the "work-made-for-hire" doctrine (rather than by assignment), as such term is defined in the 1976 Copyright Act. All Intellectual Property shall be owned by the Company irrespective of any copyright notices or confidentiality legends to the contrary that may be placed on such works by Employee or by others. Employee shall ensure that all copyright notices and confidentiality legends on all work product authored by Employee or anyone acting on his/her behalf shall conform to the Company's practices and shall specify the Company as the owner of the work. The Company hereby provides notice to Employee that the obligation to assign does not apply to an invention for which no equipment, supplies, facility, or Trade Secrets of the Company was used and which was developed entirely on the Employee's own time, unless (i) the invention relates (1) to the business of the Company, or (2) to the Company's actual or demonstrably anticipated research or development, or (ii) the invention results from any work performed by Employee for the Company.

b. <u>Keep Records</u>. Employee shall keep and maintain, or cause to be kept and maintained by anyone acting on his/her behalf, adequate and current written records of all Intellectual Property in the form of notes, sketches, drawings, computer files, reports or other documents relating thereto. Such records shall be and shall remain the exclusive property of the Company and shall be available to the Company at all times during my employment with the Company.

c. <u>Assistance</u>. Employee shall supply all assistance requested in securing for the Company's benefit any patent, copyright, trademark, service mark, license, right or other evidence of ownership of any such Intellectual Property, and will provide full information regarding any such item and execute all appropriate documentation prepared by Company in applying or otherwise registering, in the Company's name, all rights to any such item or the defense and protection of such Intellectual Property.

d. <u>Prior Inventions</u>. Employee has disclosed to the Company any continuing obligations to any third party with respect to Intellectual Property. Employee claims no rights to any inventions created prior to his/her employment for which a patent application has not previously been filed, unless he/she has described them in detail on a schedule attached to this Non-Compete Agreement.

e. <u>Trade Secret Provisions</u>. The provisions in paragraph 1 of this Non-Compete Agreement with regard to Trade Secrets and the DTSA shall apply as well in the context of the parties' Intellectual Property rights and obligations.

7. <u>Return of Company Property</u>. I agree that all documents and data accessible to me during my employment with the Company, including Confidential Information and Trade Secrets, regardless of format (electronic or hard copy), including but not limited to any Company computer, monitor, printer equipment, external drives, wireless access equipment, telecom equipment and systems ("Company Equipment"), are and remain the sole and exclusive property of the Company and/or its clients, and must

be returned to the Company upon separation or upon demand by the Company. I further agree that I will provide passwords to access such Company Equipment and I will not print, retain, copy, destroy, modify or erase Company U.S. data on Company Equipment or otherwise wipe Company Equipment prior to returning the Company Equipment.

8. <u>Consideration and Acknowledgments</u>. I acknowledge and agree that the covenants described in this Non-Compete Agreement are essential terms, and the underlying Stock Option Award would not be provided by the Company in the absence of these covenants. I further acknowledge that these covenants are supported by adequate consideration as set forth in this Non-Compete Agreement and are not in conflict with any public interest. I further acknowledge and agree that I fully understand these covenants, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms. I further acknowledge and agree that these covenants are reasonable and enforceable in all respects.

9. Enforceability; General Provisions.

(a) I agree that the restrictions contained in this Non-Compete Agreement are reasonable and necessary to protect the Company's legitimate business interests and that full compliance with the terms of this Non-Compete Agreement will not prevent me from earning a livelihood following the termination of my employment, and that these covenants do not place undue restraint on me. I further understand that the restrictions in this Non-Compete Agreement apply no matter whether my employment is terminated by me or the Company and no matter whether that termination is voluntary or involuntary.

(b) Because the Company's current base of operations is in Illinois and my connections thereto, (i) this Non-Compete Agreement shall be governed by and construed in accordance with the laws of the State of Illinois, where this Non-Compete Agreement is entered into, without giving effect to any conflict of law provisions, and (ii) I consent to personal jurisdiction and the exclusive jurisdiction of the state and federal courts of Illinois with respect to any claim, dispute or declaration arising out of this Non-Compete Agreement.

(c) In the event of a breach or a threatened breach of this Non-Compete Agreement, I acknowledge that the Company will face irreparable injury which may be difficult to calculate in dollar terms and that the Company shall be entitled, in addition to all remedies otherwise available in law or in equity, to temporary restraining orders and preliminary and final injunctions enjoining such breach or threatened breach in any court of competent jurisdiction without the necessity of posting a surety bond, as well as to obtain an equitable accounting of all profits or benefits arising out of any violation of this Non-Compete Agreement.

(d) I agree that if a court determines that any of the provisions in this Non-Compete Agreement is unenforceable or unreasonable in duration, territory, or scope, then that court shall modify those provisions so they are reasonable and enforceable, and enforce those provisions as modified.

(e) If any one or more provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement or its application is determined to be invalid, illegal, or unenforceable to any extent or for any reason by a court of competent jurisdiction, I agree that the remaining provisions (including paragraphs, subparagraphs and terms) of this Non-Compete Agreement will still be valid and the provision declared to be invalid or illegal or unenforceable will be considered to be severed and deleted from the rest of this Non-Compete Agreement. I further agree that if any court of competent jurisdiction finds any of the restrictions set forth in this Non-Compete Agreement to be overly broad and unenforceable, the restriction shall be interpreted to extend only over the

maximum time period, geographic area, or range of activities or clients that such court deems enforceable

(f) Notwithstanding the foregoing provisions of this Non-Compete Agreement, the non-competition provisions of paragraph 2 above shall not restrict Employee from performing legal services as a licensed attorney for a Competing Business to the extent that the attorney licensure requirements in the applicable jurisdiction do not permit Employee to agree to the otherwise applicable restrictions of paragraph 2.

(g) Waiver of any of the provisions of this Non-Compete Agreement by the Company in any particular instance shall not be deemed to be a waiver of any provision in any other instance and/or of the Company's other rights at law or under this Non-Compete Agreement.

(h) I agree that the Company may assign this Non-Compete Agreement to its successors and assigns and that any such successor or assign may stand in the Company's stead for purposes of enforcing this Non-Compete Agreement.

(i) I agree to reimburse the Company for all attorneys' fees, costs, and expenses that it reasonably incurs in connection with enforcing its rights and remedies under this Non-Compete Agreement, but only to the extent the Company is ultimately the prevailing party in the applicable legal proceedings.

(j) I understand and agree that, where allowed by applicable law, the time for my obligations set out in paragraphs 2 - 6 shall be extended for period of non-compliance up to an additional two (2) years following my last day of employment with the Company (for any reason).

(k) I fully understand my obligations in this Non-Compete Agreement, have had full and complete opportunity to discuss and resolve any ambiguities or uncertainties regarding these covenants before signing this Non-Compete Agreement, and have voluntarily agreed to comply with these covenants for their stated terms.

(I) I agree that all non-competition, non-solicitation, non-disclosure and use, non-recruiting, and disclosure obligations in this Non-Compete Agreement shall survive any termination of this Non-Compete Agreement and extend to the proscribed periods following my last day of employment with the Company (for any reason) and no dispute regarding any other provisions of this Non-Compete Agreement or regarding my employment or the termination of my employment shall prevent the operation and enforcement of these obligations.

(m) I understand that nothing in this Non-Compete Agreement, including the non-disclosure and nondisparagement provisions, limit my ability to file a charge or complaint with the Equal Employment Opportunity Commission, Department of Labor, National Labor Relations Board, Occupational Safety and Health Administration, Securities and Exchange Commission or any other federal, state or local governmental agency or commission. I also understand that this Non-Compete Agreement does not limit my ability to communicate with any government agencies or otherwise participate in any investigation or proceeding that may be conducted by any government agency, including providing documents or other information, without notice to the Company. Finally, I understand that nothing in this Non-Compete Agreement is intended to restrict my legally-protected right to discuss wages, hours or other working condition with co-workers, or in any way limit my rights under the National Labor Relations Act or any whistleblower act.

10. <u>Relationship of Parties</u>. I acknowledge that my relationship with the Company is "terminable at will" by either party and that the Company or I can terminate the relationship with or without cause and without following any specific procedures. Nothing contained in this Non-Compete Agreement is intended to or shall be relied upon to alter the "terminable at will" relationship between the parties. I agree that my

obligations in this Non-Compete Agreement shall survive the termination of my employment from the Company for any reason and shall be binding upon my successors, heirs, executors and representatives.

11. <u>Modifications and Other Agreements</u>. I agree that the terms of this Non-Compete Agreement may not be modified except by a written agreement signed by both me and the Company. This Non-Compete Agreement shall not supersede any other restrictive covenants to which I may be subject under an employment contract, benefit program or otherwise, such that the Company may enforce the terms of any and all restrictive covenants to which I am subject. The obligations herein are in addition to and do not limit any obligations arising under applicable statutes and common law.

12. <u>Notification</u>. I agree that in the event I am offered employment at any time in the future with any entity that may be considered a Competing Business Line, I shall immediately notify such Competing Business of the existence and terms of this Non-Compete Agreement. I also understand and agree that the Company may notify anyone attempting to or later employing me of the existence and provisions of this Non-Compete Agreement.

*** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Stock Option Award Agreement to which this Non-Compete Agreement is attached as Exhibit A, and I agree to the terms and conditions expressed in this Non-Compete Agreement.

EXHIBIT B

ADDENDUM TO THE

WALGREENS BOOTS ALLIANCE, INC. 2013 OMNIBUS INCENTIVE PLAN

STOCK OPTION AWARD AGREEMENT

In addition to the terms of the Plan and the Agreement, the Option is subject to the following additional terms and conditions to the extent you reside and/or are employed in one of the countries addressed herein. Pursuant to Section 23 of the Agreement, if you transfer your residence and/or employment to another country reflected in this Addendum, the additional terms and conditions for such country (if any) will apply to you to the extent the Company determines, in its sole discretion, that the application of such terms and conditions is necessary or advisable in order to comply with local laws, rules and/or regulations or to facilitate the operation and administration of the Option and the Plan (or the Company may establish alternative terms as may be necessary or advisable to accommodate your transfer). All defined terms contained in this Addendum shall have the same meaning as set forth in the Plan and the Agreement.

CHILE

Private Placement. The following provision shall replace Section 14 of the Agreement:

The grant of the Option hereunder is not intended to be a public offering of securities in Chile but instead is intended to be a private placement.

- a) The starting date of the offer will be the Grant Date, and this offer conforms to general ruling no. 336 of the Chilean Commission for the Financial Market;
- b) The offer deals with securities not registered in the registry of securities or in the registry of foreign securities of the Chilean Commission for the Financial Market, and therefore such securities are not subject to its oversight;

- c) The issuer is not obligated to provide public information in Chile regarding the foreign securities, since such securities are not registered with the Chilean Commission for the Financial Market; and
- d) The foreign securities shall not be subject to public offering as long as they are not registered with the corresponding registry of securities in Chile.
- a) La fecha de inicio de la oferta será el de la fecha de otorgamiento y esta oferta se acoge a la norma de carácter general n° 336 de la Comisión para el Mercado Financiero en Chile;
- b) La oferta versa sobre valores no inscritos en el registro de valores o en el registro de valores extranjeros que lleva la Comisión para el Mercado Financiero en Chile, por lo que tales valores no están sujetos a la fiscalización de ésta;
- c) Por tratar de valores no inscritos no existe la obligación por parte del emisor de entregar en chile información pública respecto de esos valores; y
- d) Esos valores no podrán ser objeto de oferta pública mientras no sean inscritos en el registro de valores correspondiente.

FRANCE

1. <u>Nature of Grant</u>. The Option is not granted under the French specific regime provided by Articles L.225-177 and seq. of the French commercial code.

2. <u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

GERMANY

No country-specific provisions.

HONG KONG

1. <u>Sale of Shares of Stock</u>. Shares of Stock purchased upon exercise of the Option are accepted as a personal investment. In the event that shares of Stock are issued in respect of the Option within six (6) months after the Grant Date, you agree that the shares of Stock may not be offered to the public or otherwise disposed of prior to the sixmonth anniversary of the Grant Date.

2. <u>IMPORTANT NOTICE</u>. WARNING: The contents of the Agreement the Addendum, the Plan, the Plan prospectus, the Plan administrative rules and all other materials pertaining to the Option and/or the Plan have not been reviewed by any regulatory authority in Hong Kong. You are hereby advised to exercise caution in relation to the offer thereunder. If you have any doubts about any of the contents of the aforesaid materials, you should obtain independent professional advice.

3. <u>Wages</u>. The Option and shares of Stock subject to the Option do not form part of your wages for the purposes of calculating any statutory or contractual payments under Hong Kong law.

IRELAND

No country-specific provisions.

ITALY

1. <u>Document Acknowledgment</u>. In accepting the Option, you acknowledge that a copy of the Plan was made available to you, and you have reviewed the Plan and the Agreement, including this Addendum, in their entirety and fully understand and accept all provisions of the Plan, the Agreement and the Addendum.

You further acknowledge that you have read and specifically approve the following provisions in the Agreement: Section 2: Vesting/Exercise/Expiration (expiration of the right to exercise the Option after the Expiration Date); Section 3: Disability (term for exercising the Option prior to the Vesting Dates in the case of a Termination of Service due to Disability); Section 4: Death (term for exercising the option prior to the Vesting Dates in the case of a Termination of Service due to death); Section 5: Retirement (term for exercising the Option prior to the Vesting Dates in the case of a Termination of Service by reason of Retirement); Section 6: Termination of Service Following a Change in Control (term for exercising the Option in the event of a Termination of Service following a Change in Control); Section 7: Other Termination of Service (term to exercise the vested Option and forfeiture of the unvested Option in other cases of Termination of Service); Section 8: Forfeiture of Outstanding Options Upon Termination for Cause or Following Termination of Service; Section 10(a): Responsibility for Taxes; Tax Withholding (liability for all Tax-Related Items related to the Option and legally applicable to the participant); Section 11: Limited Transferability (Option shall not be sold, transferred, pledged, assigned or otherwise alienated or hypothecated); Section 18: Change in Stock (right of the Company to equitably adjust the Option and the Exercise Price in the event of any change in the Stock); Section 19(j): Nature of the Option (waive any claim or entitlement to compensation or damages arising from forfeiture of the Option resulting from a Termination of Service); Section 19(I): Nature of the Option (the Company is not liable for any foreign exchange rate fluctuation impacting the value of the Option); Section 22: Non-Competition, Non-Solicitation and Confidentiality (the receipt of the Option is conditioned upon agreement of the Non-Competition, Non-Solicitation and Confidentiality Agreement attached hereto as Exhibit A); Section 23: Addendum to Agreement (the Option is subject to the terms of the Addendum); Section 25 Electronic Delivery (Company may deliver documents related to the Option or Plan electronically); Section 26: Governing Law and Jurisdiction (Agreement is governed by Illinois law without regard to any choice of law rules thereof; agreement to exclusive jurisdiction of Illinois courts); and Section 27: English Language (documents will be drawn up in English; if a translation is provided, the English version controls).

MEXICO

1. <u>Commercial Relationship</u>. You expressly recognize that your participation in the Plan and the Company's grant of the Option does not constitute an employment relationship between you and the Company. You have been granted the Option as a consequence of the commercial relationship between the Company and the Affiliate in Mexico that employs you ("WBA Mexico"), and WBA Mexico is your sole employer. Based on the foregoing, you expressly recognize that (a) the Plan and the benefits you may derive from your participation in the Plan do not establish any rights between you and WBA Mexico, (b) the Plan and the benefits you may derive from your participation in the Plan are not part of the employment conditions and/or benefits provided by WBA Mexico, and (c) any modifications or amendments of the Plan by the Company, or a termination of the Plan by the Company, shall not constitute a change or impairment of the terms and conditions of your employment with WBA Mexico.

2. <u>Extraordinary Item of Compensation</u>. You expressly recognize and acknowledge that your participation in the Plan is a result of the discretionary and unilateral decision of the Company, as well as your free and voluntary decision to participate in the Plan in accordance with the terms and conditions of the Plan, the Agreement and this Addendum. As such, you acknowledge and agree that the Company, in its sole discretion, may amend and/or discontinue your participation in the Plan at any time and without any liability. The Option, the shares of Stock subject to the Option

and the value of the same are an extraordinary item of compensation outside the scope of your employment contract, if any, and is not part of your regular or expected compensation for purposes of calculating any severance, resignation, redundancy, end of service payments, bonuses, long-service awards, pension or retirement benefits, or any similar payments, which are the exclusive obligations of WBA Mexico.

MONACO

<u>Use of English Language</u>. You acknowledge that it is your express wish that the Agreement, as well as all documents, notices and legal proceedings entered into, given or instituted pursuant hereto or relating directly or indirectly hereto, be drawn up in English. *Vous reconnaissez avoir expressément exigé la rédaction en anglais de la présente Convention, ainsi que de tous documents exécutés, avis donnés et procédures judiciaires intentées, directement ou indirectement, relatifs à, ou suite à, la présente Convention.*

NETHERLANDS

Exclusion of Claim. You acknowledge and agree that you will have no entitlement to compensation or damages insofar as such entitlement arises or may arise from your ceasing to have rights under or to be entitled to the Option, whether or not as a result of your Termination of Service (whether such termination is in breach of contract or otherwise), or from the loss or diminution in value of the Option. Upon the grant of the Option, you shall be deemed irrevocably to have waived any such entitlement.

NORWAY

No country-specific provisions.

ROMANIA

<u>Voluntary Termination of Service</u>. For the sake of clarity, a voluntary Termination of Service shall include the situation where your employment contract is terminated by operation of law on the date you reach the standard retirement age and have completed the minimum contribution record for receipt of state retirement pension or the relevant authorities award you an early-retirement pension of any type.

SPAIN

1. <u>Acknowledgement of Discretionary Nature of the Plan; No Vested Rights</u>. This provision supplements the terms of the Agreement:

In accepting the Option, you acknowledge that you consent to participation in the Plan and have received a copy of the Plan.

You understand that the Company has unilaterally, gratuitously and in its sole discretion granted an Option under the Plan to individuals who may be employees of the Company or its Affiliates throughout the world. The decision is a limited decision that is entered into upon the express assumption and condition that any grant will not economically or otherwise bind the Company or any of its Affiliates on an ongoing basis. Consequently, you understand that the Option is granted on the assumption and condition that the Option and the shares of Stock acquired upon exercise of the Option shall not become a part of any employment contract (either with the Company or any of its Affiliates) and shall not be considered a mandatory benefit, salary for any purposes (including severance compensation) or any other right whatsoever. In addition, you understand that this grant would not be made to you but for the assumptions and conditions referenced above; thus, you acknowledge and freely accept that should any or all of the assumptions be mistaken or should any of the conditions not be met for any reason, the Option shall be null and void.

Further, you understand and agree that the vesting of the Option is expressly conditioned on your continued and active rendering of service, such that upon a Termination of Service, the Option may cease vesting immediately, in whole or in part, effective on the date of your Termination of Service (unless otherwise specifically provided in Section 3, 4, 5 or 6 of the Agreement). This will be the case, for example, even if (a) you are considered to be unfairly dismissed without good cause; (b) you are dismissed for disciplinary or objective reasons or due to a collective dismissal; (c) you terminate service due to a change of work location, duties or any other employment or contractual condition, (d) you terminate service due to a unilateral breach of contract by the Company or an Affiliate. Consequently, upon a Termination of Service for any of the above reasons, you may automatically lose any rights to the Option that were not vested on the date of your Termination of Service, as described in the Plan and Agreement. In addition, you understand and agree that the post-Termination of Service exercise period specified in the Agreement shall run from the date of your Termination of Service, as determined by the Committee, in its sole discretion.

You acknowledge that you have read and specifically accept the conditions referred to in the Agreement regarding the impact of a Termination of Service on the Option.

2. <u>Termination for Cause</u>. "Cause" shall be defined as indicated in Section 8 of the Agreement, irrespective of whether the termination is or is not considered a fair termination (i.e., "despido procedente") under Spanish legislation.

3. <u>No Public Offering</u>. No "offer of securities to the public," within the meaning of Spanish law, has taken place or will take place in the Spanish territory in connection with the Option. The Plan, the Agreement (including this Addendum) and any other documents evidencing the grant of the Option have not, nor will they be registered with the *Comisión Nacional del Mercado de Valores* (the Spanish securities regulator) and none of those documents constitute a public offering prospectus.

SWITZERLAND

<u>Securities Law Notification</u>. The Option is not considered a public offering in Switzerland; therefore, it is not subject to registration in Switzerland. Neither this document nor any other materials relating to the Option constitute a prospectus as such term is understood pursuant to article 652a of the Swiss Code of Obligations, and neither this document nor any other materials relating to the Option may be publicly distributed nor otherwise made publicly available in Switzerland. Neither this document nor any other offering or marketing materials relating to the Option have been or will be filed with, or approved or supervised by, any Swiss regulatory authority (in particular, the Swiss Financial Market Supervisory Authority (FINMA)).

TURKEY

<u>Securities Law Notification</u>. The sale of shares of Stock acquired under the Plan is not permitted within Turkey. The sale of shares of Stock acquired under the Plan must occur outside of Turkey. The shares of Stock are currently traded on the Nasdaq Stock Market under the ticker symbol "WBA" and shares of Stock may be sold on this exchange.

*** *** *** *** ***

By clicking the acceptance box for this grant agreement, I acknowledge receipt of the Stock Option Award Agreement to which this Addendum is attached as Exhibit B, and I agree to the terms and conditions expressed in this Addendum.

WALGREENS BOOTS ALLIANCE, INC. EXECUTIVE RETIREMENT SAVINGS PLAN

(As Amended and Restated Effective January 1, 2020)

			Page
ARTICLE I		INTRODUCTION	1
	1.1	Name and Purpose	1
	1.2	Effective Date and Plan Year	1
ARTICLE II		DEFINITIONS	1
ARTICLE III		ELIGIBILITY AND PARTICIPATION	3
	3.1	Eligibility	3
	3.2	Status under ERISA	3
ARTICLE IV		DEFERRAL OF COMPENSATION	3
	4.1	Participant Deferral Credits	3
	4.2	Deferral Elections	3
	4.3	Vesting	4
ARTICLE V		EMPLOYER CONTRIBUTION CREDITS	4
	5.1	Amount of Employer Contribution Credits	4
	5.2	Vesting	4
ARTICLE VI		PLAN ACCOUNTING	4
	6.1	Accounts	4
	6.2	Adjustments to Participant Accounts	4
ARTICLE VII		PAYMENT OF BENEFITS	5
	7.1	Payment of Account Balances	5
	7.2	Time and Manner of Payment of Grandfathered Account Balances	5
	7.3	Time and Manner of Payment of Non-Grandfathered Account Balances	6
	7.4	Effect on Other Benefit Plans	7
	7.5	Beneficiaries and Facility of Payment	7
	7.6	Effect of Payment	8
	7.7	Withholding for Taxes	8
ARTICLE VIII		ADMINISTRATION	8
	8.1	Administration	8
	8.2	Claims Procedures	8
ARTICLE IX		AMENDMENT AND TERMINATIONS	8
	9.1	Amendment of Termination	8
ARTICLE X		MISCELLANEOUS	8
	10.1	Non-Alienation	8
	10.2	Employment Rights	9
	10.3	Trust Fund	9

		Page
10.4	Successors	9
10.5	Controlling Law	9
10.6	Severability	9
10.7	Section 409A	9

WALGREENS BOOTS ALLIANCE, INC. EXECUTIVE RETIREMENT SAVINGS PLAN

(As Amended and Restated Effective January 1, 2020)

ARTICLE I INTRODUCTION

1.1. Name and Purpose. Walgreen Co. ("Walgreens") established the Walgreen Co. Executive Deferred Profit Sharing Plan (the "Plan"), effective January 1, 1990, for the benefit of eligible employees. The Plan was amended and restated in its entirety effective January 1, 2003, was further amended effective as of January 1, 2005 and January 1, 2008, was again amended and restated in its entirety effective as of January 1, 2012, and was further amended effective as of January 1, 2013 and to be effective as of January 1, 2015. On December 31, 2014, a reorganization of Walgreens into a holding company structure (the "Reorganization") was completed. Pursuant to the Reorganization, Walgreens became a wholly owned subsidiary of a new Delaware corporation named Walgreens Boots Alliance, Inc. (the "Company"). In connection with the Reorganization, the Plan was assumed by the Company, and the Plan was amended and restated in order to reflect such assumption, effective as of December 31, 2014. The Plan was further amended and restated, effective as of February 1, 2019, and is further amended and restated, as the Walgreens Boots Alliance, Inc. Executive Retirement Savings Plan, as set forth herein. The purpose of the Plan, as amended and restated herein, is to provide eligible employees with the opportunity to defer compensation on a pre-tax basis, and to receive Employer Contribution Credits.

1.2. <u>Effective Date and Plan Year</u>. The Effective Date of the amended and restated Plan is January 1, 2020. The Plan shall be administered on a calendar year basis (the "Plan Year").

ARTICLE II DEFINITIONS

2.1. "<u>Account</u>" means the recordkeeping account established by the Administrators pursuant to Article VI to record a Participant's accrued benefit under the Plan.

2.2. "<u>Administrators</u>" means the persons appointed to administer the Plan pursuant to Section 8.1.

2.3. "Beneficiary" has the meaning set forth in Section 7.5.

2.4. "<u>Code</u>" means the Internal Revenue Code of 1986, as amended, and the regulations issued thereunder.

2.5. "**Deferral Credits**" means the portions of a Participant's base salary and/or bonus, if any, that he or she elects to defer under Article IV.

2.6. "**Deferral Election**" means an election by a Participant to defer base salary and/or bonuses in accordance with the provisions of Article IV.

2.7. "<u>Employee Deferral Subaccount</u>" means the portion of a Participant's Account consisting of Deferral Credits and Investment Adjustments thereon.

2.8. "<u>Employer Contribution Credits</u>" means the amounts credited to a Participant's Employer Subaccount pursuant to Article V.

2.9. "<u>Employer Subaccount</u>" means the portion of a Participant's Account consisting of Employer Contribution Credits and Investment Adjustments thereon.

2.10. "**ERISA**" means the Employee Retirement Income Security Act of 1974, as amended, and the regulations issued thereunder.

2.11. "<u>Grandfathered Account</u>" means the portion of a Participant's Account that was credited to the Account and became fully vested prior to January 1, 2005.

2.12. "**Investment Adjustments**" means the amount of earnings, gains, losses and expenses debited or credited to a Participant's Account pursuant to Section 6.2.

2.13. "<u>Non-Grandfathered Account</u>" means the portion of a Participant's Account that is not a Grandfathered Account.

2.14. "**Participant**" means an individual who meets the eligibility requirements set forth in Section 3.1 and is participating in one or both features of the Plan set forth in Articles IV and V.

2.15. "Plan Year" means the calendar year.

2.16. "**Retirement Savings Plan**" means, as applicable to each Participant, either the Walgreens Retirement Savings Plan or the Walgreens Puerto Rico Retirement Savings Plan, in both cases as amended and restated effective January 1, 2018, and as further amended from time to time.

2.17. "<u>Retirement Savings Plan Compensation Limitation</u>" means the limitation imposed by Section 401(a)(17) of the Code on the amount of a Participant's compensation that may be taken into account under the Retirement Savings Plan.

ARTICLE III ELIGIBILITY AND PARTICIPATION

3.1. <u>Eligibility</u>. Each employee of the Company or one of its subsidiaries who is eligible to participate in the Retirement Savings Plan shall be eligible to become a Participant in the Plan, based on such rules and procedures as may be established by the Administrator and its delegates from time to time, as follows:

- (a) if the employee's annualized base salary and target annual bonus at the time of determining eligibility (without reduction for any deferrals under this Plan) exceeds the Retirement Savings Plan Compensation Limitation, then such employee shall be eligible to defer his or her compensation in accordance with Article IV below; and
- (b) if the employee's "Salary Conversion Contributions" under Section 4.1 of the Retirement Savings Plan are limited by the Retirement Savings Plan Compensation Limitation – or would be so limited if participating in the Retirement Savings Plan – then such employee shall be eligible to receive Employer Contribution Credits in accordance with Article V below.

Notwithstanding the foregoing, an employee shall be eligible to participate in the Plan only if the Company provides such employee with written notice of his or her eligibility.

3.2. <u>Status under ERISA</u>. The Plan is an unfunded plan maintained primarily to provide deferred compensation benefits for a select group of "management or highly-compensated employees" within the meaning of Sections 201, 301, and 401 of ERISA, and therefore is exempt from the provisions of Parts 2, 3 and 4 of Title I of ERISA. The Board of Directors of the Company or its delegates may terminate the Plan or remove certain employees as Participants if it is determined by the United States Department of Labor, a court of competent jurisdiction, or an opinion of counsel, that the Plan constitutes an employee pension benefit plan within the meaning of Section 3(2) of ERISA which is not so exempt.

ARTICLE IV DEFERRAL OF COMPENSATION

4.1. <u>Participant Deferral Credits</u>. A Participant may elect to defer under the Plan by filing a Deferral Election in accordance with Section 4.2. The amount of such deferral shall not be in excess of 50 percent of the Participant's base salary (as in effect at the time of deferral) and 85 percent of the Participant's annual bonus. Deductions will be made pursuant to such Deferral Election during the Plan Year. The deferral applicable to base pay shall be reduced in substantially equal amounts from the base salary otherwise periodically payable to the Participant over the Plan Year (or over a portion of such Plan Year as deemed administratively practicable). All Deferral Credits shall be credited to the Participant's Employee Deferral Subaccount.

4.2. **Deferral Elections**. A Participant's Deferral Election shall be in writing (or by electronic means established by the Administrators), and must be made during the election period established by the Administrators which period shall end no later than the day preceding the first day of the Plan Year in which the base salary or bonus is earned. An employee who first becomes eligible to participate in this feature of the Plan on or after the first day of a Plan Year may make a Deferral Election within 30 days after he or she first becomes eligible, if and to the extent permitted by Code Section 409A; provided, however, that such Deferral Election shall only apply to base salary and bonuses earned after the Participant's Deferral Election is received by the Administrators. All Deferral Elections shall become irrevocable as of the end of the deferral election period.

4.3. <u>Vesting</u>. A Participant shall at all times be 100 percent vested in amounts credited to his or her Employee Deferral Subaccount.

ARTICLE V EMPLOYER CONTRIBUTION CREDITS

5.1. <u>Amount of Employer Contribution Credits</u>. For each Plan Year, each Participant's Employer Subaccount shall be credited with the sum of the following amounts:

(a) the additional amount that would have been allocated to the Participant's "Employer Contribution Account"

under the Retirement Savings Plan (including a share of any applicable forfeitures) if the Retirement Savings Plan Compensation Limitation did not apply; and

(b) such additional amount as determined each year by resolution of the Company's Board of Directors (or its delegate) to each Participant who qualifies for such additional credit based on the criteria set forth in such resolution.

For purposes of the foregoing calculations and credits, it shall be assumed that the Participant contributes the maximum annual amount permissible under the Retirement Savings Plan, regardless of the Participant's actual contribution level, if any.

5.2. <u>Vesting</u>. A Participant shall be vested in his or her Employer Subaccount to the same extent as the Participant is vested in his or her Employer Contribution Account under the Retirement Savings Plan.

ARTICLE VI PLAN ACCOUNTING

6.1. <u>Accounts</u>. The Company shall establish and maintain, or cause to be established and maintained, a recordkeeping account (the "Account") in the name of each Participant, which Account shall be comprised of the Employee Deferral Subaccount and the Employer Subaccount. A Participant's Account shall at all times be reflected on the Company's books as a general unsecured and unfunded obligation of the Company, and the Plan shall not give any person any right or security interest in any asset of the Company or any subsidiary of the Company nor shall it imply any trust or segregation of assets by the Company or any subsidiary of the Company.

6.2. <u>Adjustments to Participant Accounts</u>. In accordance with, and subject to, the rules and procedures that are established from time to time by the Administrators, in their sole discretion, amounts shall be credited or debited to a Participant's Account in accordance with the following rules:

- (a) <u>Investment Adjustments</u>. On and after January 1, 2020 (the "Investment Election Effective Date"), based on such procedures and forms developed by the Administrators, a Participant may elect one or more Investment Option(s), as defined below, to be used to determine the Investment Adjustments to be applied to his or her Account or designated portions thereof. In making any election described in this subsection, the Participant shall specify, subject to such minimum percentage increments as may be established by the Administrators, the percentage of his or her Account to be allocated to an Investment Option (as if the Participant was making an investment in that Investment Option with that portion of his or her Account).
- (b) <u>Default Investment Adjustments</u>. If no such elections are made by a Participant pursuant to part (a) above, the Investment Adjustments debited from or credited to a Participant's Account (i) shall continue to be based on the Investment Option(s) that apply to the Participant's Account as of the Investment Election Effective Date; or (ii) for a Participant who joins the Plan after the Investment Election Effective Date, shall be based on the Investment Option that is the default investment fund under the Retirement Savings Plan. Such adjustments shall be determined by the Administrators, and their determination shall be final and conclusive.
- (c) <u>Investment Options.</u> For purposes of this Section 6.2, "Investment Options" shall mean the investment funds offered under the Retirement Savings Plan from time to time, except for any such funds investing primarily in Company stock.
- (d) <u>Timing of Adjustments and Credits</u>. All credits and adjustments to a Participant's Account shall be made within reasonable proximity to the dates such credits and adjustments would have been made if they were effected under the Retirement Savings Plan.
- (e) <u>No Actual Investment</u>. Notwithstanding any other provision of this Plan that may be interpreted to the contrary, the Investment Options are to be used for measurement purposes only, and a Participant's election of any Investment Option(s), the allocation to his or her Account thereto, the calculation of additional amounts and the crediting or debiting of such amounts to a Participant's Account <u>shall not</u> be considered or construed in any manner as an actual investment of his or her Account in any such Investment Option. In the event that the Company, in its own discretion, decides to invest funds in any or all of the Investment Options, no Participant shall have any rights in or to such investment themselves. Without limiting the foregoing, a Participant's Account shall at all times be a bookkeeping entry only and shall not represent any investment made on his or her behalf by the Company.

ARTICLE VII PAYMENT OF BENEFITS

7.1. **Payment of Account Balances.** A Participant who incurs a separation from service with the Company, dies or becomes disabled (collectively, a "Distribution Event") shall be entitled to payment of the vested portion of his or her Account at the time and in the manner provided in Section 7.2 and/or Section 7.3 below. For purposes of this Article VII, the terms 'separation

from service' and 'disabled' shall have the meanings set forth in Code Section 409A. If a Participant is not 100 percent vested in his or her Employer Subaccount at the time of the Distribution Event, the non-vested portion shall be forfeited.

7.2. <u>**Time and Manner of Payment of Grandfathered Account Balances.</u>** The Administrators, in their sole discretion, shall determine the time and manner in which a Participant's Grandfathered Account balance shall be distributed, which may include the methods permitted under the Retirement Savings Plan.</u>

7.3. Time and Manner of Payment of Non-Grandfathered Account Balances.

- (a) Following a Participant's separation from service or disability, distribution of the portion of the Participant's Non-Grandfathered Account credited to his or her Employer Subaccount shall be made as follows:
 - (i) If the portion of the Participant's Non-Grandfathered Account balance credited to his or her Employer Subaccount (without taking into account any Company matching contributions credited following separation from service, death or disability) is less than the Lump-Sum Threshold (as defined below), then such portion of the Participant's Non-Grandfathered Account shall be paid to the Participant (or in the event of the Participant's death, to his or her Beneficiary) in one lump sum as of the date of the Distribution Event.
 - (ii) If the portion of the Participant's Non-Grandfathered Account balance credited to his or her Employer Subaccount (without taking into account any Company matching contributions credited following the Distribution Event) is equal to or greater than the Lump-Sum Threshold (as defined below), then such portion of the Participant's Non-Grandfathered Account shall be paid to the Participant (or in the event of the Participant's death, to his or her Beneficiary) in monthly installments beginning on the date of the Distribution Event. Each monthly installment shall equal the Lump-Sum Threshold divided by 12.
 - (iii) For purposes of this Section 7.3(a), "Lump-Sum Threshold" shall, for Plan Year 2019, equal \$76,000 if the Participant is age 55 or older or shall equal \$152,000 if the Participant is under age 55; provided that, beginning in 2020, these \$76,000 and \$152,000 thresholds shall be increased each year by \$2,000 and \$4,000, respectively. For purposes of this Section 7.3(a), the determination of which Lump-Sum Threshold amount shall apply and the comparison of this threshold to the Participant's Account balance shall be made as of the date of the Distribution Event.
- (b) Following a Participant's separation from service or disability, distribution of the portion of the Participant's Non-Grandfathered Account that is credited to his or her Employee Deferral Subaccount shall be made in accordance with one of the following options, as elected by the Participant:
 - (i) A single lump-sum payment, to be made as of the Participant's separation from service or disability; or
 - (ii) Annual installments over a period of 5, 10, 15 or 20 years, as elected by the Participant, with the first installment being calculated within 30 days following the Plan Year in which such separation from service or disability occurs, and such first installment payment made at the time of such calculation, except as provided below with respect to subsequent changes.

A Participant may make a distribution election under this Subsection 7.3(b) at the time the Participant makes his or her Deferral Election under Section 4.2; provided, however, that if no such election is made, the Participant shall be deemed to have elected payment under paragraph (b)(i) above. Once made, the Participant may subsequently change his or her election to an allowable alternative payout period by submitting a new election form to the Administrators, provided that any such new election form (i) shall be made at least 12 months in advance of the originally-scheduled distribution date and may not take effect for at least 12 months after the date the new election is made, (ii) shall not accelerate the time or schedule of any payment, except as permitted under Treasury regulations, and (iii) except with respect to distributions on account of death or disability, shall provide for each payment made pursuant to such distribution to be paid on a date that is not less than 5 years from the date payment would have otherwise been made.

(c) Notwithstanding any provision of the Plan to the contrary, if a Participant is deemed to be a "specified employee" within the meaning of that term under Code Section 409A(a)(2)(B), then with regard to any payment under the Plan on account of the participant's separation from service (for reasons other than death or disability), then, to the extent required by Code Section 409A, such payment or benefit shall not be made or provided until the date which is the earlier of (i) the day after the expiration of the six (6)-month period measured from the date of such separation from service, and (ii) the date of the Participant's death. Upon the expiration of the six-month delay period, all payments delayed pursuant to this provision shall be paid to the

Participant in a lump sum without interest, and all remaining payments due under this Plan shall be paid or provided in accordance with the normal payment dates specified for them.

- (d) If a Participant dies prior to the time that his or her entire Account has been distributed, such Account, or remaining Account balance, shall be distributed to the Participant's Beneficiary in a lump sum as of the Participant's death.
- (e) Distributions pursuant to this Section 7.3 shall be paid as soon as practicable after the distribution date or event specified in this Section 7.3, but in no event later than the later of (i) the last day of the calendar year in which such date or event occurs or (ii) the date that is 2 1/2 months following such date or event, and in all cases in accordance with Section 409A of the Code.

7.4. <u>Effect on Other Benefit Plans</u>. Amounts credited to or paid from the Employer Subaccount of the Plan shall not be considered to be compensation for the purposes of any qualified plan maintained by the Company or any subsidiary of the Company. The treatment of such amounts under other employee benefit plans or programs shall be determined pursuant to the provisions of such plans or programs.

7.5. **Beneficiaries and Facility of Payment**. If the Participant or his or her Beneficiary is entitled to payments under the Plan and in the opinion of the Administrators such person becomes in any way incapacitated so as to be unable to manage his or her financial affairs, the Company may make payments to the Participant's or Beneficiary's legal representative or similar person, to a custodian under the Uniform Gifts or Transfers to Minors Act of any state, or in such other manner for the benefit of the Participant or Beneficiary that the Trustees consider advisable. Any payments made in accordance with the preceding sentence shall be a full and complete discharge of any liability for such payment hereunder. Upon a Participant's death, payment under the Plan shall be made to the Participant's Beneficiary, which shall be either (i) the beneficiary or beneficiary designated under this Plan to the extent the Administrator allows for such designations and the Participant completes the applicable forms to make such designation or, otherwise, (ii) the beneficiary or beneficiaries that apply to such Participant under the Retirement Savings Plan, including as applied in the absence of any beneficiary designation under the Retirement Savings Plan. Any payments made in accordance with the preceding sentences shall be a full and complete discharge of any liability for such payment the Retirement Savings Plan. Any payments made in accordance with the preceding sentences shall be a full and complete discharge of any liability for such payments hereunder.

7.6. **Effect of Payment.** The full payment of a Participant's benefit under the Plan shall completely discharge all obligations on the part of the Company to the Participant (and the Participant's Beneficiary) with respect to the operation of the Plan, and the Participant's (and Participant's Beneficiary's) rights under the Plan shall terminate.

7.7. <u>Withholding for Taxes</u>. The Company may withhold from any payment made by it under the Plan such amount or amounts as may be required for purposes of complying with the tax withholding or other provisions of the Code or the Social Security Act or any state's income tax act or for purposes of paying any estate, inheritance or other tax attributable to any amounts payable hereunder.

ARTICLE VIII ADMINISTRATION

8.1. <u>Administration</u>. The Plan shall be administered by the committee responsible for the administration of the Retirement Savings Plan (currently the Walgreens Retirement Plans Committee), who will have, to the extent appropriate, the same powers, rights, duties, obligations and indemnity with respect to the Plan as they do with respect to the Retirement Savings Plan. Consistent with such rights under the Retirement Savings Plan, each determination provided for under the Plan shall be made by the Administrators under such procedures as may from time to time be prescribed by them, and shall be made in their absolute discretion. Any such determination shall be conclusive on all persons.

8.2. <u>Claims Procedures</u>. If a Participant or his or her beneficiary is denied all or a portion of an expected benefit under the Plan for any reason, he or she may file a claim (and thereafter appeal any denied claim) in accordance with the procedures set forth in the Summary Plan Description for the Retirement Savings Plan, which are incorporated by reference herein for this purpose.

ARTICLE IX AMENDMENT AND TERMINATIONS

9.1. <u>Amendment or Termination</u>. The Company may, in its sole discretion, terminate or amend the Plan at any time. No such termination or amendment shall change the then existing credits to or adjustments of a Participant's Account or alter his or her right to receive a distribution thereof in accordance with Article VII; provided, however, that if the Company is liquidated, it shall have the exclusive right to determine the value of each Participant's Account, as of a date established by the Administrators and to pay any unpaid distributions in any manner which such Administrators determine to be just and equitable.

ARTICLE X MISCELLANEOUS

10.1. **Non-Alienation.** All rights and benefits under the Plan are personal to the Participant, and neither the Plan nor any right or interest of a Participant or any person arising under the Plan is subject to voluntary or involuntary alienation, sale, transfer,

or assignment without the Company's consent.

10.2. <u>Employment Rights</u>. The Plan is not a contract of employment, and participation in the Plan will not give any Participant the right to be retained in the employ of the Company or any subsidiary of the Company, nor any right or claim to any benefit under the Plan, unless the right or claim has specifically accrued under the Plan.

10.3. <u>Trust Fund</u>. The Company shall be responsible for the payment of all benefits provided under the Plan. At its discretion, the Company may establish one or more trusts, with such trustees as the Administrators may approve, for the purpose of assisting in the payment of such benefits. Although such a trust shall be irrevocable, its assets shall be held for payment of all of the Company's general creditors in the event of insolvency. To the extent any benefits provided under the Plan are paid from any such trust, the Company shall have no further obligation to pay them. If not paid from the trust, such benefits shall remain the obligation of the Company.

10.4. **Successors.** Unless otherwise agreed to, the Plan is binding on and will inure to the benefit of any successor to the Company, whether by way of merger, consolidation, purchase or otherwise.

10.5. <u>**Controlling Law.</u>** To the extent not preempted by federal law, including ERISA, the Plan shall be construed in accordance with the laws of the State of Illinois.</u>

10.6. <u>Severability</u>. If any provision of the Plan shall be found to be invalid or unenforceable by a court of competent jurisdiction, the validity or enforceability of the remaining provisions of the Plan shall remain in full force and effect.

10.7. <u>Section 409A</u>. To the extent that any portion of the Plan is subject to the rules under Code Section 409A, such portion of the Plan is not intended to result in acceleration of income recognition or imposition of penalty taxes by reason of Code Section 409A, and the terms of such portion of the Plan shall be interpreted in a manner (and such portion of the Plan may be amended to the extent determined necessary or appropriate by the Company) to avoid such acceleration and penalties. The Company may modify the time at which any Account balance will be vested or distributed if it determines that such modification may be necessary to avoid acceleration of tax or imposition of penalties under Code Section 409A. For purposes of Code Section 409A, a right to receive installment payments pursuant to the Plan shall be treated as a right to receive a series of separate and distinct payments. Regardless of whether the Company modifies or fails to modify the time at which any Account balance is vested or distributed, each Participant shall be solely liable for any taxes, penalties and interest (including without limitation those imposed under Code Section 409A) incurred with respect to all amounts credited to the Participants under the Plan.

1

WALGREENS BOOTS ALLIANCE, INC. EXECUTIVE SEVERANCE AND CHANGE IN CONTROL PLAN

AMENDED AND RESTATED EFFECTIVE AUGUST 6, 2019

			Page
ARTICLE I		Statement of Purpose and Effective Date	1
	1.01	Background	1
	1.02	Purpose	1
	1.03	Effective Date	1
ARTICLE II		Definitions	
	2.01	"Accrued Annual Incentive"	1
	2.02	"Accrued Based Salary"	1
	2.03	"Accrued Obligations"	1
	2.04	"Affiliate"	1
	2.05	"Base Salary"	1
	2.06	"Board"	1
	2.07	"Cause"	2
	2.08	"CEO"	2
	2.09	"Change Date"	2
	2.10	"Change in Control"	2
	2.11	"Code"	3
	2.12	"Committee"	3
	2.13	"Company"	3
	2.14	"Disability"	3
	2.15	"Effective Date"	3
	2.16	"Employee"	3
	2.17	"Employer"	3
	2.18	"ERISA"	3
	2.19	"Exchange Act"	3
	2.20	"Good Reason"	3
	2.21	"Including"	4
	2.22	"Involuntary Termination"	4
	2.23	"Notice of Termination"	4
	2.24	"Participant"	4
	2.25	"Plan"	4
	2.26	"Plans"	4
	2.27	"Policies"	4
	2.28	"Post Change Period"	4
	2.29	"Pro-rata Annual Incentive"	4
	2.30	"Intentionally Omitted"	5
	2.31	"Section 409A Deferred Compensation"	5
	2.32	"Severance Multiple" and "Severance Period"	5
	2.33	"Target Annual Incentive"	5
	2.34	"Termination Date"	5
	2.35	"Termination of Employment"	6
	2.36	"Tier I Participant"	6
	2.37	"Tier II Participant"	6
ARTICLE III		Participation and Eligibility for Benefits	6
	3.01	Eligibility	6

			Page
	3.02	Participation	6
	3.03	Eligibility for Benefits	6
	3.04	Termination of Participation	6
ARTICLE IV		General Obligations of the Employer Upon Involuntary Termination	7
	4.01	Involuntary Termination	7
	4.02	Termination for Any Other Reason	8
ARTICLE V		Obligations of the Employer on Involuntary Termination of Certain Participants in the Post-Change Period	8
	5.01	Application	8
	5.02	Involuntary Termination in the Post-Change Period	8
	5.03	Termination on or After the Change Date for Any Other Reason	9
	5.04	Limitation on Benefits	9
ARTICLE VI		Administration	9
	6.01	The Company and Committee	9
	6.02	Delegation of Committee Authority	10
	6.03	Advisors and Agents of the Committee	10
	6.04	Records and Reports of the Committee	11
	6.05	Limitation of Liability; Indemnification	11
	6.06	Plan Expenses	11
	6.07	Service in More than One Capatcity	11
ARTICLE VII		Amendments; Termination	11
	7.01	Amendment or Termination of the Plan	11
ARTICLE VIII		Claims Procedures	11
	8.01	Filing a Claim	11
	8.02	Review of Claim Denial	12
ARTCLE IX		Release; No Mitigation; No Duplication of Benefits	12
	9.01	Release Required	12
	9.02	No Mitigation	12
	9.03	No Duplication of Benefits/Rehire	12
ARTICLE X	10.01	Miscellaneous	13
	10.01	Participant Information	13
	10.02	Electronic Media	13
	10.03	Notices	13
	10.04 10.05	No Employment Contract	13 14
	10.05	Headings Construction	14 14
	10.00	Joint and Several Liability	14
	10.07	Successors	14
	10.08	Payments to Beneficiary	14
	10.09	r dymenia to Denenolary	14

		Page
10.10	Non-Alienation of Benefits	14
10.11	Tax Matters	14
10.12	Clawback Policy	15
10.13	Governing Law	15
10.14	Severability	15

ARTICLE I

Statement of Purpose and Effective Date

1.01 **Background**. Walgreen Co. ("Walgreens"), an Illinois corporation, previously established the Walgreen Co. Executive Severance and Change in Control Plan (the "Plan"). On December 31, 2014, a reorganization of Walgreens into a holding company structure (the "Reorganization") was completed. Pursuant to the Reorganization, Walgreens became a wholly owned subsidiary of a new Delaware corporation named Walgreens Boots Alliance, Inc. (the "Company"). In connection with the Reorganization, the Plan was assumed by the Company and the Plan was amended and restated effective as of December 31, 2014, in order to reflect such assumption. The Plan, as set forth herein, is hereby amended and restated, effective as of August 6, 2019.

1.02 **Purpose**. The Plan is intended to encourage and motivate key employees to devote their full attention to the performance of their assigned duties without the distraction or concerns regarding their involuntary termination of employment. The Company believes that it is in the best interests of key employees of the Employers and the shareholders of the Company to provide financial assistance through severance payments and other benefits to eligible key employees who are involuntarily terminated. To the extent the Plan provides deferred compensation it is an unfunded plan primarily for the purposes of providing deferred compensation for a select group of management or highly compensated employees.

1.03 Effective Date. This Plan became effective as of January 1, 2013 (the "Effective Date").

ARTICLE II Definitions

When used in this Plan, the terms specified below have the following meanings:

2.01 "Accrued Annual Incentive" means the amount of any annual incentive earned in a year ended before the Termination Date, but not yet paid to a Participant as of the Termination Date, other than amounts that he or she has elected to defer or that have been automatically deferred.

2.02 "Accrued Base Salary" means the amount of a Participant's Base Salary that is accrued but unpaid as of the Termination Date, other than amounts that he or she has elected to defer.

2.03 "Accrued Obligations" means, as of any date, the sum of a Participant's Accrued Base Salary, Accrued Annual Incentive, any accrued but unpaid PTO, vacation pay, unreimbursed expenses for which proper documentation is provided, and any other vested amounts and benefits that are to be paid or provided to the Participant by the Company under the Company's plans (other than this Plan and other than any Section 409A Deferred Compensation), but which have not yet been paid or provided (as applicable).

2.04 "Affiliate" means any person with whom the Company would be considered a single employer under Sections 414(b) and 414(c) of the Code and Treas. Reg. §1.409A-3(i)(5)(ii), except that in applying Sections 1563(a)(1), (2), and (3) of the Code for purposes of determining a controlled group of corporations under Section 414(b) of the Code; the language "at least 50 percent" shall be used instead of "at least 80 percent" in each place it appears in Sections 1563(a)(1), (2), and (3) of the Code, and in applying Treas. Reg. § 1.414(c)-(2) for purposes of determining a controlled group of trades or businesses under Section 414(c) of the Code, the language "at least 50 percent" shall be used instead of "at least 80 percent" in each place it appears in Treas. Reg. § 1.414(c)-(2). Notwithstanding the foregoing, where justified by legitimate business criteria as determined by the Committee in its sole discretion, "at least 20 percent" shall be substituted for "at least 50 percent" in the preceding sentence in determining whether a Participant has a Termination of Employment.

2.05 "Base Salary" means an Employee's annual rate of salary as of any date.

2.06 "**Board**" means the Board of Directors of the Company or, from and after a Change in Control that gives rise to a surviving corporation to the Company, the Board of Directors of such surviving corporation.

2.07 "Cause" with respect to a Participant shall mean any one or more of the following, as determined by the Committee or its delegate in its sole discretion:

(a) a Participant's commission of a felony or any crime of moral turpitude;

(b) a Participant's dishonesty or material violation of standards of integrity in the course of fulfilling his or her employment duties to his or her Employer, the Company or any Affiliates;

(c) a material violation of a material written policy of the Company or his or her Employer or any Affiliate, the violation of which would be grounds for immediate termination under the applicable policy;

(d) willful and deliberate failure on the part of the Participant to perform his or her employment duties to his or her Employer, the Company or any Affiliates in any material respect, after reasonable notice of such failure and an opportunity to correct it; or

(e) failure to comply in any material respect with any applicable legal or regulatory requirements, including, but not be limited to, the Foreign Corrupt Practices Act, the securities Act of 1933, the Securities Exchange Act of 1934, the Sarbanes-Oxley Act of 2002, the Dodd-Frank Wall Street Reform and Consumer Protection Act of 2010, and the Truth in Negotiations Act, or any rules or regulations thereunder.

2.08 "CEO" means the Chief Executive Officer of the Company.

2.09 **"Change Date**" means the first date on which a Change in Control occurs before the termination of the Plan.

2.10 "**Change in Control**" means an event that meets the conditions for a "change in the ownership of a corporation" (within the meaning of Section 409A of the Code and Treas. Reg. \$1.409A-3(i)(5)(v)), a "change in the effective control of a corporation" (within the meaning of Section 409A of the Code and Treas. Reg. \$1.409A-3(i)(5)(v)) or a "change in the ownership of a substantial portion of a corporation's assets" (within the meaning of Section 409A and Treas. Reg. \$1.409A-3(i)(5)(vi)(A)) or a "change in the ownership of a substantial portion of a corporation's assets" (within the meaning of Section 409A and Treas. Reg. \$1.409A-3(i)(5)(vi)) through being one or more of the following:

(a) any one person, or more than one person acting as a group other than an employee benefit plan (or related trust) of the Company or the Company or a subsidiary (collectively "Excluded Persons"), acquires ownership of stock of the Company that, together with stock held by such person or group, constitutes more than fifty percent (50%) of the total fair market value or total voting power of the stock of the Company;

(b) any one person, or more than one person acting as a group (other than any Excluded Person), acquires (or has acquired during the twelve (12)-month period ending on the date of the most recent acquisition by such person or persons) ownership of stock of the Company that, constitutes thirty percent (30%) or more of the total fair market value or total voting power of the stock of the Company; or

(c) any one person, or more than one person acting as a group, (other than any Excluded Person) acquires (or has acquired during the twelve (12)-month period ending on date of the most recent acquisition by such person or persons) assets from the Company that have a total gross fair market value equal to or more than forty percent (40%) of the total gross fair market value of all the assets of the Company immediately before such acquisition or acquisitions; or

(d) a majority of members of the Company's Board is replaced during any twelve (12)-month period by Directors whose appointment or election is not endorsed by a majority of the members of the Company's Board before the date of the appointment or election.

Notwithstanding subsections (a) through (d), there shall not be a Change in Control if any of the foregoing events occurs, and immediately following such event: (1) all or substantially all of the individuals and entities who are the beneficial owners, respectively, of the outstanding common stock \$0.01 par value of the Company and any other equity securities of the Company that may be substituted or resubstituted for such stock ("Common Stock") and outstanding Company voting securities immediately prior to such corporate transaction will beneficially own, directly or indirectly, more than fifty percent (50%) of, respectively, the outstanding shares of Common Stock, and the combined voting power of the then outstanding voting securities entitled to vote generally in the election of directors, as the case may be, of the corporation resulting from such event (including, without limitation, a corporation which as a result of such event owns the Company or all or substantially all of the Company's assets either directly or through one or more subsidiaries) in substantially the same proportions as their ownership, immediately prior to such event, of the outstanding Company Common Stock and outstanding Company voting securities, as the case may be; (2) no person (other than an Excluded Person or such corporation resulting from such event) will beneficially own, directly or indirectly, thirty percent (30%) or more of, respectively, the outstanding shares of common stock of the corporation resulting from such event or the combined voting power of the outstanding voting securities of such corporation entitled to vote generally in the election of directors, except to the extent that such ownership existed prior to the event; and (3) individuals who were members of the incumbent Board at the time of the Board's approval of the execution of the initial agreement providing for such corporate transaction will constitute at least a majority of the members of the board of directors of the corporation resulting from such corporate transaction.

2.11 **"Code**" means the Internal Revenue Code of 1986, as amended. Reference to any provision of the Code or regulation thereunder, shall include any successor provision and any regulations and other applicable guidance or pronouncement of the Internal Revenue Service or the Department of the Treasury, and applicable case law relating to such Section of the Code.

2.12 "**Committee**" means the Compensation and Leadership Performance Committee of the Board. To the extent the Committee has delegated authority to another person or persons the term "Committee" shall refer to such other person or persons.

2.13 "Company" means Walgreens Boots Alliance, Inc. and any successor thereto.

2.14 "**Disability**" means the Participant has become disabled as provided in the long-term disability plan of the Company or an Affiliate applicable to the Participant (or which would be applicable if the Participant elected coverage under such plan).

2.15 "Effective Date" is defined in Section 1.02.

2.16 **"Employee**" means an individual who is designated as an employee of an Employer on the records of such Employer.

2.17 **"Employer**" means each of the Company and any Affiliate based in the U.S., except to the extent all Employees of any such Affiliate are excluded from Plan participation at the direction of the Committee or its delegate. The term "Employer" includes any successor to the Company or any other Employer.

2.18 **"ERISA**" means the Employee Retirement Income Security Act of 1974, as amended. Reference to any provision of ERISA shall also include any successor provision and regulations and others applicable guidance or pronouncement of a federal regulatory agency and applicable case law relating to such provision.

2.19 "Exchange Act" means the Securities Exchange Act of 1934.

2.20 **"Good Reason**" means any one or more of the following actions or omissions occurring during the Post-Change Period and for which Notice of Termination is given during the Post-Change Period:

(i) a material reduction in the Participant's base compensation;

(ii) requiring the Participant to be based at any office or location more than 50 miles from the pre-Change Date location and also farther from the Participant's residence than the pre-Change Date location;

- (iii) any material diminution in the Participant's authority, duties or responsibilities;
- (iv) any material breach of this Plan by any Employer or the Committee;

provided that, in order for there to be a Termination of Employment by a Participant for Good Reason, the Participant must notify the Participant's Employer of the event constituting such Good Reason within 90 days of the occurrence of such event, by a Notice of Termination that sets forth in reasonable detail the specific facts and circumstances claimed to provide a basis for such Good Reason termination. The Employer must have failed to cure the event constituting Good Reason within 30 days following receipt of the Notice of Termination and the Participant must terminate employment within five days after the lapse of the cure period if no cure is effected. A delay in the delivery of such Notice of Termination or in the Termination of Employment after the lapse of the cure period shall waive the right of the Participant under this Plan to terminate employment for Good Reason.

2.21 "Including" means including without limitation.

2.22 "Involuntary Termination" means the Termination of Employment of a Participant (a) initiated by the Employer other than for Cause or Disability, and (b) for a reason other than death. During the Post-Change Period, a Termination of Employment initiated by the Participant for Good Reason shall also be an Involuntary Termination. For avoidance of doubt, a Participant shall not have an Involuntary Termination of Employment if he or she (i) voluntarily resigns or retires; (ii)has a Termination of Employment because of death or Disability; or (iii) on, before the Termination of Employment (other than a Termination of Employment during the Post-Change Period), is offered employment or re-employment with the Company or an Affiliate or any successor to all or a portion of the business of the Company or Affiliate on terms and conditions substantially similar to the terms and conditions of his or her employment before his or her Termination of Employment.

2.23 "Notice of Termination" means a notice of termination delivered by the Company to a Participant or by a Participant to the Company, and which, if the Termination Date is other than the date of receipt of such Notice of Termination (and is not determined under Section 2.34(a) or (b)), the Termination Date.

2.24 **"Participant**" means an Employee who is eligible or otherwise selected to participate in the Plan in accordance with its terms and whose participation has not been terminated pursuant to Section 3.04.

2.25 "Plan" means this Walgreens Boots Alliance, Inc. Executive Severance and Change in Control Plan, as set forth herein and as from time to time amended.

2.26 "Plans" means plans, programs, or Policies of the Company or the Employer that employs a Participant.

2.27 "Policies" means policies, practices or procedures of the Company or the Employer that employs a Participant.

2.28 "**Post-Change Period**" means the period beginning on the Change Date and ending on the first anniversary of the Change Date.

2.29 **"Pro-rata Annual Incentive**" means, in respect of an Employer's fiscal year during which the Termination Date occurs, an amount equal to the product of (a) (i) in the case of a Termination Date before the Change Date, the actual annual incentive the Participant would have been paid if he or she remained employed on the payment date applicable to then-current employees, and (ii) in the case of a Termination Date or a fter the Change Date or in situations where calculation of the Participant's actual annual incentive is impracticable, the Participant's Target Annual Incentive (determined as of the Termination Date) multiplied by (b) a fraction, the

numerator of which equals the number of days from and including the first day of such fiscal year through and including the Termination Date, and the denominator of which equals 365.

2.30 "Intentionally Omitted".

2.31 "Section 409A Deferred Compensation" means a deferral of compensation that is subject to (and not otherwise exempt from) the requirements of Section 409A of the Code.

2.32 "Severance Multiple" and "Severance Period" mean:

(a) for a Tier I Participant, for a Termination Date occurring before the Change Date or after the Post-Change Period, and for a Tier II Participant, for a Termination Date occurring before, on or after the Change Date:

Title/Benefit Indicator	Severance Multiple	Severance Period
CEO	2.0x	24 months
COO, EVP	2.0x	24 months
SVP	1.0x	12 months
VP Level 5	1.0x	12 months
VP Levels 6-7	0.5x	6 months
Level 8 (Sr. Director or its equivalent)	0.5x	6 months

(b) for a Tier I Participant, for a Termination Date occurring during the Post-Change Period:

Title/Benefit Indicator	Severance Multiple	Severance Period
CEO	2.5x	30 months
COO, EVP,	2.5x	30 months
SVP	2.5x	30 months
VP Levels 5 -7	2.0x	24 months
Level 8 (Sr. Director or its equivalent)	1.0x	12 months

(c) Notwithstanding the foregoing, if on the Termination Date the Company or a Participant's Employer maintains a generally applicable severance arrangement for a broad-based group of employees with a formula that would provide severance benefits to an Participant in a greater amount than results from the application of the Severance Multiple shown above, the amount paid to such Participant under this Plan shall be adjusted so that it is not less than the amount that would be paid under such generally applicable severance arrangement. Such adjusted amount shall be paid at the time and in the form of payment in Section 4.01(a)(iii).

2.33 **"Target Annual Incentive**", as of any date, means the amount equal to the product of a Participant's Base Salary multiplied by the percentage of such Base Salary to which such Participant would be entitled as an annual incentive, based on the terms in effect on such date under any annual incentive plans for the performance period for which the annual incentive is awarded if the performance goals established pursuant to such bonus plan were achieved at the 100% (target) level as of the end of the performance period, but disregarding any reduction in Target Annual Incentive that would constitute Good Reason.

2.34 **"Termination Date**" means the effective date of the Participant's termination of employment from his or her Employer, as designated by the Employer (or by the Participant in the case of termination for Good Reason), in all cases consistent with the remaining terms and conditions of this Plan; provided, however, that:

(a) if the Participant's employment is terminated by reason of death, the Termination Date shall be the date of the Participant's death; and

(b) If the Participant's employment is terminated by reason of Disability, the Termination Date shall be the date regularly applied by the Company's Human Resource function to terminations of employment for Disability; which is generally one year after the commencement of leave of absence due to the Disability.

2.35 **"Termination of Employment**" means in respect of a Participant, a termination of employment as determined by the Committee; provided, however, that with respect to payment of any Section 409A Deferred Compensation, "Termination of Employment" shall mean "separation from service" within the meaning of Section 409A of the Code.

2.36 **"Tier I Participant**" means, unless otherwise designated by the Committee before the Change Date, a Participant who, on the Termination Date, is classified in a VP level 7 position or above (or its equivalent) under the employment level categories regularly applied by the Company's Human Resource function (but disregarding any reduction to a lower employment level category after the Change Date that would constitute Good Reason).

2.37 **"Tier II Participant**" means, unless otherwise designated by the Committee, a Participant who, on the Termination Date, is classified below a VP level 7 position (or its equivalent) under the employment level categories regularly applied by the Company's Human Resource function.

ARTICLE III Participation and Eligibility for Benefits

3.01 Eligibility.

(a) An employee is eligible for participation in this Plan if the Employee is in one of the position categories listed in Section 2.32 above (as determined under the customary practices of the Company's Human Resource function) and is employed by an Employer; provided that the Committee may determine in its discretion that an Employee who would otherwise be eligible based on the above shall not be eligible to participate in this Plan. The Committee in its discretion may also designate selected Employees who are not in one of the position categories listed in Section 2.32 to be eligible to participate in this Plan by written notice to any such Employee.

(b) Notwithstanding subsection (a), any individual who is or becomes a party to an agreement between the individual and an Employer or a participant in a severance plan offered by such Employer (collectively, "Employment Agreement") that provides for severance payments upon Termination of Employment (either before or after a Change in Control) shall not be eligible to become (or remain) a Participant in this Plan.

(c) As a condition to becoming a Participant, an Employee must first agree to restrictive covenants that are applicable to such Employee's position level, in such form as the Company may require, whether or not the Employee becomes eligible for benefits under Article IV or Article V.

3.02 **Participation**. Except as provided in Section 3.01(b), each eligible Employee shall become a Participant in the Plan on the first date (not earlier than the Effective Date) on which he or she is an eligible Employee described in Section 3.01 and satisfies the other requirements of this Article III.

3.03 **Eligibility for Benefits** A Participant becomes eligible for benefits under the Plan if the Participant has an Involuntary Termination. A Tier I Participant also becomes eligible for benefits under the Plan if the Participant has a Termination of Employment for Good Reason within one year after the Change Date.

3.04 **Termination of Participation**. If before the Termination Date a Participant is demoted below the minimum position category listed in Section 2.32 (disregarding in the case of a Tier I Participant any demotion

after the Change Date that would constitute Good Reason), such Participant shall thereupon cease to be a Participant as of the date of demotion. If upon Termination of Employment the Participant has not become eligible for a benefit under Article IV, his or her status as a Participant shall cease upon the Termination Date. Any determination as to a termination of Participation under this Section 3.04 shall be made by the Committee (or by the independent members of the Board upon the recommendation of the Committee in the case of a determination regarding the CEO).

ARTICLE IV

General Obligations of the Employer Upon Involuntary Termination

4.01 **Involuntary Termination**. If a Participant has an Involuntary Termination pursuant to a Notice of Termination given outside of a Post-Change Period, then unless Article V applies, the Employer's sole obligations to such Participant under the Plan shall be as follows:

- (a) The Employer shall pay the Participant the following:
 - (i) all Accrued Obligations;

(ii) subject to Section 9.01, the Participant's Pro-rata Annual Incentive, reduced (but not below zero) by the amount of any Annual Incentive paid to the Participant with respect to the Employer's fiscal year during which the Termination Date occurs (for example, if the Annual Incentive is paid quarterly); the Pro-rata Annual Incentive shall be paid at the same time and in the same form as the Annual Incentives for such fiscal year are paid to ongoing employees; but no later than two and one-half months after the last day of the fiscal year in which the Termination Date occurs; and

(iii) subject to Section 9.01, an amount equal to the sum of the Participant's Base Salary and Target Annual Incentive, each determined as of the Termination Date, multiplied by the applicable Severance Multiple (the "Severance Payment"). The Severance Payment shall be paid in the form of salary continuation unless the Termination Date occurs in the Post-Change Period, in which event the Severance Payment shall be paid in a single lump sum. The Severance Payment shall begin (or be made) no more than sixty days after the Termination Date, provided the applicable revocation period for the release required by Section 9.01 has expired at that time, and subject to Section 10.11(c) and Section 10.11(e). Severance Payments in the form of salary continuation payments shall be made in accordance with the Employer's regular payroll schedule beginning with the first regularly scheduled payroll date after the Termination Date; provided, however, that any Severance Payment or Payments (whether in the form of salary continuation or in a lump sum) that is or are delayed as required by Section 10.11(c) or Section 10.11(e) or to allow for the lapse of the applicable revocation period for the release required by Section 9.01 shall be accumulated without interest and paid on the first date on which payment is permitted under by Section 10.11(c) or Section 10.11(e) or after lapse of the applicable revocation period required by Section 9.01. The amount of each salary continuation payment shall be determined by dividing the Severance Payment by the number of regularly scheduled payroll periods in the Severance Period (determined without regard to any delays required by Section 10.11(c) or Section 10.11(e) or the applicable revocation period under Section 9.01.

(b) The Employer shall provide for Post-Termination of Employment nonqualified deferred compensation benefits, equity awards, and employee welfare benefits pursuant to the terms of the respective Plans and Policies under which such post-Termination of Employment benefits, awards and welfare benefits, if any, are provided, except as provided in (c) below.

(c) Subject to Section 9.01, if the Participant timely elects post-termination continuation coverage under Section 4980 of the Code ("<u>COBRA</u>") with respect to medical, vision, prescription and/or dental coverage, then the Employer shall reimburse the Participant (or pay the provider directly) for the premiums for such COBRA coverage for the Participant and his or her eligible dependents during the applicable Severance Period to the extent such premiums exceed the premiums payable for similar

employer-provided coverage by active employees. There shall be no reimbursement (or direct payment) of such premiums by the Employer for any COBRA coverage extending beyond the end of the Severance Period. Notwithstanding the forgoing, such reimbursement (or direct payment) shall cease if and when the Participant becomes eligible for medical, vision, prescription or dental coverage, respectively, from a subsequent employer, or for Medicare, or if the Participant's COBRA coverage otherwise ends.

4.02 **Termination for Any Other Reason**. If a Participant has a Termination of Employment for any reason other than as described in Section 4.01 (including termination by the Employer for Cause, termination by the Employee whether or not for Good Reason (including due to retirement), termination by the Employer or the Employee for Disability, , or termination on account of death), then, unless Article V applies, the Employer's sole obligations to such Participant under the Plan shall be to pay the Participant all Accrued Obligations determined as of the Termination Date.

ARTICLE V Obligations of the Employer on Involuntary Termination of Certain Participants in the Post-Change Period

5.01 **Application**. During the Post-Change Period a Tier I Participant shall be entitled to benefits under this Article V in lieu of benefits under Article IV.

5.02 **Involuntary Termination in the Post-Change Period**. If a Tier I Participant has an Involuntary Termination (including a Termination of Employment for Good Reason) for which a Notice of Termination is given during the Post-Change Period, then the Employer's sole obligations to such Participant under the Plan shall be as follows:

- (a) The Employer shall pay the Participant the following:
 - (i) all Accrued Obligations;

(ii) subject to Section 9.01, the Participant's Pro-rata Annual Incentive, reduced (but not below zero) by the amount of any Annual Incentive paid to the Participant with respect to the Employer's fiscal year during which the Termination Date occurs (for example, if the Annual Incentive is paid quarterly); and

(iii) subject to Section 9.01, an amount equal to the sum of the Participant's Base Salary and Target Annual Incentive, each determined as of the Termination Date, multiplied by the applicable Severance Multiple ("<u>Post-Change</u> <u>Severance Payment</u>"); provided, however, that any reduction in the Participant's Base Salary or Target annual incentive that would qualify as Good Reason shall be disregarded for this purpose.

The amount described in Section 5.01(a)(ii) and the Post-Change Severance Payment shall be paid in a lump sum payment no more than sixty days after the Termination Date, provided the applicable revocation period required for the release under Section 9.01 has expired at that time; and subject to Section 10.11(c) and Section 10.11(e), and provided further that if the Notice of Termination is given within the Post-Change Period, but the Termination Date occurs outside of the Post-Change Period, then the Post-Change Severance Payment shall be paid in the form of salary continuation payments, in accordance with Section 4.01(a)(ii).

(b) Post-Termination of Employment non-qualified deferred compensation benefits, equity awards, and employee welfare benefits shall be provided pursuant to the terms of the respective Plans and Policies under which such post-Termination of Employment benefits, awards and welfare benefits, if any, are provided, except as provided in (c) below.

(c) Subject to Section 9.01, if the Participant timely elects post-termination continuation coverage under COBRA with respect to medical, vision, prescription and/or dental coverage, then the Employer shall reimburse the Participant (or pay the provider directly) for the premiums for such COBRA

coverage for the Participant and his or her eligible dependents during the applicable Severance Period to the extent such premiums exceed the premiums payable for similar employer-provided coverage by active employees. There shall be no reimbursement (or direct payment) of such premiums by the Employer for any COBRA coverage extending beyond the end of the Severance Period. Notwithstanding the forgoing, such reimbursement (or direct payment) shall cease if and when the Participant becomes eligible for medical, vision, prescription or dental coverage, respectively, from a subsequent employer, or for Medicare, or if the Participant's COBRA coverage otherwise ends.

5.03 **Termination on or After the Change Date for Any Other Reason**. If a Participant has a Termination of Employment for which a Notice of Termination is given during the Post-Change Period, for any reason other than as described in Section 5.02 (including termination by the Employer for Cause, termination by the Employee other than for Good Reason (including due to retirement), termination by the Employer or the Employee for Disability, , or termination on account of death), then the Employer's sole obligation to the Participant under this Plan shall be to pay the Participant all Accrued Obligations determined as of the Termination Date.

5.04 Limitation on Benefits.

(a) To the extent that any payment or distribution to or for the benefit of a Participant pursuant to the terms of this Plan or any other plan, arrangement or agreement with the Company, any of its affiliated companies, any person whose actions result in a change of ownership or effective control covered by Section 280G(b)(2) of the Code or any person affiliated with the Company or such person, whether paid or payable or distributed or distributable pursuant to the terms of this Plan or otherwise (the "<u>Payments</u>") would be subject to the excise tax (the "<u>Excise Tax</u>") imposed by Section 4999 of the Code, then the Company shall reduce the Payments to the amount that is (after taking into account federal, state, local and social security taxes at the maximum marginal rates, including any excise taxes imposed by Section 4999 of the Code) one dollar less than the amount of the Payments that would subject the Participant to the Excise Tax (the "<u>Safe Harbor Cap</u>") if, and only if, such reduction would result in the Participant receiving a higher net after-tax amount. Unless a Participant shall have given prior written notice specifying a different order to the Company to effectuate the Safe Harbor Cap, the Payments to be reduced hereunder will be determined in a manner which has the least economic cost to the Participant and, to the extent the economic cost is equivalent, will be reduced in the reverse order of when the Payment would have been made to the Participant until the reduction specified herein is achieved. A Participant's right to specify the order of reduction of the Payments shall apply only to the extent that it does not directly or indirectly alter the time or method of payment of any amount that is deferred compensation subject to (and not exempt from) Section 409A.

(b) All determinations required to be made under this Section 5.04, including whether and when the Safe Harbor Cap is required and the amount of the reduction of the Payments pursuant to the Safe Harbor Cap and the assumptions to be utilized in arriving at such determination, shall be made by a public accounting firm or other nationally recognized consulting firm with expertise in Section 280G of the Code that is retained by the Company as of the date immediately prior to the Change in Control (the "<u>Calculating Firm</u>") which shall provide detailed supporting calculations both to the Company and the Participant within fifteen (15) business days of the receipt of notice from the Company or the Participant that there has been a Payment, or such earlier time as is requested by the Company shall enter into any agreement requested by the Calculating Firm in connection with the performance of the services hereunder. The Determination by the Calculating Firm shall be binding upon the Company and the Participant. The Company shall bear and pay directly all costs and expenses incurred in connection with any contests or disputes with the Internal Revenue Service relating to the Excise Tax, and the Participant shall cooperate, to the extent his or her reasonable out-of pocket expenses are reimbursed by the Company, with any reasonable requests by the Company in connection with any such contests or disputes.

ARTICLE VI Administration

6.01 The Company and Committee.

(a) The Company shall have overall responsibility for the establishment, amendment and termination of the Plan. In carrying out its responsibilities hereunder, the Company shall act through the Committee (except with respect to the CEO, with respect to whom the Company shall act through the independent members of the Board). The Committee shall have, in its discretion, the responsibilities, duties, powers and authority, assigned to it in this Plan and any responsibilities, duties, powers and authority, under this Plan that are not specifically delegated to anyone else, including the following:

(i) to determine which individuals shall be selected as Tier I Participants, and Tier II Participants.

(ii) to decide on questions concerning the Plan and the eligibility of any Participant to participate in the Plan, including whether the Participant should remain (or become) a Participant;

(iii) to determine the nature and timing of any Termination of Employment or the existence of Cause or Good Reason;

(iv) subject to any limitations under the Plan or applicable law, to make and enforce such rules and regulations and prescribe the use of such forms as it shall deem necessary for the efficient administration of the Plan;

(v) to require any person to furnish such information as it may request as a condition to receiving any benefit under the Plan;

(vi) to compute or have computed the amount of benefits that shall be payable to any person in accordance with the provisions of the Plan;

(vii) to construe and interpret the Plan and correct defects, supply omissions and reconcile inconsistencies in the Plan; and

(viii) to make all other decisions and determinations (including factual determinations) as the Committee may deem necessary or advisable in carrying out its duties and responsibilities or exercising its powers.

If the CEO is a Participant, such duties, responsibilities and powers shall be exercised with respect to him or her by independent members of the Board, which shall be the Committee unless otherwise determined by the Board. Decisions of the Committee (or such other collection of independent members of the Board with respect to the CEO) shall be final, conclusive and binding on all persons interested in the Plan, including Participants, beneficiaries and other persons claiming rights from or through a Participant.

6.02 **Delegation of Committee Authority**. Except as otherwise determined by the Committee, the officers of the Company shall administer the routine, nondiscretionary functions necessary to carry out the operations of the Plan. The Committee may delegate to officers or employees of the Company, or committees thereof, the authority, subject to such terms as the Committee shall determine, to perform such other administrative functions and exercise such administrative powers and authority, as the Committee in its discretion may determine. Such delegation may be revoked at any time.

6.03 Advisors and Agents of the Committee. The Committee may (i) authorize one or more of its members or an agent to execute or deliver any instrument, and make any payment on its behalf and (ii) utilize and cause the Company to pay for the services of associates and engage accountants, agents, clerks, legal counsel, record keepers and professional consultants (any of whom may also be serving an Employer or another Affiliate of the Company) to assist in the administration of this Plan or to render advice with regard to any responsibility under this Plan.

6.04 **Records and Reports of the Committee**. The Committee or its delegate shall maintain records and accounts relating to the administration of the Plan.

6.05 **Limitation of Liability; Indemnification**. The members of the Board and the Committee, and the persons to whom they have delegated their authority, shall have no liability with respect to any action or omission made by them in good faith nor from any action made in reliance on (i) the advice or opinion of any accountant, legal counsel, medical adviser or other professional consultant or (ii) any resolutions of the Board certified by the secretary or assistant secretary of the Company. Each member of the Board, the Committee, and each employee to whom are delegated duties, responsibilities and authority with respect to the Plan shall be indemnified, defended, and held harmless by the Company and the Employers and their respective successors against all claims, liabilities, fines and penalties and all expenses (including but not limited to attorneys' fees) reasonably incurred by or imposed on such member or employee that arise as a result of his actions or failure to act in connection with the operation and administration of the Plan, to the extent lawfully allowable and to the extent that such claim, liability, fine, penalty or expense is not paid for by liability insurance purchased by or paid for by the Company or an Employer. Notwithstanding the foregoing, the Company or an Employer shall not indemnify any person for any such amount incurred through any settlement or compromise of any action unless the Company or an Employer consent in writing to such settlement or compromise.

6.06 **Plan Expenses**. Expenses relating to the Plan before its termination shall be paid from the general assets of the Company or an Employer. Any individual who serves as a member of the Committee shall receive no additional compensation for such service.

6.07 Service in More than One Capacity. Any person or group of persons may serve the Plan in more than one capacity.

ARTICLE VII Amendments; Termination

7.01 **Amendment or Termination of the Plan**. The Company by duly adopted resolution of the Board (or of the Committee, to the extent delegated by the Board) shall have the sole right to alter, amend or terminate this Plan in whole or in part at any time and to terminate the participation of any Employee; provided, however, that:

(i) any adverse amendment or termination shall be effective only as to those Participants, if any, who have consented to such amendment or termination or who have received from the Company at least 12 months' prior written notice ("<u>Amendment Notice</u>" or "<u>Expiration Notice</u>," respectively) of such adverse amendment or termination that sets forth the date of termination or amendment ("<u>Amendment Date</u>" or "<u>Expiration Date</u>"), and

(ii) no such Amendment Notice or Expiration Notice shall be effective as to any Participant if a Change Date occurs before the Amendment Date or Expiration Date specified in the Amendment Notice or Expiration Notice.

Any purported Plan termination or amendment in violation of this Section 7.01 shall be void and of no effect.

ARTICLE VIII Claims Procedure

8.01 Filing a Claim.

(a) No claim shall be required for benefit due under the Plan. Any individual eligible for benefits under this Plan who believes he or she is entitled to additional benefits or who desires to clarify his or her right to future benefits under the Plan (<u>"Claimant"</u>) may submit his application for benefits (<u>"Claim"</u>) to the Committee (or to such other person or persons as may be designated by the Committee) in writing in such form as is provided or approved by the Committee.

(b) When a Claim has been filed properly, it shall be evaluated and the Claimant shall be notified of the approval or the denial of the Claim within 90 days after the receipt of such Claim. A Claimant shall be given a written notice in which the Claimant shall be advised as to whether the Claim is granted or denied, in whole or in part. If a Claim is denied, in whole or in part, the notice shall contain (i) the specific reasons for the denial, (ii) references to pertinent provisions of this Plan on which the denial is based, (iii) a description of any additional material or information necessary to perfect the Claim and an explanation of why such material or information is necessary, and (iv) the Claimant's right to seek review of the denial.

Review of Claim Denial. If a Claim is denied, in whole or in part, or if a Claim is neither approved nor denied within the 8.02 90-day period specified Section 8.01(b), the Claimant shall have the right, within 60 days after receipt of such denial (or after such claim to deemed denied), (i) request that the Committee (or such other person or persons as shall be designated in writing by the Committee) review the denial or the failure to approve or deny the Claim, (ii) review pertinent documents, and (iii) submit issues and comments in writing. Within 60 days after a such request is received, the Committee shall complete its review and give the Claimant written notice of its decision. The Committee shall include in its notice to Claimant the specific reasons for its decision and references to provisions of this Plan on which its decision is based. A Claimant shall have no right to seek review of a denial or benefits, or to bring any action in any court to enforce a Claim, before to his filing a Claim and exhausting his rights to review under Sections 8.01 and 8.02. No legal action for benefits under this Plan may be brought unless the action is commenced within one year from the date of the final decision on appeal has been made. If a Claimant or other interested person challenges a decision, a review by the court of law will be limited to the facts, evidence and issues presented during the claims procedure set forth above. Facts and evidence that become known to the Claimant or other interested person after having exhausted the claims procedure must be brought to the attention of the Committee for reconsideration of the claims determination. Issues not raised with the Committee will be deemed waived. Any action, suit or proceeding brought by a Claimant with respect to, or to enforce the terms of, this Plan shall be brought exclusively in the courts of the State of Illinois located in Lake County, Illinois or in the courts of the United States of America for the Northern District of Illinois.

ARTICLE IX Release; No Mitigation; No Duplication of Benefits

9.01 **Release Required**. Any and all amounts payable and benefits or additional rights provided pursuant to this Plan other than the Accrued Obligations and amounts provided under Section 4.01(b) or Section 5.02(b) shall only be payable if the Participant (or Participant's beneficiary in the event of Participant's death) timely delivers to the Employer and does not revoke a general waiver and release of claims in favor of the Company and related parties ("Company Parties") in such form and with such terms and conditions as are reasonably acceptable to the Company, and the revocation period related to such general waiver and release has expired. Such general waiver and release shall be executed and delivered (and the revocation period related thereto, if any, shall have lapsed without revocation having been made) within sixty (60) days following the Termination Date.

9.02 **No Mitigation**. No Participant shall have any duty to mitigate the amounts payable under this Plan by seeking or accepting new employment or self-employment following termination. Except as specifically otherwise provided in this Plan, all amounts payable pursuant to this Plan shall be paid without reduction regardless of any amounts of salary, compensation or other amounts that may be paid or payable to the Participant as the result of the Participant's employment by another employer or self-employment.

9.03 **No Duplication of Benefits / Rehire**. Subject to Section 10.11(f), to the extent that a Participant shall have received severance payments or other severance benefits under any other Plan or agreement of the Company or an Employer before receiving severance payments or other severance benefits pursuant to Article

IV or Article V, the severance payments or other severance benefits under such other Plan or agreement shall reduce (but not below zero) the corresponding severance payments or other severance benefits to which such Participant shall be entitled under Article IV or Article V. Notwithstanding the foregoing, if a Participant is rehired by the Company, any Affiliate, or any successor to all or a portion of the business of the Company or Affiliate, then any remaining severance payments and other severance benefits hereunder shall be discontinued upon such rehire.

To the extent that a Participant accepts payments made pursuant to Article IV or Article V, he shall be deemed to have waived his right to receive a corresponding amount of future severance payments or other severance benefits under any other Plan or agreement of the Company. Payments and benefits provided under the Plan shall be in lieu of any termination or severance payments or benefits for which the Participant may be eligible under any of the Plans or Policy of the Company or an Affiliate or under the Worker Adjustment Retraining Notification Act of 1988 or any similar statute or regulation.

ARTICLE X Miscellaneous

10.01 **Participant Information**. Each Participant shall notify the Committee of his home address and each change of home address. Each Participant shall also furnish the Committee with any other information and data that the Committee considers necessary for the proper administration of the Plan. The information provided by the Participant under this Section shall be binding on the Participant, his dependents and any beneficiary for all purposes of the Plan and the Committee shall be entitled to rely on any representations regarding personal facts made by a Participant, his dependents or beneficiary, unless such representations are known to be false.

10.02 **Electronic Media**. Under procedures authorized or approved by the Committee, any form for any notice, election, designation, or similar communication required or permitted to be given to or received from a Participant under this Plan may be communicated or made available to the Company or Participant in an electronic medium (including computer network, e-mail or voice response system) and any such communication to or from a Participant or Beneficiary through such electronic media shall be fully effective under this Plan for such purposes as such procedures shall prescribe. Any record of such communication retrieved from such electronic medium under its normal storage and retrieval parameters shall be effective as a fully authentic executed writing for all purposes of this Plan absent manifest error in the storage or retrieval process.

10.03 **Notices**. All notices and other communications under this Plan shall be in writing and delivered by hand, by nationally recognized delivery service that promises overnight delivery, or by first-class registered or certified mail, return receipt requested, postage prepaid, addressed as follows:

If to Participant, at his most recent home address on file with the Company.

If to the Company or any other Employer,

Walgreens Boots Alliance, Inc. 108 Wilmot Road Deerfield, Illinois 60015 Attn.: General Counsel

or to such other address as either party shall have furnished to the other in writing. Notice and communications shall be effective the day of receipt if delivered by hand or electronically, the second business day after deposit with an overnight delivery service if so deposited, or the fifth business day after mailing in the case of first class registered or certified mail.

10.04 **No Employment Contract**. The existence of this Plan shall not confer any legal or other rights upon any Participant to employment or continuation of employment. Employees are employees at will. The Company and each Employer reserve the right to terminate any Participant with or without cause at any time, notwithstanding the provisions of this Plan.

10.05 Headings. The headings in this Plan are for convenience of reference and shall not be given substantive effect.

10.06 **Construction**. Any masculine pronoun shall also mean the corresponding female or neuter pronoun, as the context requires. The singular and plural forms of any term used in this Plan shall be interchangeable, as the context requires.

10.07 **Joint and Several Liability**. In the event that any Employer incurs any obligation to a Participant pursuant to this Plan, such Employer, the Company and each Affiliate, if any, of which such Employer is a subsidiary shall be jointly and severally liable with such Employer for such obligation.

10.08 **Successors**. This Plan shall inure to the benefit of and be binding upon the Company, each Employer and their respective successors and assigns. The Company will require any successor (whether direct or indirect, by purchase, merger, consolidation or otherwise) to all or substantially all of the business or assets of any Employer to assume expressly and agree to comply with this Plan in the same manner and to the same extent that the Employer would be required to comply with it if no such succession had taken place. Failure to require such assumption will be a material breach of this Plan. Any successor to the business or assets of any Employer that assumes or agrees to perform this Plan by operation of law, contract, or otherwise shall be jointly and severally liable with the Employer under this Plan as if such successor were the Employer.

10.09 **Payments to Beneficiary**. If a Participant dies after becoming entitled to payments under Section 4.01 or 5.02 but before receiving all amounts to which he is entitled under this Plan, then, subject to Section 9.01, such remaining amounts shall be paid in a lump sum to one or more beneficiaries designated in writing by the Participant for the purposes of this Plan and received by the Committee before the Participant's death, which the Participant may change from time to time in the manner without the consent of any previously designated beneficiary, or if none is so designated, to the Participant's estate.

10.10 **Non-Alienation of Benefits.** Benefits payable under this Plan shall not be subject in any manner to anticipation, alienation, sale, transfer, assignment, pledge, encumbrance, charge, garnishment, execution or levy of any kind, either voluntary or involuntary, before actually being received by the Participant, and any such attempt to dispose of any right to benefits payable under this Plan shall be void.

10.11 Tax Matters.

(a) An Employer may withhold from any amounts payable under this Plan or from any other amount due a Participant any federal, state, local and other income, employment and other taxes that are required to be withheld pursuant to any applicable law or regulation.

(b) The intent of the Employers is that payments and benefits under this Plan are exempt from or comply with Section 409A of the Code and, accordingly, to the maximum extent permitted, this Plan shall be interpreted in accordance with that intent. To the extent that any provision hereof is modified in order to comply with Section 409A of the Code, such modification shall be made in good faith and shall, to the maximum extent reasonably possible, maintain the original intent and economic benefit to the Participant and the Employer of the applicable provision without violating the provisions of Section 409A of the Code. In no event whatsoever shall the Company or any Employer be liable for any additional tax, interest or penalty that may be imposed on a Participant or Employee by Section 409A of the Code or damages for failing to comply with Section 409A of the Code.

(c) If a Participant is deemed on the Termination Date to be a "specified employee" within the meaning of that term under Section 409A(a)(2)(B) of the Code, then with regard to any payment or the provision of any benefit that is considered "nonqualified deferred compensation" under Section 409A of the Code payable on account of a "separation from service," to the extent required by Section 409A of the Code, such payment or benefit shall not be made or provided until the date which is the earlier of (i) the day after the expiration of the six (6)-month period measured from the date of such "separation from service" of the Employee, and (ii) the date of the Employee's death. Upon the expiration of the six-month delay period, all payments and benefits delayed pursuant to this provision (whether they would

have otherwise been payable in a single sum or in installments in the absence of such delay) shall be paid or reimbursed to the Employee in a lump sum without interest, and all remaining payments and benefits due under this Plan shall be paid or provided in accordance with the normal payment dates specified for them herein.

(d) To the extent that reimbursements or other in-kind benefits under this Plan constitute "nonqualified deferred compensation" for purposes of Section 409A of the Code, (A) all expenses or other reimbursements hereunder shall be made on before to the last day of the taxable year following the taxable year in which such expenses were incurred by the Participant, (B) any right to reimbursement or in-kind benefits shall not be subject to liquidation or exchange for another benefit, and (C) no such reimbursement, expenses eligible for reimbursement, or in-kind benefits provided in any taxable year shall in any way affect the expenses eligible for reimbursement, or in-kind benefits to be provided, in any other taxable year.

(e) For purposes of Section 409A of the Code, the Participant's right to receive installment payments pursuant to this Plan shall be treated as a right to receive a series of separate and distinct payments. Whenever this Plan specifies a payment period with reference to a number of days, the actual date of payment within the specified period shall be within the sole discretion of the Employer; provided that if the timing of the payment is contingent on the lapse or expiration of the revocation period for the release required under Section 9.01 and such revocation period could, as of the Termination Date, lapse either in the same year as the Termination Date or in the following year, the actual date of payment within the specified period shall be in such following year.

(f) Notwithstanding any other provision of this Plan to the contrary, in no event shall any payment or benefit under this Plan that constitutes "nonqualified deferred compensation" for purposes of Section 409A of the Code be subject to offset by any other amount unless such offset would not trigger additional taxes and penalties under Section 409A of the Code.

10.12 **Clawback Policy**. Notwithstanding anything to the contrary herein, all compensation paid to a Participant shall be subject to forfeiture, recovery by Company or other action pursuant to any clawback or recoupment policy which the Company may adopt from time to time, including any policy adopted and maintained to comply with the Dodd-Frank Wall Street Reform and Consumer Protection Act and implementing rules and regulations thereunder, or as otherwise required by applicable law.

10.13 **Governing Law**. The provisions of this Plan shall be governed, construed and administered in accordance with the laws of the State of Illinois, other than its laws respecting choice of law, except to the extent preempted by federal law.

10.14 **Severability**. If any one or more Articles, Sections or other portions of this Plan are declared by any court or governmental authority to be unlawful or invalid, such unlawfulness or invalidity shall not serve to invalidate any Article, Section or other portion not so declared to be unlawful or invalid; provided that if the release required under Section 9.01 is declared to be unlawful or unenforceable, then no payments shall be made the payment of which is subject to such release, and the Participant shall forthwith restore to the Employer any payments previously made that were subject to such release. Any Article, Section or other portion so declared to be unlawful or invalid shall be construed so as to effectuate the terms of such article, section or other portion to the fullest extent possible while remaining lawful and valid.

5 July 2019

PRIVATE & CONFIDENTIAL

Alex Gourlay 108 Wilmot Road Deerfield, IL 60015

Dear Alex,

Extension to your Assignment to Walgreen Co.

This letter is to confirm the extension of your secondment to Walgreen Co. under the secondment letter agreement dated 26 September 2013 between you and Walgreens Boots Alliance Services Limited (formerly Alliance Boots Management Services Ltd.), as previously extended per letter agreement dated 27 January 2016, 27 March 2017, 13 July 2017 & 1 August 2018 (collectively, the "Agreement"). This extension is for an additional 12 months, with an end date of 31 July 2020.

All assignment terms and conditions remain unchanged, as detailed in the Agreement.

Please sign below to confirm that you have read, understand and agree to this extension of the Agreement.

Yours sincerely,

/s/ Kathleen Wilson-Thompson

Dated: October 22, 2019

Kathleen Wilson-Thompson, Executive Vice President - Global Chief Human Resources Officer, WBA Signed on behalf of Walgreens Boots Alliance Services Limited

PROPRIETARY AND CONFIDENTIAL

I confirm that I have read, understand and agree to be bound by the contents of this extension letter.

Alex Gourlay

/s/ Alex Gourlay	
Signed	

.....July 11, 2019..... Date signed

Exhibit 21

Certain subsidiaries of Walgreens Boots Alliance, Inc. as of August 31, 2019 and their respective state of incorporation or organization are listed below. The names of certain other subsidiaries have been omitted because, considered in the aggregate as a single subsidiary, they would not constitute, as of August 31, 2019, a "significant subsidiary" as that term is defined in Rule 1-02(w) of Regulation S-X.

Name	State or Country of Incorporation
Walgreen Arizona Drug Co.	Arizona
Walgreens Boots Alliance Holdings LLC	Delaware
Waltrust Properties, Inc.	Delaware
Walgreens Specialty Pharmacy, LLC	Delaware
Prime Therapeutics Specialty Pharmacy LLC	Delaware
WBA Investments, Inc.	Delaware
WBA Financial, LLC	Delaware
United Company of Pharmacists SAE	Egypt
Alliance Healthcare Répartition	France
Alliance Healthcare Deutschland AG	Germany
Walgreens Boots Alliance (Hong Kong) Investments Limited	Hong Kong
Walgreen Co.	Illinois
Bond Drug Company of Illinois, LLC	Illinois
Boots Retail (Ireland) Limited	Ireland
Walgreen Louisiana Co., Inc.	Louisiana
Walgreen International S.à r.l.	Luxembourg
Superior Luxco 1 S.à r.l.	Luxembourg
Walgreen Investments Luxembourg S.à r.l.	Luxembourg
WBA Luxembourg 7 S.à.r.l.	Luxembourg
WBA Luxembourg 6 S.à r.l.	Luxembourg
Farmacias Benavides S.A.B. de C.V.	Mexico
Walgreen Eastern Co., Inc.	New York
Duane Reade	New York
Walgreen of Puerto Rico, Inc.	Puerto Rico
Alliance Healthcare Romania SRL	Romania
Alliance Healthcare España S.A.	Spain
Alliance Healthcare Ecza Deposu Anonim Şirketi	Turkey
Boots UK Limited	United Kingdom
The Boots Company PLC	United Kingdom
Alliance Healthcare (Distribution) Limited	United Kingdom
Alliance Boots Holdings Limited	United Kingdom
WBA Acquisitions UK Holdco 7 Limited	United Kingdom
Superior Acquisitions Limited	United Kingdom
WBAD Holdings Limited	United Kingdom
WBA International Limited	United Kingdom
Walgreens Boots Alliance Limited	United Kingdom
WBA Acquisitions UK Topco Limited	United Kingdom
Superior Holdings Limited	United Kingdom
Walgreens Boots Alliance Scottish LP	United Kingdom

CONSENT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

We consent to the incorporation by reference in Registration Statement Nos. 333-201327 and 333-198768 on Form S-8 and Registration Statement Nos. 333-228923 and 333-209569 on Form S-3 of our reports dated October 28, 2019 related to the consolidated financial statements of Walgreens Boots Alliance, Inc. (the "Company") and the effectiveness of the Company's internal control over financial reporting, appearing in this Annual Report on Form 10-K of the Company for the year ended August 31, 2019.

/s/ DELOITTE & TOUCHE LLP

Chicago, Illinois

October 28, 2019

CERTIFICATION

I, Stefano Pessina, certify that:

- 1. I have reviewed this annual report on Form 10-K of Walgreens Boots Alliance, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/

Stefano Pessina

Chief Executive Officer

Date: October 28, 2019

Stefano Pessina

CERTIFICATION

I, James Kehoe, certify that:

- 1. I have reviewed this annual report on Form 10-K of Walgreens Boots Alliance, Inc.;
- Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rule 13a-15(f) and 15d-15(f)) for the registrant and have:
 - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

/s/	James Kehoe	Global Chief Financial Officer	Date: October 28, 2019
	James Kehoe		

CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

In connection with the Annual Report of Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), on Form 10-K for the year ended August 31, 2019 as filed with the Securities and Exchange Commission (the "Report"), I, Stefano Pessina, Chief Executive Officer of the Company, certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, to the best of my knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

<u>/s/ Stefano Pessina</u> Stefano Pessina Chief Executive Officer Dated: October 28, 2019

A signed original of this written statement required by Section 906 has been provided to Walgreens Boots Alliance, Inc. and will be retained by Walgreens Boots Alliance, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.

CERTIFICATION PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002 (18 U.S.C. SECTION 1350)

In connection with the Annual Report of Walgreens Boots Alliance, Inc., a Delaware corporation (the "Company"), on Form 10-K for the year ended August 31, 2019 as filed with the Securities and Exchange Commission (the "Report"), I, James Kehoe, Global Chief Financial Officer of the Company, certify, pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, to the best of my knowledge:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

<u>/s/ James Kehoe</u> James Kehoe Global Chief Financial Officer Dated: October 28, 2019

A signed original of this written statement required by Section 906 has been provided to Walgreens Boots Alliance, Inc. and will be retained by Walgreens Boots Alliance, Inc. and furnished to the Securities and Exchange Commission or its staff upon request.